ANGLIA RUSKIN UNIVERSITY

FACULTY OF SCIENCE AND ENGINEERING

THE EFFECT OF SPECIES ON PERCEPTIONS OF ANIMALS: PREDICTORS AND CAUSES OF PET SPECIESISM

SARAH MAY GRADIDGE

A thesis in partial fulfilment of the requirements of Anglia Ruskin University for the degree

of Doctor of Philosophy

Submitted: October 2022

Acknowledgements

I would firstly like to thank my family and friends for all their support and encouragement during my PhD, especially Chris, Mum and John, Dad and Jeannie, and Nessa and Steph. I could not have done this PhD without your support. I would also like to thank my supervisors Dr Magdalena Zawisza, Dr Annelie Harvey, Prof Viren Swami, and Prof Daragh McDermott for their excellent supervision and informing my development as a researcher. Thanks also to my annual review and/or confirmation of registration panel members Dr Jane Aspell, Dr Jacob Dunn, Dr Lewis Goodings, and Dr Andrew Smith for their constructive feedback, which has aided the project. Finally, I would like to thank the School of Psychology and Sport Science at Anglia Ruskin University for providing financial support to cover reimbursement to participants for taking part in studies.

ANGLIA RUSKIN UNIVERSITY

ABSTRACT

FACULTY OF SCIENCE AND ENGINEERING

DOCTOR OF PHILOSOPHY

THE EFFECT OF SPECIES ON PERCEPTIONS OF ANIMALS: PREDICTORS AND CAUSES OF PET SPECIESISM

SARAH MAY GRADIDGE

October 2022

What was Investigated and Why

Dogs and pigs share multiple traits, yet dogs are typically companions whilst pigs are food, a phenomenon termed pet speciesism. Pet speciesism can harm animal welfare due to meat consumption and associated climate change. Whilst previous research evidences pet speciesism, research has yet to identify why pet speciesism occurs. This thesis therefore explores possible causes drawn from previous literature, so future interventions can reduce or prevent pet speciesism: behavioural self-relevance, subjective self-relevance, familiarity, similarity to humans, pet status, and profit status.

How the Topic was Investigated

Pet speciesism is operationalised here across six psychological dimensions from the Stereotype Content Model (SCM) and Behaviours from Intergroup Affect and Stereotypes (BIAS) map: Warmth, competence, active help, active harm, passive help, and passive harm. I explored the above possible causes through six studies: Firstly, a regression identifying which variables were significantly associated with pet speciesism, followed by five experiments manipulating potential causes of pet speciesism.

What was Found

Behavioural self-relevance, subjective self-relevance, familiarity, similarity, and pet status were significant regressors. The causal experiments found that: Neither behavioural nor subjective self-relevance caused pet speciesism. Conversely, similarity significantly caused pet speciesism in most SCM/BIAS map dimensions. Familiarity could not be effectively manipulated. Finally, pet status may significantly cause pet speciesism, but only in limited SCM/BIAS map dimensions.

What Conclusions were Drawn

This thesis concludes that similarity causes pet speciesism in most SCM/BIAS map dimensions, whereas pet status may cause pet speciesism only in limited SCM/BIAS map dimensions. The thesis also highlights unexpected null effects of behavioural and subjective self-relevance and failed familiarity manipulations. Overall, this thesis uniquely tests extrinsic causes of pet speciesism, with extensive theoretical (demonstrating limitations of previous literature) and practical (informing interventions) implications. The research may inform interventions which decrease pet speciesism and reduce meat consumption.

Keywords: pet speciesism, Stereotype Content Model, BIAS map, dog, pig

Acknowledgements	i
List of Figures	ix
List of Tables	X
Notation	. xiii
Chronology of Events	. xiv
List of Appendices	XV
Third Party Copyright Declaration	. xvi
Chapter 1. Introduction	1
1.1. Dog vs. Pig Pet Speciesism	1
1.2. Framework for Measuring Pet Speciesism	4
1.3. Causes of Pet Speciesism	6
1.3.1. Self-Relevance	10
1.3.2. Familiarity	13
1.3.3. Similarity to Humans	15
1.3.4. Species Status	17
1.4. Gaps in the Literature and Research Questions (RQs)	18
Chapter 2. Study 1: Exploring Regressors of Pet Speciesism	20
2.1. Introduction	20
2.2. Methods	26
2.2.1. Participants	26
2.2.2. Design	27
2.2.3. Materials	27
2.2.4. Procedure	32
2.3. Analyses	33
2.3.1. General Analytical Strategy Across Studies	33
2.3.2. Statistical Assumptions to Test H1-H5: Exploring Dog vs. Pig Differences Acr Warmth, Competence, Familiarity, Similarity, Profit Status and Pet Status (MANOV	coss A) 35
2.3.3. Statistical Assumptions to Test H6-H12: Exploring Regressors of Dogs' and Pi Warmth and Competence (Multiple Regressions)	igs' 36
2.3.4. Testing H1-H5: Exploring Dog vs. Pig Differences in Warmth, Competence, Familiarity, Similarity, Profit Status and Pet Status	37
2.3.5. Testing H6-H12: Exploring Regressors of Dogs' and Pigs' Warmth and Competence	39
2.4. Discussion	44
2.4.1. Discussion of Findings	44

Table of Contents

2.4.2. Limitations of the Current Study	49
2.5. Conclusion	50
Chapter 3. Study 2: Causal Exploration of Self-Relevance	52
3.1. Introduction and Extension from Study 1	52
3.2. Methods	57
3.2.1. Participants	57
3.2.2. Design	58
3.2.3. Materials	59
3.2.4. Procedure	62
3.3. Analyses	63
3.3.1. Statistical Assumptions to Test H1-H4: Exploring Effects of Behavioural and Subjective Self-Relevance on SCM and BIAS Map DVs (2x2x2 MANCOVA)	64
3.3.2. Statistical Assumptions to Test H5 and Exploratory Analysis: Exploring Effect Behavioural and Subjective Self-Relevance on Post-Manipulation Dissonance, Post- Evaluation Dissonance and Reported Weekly Meat Consumption (2x2 MANCOVA)	cts of - .) 65
3.3.3. Testing H1-H4: Exploring Effects of Behavioural and Subjective Self-Relevant on SCM and BIAS Map DVs	nce 66
3.3.4. Testing H5 and Exploratory Analysis: Exploring Effects of Behavioural and Subjective Self-Relevance on Post-Manipulation Dissonance, Post-Evaluation Dissonance and Reported Weekly Meat Consumption	68
3.3.5. Testing H6-H9: Exploring Mediational Relationships Between Behavioural ar Subjective Self-Relevance, Dissonance, Warmth and Active Behavioural Intentions.	nd 71
3.4. Discussion	75
3.4.1. Discussion of Findings	75
3.4.2. Limitations of the Current Study and Directions for Future Research	79
3.4.3. Conclusion	80
Chapter 4. Pilot Studies and Study 3 Exploring Behavioural and Subjective Self-Relevan	nce 82
4.1. Introduction	82
4.2. Follow-Up/Pilot Study 1a to Test Behavioural and Subjective Self-Relevance Manipulations	82
4.2.1. Introduction and Extension from Study 2	82
4.2.2. Methods	83
4.2.3. Analyses	88
4.2.4. Discussion and Conclusion	91
4.3. Pilot Study 2a to Test Amended Behavioural and Subjective Self-Relevance	
Manipulations	91
4.3.1. Introduction	91

4.3.2. Methods	
4.3.3. Analyses	
4.3.4. Discussion and Conclusion	100
4.4. Pilot Study 3a to Test Behavioural Self-Relevance Manipulation	101
4.4.1. Introduction	101
4.4.2. Methods	102
4.4.3. Analyses	107
4.4.4. Discussion and Conclusion	109
4.5. Pilot Study 4a to Test Behavioural Self-Relevance Manipulation	109
4.5.1. Introduction	109
4.5.2. Methods	
4.5.3. Analyses	
4.5.4. Discussion and Conclusion	
4.6. Study 3: Applying Self-Relevance to Dogs and Pigs	
4.6.1. Introduction and Extension from Studies 1-2 and Pilot Studies	117
4.6.2. Methods	120
4.6.3. Analyses	
4.6.4. Discussion	
4.7. Overall Discussion	
Chapter 5. Pilot Study and Study 4 Exploring Familiarity and Similarity	
5.1. Introduction	
5.2. Pilot Study 1b to Test Familiarity and Similarity Manipulations	161
5.2.1. Introduction	
5.2.2. Methods	
5.2.3. Analyses	
5.2.4. Discussion and Conclusion	
5.3. Study 4: Causal Exploration of Familiarity and Similarity	
5.3.1. Introduction and Extension from Study 1	
5.3.2. Methods	
5.3.3. Analyses	
5.3.4. Discussion	
5.4. Overall Discussion	
Chapter 6. Pilot Studies for Familiarity and Study 5 Exploring Similarity	
6.1. Introduction	
6.2. Pilot Study 2b to Test Amended Familiarity Manipulation	

6.2.1. Introduction	
6.2.2. Methods	
6.2.3. Analyses	
6.2.4. Discussion and Conclusion	
6.3. Pilot Study 3b to Test Amended Familiarity Manipulation	
6.3.1. Introduction	
6.3.2. Methods	
6.3.3. Analyses	
6.3.4. Discussion and Conclusion	
6.4. Pilot Study 4b to Test Amended Familiarity Manipulation	
6.4.1. Introduction	
6.4.2. Methods	
6.4.3. Analyses	
6.4.4. Discussion and Conclusion	
6.5. Study 5: Applying Similarity to Dogs and Pigs	
6.5.1. Introduction and Extension from Studies 1 and 4	
6.5.2. Methods	
6.5.3. Analyses	
6.5.4. Discussion	
6.6. Overall Discussion	
Chapter 7. Pilot Study and Study 6 Exploring Pet Status	
7.1. Introduction	
7.2. Pilot Study 1c to Test Pet Status Manipulation	
7.2.1. Introduction	
7.2.2. Methods	
7.2.3. Analyses	
7.2.4. Discussion and Conclusion	
7.3. Study 6: Causal Exploration of Pet Status	
7.3.1. Introduction and Extension from Study 1	
7.3.2. Methods	
7.3.3. Analyses	
7.3.4. Discussion	
7.4. Overall Discussion	
Chapter 8. Discussion and Conclusions	
8.1. Discussion of all Findings	

8.1.1. Effects of Species on SCM/BIAS Map Variables (Evidence for Pet Speciesism) 29	6
8.1.2. Exploring Associations Between Warmth, Competence and Behavioural Intention (Evidence for the BIAS map as an Extension of SCM)	s 8
8.1.3. Exploring Possible Causes of Pet Speciesism: Behavioural Self-Relevance, Subjective Self-Relevance, Familiarity, Similarity, and Pet Status	9
8.1.4. Exploring Mediational Relationships Between Possible Causes/Species and Behavioural Intentions Through Warmth/Competence	7
8.2. Theoretical and Practical Implications	8
8.2.1. Theoretical Implications	8
8.2.2. Practical Implications	0
8.3. Limitations of the Project and Associated Directions for Future Research	2
8.3.1. Limitations in Scope	2
8.3.2. Methodological Limitations	5
8.4. Overall Conclusions	7
References	9
Appendices	0
Appendix 1. Study 1 Full Measures	0
Appendix 2. Principal Axis Factor Analyses	2
Appendix 3. Study 1 Participant Advertisements	5
Appendix 4. Statistics Tables	8
Appendix 5. Study 2 Participant Advertisements	2
Appendix 6. Images for Self-Relevance Manipulation	5
Appendix 7. Study 3 Prolific Participant Advertisement	8
Appendix 8. Study 4 Participant Advertisements	9
Appendix 9. Study 5 Prolific Participant Advertisement	1
Appendix 10. Study 6 Participant Advertisements	2

Figure	Page Number	
Figure 1. The SCM stereotypes of animals per	5	
Sevillano and Fiske (2019b).		
Figure 2. Pre- and post-manipulation	146	
mediational models of the species X1 variable		
(pig vs. non-pig) on active help through		
warmth.		
<i>Figure 3</i> . Pre- and post-manipulation	147	
mediational models of the species X2 variable		
(tree kangaroo vs. non-tree kangaroo) on active		
help through warmth.		
Figure 4. Pre- and post-manipulation	148	
mediational models of the species X1 variable		
(pig vs. non-pig) on active harm through		
warmth.		
<i>Figure 5</i> . Pre- and post-manipulation	149	
mediational models of the species X2 variable		
(tree kangaroo vs. non-tree kangaroo) on active		
harm through warmth.		
<i>Figure 6.</i> Pre- and post-manipulation	150	
mediational models of the species X1 variable		
(pig vs. non-pig) on passive help through		
competence.		
<i>Figure 7.</i> Pre- and post-manipulation	151	
mediational models of the species X2 variable	101	
(tree kangaroo vs. non-tree kangaroo) on		
passive help through competence.		
<i>Figure 8</i> . Mediational model of similarity on	198	
active help through warmth.		
Figure 9. Mediational model of similarity on	251	
active help through warmth.		
Figure 10. Mediational model of similarity on	252	
passive help through competence.		
<i>Figure 11.</i> Pre- and post-manipulation	253	
mediational models of species on active help		
through warmth.		
<i>Figure 12</i> . Pre- and post-manipulation	254	
mediational models of species on active harm		
through warmth.		
<i>Figure 13</i> . Mediational model of pet status on	287	
post-manipulation active harm through post-	-	
manipulation warmth.		

List of Figures

Table	Page Numbers
Table 1. Descriptive statistics for effect of	39
species on warmth, competence, familiarity,	
similarity, profit status, and pet status.	
Table 2. Participants within each condition for	58
Study 2.	
<i>Table 3.</i> Descriptive statistics for effect of	70
behavioural self-relevance on post-manipulation	
dissonance, post-evaluation dissonance, and	
reported weekly meat consumption.	
<i>Table 4.</i> Descriptive statistics for effect of	71
subjective self-relevance on post-manipulation	
dissonance, post-evaluation dissonance, and	
reported weekly meat consumption.	
<i>Table 5</i> . Participants within each condition for	84
Pilot Study 1a.	
<i>Table 6</i> . Participants within each condition for	93
Pilot Study 2a	
Table 7 Participants within each condition for	122
Study 3	
Table 8 Descriptive statistics for interaction	138
effect of behavioural self-relevance and species	100
on measured behavioural self-relevance	
Table 9 Descriptive statistics for interaction	139
effect of behavioural self-relevance and	137
subjective self-relevance on measured	
behavioural self-relevance	
Table 10 Descriptive statistics for effect of	143
species on warmth competence active help	175
active harm passive help passive harm	
Table 11 Descriptive statistics for interaction	144
effect of behavioural self-relevance and species	177
on discomfort	
Table 12 Participants within each condition for	163
Pilot Study 1h	105
Table 13 Descriptive statistics for interaction	170
affect of familiarity and similarity	170
manipulations on measured familiarity	
Table 14 Descriptive statistics for interaction	171
offoot of familiarity and similarity	1/1
monipulations on monopured similarity	
Table 15 Douticinents within each condition for	179
<i>Tuble 15.</i> Participants within each condition for	178
Study 4. Table 16 Descriptive statistics for effect of	100
similarity manipulation on massured similarity	190
summarity manipulation on measured similarity	
valiables. Table 17 Descriptive statistics for interaction	102
affoot of familiarity/similarity on dition and	192
effect of familiarity/similarity condition and	
time on warmth.	

List of Tables

Table 18. Descriptive statistics for interaction	193
effect of familiarity/similarity condition and	
time on competence.	
Table 19. Descriptive statistics for interaction	194
effect of familiarity/similarity condition and	
time on active help.	
Table 20. Descriptive statistics for interaction	195
effect of familiarity/similarity condition and	
time on passive help.	
Table 21. Participants within each condition for	211
Pilot Study 2b.	
Table 22. Participants within each condition for	216
Pilot Study 3b.	
Table 23. Participants within each condition for	222
Pilot Study 4b.	
Table 24. Participants within each condition for	232
Study 5.	
Table 25. Descriptive statistics for interaction	241
effect of similarity condition and time on	
perceived similarity to humans.	
Table 26. Descriptive statistics for interaction	242
effect of similarity condition and time on	
perceived shared emotional traits with humans.	
Table 27. Descriptive statistics for interaction	244
effect of similarity condition and time on	
perceived shared cognitive traits with humans.	
Table 28. Descriptive statistics for interaction	245
effect of similarity condition and time on	
warmth.	
Table 29. Descriptive statistics for interaction	246
effect of similarity condition and time on	
competence.	
Table 30. Descriptive statistics for interaction	247
effect of similarity condition and time on active	
help.	
Table 31. Descriptive statistics for interaction	248
effect of similarity condition and time on active	
harm.	
Table 32. Descriptive statistics for effect of	249
species on similarity measures, warmth,	
competence, active help, passive help, active	
harm, and passive harm.	
Table 33. Participants within each condition for	265
Pilot Study 1c.	
Table 34. Participants within each condition for	273
Study 6.	
Table 35. Descriptive statistics for interaction	282
effect of pet status condition and time on	
warmth.	

Table 36 Descriptive statistics for interaction	283
effect of pet status condition and time on active	285
harm	
Table 37 Descriptive statistics for interaction	284
effect of pet status condition and time on	264
passive harm	
Table 11 Information statistics for Study 1	358 360
multiple regressions on dog warmth dog	338-300
competence nig warmth and nig competence	
Table 42 Inferential statistics for Study 2 2x2x2	360 361
MANCOVA on SCM/BIAS man variables	500-501
Table 43 Inferential statistics for Study 2.2x2	362
MANCOVA on reported weekly meat	502
consumption post-manipulation dissonance and	
nost-evaluation dissonance	
Table 44 Inferential statistics for Study 3 3x?	363
ANCOVA for subjective self-relevance	505
manipulation check	
Table 45 Inferential statistics for Study 3 3x2x2	363-366
MANCOVA on SCM/BIAS man variables	505 500
reported weekly meat consumption discomfort	
and behavioural self-relevance	
Table A6 Inferential statistics for Study 4 one-	367
way ANCOVA for familiarity manipulation	507
check	
<i>Table A7</i> . Inferential statistics for Study 4 one-	367
way MANCOVA for similarity manipulation	
checks	
<i>Table A8.</i> Inferential statistics for Study 4 4x2	368
MANCOVA on SCM/BIAS map variables	
<i>Table A9.</i> Inferential statistics for Study 5 2x2x2	369-370
MANCOVA on SCM/BIAS map variables and	
similarity manipulation checks	
Table A10. Inferential statistics for Study 6 one-	370
way ANCOVA for pet status manipulation	
check	
Table A11. Inferential statistics for Study 6 2x2	371
MANCOVA on SCM/BIAS map variables	

Notation

BIAS map = the Behaviours from Intergroup Affect and Stereotypes map

- DV = dependent variable
- IV = independent variable
- PAFA = principal axis factor analysis
- RQ = research question
- SCM = the Stereotype Content Model

Chronology of Events

Studies within this thesis were conducted in the following order:

- Study 1: October and November 2019
- Study 2: April and May 2020
- Study 4: January and February 2021
- Study 3: November 2021
- Study 6: March 2021 and November 2021
- Study 5: November and December 2021

List of Appendices

Appendix	Page Numbers
Appendix 1. Study 1 Full Measures	350-351
Appendix 2. Principal Axis Factor Analyses	352-354
Appendix 3. Study 1 Participant Advertisements	355-357
Appendix 4. Statistics Tables	358-371
Appendix 5. Study 2 Participant Advertisements	372-374
Appendix 6. Images for Self-Relevance	375-377
Manipulation	
Appendix 7. Study 3 Prolific Participant	378
Advertisement	
Appendix 8. Study 4 Participant Advertisements	379-380
Appendix 9. Study 5 Prolific Participant	381
Advertisement	
Appendix 10. Study 6 Participant	382-384
Advertisements	

Third Party Copyright Declaration

The Effect of Species on Perceptions of Animals: Predictors and Causes of Pet Speciesism

Sarah May Gradidge

Attention is drawn to the fact that copyright of this thesis rests with

- (i) Anglia Ruskin University for one year and thereafter with
- (ii) Sarah May Gradidge

This copy of the thesis has been supplied on condition that anyone who consults it is bound by copyright.

This work may:

- (i) be made available for consultation with Anglia Ruskin University Library, or
- be lent to other libraries for the purpose of consultation or may be photocopied for such purposes
- (iii) be made available in Anglia Ruskin University's repository and made available on open access worldwide for non-commercial educational purposes, for an indefinite period.

Chapter 1. Introduction

1.1. Dog vs. Pig Pet Speciesism

Differences in how dogs and pigs are treated within countries such as the United Kingdom are stark: Approximately 34% of households in the United Kingdom have a pet dog (Bedford, 2022) and dogs are commonly called 'man's best friend', yet hundreds of thousands of pigs are slaughtered each month for meat in the United Kingdom (e.g., 941,000 in June 2022 alone; Department for Environment, Food & Rural Affairs [DEFRA], 2022). These stark differences in how dogs and pigs are treated occur despite their multiple shared attributes: For example, pigs show comparable skill to dogs at cognitive abilities like spatial navigation and memory (Arts et al., 2009; Smith et al., 2021), learning (Broom et al., 2009; Fukuzawa & Igarishi, 2017), and recognition of individual conspecifics (McLeman et al., 2005, 2008; Racca et al., 2010). Both animals reside in socially intricate groups (Marino & Colvin, 2015), and engage in social play (Horback, 2014; Ward et al., 2008). Dogs and pigs also demonstrate similar emotionality, especially within a social context (Csoltova & Mehinagic, 2020; Reimert et al., 2013, 2015). Additionally, dogs and pigs share similar appearances, with both animals being quadruped mammals of relatively similar size. The vast differences in treatment of dogs and pigs within countries such as the United Kingdom, despite dogs' and pigs' multiple shared characteristics, may be arising from *prejudice*: Unjustified and unfair bias against, or in favour of, certain groups. Prejudice against or towards animals due to their taxonomic classification is termed speciesism (Singer, 1990), which takes two main forms: anthropocentric speciesism (prejudice against all animals in favour of humans) and pet speciesism (Caviola & Capraro, 2020), with the latter being the focus of this thesis.

Pet speciesism constitutes any form of prejudice in favour of 'pet' animals like dogs or cats, and against 'non-pet' animals like pigs or cows. Pet speciesism can therefore be

measured by comparing any pet animal to any non-pet animal (e.g., cat vs. pig, dog vs. cow). However, dogs and pigs represent a natural 'matched pair' to investigate pet speciesism, as these two species share multiple similarities in cognition, appearance, behaviour, emotionality, and sociality as seen above, yet receive highly differential treatment within countries like the United Kingdom. Additionally, dog vs. pig pet speciesism has already been evidenced across multiple psychological domains, including liking for dogs over pigs (Caviola & Capraro, 2020), viewing dogs as being more capable of feeling emotions than pigs (Bilewicz et al., 2011), feeling greater empathy towards and willingness to help dog (vs. pig) victims (Gradidge et al., 2021b), and expressing less *victim derogation* (negative perceptions of a victim) and less *second-hand forgiveness* (forgiveness for the perpetrator who harmed a victim) towards dog (vs. pig) victims (Gradidge et al., 2021b). I therefore focus on dog vs. pig pet speciesism (hereon pet speciesism) throughout this thesis.

Pet speciesism holds extremely negative consequences for pigs, including loss of their lives to deliberate human activities such as meat production, and severe negative welfare implications within meat production, such as farrowing crates (McCulloch, 2022) and tail docking (De Briyne et al., 2018). Additionally, meat consumption (including from pigs) holds wider global negative consequences by significantly contributing to climate change (González et al., 2020), such as through greenhouse gas emissions (Godfray et al., 2018). Climate change in turn has disastrous direct and indirect consequences for human welfare. Directly, climate change is leading and will lead to increasingly frequent and intense natural disasters like extreme flooding, storms, and heatwaves (Intergovernmental Panel on Climate Change [IPCC], 2021; Philip et al., 2021). Climate change has been projected to cause up to 83 million deaths at 4.1°C of global warming above pre-industrial levels or nine million deaths at 2.4°C by 2100 (Bressler, 2021), and climate change will inevitably create 'climate refugees' (i.e., people displaced from their homes due to natural disasters; Berchin et al.,

2017; Islam & Khan, 2018). Indirectly, climate change will harm the welfare of wild animals, such as through biodiversity loss and ecosystem collapse (Canadell & Jackson, 2021; Machovina et al., 2015), and by compounding negative consequences of habitat loss (Mantyka-Pringle et al., 2012). Negative effects of climate change on wild animals (e.g., loss of biodiversity) in turn harms humans, such as by increasing the risk of zoonotic diseases to humans (Keesing & Ostfeld, 2021), and thereby causing human deaths and increasing the risk of future pandemics. Reducing meat consumption is thus crucial for meeting global climate targets (Springmann et al., 2018).

One tool for reducing (pig) meat consumption, and thereby protecting human, pig, and wild animal welfare, is to improve perceptions of pigs by targeting causes of pet speciesism. That is, by identifying exactly why pet speciesism occurs, interventions can be developed to remove these causes, and thus reduce or prevent pet speciesism. Alongside possibly reducing (pig) meat consumption, this approach may have broader positive effects for pigs, such as protections within scientific research, especially amid growing interests in organ transplantation from pigs (Koplin & Wilkinson, 2019; Shah & Han, 2022), policymakers implementing policies to better protect pig welfare (e.g., eliminating farrowing crates), and the public advocating for pig welfare (Garner, 1995, 2008; Munro, 2005). However, previous research has not yet elucidated exactly why pet speciesism occurs, limiting the ability of interventions to reduce pet speciesism and thereby improve perceptions of pigs. Considering limited previous research, and the urgent and crucial need to reduce pet speciesism, this thesis thus tests possible causes of pet speciesism and is designed to inform future interventions to reduce or prevent pet speciesism. Before discussing possible causes of pet speciesism, I first discuss how this thesis operationalises pet speciesism.

1.2. Framework for Measuring Pet Speciesism

Pet speciesism is prejudice towards certain species of animals ('pet' animals like dogs) and against other species ('non-pet' animals like pigs; Caviola & Capraro, 2020). Pet speciesism can be measured across any psychological dimension, including liking (Caviola & Capraro, 2020), emotional attribution (Bilewicz et al., 2011), and empathy (Gradidge et al., 2021b). However, a robust and consistently used measure of pet speciesism has not yet been developed, which undermines investigations of pet speciesism. Social psychological research provides consistent and robust measurements of peoples' perceptions of others through the Stereotype Content Model (SCM; Fiske, 1998; Fiske, et al., 1999). Importantly, this model applies to animals (Sevillano & Fiske, 2016a, 2016b, 2019), and thus represents an appropriate framework to measure pet speciesism. The SCM proposes peoples' perceptions of others (human or non-human) consist of two dimensions: warmth (how much another being's intent is perceived as being positive) and *competence* (how much another being is perceived as being capable of enacting intent; Fiske, 1998; Fiske et al., 1999). Applied to animals, warmth is a perceived tendency of an animal towards friendly or aggressive intent, whilst perceptions of competence can be informed by an animal's friendly (e.g., wagging tail) or aggressive (e.g., biting) behaviours or capabilities (Sevillano & Fiske, 2016a, 2016b).

The SCM, as applied to humans, indicates stereotypes are formed from warmth and competence judgements, creating four stereotypes: *contemptuous* (low warmth, low competence; e.g., people in poverty), *envious* (low warmth, high competence; e.g., wealthy people), *paternalistic* (high warmth, low competence; e.g., elderly people), and *admired* (high warmth, high competence; e.g., *ingroup* members, whereby ingroups are groups to which a person belongs and/or feels identification with; Fiske et al., 2002). Similar stereotypes apply to animals: '*pests*'/*contemptible stereotype* (low warmth, low competence, e.g., rats), '*predators*'/*threatening-awe stereotype* (low warmth, high competence, e.g.,

tigers), '*prey*'/*subordination stereotype* (moderate warmth, low competence, e.g., pigs), and '*companions*'/*protective stereotype* (high warmth, high competence, e.g., dogs; Sevillano & Fiske, 2016b; Figure 1). Low warmth and/or competence of a being denotes prejudice against them, whilst high warmth and competence indicates prejudice in favour of them.



Figure 1. The SCM stereotypes of animals per Sevillano and Fiske (2019b).

The SCM represents an appropriate, robust and consistent measurement of pet speciesism for three reasons: a) the model applies to animals (Sevillano & Fiske, 2016b, 2019), b) established measures from the model have adequate reliability and validity (Diamantopoulos et al., 2017; Sevillano & Fiske, 2016b), and c) the model extends to behavioural intentions through the *Behaviours from Intergroup Affect and Stereotypes (BIAS) map* (Cuddy et al., 2007), whereby warmth and competence evaluations of a being inform active and passive behavioural intentions respectively towards that being. Specifically, 'warm' people or animals are typically subject to greater *active help* (intentional and effortful aid) and less *active harm* (intentional and effortful harm; Cuddy et al., 2007; Sevillano & Fiske, 2016b, 2019). Meanwhile, 'competent' people or animals are typically subject to

greater *passive help* (aid which requires minimal effort) and less *passive harm* (exclusionary harmful behaviour which requires minimal effort; Cuddy et al., 2007; Sevillano & Fiske, 2016b, 2019).¹ Combined implementation of the SCM/BIAS map within this thesis thus enables pet speciesism gaps in perceptions of dogs vs. pigs (warmth, competence), and behavioural intentions towards dogs vs. pigs (active help, passive help, active harm, passive harm), to be explained.

Having established the psychological framework utilised to measure pet speciesism, I now consider possible causes of pet speciesism.

1.3. Causes of Pet Speciesism

Humans' perceptions of animals, including dogs and pigs, likely arise from two main sources: '*intrinsic*' factors referring to variables inherent to an animal (e.g., appearance, behaviour), and '*extrinsic*' factors referring to variables imposed onto an animal by humans (e.g., familiarity with an animal; Serpell, 2004). Research on intrinsic factors which inform people's positive perceptions of dogs is abundant. Dogs are viewed positively due to intrinsic factors such as their appearance (Archer & Monton, 2011; Kaminski et al., 2019), their evolved behaviour towards humans (e.g., responsiveness; Bray et al., 2021; Johnston et al., 2017; Pérez Fraga et al., 2021), and their innate sociability, predating to their wolf ancestors (Cordoni & Pelagi, 2019). Whilst dogs and pigs share positive intrinsic factors (e.g., Marino & Colvin, 2015), like their high sociability (Cooper et al., 2003; Podgórski et al., 2014), which should inform positive perceptions of both animals, dogs are superior to pigs in some

¹Warmth can also sometimes inform passive behavioural intentions (i.e., greater warmth increasing passive help and reducing passive harm; Cuddy et al., 2007; Sevillano & Fiske, 2019), whilst competence can sometimes also inform active behavioural intentions (i.e., greater competence increasing active help and reducing active harm; Sevillano & Fiske, 2019). These findings occur especially for ambivalent groups (e.g., animals which are deemed highly warm but low in competence or vice versa; Sevillano & Fiske, 2019). However, these findings are not always replicated (e.g., Sevillano & Fiske, 2016b).

intrinsic factors, such as their responsiveness to humans (Gerencsér et al., 2019; Pérez Fraga et al., 2021). Thus, intrinsic factors are likely contributing at least partly to pet speciesism.

Yet, only focussing on intrinsic factors raises two main issues. Firstly, intrinsic factors are inflexible: Changing intrinsic factors in interventions is either impossible or raises significant ethical issues. For example, intrinsic factors, like appearance and behaviour, can technically be amended through human-controlled selective breeding of animals, but this breeding raises ethical concerns (e.g., brachycephalic dog breeds; O'Neill et al., 2020; Packer et al., 2019) and requires generations of animals to implement the change. Additionally, selective breeding for intrinsic factors deemed positive by humans is not beneficial for improving perceptions of non-domesticated animals (e.g., animals outside the remit of human-controlled selective breeding) and may reinforce the superiority of certain intrinsic attributes over others. Conversely, extrinsic factors are more conducive to change in interventions as these factors are not inherent to species. For example, if pigs are viewed negatively because people are not as familiar with pigs as they are with dogs, then increasing familiarity by interacting with pigs would represent a feasible and ethical intervention to reduce pet speciesism.

Secondly, peoples' perceptions of other humans are frequently informed by social psychological (extrinsic) factors, like perceived similarity to the self and familiarity with the person (e.g., Pettigrew & Tropp, 2006; Rodríguez-Pérez et al., 2011). These social psychological cues also likely apply to perceptions of animals and to pet speciesism specifically. For instance, dogs are not universally deemed companion animals across cultures and history: Dogs have been and/or are currently consumed as meat within many cultures around the world (e.g., China and Hong Kong, Hurley, 2016, Li et al., 2017, Poon, 2014; some Native American groups, Roberts, 2017; South Korea, Oh & Jackson, 2011, Podberscek, 2009; Spain, Vallejo et al., 2017; the Yup'ik people of Alaska, McManus-Fry et

al., 2018), and killing and consumption of dogs is commonplace within some contemporary societies (Gray & Young, 2011). Historically, dogs have been kept for functions like protection and hunting instead of companionship (Herzog, 2014), alongside being used as a source of meat, such as during the Bronze Age (Price et al., 2020), and Iron Age (Horard-Herbin et al., 2018). Pigs are also not universally deemed 'food' across cultures: For instance, pigs are kept as pets in some contemporary societies (Gray & Young, 2011), including the Aymara people from South America, Ekari people from Indonesia, Hokkien Taiwanese people, Trobriand Islanders from Papua New Guinea, and Tucano people from South America (P. Gray, personal communication, June 28, 2022).

There are also within-culture differences and deviations in perceptions of dogs and pigs. For example, despite dogs' exalted status as 'man's best friend', an estimated 14% of people within the United Kingdom are either a little (11%) or a lot (3%) afraid of dogs (YouGov, 2014). Abuse against dogs is still prevalent, with over 92,000 reports of abuse against dogs received by the RSPCA in 2021 (RSPCA, 2022). Additionally, despite pigs' widespread usage in meat production within countries like the United Kingdom and United States, keeping 'micro-pigs' as pets still arose as a trend within these countries (Curnutte, 2014).

Thus, despite intrinsic factors working in favour of dogs and against pigs, like dogs' greater responsiveness to humans (Pérez Fraga et al., 2021), perceptions of dogs and pigs are not fixed and can vary across cultures, history, and individuals. Therefore, dogs being deemed companion animals and pigs being deemed 'food' (pet speciesism) is not a default position informed solely by dogs' and pigs' intrinsic characteristics. Instead, where pet speciesism does occur, it is likely to arise also from extrinsic factors, such as social learning that dogs are 'pets' (*pet status*) and pigs are 'food' (*profit status*), learnt personal habits and investment in consuming pig meat (*behavioural self-relevance* and *subjective self-relevance*),

and personal experiences with dogs and pigs (e.g., *familiarity* with each animal and how much each animal is deemed *similar* to humans). Indeed, anthropocentric speciesism (prejudice against all animals in favour of humans) is seemingly learnt during childhood or adolescence, as five-to-nine-year-old children exhibit far lower anthropocentric speciesism than adults (Wilks et al., 2021). Although such learning has not yet been evidenced for pet speciesism specifically, this research indicates people learn their perceptions of animals through culture, and thus suggests learnt extrinsic factors inform pet speciesism. Overall, incorporating extrinsic factors into explaining pet speciesism allows for a broader explanation of why pet speciesism is occurring, accounts for cultural, historical, and individual diversity in perceptions of dogs and pigs, and allows for the development of feasible interventions to reduce or prevent pet speciesism.

Despite the benefits of exploring extrinsic factors, these factors have not yet been applied to pet speciesism. Thus, for the first time, this thesis explores the following extrinsic factors which may cause pet speciesism, with a view to informing future research which develops interventions to reduce or prevent pet speciesism: *behavioural self-relevance* of pigs (whether or how much someone consumes pig meat), *subjective self-relevance* of pigs (whether or how much someone enjoys consuming pig meat), *familiarity* (quantity and/or quality of contact with an animal), *similarity to humans* (how much animals are viewed as possessing human-like characteristics), *pet status* (how much an animal is deemed a 'pet') and *profit status* (how much an animal is deemed 'profitable' for humans). These possible extrinsic causes have been selected from social psychological (familiarity, similarity) or speciesism (behavioural self-relevance, subjective self-relevance, pet status, profit status) literature bases, and have all previously been linked to perceptions of animals (self-relevance, Piazza & Loughnan, 2016; pet status and profit status, Signal et al., 2018; familiarity, Auger & Amiot, 2019b; similarity, Prguda & Neumann, 2014). Additionally, these variables capture

a range of extrinsic factors arising from different sources, such as social learning (pet status, profit status), learnt personal habits (behavioural self-relevance, subjective self-relevance) and personal experiences (familiarity, similarity). Research has either not yet tested whether these extrinsic factors causally inform perceptions of animals and/or research has not yet tested whether these extrinsic factors cause pet speciesism specifically. I discuss these extrinsic factors individually and in more detail below.

1.3.1. Self-Relevance

The term 'self-relevance' has traditionally been used to refer to whether or how much someone uses (e.g., consumes) an animal for personal benefit (e.g., for meat; Piazza & Loughnan, 2016). For instance, cows are self-relevant for people who consume meat from cows. Self-relevance has typically been theorised to cause negative perceptions of selfrelevant animals, like pigs (see Gradidge et al., 2021a for further detail). Specifically, most people wish to avoid harming pigs and other animals yet using pigs and other 'food' animals for personal benefit (e.g., consuming them as meat) harms them by default. This conflict is theorised to cause feelings of discomfort, termed 'cognitive dissonance' (hereon dissonance), which people are motivated to alleviate (Loughnan et al., 2014; Rothgerber, 2020). Thus, people typically choose either to stop consuming pigs and other animals and view the animals positively ('moral engagement'), or to continue consuming pigs and other animals and view the animals negatively ('moral disengagement'; Gradidge et al., 2021a; Loughnan et al., 2014). As most people continue to consume meat (e.g., 86-91% of the United Kingdom population; Johnson, 2022), many people morally disengage from pigs and other 'food' animals by viewing these animals negatively (Gradidge et al., 2021a). Therefore, (pig) meat consumers are motivated to view pigs negatively to alleviate their discomfort, termed 'motivated cognition' (Piazza & Loughnan, 2016). Self-relevance as a possible cause of pet speciesism would thereby operate through a) pigs being self-relevant to most people, as pigs

are regularly consumed as meat (e.g., DEFRA, 2022), and b) pigs' self-relevance causing meat consumers to view pigs as lacking warmth and competence, and to have more negative behavioural intentions towards them.

Previous research indicates self-relevance indeed contributes to negative perceptions of self-relevant animals. For instance, when participants consume dried beef (vs. dried nuts), they perceive cows as having lower *moral status* (how much an animal is deemed worthy of moral consideration) and feel less responsibility to have moral concern for animals (Loughnan et al., 2010). That is, making cows self-relevant (through consuming dried beef vs. not) causes cows to be viewed more negatively (through lower moral status and reduced moral concern for animals) as a way to reduce one's moral discomfort and justify one's consumption of cows.

Applied to pigs specifically, people typically ignore positive intelligence information about self-relevant animals (pigs) when forming their judgements of pigs' moral status, whilst not ignoring such intelligence information about non-self-relevant animals like fictional animals or tapirs (Piazza & Loughnan, 2016). Specifically, how intelligent (or not) an animal is presented as is important for informing the animal's perceived moral status, whereby more intelligent animals are deemed as having greater moral status, unless the animal is self-relevant. These findings indicate people are motivated not only to view selfrelevant animals negatively, but also to disregard positive information about self-relevant animals.

However, research exploring causal effects of self-relevance on perceptions of animals is limited, and, when self-relevance has been causally manipulated (e.g., Loughnan et al., 2010), it has not yet been applied to pigs nor to the SCM/BIAS map specifically. Furthermore, previous self-relevance literature has solely explored investment in meat

consumption through actual behaviour rather than investment in meat consumption through psychological attitudes towards meat (e.g., liking for meat). I thus differentiate here between traditional self-relevance (which I term '*behavioural self-relevance*'), describing behavioural investment in meat consumption (e.g., actual meat consumption), and a related but distinct possible form of self-relevance, which I term '*subjective self-relevance*', referring to psychological investment in meat consumption (e.g., liking for meat).

Regarding 'subjective self-relevance', previous research indicates liking meat is frequently associated with, and may therefore impact, perceptions of animals. For instance, meat consumption is often described as 'pleasurable' or 'nice' (Hyers, 2006; Macdiarmid et al., 2016; Piazza et al., 2015), and greater meat enjoyment is linked to greater denial of animals' ability to suffer (Monteiro et al., 2017). Whilst perceived 'niceness' of meat is typically employed as a strategy to justify meat consumption (Piazza et al., 2015), perceived 'niceness' may also enhance commitment to meat consumption, and thus necessitate stronger devaluing of self-relevant animals to alleviate stronger dissonance. Subjective self-relevance should thus work in an identical way to behavioural self-relevance due to motivated cognition: The greater psychological investment in meat (e.g., liking for meat) for a particular animal, the more negatively that animal should be viewed. Subjective self-relevance would therefore cause pet speciesism through a) pigs being subjectively self-relevant to most people, as most people enjoy consuming pig meat, and b) pigs' subjective self-relevance causing meat enjoyers to view pigs as lacking warmth and competence, and to have more negative behavioural intentions towards them. Behavioural and subjective self-relevance hypothetically share the same underlying motivational process, whereby investment in meat consumption motivates people to view meat animals negatively, yet behavioural and subjective self-relevance differ in their exact type of investment: Behavioural self-relevance is motivated by actual consumption of meat, whilst subjective self-relevance is motivated by

enjoyment of meat. Extending upon previous literature, this thesis uniquely aims to test, firstly, whether behavioural and subjective self-relevance are associated with pet speciesism (Study 1) and, secondly, whether behavioural and subjective self-relevance cause pet speciesism (Studies 2-3).

1.3.2. Familiarity

Familiarity refers to the amount (quantity) or perceived value (quality) of contact with another being (Auger & Amiot, 2016). Familiarity is important for positively informing peoples' perceptions of fellow humans (Reis et al., 2011), including through actual (Pettigrew & Tropp, 2006) and imagined (Vezzali et al., 2012) intergroup contact, whereby interacting with (or imagining interacting with) members of human *outgroups* (groups to which a person does not belong and/or feel identification with) improves feeling towards them. Even familiarity (vs. unfamiliarity) of human faces alone increases their likeability (Harmon-Jones & Allen, 2001), and people view familiar (vs. unfamiliar) faces as being happier (Carr et al., 2017; Claypool et al., 2007). Additionally, familiarity (vs. unfamiliarity) of a face enables better recognition of emotional expression on that face (Li et al., 2019). Such findings indicate people are biased in favour of familiar others (e.g., Zebrowitz et al., 2007). People are also biased against unfamiliar others (Zebrowitz et al., 2007), and greater familiarity with outgroups can reduce prejudice against them, including transgender people (Flores et al., 2018), and people of different ethnicities to one's own (Zebrowitz et al., 2008).

Positive effects of familiarity have been explained by multiple theories, including the *mere exposure effect* (Zajonc, 1968) and *intergroup contact theory* (Pettigrew, 1998; see Swami, 2021, for alternative theories). The mere exposure effect suggests familiarity enhances liking due to repeated exposure to a stimulus. This repeated exposure is theorised to enable easier recognition of the stimulus (*'perceptual fluency'*), and this positive experience of 'perceptual fluency' is erroneously attributed to liking for the stimulus (Bornstein &

D'Agostino, 1992, 1994). Alternatively, intergroup contact theory suggests familiarity increases liking as (positive) intergroup contact provides the opportunity to learn that the outgroup is not threatening and thereby reduces intergroup anxiety (Pettigrew & Tropp, 2008). Alongside decreasing negative affect, intergroup contact can increase positive affect (Tausch & Hewstone, 2010), subsequently enhancing empathy towards the outgroup (Pettigrew & Tropp, 2008).

In line with these theories, positive effects of familiarity have been extended to animals. For example, *imagined intergroup contact* with dogs and cows (i.e., imagining interacting with these animals) increases inclusiveness of animals into the self and encourages positive behavioural intentions towards animals (Auger & Amiot, 2019b). Greater contact with animals also bolsters identification with them (Auger & Amiot, 2016), alongside enhancing inclusiveness of animals within the self and lowering animal-related intergroup anxiety, thus predicting more positive attitudes and behavioural intentions towards animals as a whole and towards animal subgroups (Auger & Amiot, 2019a). Greater familiarity with animals has also been linked to viewing animals as capable of experiencing a wider range of emotions, both for animals as a whole and for specific species (Morris et al., 2012). In addition, greater familiarity with an animal is associated with more positive perceptions of the animal (Possidónio et al., 2019). Whilst this previous research indicates positive relationships between familiarity and perceptions of animals, this research has not yet been applied to pet speciesism specifically nor to the SCM/BIAS map. In addition, causal effects of familiarity have rarely been investigated, and, where positive causal effects have been determined (Auger & Amiot, 2019b), effects of familiarity have been tested on perceptions of all animals or perceptions of animal subgroups (e.g., 'farm' animals), but not yet perceptions of specific species.

Due to the high prevalence of dogs kept as pets within households (Bedford, 2022), combined with positive effects of familiarity on perceptions of animals (Auger & Amiot, 2019b), a) dogs should be deemed more familiar than pigs, and b) dogs' greater familiarity should inform their greater warmth and competence and more positive behavioural intentions towards them. This thesis uniquely tests these hypotheses by exploring, firstly, if familiarity is associated with pet speciesism (Study 1) and, secondly, if familiarity causes pet speciesism (Study 4).

1.3.3. Similarity to Humans

Similarity to humans is how much animals are perceived as possessing human-like characteristics (Amiot et al., 2020). Perceived similarity typically informs more positive perceptions of other humans in an interpersonal context (although see Swami, 2021 for exceptions). For instance, people typically prefer others they perceive as similar (vs. dissimilar) to themselves (Montoya et al., 2008). People also typically attribute perceived similar (vs. less similar) others as being more capable of experiencing secondary emotions, such as pride, nostalgia, or compassion (Rodríguez-Pérez et al., 2011). As secondary emotions are deemed more 'human-like' than primary emotions such as anger, joy, or fear (Demoulin et al., 2004; Leyens et al., 2001), greater attribution of secondary emotions due to similarity thus increases others' perceived humanity (Gaunt et al., 2002; Leyens et al., 2001). As attribution of secondary emotions increases pro-social behavioural intentions (Vaes et al., 2002), people may theoretically act more pro-socially towards similar (vs. dissimilar) others due to similar others being deemed capable of experiencing secondary emotions. Conversely, dissimilarity decreases others' perceived humanity (dehumanization), by denying dissimilar others the capacity to experience 'human-like' secondary emotions (Leyens et al., 2001). Indeed, peoples' preferences against or towards others may be driven by similarity and dissimilarity, whereby dissimilarity leads people to exclude dissimilar others from one's

social circle and similarity informs attraction within the remaining social circle (Byrne et al., 1986; see Swami, 2021).

As applied to animals, similarity improves perceptions of them: People consistently prefer (Batt, 2009; Kozachenko & Piazza, 2021) and empathise more with (Harrison & Hall, 2010; Miralles et al., 2019; Prguda & Neumann, 2014; Westbury & Neumann, 2008) animals which are more (vs. less) biologically and behaviourally similar to humans. Greater perceived similarity of an animal to humans is also linked to more positive perceptions of the animal (Possidónio et al., 2019). Additionally, emphasising similarities of animals to humans (vs. similarities of humans to animals) increases animals' perceived ability to feel sensations, which in turn increases moral inclusivity for animals (Bastian et al., 2012b). This greater moral inclusivity itself predicts lower speciesism and greater moral concern for the (animal) outgroup (Bastian et al., 2012b).

A detailed causal analysis found that describing animals as having (vs. lacking) mental characteristics, like emotions and cognitive capabilities, increased *mind attribution* towards animals (how much an animal is viewed as possessing a mind), with some characteristics influencing perceptions of animals more than others (e.g., spatial reasoning informing greater perceived capability for thinking; Leach et al., 2021). Whilst this research suggests highlighting mental capacities of animals, and thus making animals' shared characteristics with humans potentially salient, informs perceptions of animals, this research did not analyse (dis)similarity to humans specifically. Additionally, findings were not analysed in relation to individual species (e.g., dogs vs. pigs), limiting the ability to say how much similarity applies to pet speciesism specifically, and this research has not yet been applied to the SCM/BIAS map. Further, whilst this research measured perceived immorality of consuming the animal and feelings of guilt, alongside moral perceptions of animals (Leach

et al., 2021), the research did not assess behavioural intentions towards animals, thereby limiting generalisability to real-world behaviour.

Due to dogs' status as 'psychological kin' (Topolski et al., 2013), and frequent classification of pet animals as ingroup members (McConnell et al., 2019), a) dogs should be deemed more similar to humans than pigs are, and b) dogs' greater similarity should inform their greater warmth and competence and more positive behavioural intentions towards them. This thesis uniquely tests these hypotheses by exploring, firstly, if similarity is associated with pet speciesism (Study 1) and, secondly, if similarity causes pet speciesism (Studies 4-5).

1.3.4. Species Status

Pet speciesism may also arise from the status an animal holds within the society the animal resides in (*categorisation*'; Bratanova et al., 2011). Specifically, animals can be categorised as companion animals (*pet status*), animals used for human benefit (*profit status*), or animals which are annoying or dangerous to humans (*pest status*). Of relevance to pet speciesism are pet status and profit status. Overall, people empathise more with (Signal et al., 2018), have more positive attitudes towards (Hazel et al., 2011; Robbins et al., 2021; Signal et al., 2018; Taylor & Signal, 2009), perceive more emotions in (Wilkins et al., 2015), and attribute greater mind to (Robbins et al., 2021) 'pet' animals over 'profit' animals or 'pests'. Additionally, mere labelling of animals as 'edible' (vs. 'non-edible') or 'food' (vs. not 'food'), which both manipulate animals' profit status, negatively impact perceptions of animals, such as by reducing animals' perceived capacity to suffer (Bastian et al., 2012a; Bilewicz et al., 2016; Bratanova et al., 2011). Equivalent studies causally manipulating pet status have not yet been conducted, and profit status findings have not yet been applied to pet speciesism specifically. Additionally, neither pet nor profit status have been investigated utilising the SCM/BIAS map. Thus, this thesis uniquely extends previous findings by

causally manipulating pet and profit status to investigate pet speciesism using the SCM/BIAS map.

Due to the ubiquitous presence of dogs within households in countries such as the United Kingdom (Bedford, 2022), compared to pigs' inability to be legally classified as pets (DEFRA, 2014), a) dogs should be viewed as having greater pet status than pigs (López-Cepero et al., 2021), and b) dogs' greater pet status should inform their greater warmth and competence and more positive behavioural intentions towards them. Additionally, due to pigs' commonplace usage in meat production (DEFRA, 2022), compared to sales of dog meat being illegal (Ares & Sutherland, 2019), a) pigs should be viewed as having greater profit status than dogs (López-Cepero et al., 2021), and b) pigs' greater profit status should inform their lower warmth and competence and more negative behavioural intentions towards them. This thesis uniquely tests these hypotheses by exploring, firstly, whether pet and profit status are associated with pet speciesism (Study 1) and, secondly, whether pet status causes pet speciesism (Study 6). Study 6 did not assess causal effects of profit status due to lack of significant relationships between profit status and pet speciesism within Study 1.

1.4. Gaps in the Literature and Research Questions (RQs)

This thesis builds upon speciesism, social psychological, and human-animal interaction ('anthrozoological') literature to determine extrinsic causes of pet speciesism for the first time. Due to greater flexibility in altering extrinsic (vs. intrinsic) factors, determining extrinsic causes of pet speciesism has the potential to inform interventions to decrease or prevent pet speciesism, and thus to reduce (pig) meat consumption, improve pig welfare, and enhance advocacy for pigs. Yet, previous literature has focussed exclusively on intrinsic causes of pet speciesism (e.g., appearance; Archer & Monton, 2011), instead of extrinsic causes. Possible extrinsic causes include behavioural and subjective self-relevance, familiarity and similarity, pet status and profit status. Effects of these variables on

perceptions of animals have not yet been tested causally and/or these variables have not yet been applied to pet speciesism specifically or to the SCM/BIAS map. The gaps in literature lead to the following RQs: a) What variables are associated with pet speciesism? (Study 1) and b) What variables cause pet speciesism? (Studies 2-6). Therefore, this thesis aims, firstly, to determine which variables (behavioural self-relevance, subjective self-relevance, familiarity, similarity, pet status, profit status) are significantly associated with pet speciesism (Study 1) and, secondly, to explore which variables significantly cause pet speciesism (Studies 2-6). Specifically, the thesis tests causal effects of behavioural and subjective selfrelevance (Studies 2-3), familiarity and similarity (Studies 4-5), and pet status (Study 6), on perceptions of an unknown animal (to test if these variables causally impact perceptions of animals more generally) or dogs and pigs (to apply to pet speciesism specifically). Possible causal effects are tested only for variables identified as significant regressors of pet speciesism within Study 1. This thesis is designed to inform interventions to reduce or prevent pet speciesism, meaning variables must demonstrate a robust causal effect on behavioural intentions to effectively inform interventions. Robust causal effects are defined as those arising in variables which do not have floor or ceiling effects and variables which have adequate scale reliability. As such, possible causal effects on SCM/BIAS map perceptions of dogs and pigs are tested only for variables which are found to robustly inform behavioural intentions towards unknown animals.²

²An exception is made for variables which do not robustly inform behavioural intentions and yet also are not successfully manipulated (as determined by failed manipulation checks), whereby amended manipulations are employed to determine causal effects of these variables on perceptions of unknown animals, dogs, and pigs.
Chapter 2. Study 1: Exploring Regressors of Pet Speciesism³

2.1. Introduction

As discussed within Chapter 1, this thesis aims to determine why pet speciesism occurs in order to inform interventions to reduce or prevent pet speciesism, thereby decreasing (pig) meat consumption and protecting pig welfare. Whilst previous research has identified intrinsic causes of pet speciesism (e.g., responsiveness, Pérez Fraga et al., 2021), cultural, historical, and individual differences in perceptions of dogs and pigs indicate perceptions of these animals are also informed by extrinsic factors. Moreover, extrinsic factors are more flexible to change, and thus more easily manipulated in interventions. This study thereby forms the foundation for subsequent studies within this thesis by determining significant extrinsic regressors of pet speciesism, which are tested for causal roles in pet speciesism within Studies 2-6. Pet speciesism is measured within this study (and throughout subsequent studies) via dogs' and pigs' warmth and competence, whereby warmth and competence are measured through the warmth and competence subscales from Sevillano and Fiske (2016b), shortened and adapted from Fiske et al. (2002). Higher warmth and/or competence for dogs and pigs indicates more positive perceptions of that species.

Before exploring regressors of pet speciesism, I first aim to explore if a) pet speciesism is indeed evidenced in terms of warmth and competence, in line with the SCM (Sevillano & Fiske, 2016b), and; b) assumptions about dogs being deemed more familiar, more similar to humans, less as 'profit' animals, and more as 'pet' animals compared to pigs (discussed in Chapter 1 and below) are supported, whereby familiarity, similarity, pet status, and profit status are all measured through single items asking how familiar/similar/how much

³This study has been published in *People and Animals: The International Journal of Research and Practice* (*PAIJ*): Gradidge, S., Zawisza, M., Harvey, A. J., & McDermott, D. T. (2022). Farmyard animal or best friend? Exploring predictors of dog vs. pig pet speciesism. *People and Animals: The International Journal of Research and Practice*, *5*(1), 11. <u>https://docs.lib.purdue.edu/paij/vol5/iss1/11</u>

of a pet/how much of a 'profit' animal the participant considers dogs and pigs to be. First, following previous SCM research (Sevillano & Fiske, 2016b), I hypothesise that:

H1: Dogs will be deemed significantly warmer (**a**) and more competent (**b**) than pigs (i.e., pet speciesism will be evidenced).

Second, due to the high prevalence of dogs kept as pets within households (Bedford, 2022) and therefore high levels of day-to-day contact with dogs, I predict that:

H2: Dogs will be deemed significantly more familiar to humans than pigs are.

Third, due to dogs' status as 'psychological kin' (Topolski et al., 2013), and frequent classification of pet animals as ingroup members (McConnell et al., 2019), I hypothesise that:

H3: Dogs will be deemed significantly more similar to humans than pigs are.

Fourth, due to pigs' commonplace usage in meat production (DEFRA, 2022), compared to sales of dog meat being illegal (Ares & Sutherland, 2019), I predict that:

H4: Pigs will be viewed significantly more as profit animals than dogs.

Fifth, due to the ubiquitous presence of dogs within households in countries such as the United Kingdom (Bedford, 2022), compared to pigs' inability to be legally classified as pets (DEFRA, 2014), I hypothesise that:

H5: Dogs will be viewed significantly more as pet animals than pigs.

After exploring evidence for pet speciesism and the assumptions discussed above, this study will explore possible regressors of pet speciesism, whereby possible regressors are informed by social psychological and speciesism literature bases as discussed within Chapter 1 and below: behavioural self-relevance, subjective self-relevance, similarity, familiarity, pet status, and profit status. Subjective self-relevance is measured through the Product

Involvement Scale (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009) adapted to refer to pig products specifically. Behavioural self-relevance is measured through a single item asking how many pig products the participant consumes per week, whilst familiarity, similarity, pet status, and profit status are also all measured through single items asking how familiar/similar/how much of a pet/how much of a 'profit' animal the participant considers dogs and pigs to be.

In regard to regressors, this study aims to, firstly, determine whether consumption and enjoyment of pig meat (behavioural and subjective self-relevance) are associated with more negative perceptions of pigs, in line with previous self-relevance literature (Loughnan et al., 2010; Piazza & Loughnan, 2016). Specifically, due to cognitive dissonance and motivated cognition (Loughnan et al., 2010; Piazza & Loughnan, 2016; Rothgerber, 2020), people may be motivated to view pigs negatively (in terms of warmth and competence) when they consume pigs (behavioural self-relevance) and/or when they like consuming pigs (subjective self-relevance), as devaluing of pigs justifies them being consumed. Before exploring the above causally in later studies, this study thus aims to first explore any correlational relationship between behavioural and subjective self-relevance of pigs and pigs' warmth and competence, hypothesising that:

H6: Greater behavioural (**a**) and subjective (**b**) self-relevance will be significantly associated with lower warmth and competence of pigs.

Secondly, this study aims to investigate whether familiarity with, and perceived similarity of, dogs and pigs are associated with more positive perceptions of both species, expanding upon previous social psychological literature (Auger & Amiot, 2019b; Leach et al., 2021). Specifically, people typically prefer animals with which they are more familiar (e.g., through imagined contact; Auger & Amiot, 2019b). Research on familiarity with human

outgroups suggests positive effects of familiarity through intergroup contact operate through reducing negative affect (e.g., intergroup anxiety; Pettigrew & Tropp, 2008), and increasing positive affect (Tausch & Hewstone, 2010), towards the outgroup. Therefore, presumed greater day-to-day contact and familiarity with dogs (see H2 above) may enhance positive affect, and reduce negative affect, towards dogs, thus improving dogs' perceived warmth and competence (compared to pigs). Regarding similarity, animals are viewed more positively when they are deemed similar to humans (vs. not; Batt, 2009; Kozachenko & Piazza, 2021; Possidónio et al., 2019), or when described as having 'human-like' mental characteristics (vs. not; Leach et al., 2021). Such similarity to humans enables 'humanisation' of animals by encouraging inclusion of animals within our moral circle, and thereby enhances moral concern for them (Bastian et al., 2012): As such, positive effects of similarity may extend to the SCM, whereby an animals' greater perceived similarity to humans (e.g., dogs compared to pigs; see H3 above) would theoretically inform the animal's greater perceived warmth and competence. Before exploring the above causally in later studies, this study aims to explore first any correlational relationship between familiarity/similarity of dogs and pigs, and their warmth and competence, predicting that:

H7: Greater familiarity with either dogs (a) or pigs (b) will be significantly associated with greater warmth and competence of that species.

H8: Greater similarity with either dogs (**a**) or pigs (**b**) will be significantly associated with greater warmth and competence of that species.

Thirdly, this study explores whether the degree to which dogs and pigs are deemed companion animals (pet status), or as animals used for human benefit (profit status), are associated with more positive or negative perceptions respectively of dogs and pigs, corroborating previous species status literature (Signal et al., 2018; Taylor & Signal, 2009).

Specifically, people view pet animals more positively than profit animals (Signal et al., 2018; Taylor & Signal, 2009), whereby the simple labelling of an animal as a 'pet' engenders positive perceptions of the animal, and the simple labelling of an animal as a 'profit' animal engenders negative perceptions of the animal. These findings are underpinned by 'categorisation' theory, whereby the mere categorisation of an animal into a group (i.e., 'pet' or 'profit') informs how people perceive the animal (Bratanova et al., 2011). As such, presumed labelling of pigs as 'profit' animals (see **H4** above) and dogs as 'pets' (see **H5** above) may explain why pigs are viewed as less warm and competent compared to dogs. Before exploring the above causally in a later study, this study aims to first explore any correlational relationship between pet/profit status of dogs and pigs, and their warmth and competence, hypothesising that:

H9: Greater pet status of either dogs (a) or pigs (b) will be significantly associated with greater warmth and competence of that species.

H10: Greater profit status of either dogs (**a**) or pigs (**b**) will be significantly associated with lower warmth and competence of that species.

Finally, this study aims to assess if individual differences variables *(support for animal utility, empathy towards animals)* are associated with perceptions of dogs and pigs, and therefore whether they may moderate effects of pet speciesism. Specifically, *support for animal utility* refers to how much people morally approve of using animals for human benefit (Kendall et al., 2006), whilst *empathy towards animals* refers to how much someone feels concern and care for animals; Powell, 2010). Whilst there are alternative individual differences variables which likely impact and/or inform pet speciesism (e.g., *social dominance orientation*; Dhont et al., 2014), these two specific variables were selected from extant speciesism and human-animal interaction literature for three main reasons: a) these

variables focus on perceptions of animals specifically, rather than broader prejudicial attitudes and/or perceptions of hierarchy, which do not reflect animal-directed prejudice specifically, b) they capture dichotomous and opposite perceptions of animals, with support for animal utility measuring negative, domineering perceptions of animals and empathy towards animals measuring positive, caring perceptions of animals, and c) they bridge the gap between anthropocentric and pet speciesism by exploring how broader perceptions of animals' inferiority or equality to humans (anthropocentric speciesism) apply to perceptions of different types of species (pet speciesism). Additionally, both individual differences variables may moderate pet speciesism: For example, individuals with a stronger belief in humans being superior to other animals view animals more negatively (Monteiro et al., 2017), especially 'food' animals (Krings et al., 2021), and may thus exhibit greater pet speciesism. In contrast, empathy towards animals correlates with more positive views of animals (Apostol et al., 2013; Hills, 1995; Signal & Taylor, 2007; Taylor & Signal, 2005), and may thus result in lower pet speciesism. Neither support for animal utility nor empathy towards animals have yet been applied to perceptions of dogs and pigs, pet speciesism, nor to the SCM/BIAS map. The current study therefore uniquely tests if empathy towards animals and support for animal utility are associated with pet speciesism, whereby empathy towards animals and support for animal utility are both measured through scales as follows respectively: Empathy Towards Animals Scale (Powell, 2010, adapted from Davis, 1980, 1983) consisting of two subscales (empathic concern and perspective-taking), and Animal Utility Scale (Kendall et al., 2006). Based on the previously discussed previous research, I predict that:

H11: Greater empathy for animals will be significantly associated with greater warmth and competence of dogs and pigs.

H12: Greater support for animal utility will be significantly associated with lower warmth and competence of dogs and pigs.

2.2. Methods

2.2.1. Participants

Two-hundred-and-seventy-six participants were recruited via volunteer sampling. Thirty-nine participants were excluded for providing partial data only (e.g., not completing all scales), whilst another five participants were excluded for failing the attention check. The regressions included diet and gender as covariates, as both variables influence perceptions of animals (Apostol et al., 2013; Herzog, 2007; Lund et al., 2016; Rothgerber, 2015; see Section 2.3 for further discussion). To dummy code diet into non-meat consumer (coded as zero, including vegans, vegetarians, and pescatarians; n=74) and meat consumer (coded as one; n=138), 12 participants were excluded in total due to being flexitarians (n=5), indicating they would rather not say (n=3), not consuming pig meat (n=2), providing an 'other' response (*n*=1), or giving no response (*n*=1). To dummy code gender into women (coded as zero; n=172) and men (coded as one; n=40), eight participants were excluded due to indicating they would rather not say (n=4), being non-binary (n=3), or providing an 'other' response (*n*=1). Exclusions left a sample size of 212 for the regressions (81.1% women, 18.9% men; M_{age} =28.71, SD_{age} =10.93, age range=18-66). Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For dietary group: 65.1% (meat consumer), 13.7% (vegan), 12.7% (vegetarian), and 8.5% (pescatarian). For nationality: 47.2% (British), 12.3% (United States), 4.7% (Malaysian), 3.8% (Portuguese), 2.8% each (French, German), and 26.4% other. For ethnicity: 75.7% (White), 13.3% (Asian), 5.8% (mixed), 3.3% (Black), and 1.9% other. For country of residence ($M_{duration} = 21.82$, $SD_{duration} = 14.69$, range: 1 month to

66 years): 59% (United Kingdom), 15.1% (United States), 4.2% (Malaysia), 2.4% each (Australia, France), and 16.9% other.

The sample size of 212 exceeds the minimum required sample size of 131 per an *a priori* power analysis conducted via G*Power for a multiple regression analysis ('fixed model, R^2 deviation from zero', medium effect size of $f^2 = .15$, 13 regressors, $\alpha = .05$, power = .8). A medium effect size has been chosen throughout this thesis for *a priori* power analyses, due to previous pet speciesism findings being typically of medium size (Gradidge et al., 2021b).

2.2.2. Design

This study follows a regression design with 13 regressors (behavioural and subjective self-relevance of pigs, familiarity, similarity, pet status and profit status of dogs and pigs, empathic concern towards animals, perspective taking of animals, support for animal utility), and four outcome variables (dogs' warmth, dogs' competence, pigs' warmth, pigs' competence). Perceptions of dogs and pigs were analysed separately to determine if perceptions of animals differ across species. Diet and gender were included as covariates for the regressions.

2.2.3. Materials

2.2.3.1. Empathy Towards Animals (Powell, 2010, adapted from Davis, 1980, 1983).

Empathy towards animals was measured via the Empathy Towards Animals Scale (Powell, 2010; adapted from the Interpersonal Reactivity Index; Davis, 1980, 1983; Appendix 1a), whereby participants answered 12 items on a Likert scale from one (*'not at all'*) to five (*'very much'*). A sample item is *'I often have tender, concerned feelings for animals who suffer misfortune'*. Statements two, four, five, and ten (e.g., *'I sometimes find it*

difficult to see things from an animal's point of view') were reverse-scored. The Interpersonal Reactivity Index has adequate test-retest reliability (r=.61-81; Davis, 1980), and adequate convergent and discriminant validity (Davis, 1983). The scale consists of two subscales: empathic concern (α s=.68-.73; Davis, 1980) and perspective-taking (α s=.71-.75; Davis, 1980). Thus, items one to seven were summed to calculate an 'empathic concern' score, and items eight to 12 were summed to calculate a 'perspective taking' score, with higher scores on either subscale indicating higher levels of that variable. Reliability was adequate within this study (empathic concern: ω =.82; 95% BCa CI [.74, .87]; perspective-taking: ω =.84; 95% BCa CI [.79, .88]), whereby adequate reliability is interpreted throughout this thesis as $\omega \ge .7$ (Dunn et al., 2014). Hierarchical ω is reported throughout this thesis due to concerns with Cronbach's α , such as its assumption that items are perfectly correlated (Dunn et al., 2014), and α being easily impacted by the number of scale items (Vaske et al., 2017). Additionally, hierarchical ω is used instead of coefficient ω , as hierarchical ω allows for correlations between errors (Kelley & Pornprasertmanit, 2016). ω and its associated bootstrapped 95% confidence interval (1,000 bootstrap samples) were calculated through the R MBESS package (Kelley, 2022).

2.2.3.2. Support for Animal Utility (Kendall et al., 2006).

Support for animal utility was measured via the Animal Utility Scale (Kendall et al., 2006; α =.65 from Kendall et al., 2006; Appendix 1b), whereby participants answered three items on a Likert scale from one ('*strongly disagree*') to seven ('*strongly agree*'). Items were summed to provide a support for animal utility score, with higher scores indicating greater support for animal utility. A sample item is '*It is acceptable to use animals to test consumer products such as soaps, cosmetics, and household cleaners*'. No items were reverse-scored. The scale has adequate validity (Cembalo et al., 2016). However, reliability was inadequate within this study (ω =.58; 95% BCa CI [.45, .68]). I therefore ran a principal axis factor

analysis (PAFA; see Appendix 2a for further detail) with direct oblimin rotation to assess factorial validity, alongside assessing inter-item correlations to further test reliability, and communality values to test common variance. Throughout this thesis, communality values, factor loadings, and inter-item correlations were considered conjointly when deciding whether to retain or remove an item from a scale. All three items adequately loaded onto one factor (\geq .45, whereby adequate loading is \geq .4; Field, 2018), indicating acceptable factorial validity. Communality was adequate for the second item (.66; whereby values should ideally be \geq .4; Tabachnick & Fidell, 2019), but lower for the first (.2) and third (.28) items. However, the first and third items had acceptable factor loadings, and communality values can still be considered adequate (though not ideal) if \geq .2 (Child, 2006). Additionally, whilst item one and item three correlated inadequately with each other, *r*=.24 (whereby inadequate inter-item correlation is $r \leq$.3; Field, 2018), both items correlated adequately with item two, $rs \geq$.36. I therefore retained all items. However, findings with this scale should be interpreted with caution due to inadequate reliability.

2.2.3.3. Perceived Familiarity of Dogs and Pigs.

Perceived familiarity of dogs and pigs was measured by a single item developed by the researcher '*How familiar do you perceive the following animals [dogs/pigs] to be to you?*' on a Likert scale from one ('*not at all*') to five ('*very much*'). A higher score indicates greater familiarity. This item was informed by previous research, which has measured familiarity of animals using a similar single item (Possidónio et al., 2019).

2.2.3.4. Perceived Similarity of Dogs and Pigs.

Perceived similarity of dogs and pigs was measured by a single item developed by the researcher '*How similar do you perceive the following animals [dogs/pigs] to be to humans?*' on a Likert scale from one ('*not at all*') to five ('*very much'*). A higher score

indicates greater similarity. This item was informed by previous research, which has measured similarity of animals using a similar single item (Possidónio et al., 2019).

2.2.3.5. Perceived Pet Status of Dogs and Pigs.

As a scale to measure pet status has not yet been developed, perceived pet status of dogs and pigs was measured by a single item developed by the researcher which holds face validity: '*How much do you perceive the following animals [dogs/pigs] to be a "pet" animal (an animal that is kept within a household as a companion)*?' on a Likert scale from one ('*not at all*') to five ('*very much*'). A higher score indicates greater pet status.

2.2.3.6. Perceived Profit Status of Dogs and Pigs.

As a scale to measure profit status has not yet been developed, perceived profit status of dogs and pigs was measured by a single item developed by the researcher which holds face validity: '*How much do you perceive the following animals [dogs/pigs] to be a "profit" animal (an animal that is used in some way for human consumption, e.g., for meat, leather or animal testing)*?' on a Likert scale from one ('*not at all*') to five ('*very much*'). A higher score indicates greater profit status.

2.2.3.7. Subjective Self-Relevance of Pigs (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009).

Subjective self-relevance of pigs was measured via an adapted version of the Product Involvement Scale (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009; α =.86 from Kim, 2006; Appendix 1c), whereby participants indicated their level of agreement with three statements regarding pig products (e.g., ham, pork, bacon) on a Likert scale from one (*'strongly disagree'*) to seven (*'strongly agree'*). This scale was adapted here by a) utilising a Likert scale instead of a semantic differential to reduce cognitive effort and ease scale completion (Friborg et al., 2006), and b) referring to pig products specifically. A sample item is *'I am very interested*

in products made from pigs (e.g., pork, ham)'. Due to the adaptations, I ran a PAFA (Appendix 2b) with direct oblimin rotation to test factorial validity, alongside assessing inter-item correlations to test reliability, and communality values to test common variance. The PAFA revealed inadequate loading of the third item '*I am not indifferent to products made from pigs* (e.g., pork, ham)' onto the factor (.32), whilst the first two items adequately loaded onto one factor (\geq .87). The third item also correlated inadequately with the first and second items, *rs*=.27, and had inadequate communality (.1). Therefore, I removed the third item, creating a two-item subjective self-relevance index. Higher scores indicate greater subjective self-relevance of pigs (i.e., greater involvement with pig products). Neither item was reverse-scored. Reliability was adequate within this study per the Spearman-Brown coefficient (r_{sb} =.86). I report the Spearman-Brown coefficient to check reliability for two-item scales throughout this thesis, per Eisinga et al. (2012). Adequate reliability is interpreted throughout as $r_{sb} \geq .7$.

2.2.3.8. Behavioural Self-Relevance of Pigs.

As a scale to measure behavioural self-relevance has not yet been developed, behavioural self-relevance of pigs was measured via a single item developed by the researcher '*How many days a week do you eat products made from pigs (e.g., ham, pork, sausages, bacon)*?' from zero to seven days per week. A higher score indicates greater behavioural self-relevance (i.e., greater consumption of pig products).

2.2.3.9. Warmth and Competence of Dogs and Pigs (Sevillano & Fiske, 2016b).⁴

Warmth (α =.83 from Sevillano & Fiske, 2016b) and competence (α =.87 from Sevillano & Fiske, 2016b) were measured via shortened warmth and competence subscales

⁴Unlike all subsequent studies, behavioural intentions are not measured in the current study, as inclusion of behavioural intentions would significantly increase the number of regression analyses required (from four to 12), and thus increase Type I error.

(Sevillano & Fiske, 2016b, adapted from Fiske et al., 2002). Participants indicated how much they viewed dogs and pigs as '*warm*', '*well-intentioned*' and '*friendly*' (warmth), and '*competent*', '*skillful*' and '*intelligent*' (competence), on a Likert scale from one ('*not at all*') to five ('*extremely*'). Items within each subscale were summed to provide warmth and competence scores, with higher scores indicating greater levels of that variable. No items were reverse-scored. The warmth and competence subscales have adequate discriminant and convergent validity (Diamantopoulos et al., 2017). Reliability was adequate across species within this study (dog warmth: ω =.88, 95% BCa CI [.82, .91]; dog competence: ω =.88, 95% BCa CI [.84, .9]; pig warmth: ω =.89, 95% BCa CI [.85, .92]; pig competence: ω =.91, 95% BCa CI [.87, .93]).

2.2.4. Procedure

2.2.4.1. General Procedure.

All studies throughout this thesis shared procedural details as follows: All studies were approved by Anglia Ruskin University's (ARU's) Psychology and Sport Science School Research Ethics Panel per British Psychological Society ethical code of conduct and General Data Protection Regulation legislation. Participants had to be 18+. Studies were conducted online via Qualtrics. Before taking part, participants read a participant information sheet, followed by providing informed consent. After reading and completing measures and manipulations specific to each study, participants could indicate their thoughts on the study, reported technical difficulties, and offered final comments via a textbox, before being debriefed.

2.2.4.2. Study 1 Procedure.

Participants were recruited via social media, posters, flyers, and Sona (Appendix 3). The study was conducted in October and November 2019 as a voluntary, open survey. After providing informed consent, participants completed the scales in the order listed in Section

2.4.3., followed by answering demographic questions, before being debriefed. Participants completed an attention check '*If you are reading this statement, please choose option 3* "*Somewhat*"" within the Empathy Towards Animals scale to ensure participants were reading the statements. Five participants were excluded for failing this attention check. Seven participants reported technical difficulties, but responses were complete and maintained within analyses. ARU undergraduate psychology students (n=13) received 0.25 Sona credits as reimbursement.

2.3. Analyses

A one-way (species: dog vs. pig) within-subjects multivariate analysis of variance (MANOVA) was conducted with warmth, competence, familiarity, similarity, pet status, and profit status as the dependent variables (DVs).

Four multiple regressions with 13 regressors each (behavioural and subjective selfrelevance of pigs, familiarity, similarity, pet and profit status of dogs and pigs, empathic concern towards animals, perspective taking of animals, support for animal utility) were run on the four outcome variables: dog warmth, dog competence, pig warmth, and pig competence. As extensive literature has found that women and men (Apostol et al., 2013; Herzog, 2007), and meat consumers and non-meat consumers (Lund et al., 2016; Rothgerber, 2015), respectively, differ in how they perceive animals, participant diet and gender may impact regression findings. As such, diet and gender were included as covariates within these multiple regressions, in order to adjust the regression model to account for these variables, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022).

2.3.1. General Analytical Strategy Across Studies

Throughout this thesis, most tests of differences are conducted utilising MANOVA, whereby DVs are combined into a single multivariate construct, instead of ANOVA, due to

conceptual and/or statistical relationships between DVs. Conceptual relationship refers to when DVs arise from the same theoretical framework and/or are subscales within the same overall scale, whilst statistical relationship refers to either positive or negative moderate correlations between DVs (whereby moderate correlations are defined within this thesis as rs $\geq \pm .3$; Field, 2018). When DVs are conceptually and/or statistically related, they should be combined into a single construct within MANOVA to account for the underlying relationships between the DVs (Field, 2018; Tabachnick & Fidell, 2019). Additionally, due to the presence of multiple DVs throughout the studies in this thesis, conducting a MANOVA (as opposed to separate ANOVAs for each and every DV) controls for Type I error (Field, 2018; Tabachnick & Fidell, 2019). Conceptual and statistical relationships between DVs specific to each study are reported in the relevant study chapter. For the current study, warmth and competence conceptually arise from the same theoretical framework (SCM) and constitute subscales within the same overall scale. Pet status and profit status also conceptually arise from the same theoretical framework of categorisation. Statistically, many of the DVs correlated with each other equal to or stronger than $\pm .3$, as assessed through Pearson correlations. Therefore, MANOVA was deemed appropriate within the current study.

All MANOVAs throughout this thesis were conducted using the following procedures: Normality for each DV across each condition was tested as a proxy for multivariate normality, whereby normality was assessed through Kolmogorov-Smirnov tests of normality and skewness across conditions. DVs which failed Kolmogorov-Smirnov tests but had skewness within acceptable boundaries (-2 to 2; Kim, 2013; West et al., 1995) were judged to be acceptably normal. Where normality was violated, I proceeded with the MANOVA, as MANOVA is robust to non-normality (Blanca et al., 2017), especially when group sizes are approximately equal (Field et al., 2012). Multivariate outliers were assessed by Mahalanobis distance values. Multicollinearity between DVs across conditions was

assessed through Pearson correlations. Linear relationships between DVs across conditions were assessed by visually inspecting scatterplots. If this assumption was not met, I proceeded with the MANOVA, but the analysis may suffer from lower statistical power (Tabachnick & Fidell, 2014). For MANOVAs including between-subjects IVs, homogeneity of variances is assessed using Levene's test for equality of variances, whilst homogeneity of covariances is assessed using Box's M test. Where homogeneity of variances is not present, analyses should be robust to heterogeneity of variances due to approximately equal numbers of participants per condition (Field et al., 2012; Field, 2018). Where homogeneity of covariances is not present, statistics report Pillai's trace instead of Wilks' lambda.

Significant multivariate effects from MANCOVAs were followed up with univariate ANCOVAs for each DV, unless otherwise stated. *Post hoc* pairwise comparisons with a Bonferroni adjustment were in turn run to investigate some significant findings from the follow-up ANCOVAs (e.g., interactions or main effects for IVs with more than two levels). For Studies 2-6, mediation analyses were conducted to test the full BIAS map model. Effect sizes are mostly reported throughout this thesis using η_p^2 , defined as small (η_p^2 =.01), medium (η_p^2 =.06), or large (η_p^2 =.14; Richardson, 2011). For significant *post hoc* pairwise comparisons, Cohen's *d* is reported instead, defined as small (d=.2), medium (d=.5), or large (d=.8; Cohen, 1988).

2.3.2. Statistical Assumptions to Test H1-H5: Exploring Dog vs. Pig Differences Across Warmth, Competence, Familiarity, Similarity, Profit Status and Pet Status (MANOVA)⁵

All DVs failed normality tests, ps < .05, but all DVs had acceptable skewness, except for pet status in the dog condition. Due to excessive skewness, multiple extreme univariate

⁵Note that, as diet and gender are not included as IVs or covariates within this analysis, the sample size of N=232 is utilised for this analysis. That is, 20 participants who were excluded from the regression analyses in order to allow diet and gender to be dummy coded did not need to be excluded for this MANOVA analysis also.

outliers, and floor effects in pet status for the dog condition, I also ran a MANOVA excluding pet status as a DV, which did not change multivariate conclusions. I therefore report the MANOVA including pet status.

All DVs except pet status either did not have univariate outliers in either condition (competence, similarity), or had non-extreme univariate outliers in the dog condition which did not cause excessive skewness (warmth, familiarity, profit status). There were multivariate outliers for six DVs, critical values \geq 22.46, p < .001. Running the MANOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in pet status. However, running the MANOVA including and excluding univariate and multivariate outliers without pet status as a DV indicated multivariate conclusions did not change, and I therefore report the MANOVA including univariate and multivariate outliers.

There was no multicollinearity, $rs \le .71$. Linear relationships between the DVs were not present across each condition. Running follow-up univariate ANOVAs including and excluding univariate outliers for warmth, familiarity, and profit status did not change conclusions. ANOVAs are therefore reported including univariate outliers. A follow-up nonparametric sign test with continuity correction was run instead of a follow-up ANOVA for pet status, due to excessive skewness, multiple extreme outliers, and floor effects. A nonparametric sign test was run for pet status instead of the alternative non-parametric Wilcoxon signed-rank test, due to failed assumption of symmetrical distribution.

2.3.3. Statistical Assumptions to Test H6-H12: Exploring Regressors of Dogs' and Pigs' Warmth and Competence (Multiple Regressions)

Residuals were approximately normally distributed, as assessed through histograms and P-P plots. Outliers and leverage values (\geq .2) were present. However, excluding outliers and leverage values did not alter main conclusions for three regressions (dogs' warmth, dogs'

competence, pigs' warmth). I therefore report the regressions including these outliers and leverage values. Excluding outliers and leverage values did change two conclusions for the regression on pigs' competence (Footnotes 8-9). However, I report this regression including outliers and leverage values to reflect the unamended dataset. There were no highly influential points, as assessed by Cook's distance. There was no multicollinearity between regressors, as assessed through Pearson's correlations ($rs \le .61$) and VIFs (≤ 2.24). Homoscedasticity and linearity assumptions were approximately met, as determined through inspection of scatterplots of residuals and partial regression plots. Durbin-Watson statistics were within acceptable boundaries for all regressions, $1.85 \le$ Durbin-Watson ≤ 1.99 .

2.3.4. Testing H1-H5: Exploring Dog vs. Pig Differences in Warmth, Competence, Familiarity, Similarity, Profit Status and Pet Status⁶

A one-way (species: dog vs. pig) within-subjects MANOVA was run on warmth, competence, familiarity, similarity, pet status, and profit status to test **H1-H5**. Descriptive statistics for effects of species are shown in Table 1.

2.3.4.1. H1-H5: Multivariate Effects of Species on the Combined DVs.

Species had a significant main effect on the combined DVs, F(6, 226) = 128.05, p < .001, $\eta_p^2 = .77$ (large-sized). I therefore followed up this significant multivariate effect with univariate ANCOVAs on the DVs (or non-parametric sign test with continuity correction for pet status) below.

⁶As single Likert items can be argued to be non-parametric (Bishop & Herron, 2015), I also ran two nonparametric Wilcoxon signed-rank tests with species (dog vs. pig) as the IV and familiarity and similarity as the DVs. I also ran a non-parametric sign test with continuity correction with species (dog vs. pig) as the IV and profit status as the DV. These non-parametric analyses revealed identical results to the MANOVA.

2.3.4.2. H1a: Univariate Effects of Species on Warmth.

As predicted: There was a significant effect of species on warmth, F(1, 231) =195.81, p < .001, $\eta_p^2 = .46$ (large-sized), whereby dogs were evaluated as significantly warmer than pigs, supporting H1a.

2.3.4.3. H1b: Univariate Effects of Species on Competence.

As predicted: There was a significant effect of species on competence, F(1, 231) = 69.42, p < .001, $\eta_p^2 = .23$ (large-sized), whereby dogs were evaluated as significantly more competent than pigs, supporting H1b.

2.3.4.4. H2: Univariate Effects of Species on Familiarity.

As predicted: There was a significant effect of species on familiarity, F(1, 231) = 231.64, p < .001, $\eta_p^2 = .5$ (large-sized), whereby dogs were evaluated as significantly more familiar than pigs were, supporting H2.

2.3.4.5. H3: Univariate Effects of Species on Similarity.

As predicted: There was a significant effect of species on similarity, F(1, 231) = 61.33, p < .001, $\eta_p^2 = .21$ (large-sized), whereby dogs were evaluated as significantly more similar to humans than pigs were, supporting H3.

2.3.4.6. H4: Univariate Effects of Species on Profit Status.

As predicted: There was a significant effect of species on profit status, F(1, 231) = 349.31, p < .001, $\eta_p^2 = .6$ (large-sized), whereby pigs were deemed profit animals significantly more than dogs, supporting H4.

2.3.4.7. H5: Effect of Species on Pet Status.

As predicted: There was a significant effect of species on pet status, z = 13.65, p < .001, r=.9 (large-sized; Cohen, 1988), whereby dogs were deemed to be pets significantly more than pigs, supporting H5.

Table 1.

Trofit Status, and Tet Status.				
DV	Dog Mean	Dog SD	Pig Mean	Pig SD
Warmth***	13.14	2.23	10.68	3.03
Competence***	12.45	2.43	10.8	3.19
Familiarity***	4.19	1.02	3.06	1.2
Similarity***	3.67	1.19	3.1	1.15
Profit Status***	1.88	1.12	3.75	1.52
Pet Status***	5 ^a	а	3 ^a	а

Descriptive Statistics for Effect of Species on Warmth, Competence, Familiarity, Similarity, Profit Status, and Pet Status.

Note: ^aMedian is reported for pet status rather than mean, and the *SD* cannot be reported for pet status, due to use of a sign test with continuity correction rather than an ANOVA. *** $p \le .001$

2.3.5. Testing H6-H12: Exploring Regressors of Dogs' and Pigs' Warmth and

*Competence*⁷

Four multiple regressions with 13 regressors each (behavioural and subjective selfrelevance of pigs, familiarity, similarity, pet and profit status of dogs and pigs, empathic concern towards animals, perspective taking of animals, support for animal utility), including diet and gender as covariates, were run on dog warmth, dog competence, pig warmth, and pig competence to test **H6-H12** (see Table A1 in Appendix 4 for full inferential statistics). The overall model was significant for each outcome variable, ps < .05. Individual regressor results are reported below.

2.3.5.1. H6a: Association Between Behavioural Self-Relevance of Pigs and Pigs' Warmth/Competence.

Contrary to H6a: Behavioural self-relevance was significantly positively associated with pigs' warmth, F(1, 196) = 7.15, p = .01, $\eta_p^2 = .04$ (small-sized), B = .36, SE = .13. Specifically, pigs' greater warmth was associated with participants reporting greater

⁷As single Likert items can be deemed non-parametric (Bishop & Herron, 2015), I also ran non-parametric ordinal logistic regressions. These non-parametric analyses revealed identical results.

consumption of pig food products. Also not supporting H6a: Behavioural self-relevance of pigs did not significantly link to pigs' competence, F(1, 196) = .48, p = .49, $\eta_p^2 = .002$, B = .01, SE = .14.

2.3.5.2. H6b: Association Between Subjective Self-Relevance of Pigs and Pigs' Warmth/Competence.⁸

Agreeing with H6b: Subjective self-relevance was significantly negatively associated with pigs' warmth, F(1, 196) = 5.08, p = .03, $\eta_p^2 = .03$ (small-sized), B = -.31, SE = .14. That is, pigs' lower warmth was associated with participants reporting greater enjoyment from consuming pig food products. Not supporting H6b: Subjective self-relevance of pigs did not significantly link to pigs' competence, F(1, 196) = 3.48, p = .06, $\eta_p^2 = .02$, B = -.27, SE = .15.

2.3.5.3. H7a: Association Between Familiarity with Dogs and Dogs'

Warmth/Competence.

Supporting H7a: Familiarity with dogs was significantly positively associated with dogs' warmth, F(1, 196) = 14.77, p < .001, $\eta_p^2 = .07$ (medium-sized), B = .56, SE = .15, whereby dogs' greater familiarity was associated with dogs' greater warmth. Not supporting H7a: Familiarity with dogs did not significantly link to dogs' competence, F(1, 196) = .06, p = .81, $\eta_p^{2} < .001$, B = -.05, SE = .18.

2.3.5.4. H7b: Association Between Familiarity with Pigs and Pigs'

Warmth/Competence.

Agreeing with H7b: Familiarity with pigs was significantly positively associated with pigs' competence, F(1, 196) = 7.04, p = .01, $\eta_p^2 = .04$ (small-sized), B = .56, SE = .21, whereby pigs' greater familiarity was associated with pigs' greater competence. Not

⁸When excluding outliers and leverage values, subjective self-relevance of pigs was significantly negatively associated with pigs' competence, F(1, 186) = 6.31, p = .01, $\eta_p^2 = .03$, B = -.35, SE = .14, whereby greater subjective self-relevance of pigs was linked to pigs' lower competence.

supporting H7b: Familiarity with pigs did not significantly link to pigs' warmth, F(1, 196) = 1.07, p = .3, $\eta_p^2 = .01$, B = .21, SE = .2.

2.3.5.5. H8a: Association Between Similarity of Dogs and Dogs'

Warmth/Competence.

Supporting H8a: Perceived similarity of dogs was significantly positively associated with dogs' warmth, F(1, 196) = 10.17, p = .002, $\eta_p^2 = .05$ (small-to-medium-sized), B = .43, SE = .13, and dogs' competence, F(1, 196) = 5.23, p = .02, $\eta_p^2 = .03$ (small-sized), B = .38, SE = .17. That is, dogs' greater perceived similarity to humans was associated with dogs' greater warmth and competence.

2.3.5.6. H8b: Association Between Similarity of Pigs and Pigs'

Warmth/Competence.

Supporting H8b: Perceived similarity of pigs was significantly positively associated with pigs' warmth, F(1, 196) = 5.12, p = .03, $\eta_p^2 = .03$ (small-sized), B = .5, SE = .22, and pigs' competence, F(1, 196) = 11.95, p < .001, $\eta_p^2 = .06$ (medium-sized), B = .79, SE = .23. That is, pigs' greater perceived similarity to humans was associated with pigs' greater warmth and competence.

2.3.5.7. H9a: Association Between Pet Status of Dogs and Dogs'

Warmth/Competence.

Agreeing with H9a: Pet status of dogs was significantly positively associated with dogs' warmth, F(1, 196) = 14, p < .001, $\eta_p^2 = .07$ (medium-sized), B = .72, SE = .19, and dogs' competence, F(1, 196) = 7.81, p = .01, $\eta_p^2 = .04$ (small-sized), B = .66, SE = .24. That is, dogs' greater pet status was associated with dogs' greater warmth and competence.

2.3.5.8. H9b: Association Between Pet Status of Pigs and Pigs'

Warmth/Competence.

Supporting H9b: Pet status of pigs was significantly positively associated with pigs' warmth, F(1, 196) = 13.69, p < .001, $\eta_p^2 = .07$ (medium-sized), B = .64, SE = .17, and pigs' competence, F(1, 196) = 8.28, p = .004, $\eta_p^2 = .04$ (small-sized), B = .52, SE = .18. That is, pigs' greater pet status was associated with pigs' greater warmth and competence.

2.3.5.9. H10a: Association Between Profit Status of Dogs and Dogs'

Warmth/Competence.

Supporting H10a: Profit status of dogs was significantly negatively associated with dogs' warmth, F(1, 196) = 11.59, p < .001, $\eta_p^2 = .06$ (medium-sized), B = -.39, SE = .12. That is, dogs' greater profit status was associated with dogs' lower warmth. Not supporting H10a: Perceived profit status of dogs did not significantly link to dogs' competence, F(1, 196) = .95, p = .33, $\eta_p^2 = .01$, B = -.14, SE = .14.

2.3.5.10. H10b: Association Between Profit Status of Pigs and Pigs'

Warmth/Competence.

Not supporting H10b: Profit status of pigs did not significantly link to pigs' warmth, $F(1, 196) = 1.18, p = .28, \eta_p^2 = .01, B = -.15, SE = .14$, or pigs' competence, F(1, 196) = .003, $p = .96, \eta_p^2 < .001, B = -.01, SE = .15$.

2.3.5.11. H11a-b: Association Between Empathic Concern towards Animals and Dogs'/Pigs' Warmth and Competence.⁹

Partly supporting H11a: Empathic concern towards animals was significantly positively associated with dogs' warmth, F(1, 196) = 12.04, p < .001, $\eta_p^2 = .06$ (medium-

⁹When excluding outliers and leverage values, empathic concern was significantly positively associated with pigs' competence, F(1, 186) = 6.55, p = .01, $\eta_p^2 = .03$, B = .13, SE = .05, whereby greater empathic concern for animals was linked to pigs' greater competence.

sized), B = .12, SE = .03, and dogs' competence, F(1, 196) = 9.73, p = .002, $\eta_p^2 = .05$ (small-to-medium-sized), B = .13, SE = .04. That is, greater empathic concern for animals was associated with dogs' greater warmth and competence. Also partly consistent with H11b: Empathic concern towards animals was significantly positively associated with pigs' warmth, F(1, 196) = 7.69, p = .01, $\eta_p^2 = .04$ (small-sized), B = .14, SE = .05, whereby greater empathic concern for animal was associated with pigs' greater warmth. However, partly not supporting H11b, empathic concern towards animals did not significantly link to pigs' competence, F(1, 196) = 2.52, p = .11, $\eta_p^2 = .01$ (small-sized), B = .08, SE = .05.

2.3.5.12. H11a-b: Association Between Perspective Taking of Animals and Dogs'/Pigs' Warmth and Competence.

Partly in support of H11a: Perspective taking of animals was significantly positively associated with dogs' competence, F(1, 196) = 3.96, p = .048, $\eta_p^2 = .02$ (small-sized), B = .07, SE = .04. That is, being able to take the perspective of animals was linked to dogs' greater competence. However, **partly not supporting H11a:** Perspective taking of animals was not significantly associated with dogs' warmth, F(1, 196) = 1.12, p = .29, $\eta_p^2 = .01$, B = .03, SE = .03. Also, **not supporting H11b:** Perspective taking of animals was not significantly associated with pigs' warmth, F(1, 196) = .04, p = .85, $\eta_p^2 < .001$, B = -.01, SE = .05, or pigs' competence, F(1, 196) = .37, p = .54, $\eta_p^2 = .002$, B = .03, SE = .05.

2.3.5.13. H12a-b: Association Between Support for Animal Utility and

Dogs'/Pigs' Warmth and Competence.

Not supporting H12a and H12b: Support for animal utility was not significantly associated with dogs' warmth, F(1, 196) = 1.16, p = .28, $\eta_p^2 = .01$, B = -.05, SE = .04, or dogs' competence, F(1, 196) = .87, p = .35, $\eta_p^2 = .004$, B = -.05, SE = .05, nor pigs' warmth, F(1, 196) = 2.41, p = .12, $\eta_p^2 = .01$, B = -.1, SE = .07, or pigs' competence, F(1, 196) = .66, p = .42, $\eta_p^2 = .003$, B = -.06, SE = .07.

2.4. Discussion

2.4.1. Discussion of Findings

2.4.1.1. H1-H5: Evidence for Dog vs. Pig Differences in Warmth, Competence, Familiarity, Similarity, Profit Status and Pet Status (Pet Speciesism).

Agreeing with previous literature (Bilewicz et al., 2011; Caviola & Capraro, 2020; Gradidge et al., 2021b), pet speciesism was evidenced in the context of the SCM/BIAS map, whereby dogs were deemed warmer (H1a; large-sized) and more competent (H1b; largesized) than pigs. Beyond the SCM/BIAS map, dogs were also deemed more familiar (H2; large-sized), more similar to humans (H3; large-sized), less as profit animals (H4; largesized) and more as pet animals (H5; large-sized) than pigs. These findings thus indicate familiarity, similarity, profit status, and pet status could all explain pet speciesism, whereby dogs' greater familiarity, similarity, and pet status, and lower profit status, compared to pigs, may explain why dogs are deemed warmer and more competent than pigs (i.e., why pet speciesism occurs). All pet speciesism findings were large-sized, aligning with previous pet speciesism findings (e.g., Gradidge et al., 2021b).

2.4.1.2. H6a-b: Behavioural and Subjective Self-Relevance of Pigs as Regressors of Pigs' Warmth/Competence.

Referring to regressors of pet speciesism, the current study found that, partially consistent with **H6b**, and agreeing with motivated cognition literature (Loughnan et al., 2010; Piazza & Loughnan, 2016), greater subjective self-relevance was associated with pigs' lower warmth (small-sized). Conversely, greater behavioural self-relevance was associated with pigs' greater warmth (small-sized), which was the opposite direction to predicted (**H6a**), and contrary to previous literature (Loughnan et al., 2010; Piazza & Loughnan, 2016). Partially not supporting **H6a-H6b**, neither behavioural nor subjective self-relevance linked to pigs' competence. These findings thus suggest that a) behavioural and subjective self-relevance

link to warmth only and not competence, and b) behavioural and subjective self-relevance differentially relate to warmth (behavioural self-relevance positively, subjective selfrelevance negatively). However, these findings were only small-sized, so may have lower practical significance.

The unexpected direction of the relationship between behavioural self-relevance and warmth may be explained by two main possibilities: a) people are morally engaging instead of disengaging (e.g., viewing pigs as more, rather than less, warm as a response to dissonance; see Gradidge et al., 2021a for more information on moral engagement), or b) findings are caused by a third variable due to current findings being non-causal. For instance, people, especially women, can under-report their meat consumption as a strategy to reduce dissonance, by distancing oneself from one's behaviour (Rothgerber, 2019). Thus, as this study is non-causal, and most of the sample is women, participants may have resolved dissonance by under-reporting their pig product consumption and viewing pigs as low in warmth. However, socially desirable responding would theoretically extend to other measures, such as under-reporting one's liking for pig meat (subjective self-relevance), which does not appear to be the case within this study. Therefore, to clarify discrepant findings, Studies 2-3 assess causal effects of behavioural and subjective self-relevance on warmth. Additionally, Studies 2-3 measure reported weekly meat consumption to test for any under-reporting of meat consumption across behavioural and subjective self-relevance conditions.

2.4.1.3. H7: Familiarity as a Regressor of Warmth and Competence.

The current study found greater familiarity with dogs was associated with dogs' greater warmth (medium-sized), whilst greater familiarity with pigs was associated with pigs' greater competence (small-sized). These findings agree with **H7** for dogs' warmth and pigs' competence but may not support **H7** for dogs' competence and pigs' warmth. Additionally, the relationship between familiarity with pigs and pigs' competence was only small-sized, so

may have lower practical significance. However, these findings do indicate that familiarity has differing relationships with warmth and competence depending on target species. Whilst positive links between familiarity with dogs or pigs and perceptions of them were expected due to previous literature (Auger & Amiot, 2015, 2016, 2019a, 2019b), differential relationships between familiarity and dogs' warmth vs. pigs' competence were not expected. These differential relationships with familiarity will need to be determined as causal before extensive explanations can be explored for these relationships (Study 4).

2.4.1.4. H8: Similarity as a Regressor of Warmth and Competence.

Supporting **H8** and previous research (Batt, 2009; Leach et al., 2021), this study found greater perceived similarity of dogs or pigs was associated with that species' increased warmth and competence (small-sized for dogs' competence and pigs' warmth; small-tomedium-sized for dogs' warmth; medium-sized for pigs' competence). Findings for dogs' competence and pigs' warmth were only small-sized, so may have lower practical significance. Studies 4-5 expanded upon current findings to assess causal effects of similarity on warmth, competence, and behavioural intentions.

2.4.1.5. H9: Pet Status as a Regressor of Warmth and Competence.

Agreeing with **H9** and previous literature (Signal et al., 2018; Taylor & Signal, 2009), the current study found dogs' or pigs' greater pet status was associated with increased warmth and competence of that species (small-sized for pigs' and dogs' competence; medium-sized for dogs' and pigs' warmth). Findings for dogs' and pigs' competence were only small-sized, so may have lower practical significance. Due to the current findings, the relationship between an animal's pet status and the animals' warmth and competence was tested for causality in Study 6.

2.4.1.6. H10a-b: Profit Status as a Regressor of Warmth and Competence.

The current study found dogs' greater profit status was associated with dogs' decreased warmth (medium-sized), partially supporting **H10a**. However, not supporting **H10a**, profit status of dogs did not link to dogs' competence. The relationship between dogs' profit status and dogs' warmth (yet not dogs' competence) is in line with previous research (Signal et al., 2018; Taylor & Signal, 2009). However, not supporting previous research and **H10b**, pigs' profit status did not link to pigs' warmth or competence. Current findings for profit status thus mostly do not support previous correlational research on the relationship between profit status and perceptions of animals (Signal et al., 2018; Taylor & Signal, 2009), nor causal research on effects of labelling animals as 'food' vs. not 'food' (Bastian et al., 2012a). Therefore, profit status cannot explain why pigs are viewed more negatively than dogs: That is, whilst pigs are indeed deemed profit animals more than dogs, viewing pigs as a profit animal is not associated with their decreased warmth and competence. Due to these null findings, I did not test for causal effects of profit status on perceptions of animals in subsequent experiments.

2.4.1.7. H11: Empathic Concern and Perspective Taking as Regressors of Warmth and Competence.

Greater empathic concern towards animals was associated with greater warmth of dogs and pigs as expected (small-sized for pigs; medium-sized for dogs), partially consistent with **H11** and previous literature (Apostol et al., 2013). Additionally, greater empathic concern towards animals was associated with dogs' greater competence (small-to-mediumsized), also partly supporting **H11**. However, empathic concern did not link to pigs' competence, not supporting **H11**. Additionally, greater perspective taking of animals was associated with dogs' greater competence as expected (small-sized), partly consistent with **H11** and previous literature (Apostol et al., 2013). However, perspective taking of animals

unexpectedly did not link to dogs' warmth, nor to pigs' warmth or competence, not supporting H11.

As previous research has focussed on overall empathy towards animals (Apostol et al., 2013; Hills, 1995), these findings expand upon previous literature to indicate the relationships between empathy and perceptions of animals depend on the type of empathy (empathic concern, perspective taking), type of perception (warmth, competence), and species (dog or pig). Future research should test the types of empathy on different types of perceptions of animals (e.g., behavioural intentions, moral status, mind attribution), and across perceptions of different species, to investigate boundary conditions of where and when empathy towards animals is linked to more positive perceptions of them. However, current findings indicate empathy towards animals (as measured through empathic concern and perspective taking) does not improve most perceptions of pigs and thereby that empathy towards animals does not lower pet speciesism. I therefore did not test empathy towards animals within further studies in this thesis. Additionally, some findings were only smallsized, so may have lower practical significance.

2.4.1.8. H12: Support for Animal Utility as a Regressor of Warmth and Competence.

Support for animal utility did not link to dogs' and pigs' warmth and competence, not supporting **H12**. These findings may therefore not support previous research (Krings et al., 2021; Monteiro et al., 2017), and suggest that support for animal utility cannot causally impact perceptions of dogs and pigs. The discrepancy in findings between this study and previous literature may arise from differences in measurements of the variables. Specifically, previous research has assessed beliefs in human dominion over animals (Monteiro et al., 2017), whilst this study assessed support for animal utility only. These variables may be related but differ in their valuing of animals: For example, both human dominion over

animals and support for animal utility may be linked to anthropocentric speciesism (viewing humans as warmer and more competent than other animals), whilst only human dominion over animals may be linked to not valuing animals at all (e.g., viewing them as extremely low in warmth and competence). As this study found null findings for support for animal utility, I did not test this variable further within subsequent studies in this thesis.

2.4.2. Limitations of the Current Study

The current study has limitations as follows, which are addressed in Studies 2-6: a) lack of causality, b) effects of bias on perceptions of animals, and c) lack of exploration of behavioural intentions.

Firstly, this study only investigated regressors of pet speciesism and does not measure causality. Thus, due to the cross-sectional data within this study, it is impossible to determine causal effects of the regressors on pet speciesism. Subsequent experiments within this thesis therefore intend to determine causality of the significant regressors on pet speciesism. Further studies thus causally manipulate behavioural and subjective self-relevance (Studies 2-3), familiarity (Study 4), similarity (Studies 4-5), and pet status (Study 6), to assess causal effects on perceptions of animals.

Secondly, this study explores perceptions of dogs' and pigs' warmth and competence. Perceptions of known animals may be influenced by bias or preconceptions of these animals, whereby bias and preconceptions act as moderators which obscure the actual relationships between the regressors and pet speciesism. To accurately determine causes of pet speciesism in subsequent experiments, studies will need to causally manipulate possible causal variables with an unknown animal whom participants are less biased towards or against (Studies 2-4 and 6), a method called the '*novel animal paradigm*' (Piazza et al., 2014; Piazza & Loughnan, 2016; Sytsma & Machery, 2012; Section 3.2.3.1.).

Thirdly, this study only assessed perceptions of animals (warmth, competence), without measuring behavioural intentions. As the BIAS map extends from the SCM, whereby warmth and competence typically inform active and passive behavioural intentions respectively (Cuddy et al., 2007), subsequent studies need to also measure behavioural intentions (Studies 2-6). Measurement of these behavioural intentions may enable easier application of significant causes to real-world behaviour (e.g., Webb & Sheeran, 2006), and thus improve real-world applications of interventions to reduce or prevent pet speciesism.

2.5. Conclusion

Firstly, the current study finds evidence for pet speciesism in the context of the SCM/BIAS map, alongside pet speciesism in familiarity, similarity, pet status, and profit status. Secondly, this study finds that pet status, similarity, familiarity, behavioural self-relevance, and subjective self-relevance are all associated with perceptions of dogs and/or pigs. These variables were thus all tested for causality in Studies 2-6. The lack of significant relationship between profit status and perceptions of pigs indicates profit status is incapable of explaining or reducing pet speciesism. Specifically, whilst pigs are deemed profit animals more than dogs, profit status is not associated with pigs' decreased warmth and competence. Regarding individual differences variables, support for animal utility was not associated with perceptions of dogs and pigs. Meanwhile, empathic concern towards animals was associated with dogs' and pigs' greater warmth, and dogs' greater competence, but not with pigs' competence. Perspective taking of animals was associated with dogs' greater competence, but not with dogs' warmth, pigs' warmth, or pigs' competence.

Overall, the current study evidences pet speciesism, and identifies significant regressors of dogs' and pigs' warmth and competence, which are investigated causally in subsequent experiments. This study has limitations including lack of causality, potentially moderating effects of bias in perceptions of dogs and pigs, and lack of measurement of

behavioural intentions. Subsequent experiments within this thesis therefore causally manipulate significant regressors from the current study and apply these findings to unknown animals and behavioural intentions as follows: behavioural and subjective self-relevance (Studies 2-3), familiarity (Study 4), similarity (Studies 4-5), and pet status (Study 6).

Chapter 3. Study 2: Causal Exploration of Self-Relevance

3.1. Introduction and Extension from Study 1

Study 1 found that behavioural and subjective self-relevance were associated with pigs' greater and lower warmth respectively, while neither linked to pigs' competence. Whilst the relationship between subjective self-relevance and warmth aligns with previous self-relevance literature and motivated cognition (Loughnan et al., 2010; Piazza & Loughnan, 2016), the relationship between behavioural self-relevance and warmth occurred in the opposite direction to predicted. That is, whilst people who more greatly enjoy consuming pigs viewed pigs as less warm, as expected, people who consume pigs more often unexpectedly viewed pigs as warmer. This finding stands in contrast to motivated cognition theory, as it would be expected that greater consumption of pigs would lead to greater cognitive dissonance, therefore causing people to view pigs as less warm. For instance, actual consumption of beef caused people to view cows as having lower moral status (Loughnan et al., 2010), and such a finding would be expected to extend to other perceptions of animals, such as the SCM (i.e., warmth).

The current study therefore extends the equivocal findings from Study 1 by testing whether these relationships are causal; that is, whether behavioural and subjective self-relevance are causally affect warmth, whereby behavioural and subjective self-relevance are causally manipulated within the current study through text manipulations regarding a 'target animal' previously unknown to the participant. Specifically, participants are presented with an imaginary scenario where they are offered meat from an unknown animal, and they are asked to imagine either that they like/dislike this meat (manipulation of subjective self-relevance) and that they agree/refuse to consume this meat (manipulation of behavioural self-relevance). An unknown animal is used for these text manipulations as a modified version of the '*novel animal paradigm*' (Piazza et al., 2014; Piazza & Loughnan, 2016; Sytsma &

Machery, 2012), whereby characteristics of a novel and typically fictional animal are causally manipulated to assess unmoderated effects of amended characteristics on perceptions of the animal. Novel fictional animals are employed instead of known animals to reduce moderating effects of bias or preconceptions on perceptions of the animal. For the current study, and subsequent studies (Studies 3-4 and 6), a real but unknown animal is utilised instead of the typically used fictional animal ('trablans'; Piazza & Loughnan, 2016) to enable a photograph of the animal to be included. Inclusion of such a photograph may enhance effectiveness of the imagined scenarios (Miller & Stoica, 2004).

Therefore, the current study utilises these behavioural and subjective self-relevance manipulations to explore if Study 1 findings regarding self-relevance and warmth are causal, rather than correlational, and to see if motivated cognition theory is supported or not. Note that warmth is measured with the same items from Study 1, which originate from Sevillano and Fiske (2016b).

For subjective self-relevance, in line with above discussed Study 1 findings and motivated cognition literature (e.g., Loughnan et al., 2010), it is hypothesised that:

H1: Animals portrayed as subjectively self-relevant (vs. not) will be seen as significantly less warm.

For behavioural self-relevance, a non-directional hypothesis is proposed, whereby findings may align either with Study 1 or with previous motivated cognition literature (e.g., Loughnan et al., 2010). I therefore predict that:

H2: Animals portrayed as behaviourally self-relevant (vs. not) will either be seen as significantly less warm (**a**, per previous literature; Piazza & Loughnan, 2016), or significantly more warm (**b**, per Study 1).

Note that hypotheses are not made for competence, as Study 1 found a lack of significant relationships between behavioural and subjective self-relevance and competence. However, competence is still measured within the current study, again utilising the same items from Study 1 (Sevillano & Fiske, 2016b).

Alongside testing causal effects of behavioural and subjective self-relevance on warmth, the current study extends these findings to behavioural intentions in line with the BIAS map (Cuddy et al., 2007). Specifically, the current study aims to investigate if behavioural and subjective self-relevance causally affect *active help* (intentional and effortful aid towards the animal), *active harm* (intentional and effortful damage inflicted upon the animal), *passive help* (aid towards the animal without active effort required), or *passive harm* (exclusionary behaviour towards the animal without active effort required), whereby these behavioural intentions are measured within the current study through BIAS map subscales from Sevillano and Fiske (2016b). The BIAS map posits that warmth informs greater active help and less active harm, whilst competence informs greater passive help and less passive harm (Cuddy et al., 2007). Thus, agreeing with the BIAS map, relationships between behavioural and subjective self-relevance and warmth within Study 1 should extend to effects on active help and active harm in the current study. As such, congruent with the BIAS map, I hypothesise that:

H3: If subjectively self-relevant animals (vs. not) are deemed significantly less warm (i.e., H1 is evidenced), subjectively self-relevant animals (vs. not) will be subject to significantly more active harm (a) and less active help (b).

H4: If behaviourally self-relevant animals (vs. not) are deemed significantly less warm (i.e., H2a is evidenced), behaviourally self-relevant animals (vs. not) will be subject to significantly more active harm (a) and less active help (b).

Beyond assessing causal effects of behavioural and subjective self-relevance, the current study also aims to directly test the *motivated cognition model*. Specifically, this model states self-relevance of the target animal should cause *dissonance* (moral discomfort), due to a contradiction between valuing animals and consuming (or enjoying consuming) them (Gradidge et al., 2021a; Loughnan et al., 2014; Rothgerber, 2020), whereby dissonance is measured within the current study through the Dissonance Affect Questionnaire (negative affect subscale; Harmon-Jones, 2000) both directly after the self-relevance manipulation (post-manipulation dissonance) and after evaluating how warm/competent the animal is (post-evaluation dissonance). As most people like animals and do not want to hurt them (Loughnan et al., 2014), yet within the current study may need to imagine consuming and/or liking meat from an animal, I predict that:

H5: Behaviourally or subjectively self-relevant animals (vs. not) will significantly increase post-manipulation dissonance.

This dissonance in turn should motivate participants to view the target animal more negatively (e.g., through lower warmth) to alleviate the uncomfortable experience of dissonance and justify one's consumption of, or enjoyment of, consuming animals (Gradidge et al., 2021a). As such, to explore the role of dissonance in informing negative perceptions of animals, I hypothesise that:

H6: Post-manipulation dissonance will significantly mediate relationships between behavioural and subjective self-relevance and warmth. That is, behaviourally (**a**) or subjectively (**b**) self-relevant animals (vs. not) will significantly increase postmanipulation dissonance, which will in turn significantly reduce the animal's postmanipulation warmth.
Additionally, to explore if this devaluing of animals through their decreased perceived warmth fulfils the purpose of alleviating dissonance (Gradidge et al., 2021a), I predict that:

H7: The greater post-manipulation dissonance, the lower post-evaluation dissonance, due to the animal's decreased post-manipulation warmth (mediation). That is, warmth will be deliberately significantly decreased by participants as part of motivated cognition to reduce dissonance.

In addition to the above, inclusion of measurements for behavioural intentions within this study also enables replicability of the BIAS map to animals to be tested (Sevillano & Fiske, 2016b, 2019). That is, it can be tested directly if any effects of behavioural and subjective self-relevance on active behavioural intentions are mediated through warmth, in line with both general SCM and BIAS map literature (Cuddy et al., 2007) and SCM/BIAS map literature applied to animals specifically (Sevillano & Fiske, 2016b, 2019). As warmth typically informs active behavioural intentions (Sevillano & Fiske, 2016b, 2019), it is hypothesised that:

H8: Post-manipulation warmth will significantly mediate relationships between behavioural self-relevance and post-manipulation active harm and post-manipulation active help. That is, behaviourally self-relevant animals (vs. not) will significantly decrease post-manipulation warmth, which will in turn significantly increase postmanipulation active harm (**a**) and decrease post-manipulation active help (**b**).

H9: Post-manipulation warmth will significantly mediate relationships between subjective self-relevance and post-manipulation active harm and post-manipulation active help. That is, subjectively self-relevant animals (vs. not) will significantly decrease post-manipulation warmth, which will in turn significantly increase post-manipulation active harm (**a**) and decrease post-manipulation active help (**b**).

Finally, reported weekly meat consumption is included as an exploratory variable within this study to better enable detection of meat consumption under-reporting across behavioural and subjective self-relevance conditions (Rothgerber, 2019), whereby this reporting is measured through a single item '*How many days a week do you consume meat*?'. As reported weekly meat consumption is included as an exploratory variable, no hypotheses are made for this variable.

3.2. Methods

3.2.1. Participants

Three-hundred-and-eighteen participants were recruited via volunteer sampling. Fiftyseven participants were excluded for giving partial responses, 18 participants for failing an attention check, and one participant due to technical difficulties. All analyses included gender and diet as covariates, as both variables influence perceptions of animals (Apostol et al., 2013; Herzog, 2007; Lund et al., 2016; Rothgerber, 2015; see Section 3.3 for further discussion). To dummy code gender into women (coded as zero; n=169) and men (coded as one; n=58), five participants were excluded due to being non-binary (n=2), giving no response (n=2), or indicating they would rather not say (n=1). To dummy code diet into meat consumer (coded as one; n=158) and non-meat consumer (coded as zero; n=69, including vegans, vegetarians, and pescatarians), 11 participants were excluded due to being reducetarian (n=5), flexitarian (n=3), giving no response (n=1), answering N/A (n=1), or indicating they would rather not say (n=1). Exclusions left a sample size of 227 (74.4%) women, 25.6% men; $M_{age}=27.76$, $SD_{age}=9.35$, age range=18-69), which exceeds the minimum required sample size of 179 per a G*Power a priori power analysis for a MANOVA analysis ('repeated measures, within-between interaction', medium effect size *f*=.25, four groups, two measurements, α =.05, power = .8). Participants within each condition are shown in Table 2. Demographics of the sample are reported as follows in percentages,

with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For dietary group: 69.6% (meat consumer), 17.6% (vegetarian), 7.5% (vegan), and 5.3% (pescatarian). For nationality: 54.6% (United Kingdom/British), 9.3% (Indian), 4% (German), 2.6% (no response), 2.2% each (Portuguese, United States), and 25.1% other. For ethnicity: 73.8% (White), 16.7% (Asian), 4.3% (mixed), and 5.2% other. For country of residence ($M_{duration}$ =21.06, $SD_{duration}$ =13.56, range: less than one month to 69 years): 70.4% (United Kingdom), 7.9% (India), 3.5% (Germany), 3.1% each (The Netherlands, United States), and 12% other. For religion: 45% (no religion or N/A), 26.8% (Christianity), 7.5% (Hinduism), 4.4% (agnosticism), 3.5% each (atheism, no response), 3.1% (Islam), and 6.2% other.

Table 2.

Participants Within Each Condition for Study 2.

Sen-Kelevance Number	of
ondition Participa	ints
Present 54	
Absent 61	
Absent 62	
Present 65	
F	Present 65

3.2.2. Design

This experiment follows a 2(behavioural self-relevance: present vs. absent; betweensubjects) x 2(subjective self-relevance: present vs. absent; between-subjects) x 2(time: prevs. post-manipulation; within-subjects) mixed MANCOVA design, with warmth, competence, active help, passive help, active harm, and passive harm as the DVs. Reported weekly meat consumption, post-manipulation dissonance, and post-evaluation dissonance were included as DVs in a 2(behavioural self-relevance: present vs. absent; between-subjects) x 2(subjective self-relevance: present vs. absent; between-subjects MANCOVA. Gender and diet were included as covariates.

3.2.3. Materials

3.2.3.1. Novel Animal Paradigm (Piazza et al., 2014; Piazza & Loughnan, 2016; Sytsma & Machery, 2012).

The current study employed an adapted version of the novel animal paradigm (Piazza et al., 2014; Piazza & Loughnan, 2016; Sytsma & Machery, 2012), whereby stated attributes (e.g., intelligence, harmfulness) of a fictional animal are causally manipulated to assess their unmoderated effects on perceptions of the animal. A fictional animal is typically utilised instead of a real animal to avoid prior perceptions of the animal and bias moderating effects of these causal manipulations on subsequent perceptions of the animal. An adapted version of the paradigm was implemented within this study, whereby a real animal (Bennett's tree kangaroo [hereon tree kangaroo], fossa or tamandua) was used instead of a fictional animal to enable a photograph of the animal to be utilised. To enable further elimination of bias, real animals were chosen for the current study and subsequent studies (Studies 3-4 and 6) which are likely to be unknown to most participants due to their relative obscurity (e.g., lack of exposure in media and remoteness from most Western human populations, e.g., Bratanova et al., 2011). In addition, tree kangaroos have been effectively utilised in prior research employing the novel animal paradigm (Bratanova et al., 2011). This animal therefore represents an appropriate target animal. Lack of familiarity with the animal was explicitly checked by asking participants if they recognised the animal ('Do you recognise this animal?' Options: 'Yes' or 'No'), with participants being directed to the next animal if they answered 'yes' and excluded from the study if they recognised all animals. Participants completed the entirety of the study about their target animal (tree kangaroo, fossa, or tamandua) only.

3.2.3.2. Animal Photographs.

A photograph was implemented for each target animal to enhance participants' ability to visualise the imagined scenario (Miller & Stoica, 2004), whereby the tree kangaroo

photograph was sourced from <u>https://www.shutterstock.com/image-photo/close-shot-cute-</u> tree-kangaroo-168261035, the fossa photograph sourced from

https://www.flickr.com/photos/mathiasappel/19504925051, and the tamandua photograph sourced from https://tinyurl.com/2jazjz3s. Photographs faced the animal directly and included the animal as the central focus of the photograph.

3.2.3.3. Behavioural and Subjective Self-Relevance Text Manipulations.

Behavioural and subjective self-relevance were manipulated within this study through an imagined scenario about the participant's target animal as follows: '*Imagine that you like/dislike tree kangaroo meat. You have been offered tree kangaroo meat again. Imagine you agree/refuse to consume the tree kangaroo meat*'. Subjective self-relevance is manipulated through use of the word '*like*' (subjective self-relevance) or '*dislike*' (lack of subjective self-relevance). Behavioural self-relevance is manipulated through use of the word '*agree*' (behavioural self-relevance) or '*refuse*' (lack of behavioural self-relevance). These text manipulations were developed by the current researcher and are informed by previous text and imagined scenarios with animals (Bratanova et al., 2011).

3.2.3.4. Warmth and Competence (Sevillano & Fiske, 2016b).

Warmth and competence were measured with the same items from Study 1 on a Likert scale from one ('*not at all*') to five ('*extremely*'): *warm*', '*well-intentioned*', '*friendly*' (warmth), and '*competent*', '*skillful*', '*intelligent*' (competence; Sevillano & Fiske, 2016b). Items within each subscale were summed to provide warmth and competence scores. Reliability was adequate within this study for warmth (pre-manipulation ω =.85, 95% BCa CI [.79, .89]; post-manipulation ω =.89, 95% BCa CI [.84, .92]), and competence (premanipulation ω =.88, 95% BCa CI [.84, .9]; post-manipulation ω =.91, 95% BCa CI [.88, .93]).

3.2.3.5. Behavioural Intentions Towards the Animal (Sevillano & Fiske, 2016b).

Behavioural intentions were measured based on how willing, on a Likert scale from one ('not at all willing') to five ('extremely willing'), participants were to do the following behaviours to their target animal (Sevillano & Fiske, 2016b): 'support', 'help', 'behave *friendly towards*', and *'interact with*' the animal (active help; α =.93 from Sevillano & Fiske, 2016b), 'sustain' and 'conserve' the animal (passive help; α =.8 from Sevillano & Fiske, 2016b), 'kill', 'injure', 'exterminate', 'trap', and 'reject' the animal (active harm; α =.93 from Sevillano & Fiske, 2016b), 'let [the animal] die off' and 'ignore' the animal (passive harm; α =.74 from Sevillano & Fiske, 2016b). No items were reverse-scored. Items within each subscale were summed to provide active help, passive help, active harm, and passive harm scores. Higher scores on any subscale indicates higher levels of that variable. Reliability was adequate within this study for active help (pre-manipulation ω =.83, 95% BCa CI [.76, .87]; post-manipulation ω =.84, 95% BCa CI [.78, .88]), passive help (pre-manipulation r_{sb} =.77; post-manipulation r_{sb} =.81), and active harm (pre-manipulation ω =.88, 95% BCa CI [.72, .94]; post-manipulation ω =.9, 95% BCa CI [.71, .97]). Reliability was inadequate within this study for passive harm (pre-manipulation r_{sb} =.58; post-manipulation r_{sb} =.69). As the passive harm subscale contains two items, I did not run a PAFA for this subscale, and I was unable to remove an item to improve reliability. Instead, I retained both items in the subscale, and findings using this subscale should be interpreted with caution due to inadequate reliability.

3.2.3.6. Dissonance (Harmon-Jones, 2000).

Dissonance was measured via the Dissonance Affect Questionnaire (negative affect subscale; Harmon-Jones, 2000), whereby participants rated, on a Likert scale from one ('*not at all*') to five ('*extremely*'), how '*uncomfortable*', '*uneasy*', and '*bothered*' they currently felt (α =.7 from Harmon-Jones, 2000). No items were reverse-scored. Items were summed to provide a dissonance score. A higher score indicates greater dissonance. Reliability was

adequate within this study: post-manipulation (ω =.97, 95% BCa CI [.95, .98]), postevaluation (ω =.95, 95% BCa CI [.93, .97]). Dissonance was measured twice within this study: once following the behavioural and subjective self-relevance manipulation (postmanipulation), and once following warmth, competence, and behavioural intentions ratings (post-evaluation).

3.2.3.7. Reported Weekly Meat Consumption.

Reported weekly meat consumption was measured through the single item '*How many days a week do you consume meat*?', from zero to seven days per week.

3.2.4. Procedure

Participants were recruited via social media, posters/flyers, and Sona (Appendix 5). The experiment was conducted in April and May 2020 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos (alongside a photograph of the animal), followed by a fossa and tamandua if participants recognised any previous animal. Animal species names were used throughout the study. The experiment either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected towards the end of the experiment and did not participate further. Following initial recognition stage, participants rated the target animal's warmth and competence, alongside behavioural intentions towards the animal (all scales and items within scales presented in randomised order).

After initial ratings, participants were randomly assigned to read one of the four behavioural and subjective self-relevance text manipulations described in Section 3.2.3.3. regarding their target animal. Participants subsequently indicated their dissonance and reported weekly meat consumption. Participants then again gave their warmth, competence, and behavioural intentions ratings (all scales and items within scales presented in randomised

order), followed by re-stating their current dissonance. Participants completed the attention check '*Please choose 3 somewhat*' in the post-manipulation warmth and competence subscales. Eighteen participants were excluded for failing this attention check. Finally, participants provided demographics before being debriefed. One participant was excluded based on technical difficulties due to stating the animal photograph did not load. An additional six participants reported technical difficulties, but responses were complete and maintained within analyses. ARU undergraduate psychology students (*n*=13) received 0.25 Sona credits as reimbursement. All other participants could enter a prize draw to win one of two £50 Amazon gift vouchers if they wished.

3.3. Analyses

A 2(behavioural self-relevance: present vs. absent) x 2(subjective self-relevance: present vs. absent) x 2(time: pre- vs. post-manipulation) mixed MANCOVA was conducted with warmth, competence, active help, passive help, active harm, and passive harm as the DVs. As all of the DVs were conceptually related (from the SCM/BIAS map and subscales of the same overall scales), alongside many DVs being statistically related through moderate correlations, $rs \ge .3$, MANCOVA was deemed appropriate for this analysis. An additional 2(behavioural self-relevance: present vs. absent) x 2(subjective self-relevance: present vs. absent) between-subjects MANCOVA was conducted on reported weekly meat consumption, post-manipulation dissonance, and post-evaluation dissonance as the DVs. As two of the DVs were conceptually related (i.e., post-manipulation dissonance and post-evaluation dissonance are the same variable measured at different timepoints), as well as some moderate correlations between the DVs, $rs \ge .3$, MANCOVA was deemed appropriate for this analysis.

Gender and diet were again included as covariates within these MANCOVAs. Although the current study randomly allocates participants to conditions, and therefore diet and gender are approximately evenly balanced across conditions, inclusion of these

covariates still increases power of the analyses by controlling for and partialling out any relationship between these covariates and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as both diet and gender have previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007; Lund et al., 2016; Rothgerber, 2015), including diet and gender as covariates allows for any theoretically-informed relationships between diet and gender, and warmth/competence perceptions of dogs and pigs and behavioural intentions towards them (the outcome variables) to be statistically controlled for.

3.3.1. Statistical Assumptions to Test H1-H4: Exploring Effects of Behavioural and Subjective Self-Relevance on SCM and BIAS Map DVs (2x2x2 MANCOVA)

Most DVs, except for pre-manipulation competence when both behavioural and subjective self-relevance were present, failed normality tests, ps < .05. However, most DVs had acceptable skewness, except for active harm across conditions and passive harm in all post-manipulation conditions except when behavioural and subjective self-relevance were both present. Due to excessive skewness, multiple extreme univariate outliers, and floor effects in active harm and passive harm, I also ran a MANCOVA excluding active harm and passive harm as DVs, which did not change multivariate conclusions. I therefore report the MANCOVA including active harm and passive harm.

All DVs except active harm and passive harm had non-extreme univariate outliers in at least one condition, which did not cause excessive skewness. There were multivariate outliers for six DVs, critical values ≥ 22.46 , p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in active harm and passive harm. However, running the MANCOVA including and excluding univariate and multivariate outliers without active harm and passive harm as DVs indicated multivariate conclusions did not change, and I therefore report the MANCOVA including univariate and multivariate outliers. There was no multicollinearity, $rs \le .73$. Linear relationships were mostly present. Running follow-up univariate ANCOVAs including and excluding univariate outliers did not change conclusions for active help and warmth, whilst one minor finding was changed for competence (Footnote 10), and for passive help (Footnote 11). I report these ANCOVAs including outliers to reflect the unamended dataset. Excessive skewness, multiple univariate outliers, and floor effects in active harm and passive harm meant assumptions for the ANCOVAs on these two DVs were violated. However, I proceeded with these ANCOVAs due to robustness of ANCOVA to non-normality and lack of non-parametric alternatives. Results from these ANCOVAs should be interpreted with caution.

Homogeneity of variances was present for all DVs, ps > .05, except active harm and passive harm, ps < .05. Homogeneity of covariances was not present, p < .001.

3.3.2. Statistical Assumptions to Test H5 and Exploratory Analysis: Exploring Effects of Behavioural and Subjective Self-Relevance on Post-Manipulation Dissonance, Post-Evaluation Dissonance and Reported Weekly Meat Consumption (2x2 MANCOVA)

All DVs failed normality tests, ps < .05, but all DVs had acceptable skewness. Postmanipulation dissonance and reported weekly meat consumption did not have univariate outliers, whilst post-evaluation dissonance did have univariate outliers. However, these univariate outliers were not extreme, and they did not excessively skew post-evaluation dissonance. There were no multivariate outliers for three DVs, critical values < 16.27, p >.001. Running the MANCOVA including and excluding univariate outliers did not change conclusions. I therefore report this analysis including univariate outliers.

There was no multicollinearity, $rs \le .54$. Linear relationships between the DVs were not present across each condition. Running follow-up univariate ANCOVAs including and excluding univariate outliers for post-evaluation dissonance did not change conclusions. I therefore report this ANCOVA including univariate outliers. Homogeneity of variances was present for all DVs, ps > .05. Homogeneity of covariances was present, p > .001.

3.3.3. Testing H1-H4: Exploring Effects of Behavioural and Subjective Self-Relevance on SCM and BIAS Map DVs

A 2(behavioural self-relevance: present vs. absent; between-subjects) x 2(subjective self-relevance: present vs. absent; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed MANCOVA, including diet and gender as covariates, was run on warmth, competence, active help, passive help, active harm, and passive harm to test **H1-H4** (see Table A2 in Appendix 4 for full inferential statistics). To be deemed to have a causal effect in line with **H1-H4**, behavioural self-relevance and subjective self-relevance must interact with time on the combined DVs, as significant main effects of behavioural and/or subjective self-relevance without interacting with time indicate a randomly occurring difference between groups even before reading the self-relevance manipulations.

3.3.3.1. Multivariate Effects of Behavioural and Subjective Self-Relevance on the Combined DVs.

Not supporting hypotheses (**H1-H4**), neither behavioural self-relevance, F(6, 216) = .76, p = .6, $\eta_p^2 = .02$, nor subjective self-relevance, F(6, 216) = 1.08, p = .38, $\eta_p^2 = .03$, significantly interacted with time to have an effect on the combined DVs. As a result, inferential and descriptive statistics for follow-up ANCOVAs on the separate DVs with behavioural and subjective self-relevance as IVs are not reported. However, there were significant main effects of diet, F(6, 216) = 2.36, p = .03, $\eta_p^2 = .06$ (medium-sized), gender, F(6, 216) = 3.31, p = .004, $\eta_p^2 = .08$ (medium-sized), and time, F(6, 216) = 2.71, p = .02, $\eta_p^2 = .07$ (medium-sized), on the combined DVs. I followed up these significant multivariate effects with univariate ANCOVAs on the DVs separately. As these findings relate only to

covariates and to time (rather than the primary variables of behavioural and subjective selfrelevance), only significant univariate findings regarding these variables are reported below.

3.3.3.2. Univariate Effect of Time on Warmth.

There was a significant main effect of time on warmth, F(1, 221) = 9.91, p = .002, $\eta_p^2 = .04$ (small-sized), whereby the animal was deemed significantly warmer post-manipulation (*M*=10.35, *SD*=2.64) than pre-manipulation (*M*=9.97, *SD*=2.45).

3.3.3.3. Univariate Effects of Diet and Gender on Competence.¹⁰

There were significant main effects of gender, F(1, 221) = 6.55, p = .01, $\eta_p^2 = .03$ (small-sized), and diet, F(1, 221) = 5.16, p = .02, $\eta_p^2 = .02$ (small-sized), on competence, whereby women viewed animals as significantly more competent than men did premanipulation, B = -.99, SE = .37, p = .01, and post-manipulation, B = -.8, SE = .37, p = .03, and non-meat consumers viewed animals as significantly more competent than meat consumers did pre-manipulation, B = -.74, SE = .34, p = .03, and post-manipulation, B = -.73, SE = .34, p = .03.

3.3.3.4. Univariate Effects of Time and Diet on Active Help.

Time had a significant effect on active help, F(1, 221) = 5.43, p = .02, $\eta_p^2 = .02$ (small-sized), whereby active help increased from pre-manipulation (*M*=15.88, *SD*=3.38) to post-manipulation (*M*=16.16, *SD*=3.31). There was also a significant main effect of diet on active help, F(1, 221) = 7.75, p = .01, $\eta_p^2 = .03$ (small-sized), whereby non-meat consumers showed significantly greater active help than meat consumers, pre-manipulation, B = -1.14, SE = .49, p = .02, and post-manipulation, B = -1.4, SE = .47, p = .003.

¹⁰When excluding univariate outliers, the main effect of diet on competence was no longer significant, F(1, 217) = 3.57, p = .06, $\eta_p^2 = .02$ (small-sized). Additionally, diet still had a significant effect pre-manipulation, B = -.64, SE = .32, p = .0497, but no longer had a significant effect post-manipulation, B = -.53, SE = .32, p = .1.

3.3.3.5. Univariate Effect of Diet on Passive Help.¹¹

There was a significant main effect of diet on passive help, F(1, 221) = 9.99, p = .002, $\eta_p^2 = .04$ (small-sized), whereby non-meat consumers showed significantly greater passive help than meat consumers, pre-manipulation, B = -.65, SE = .25, p = .01, and postmanipulation, B = -.87, SE = .26, p = .001.

3.3.3.6. Univariate Effect of Gender on Active Harm.

There was a significant main effect of gender on active harm, F(1, 221) = 8.91, p = .003, $\eta_p^2 = .04$ (small-sized), whereby men were significantly more willing to actively harm animals than women were pre-manipulation, B = .88, SE = .31, p = .01, and post-manipulation, B = .84, SE = .29, p = .004.

3.3.3.7. Univariate Effect of Time on Passive Harm.

There was a significant main effect of time on passive harm, F(1, 221) = 4.07, p = .045, $\eta_p^2 = .02$ (small-sized), whereby passive harm decreased from pre-manipulation (*M*=2.8, *SD*=1.15) to post-manipulation (*M*=2.63, *SD*=1.17).

3.3.4. Testing H5 and Exploratory Analysis: Exploring Effects of Behavioural and Subjective Self-Relevance on Post-Manipulation Dissonance, Post-Evaluation Dissonance and Reported Weekly Meat Consumption

A 2(behavioural self-relevance: present vs. absent) x 2(subjective self-relevance: present vs. absent) between-subjects MANCOVA, including diet and gender as covariates, was run on post-manipulation dissonance (to test **H5**), post-evaluation dissonance, and reported weekly meat consumption (see Table A3 in Appendix 4 for full inferential statistics).

¹¹When excluding univariate outliers, a main effect of time on passive help became significant, F(1, 214) = 5.96, p = .02, $\eta_p^2 = .03$ (small-sized). Specifically, participants were significantly more willing to passively help the animal post-manipulation (M=8.4, SD=1.66) than pre-manipulation (M=8.29, SD=1.65).

Descriptive statistics for effects of behavioural and subjective self-relevance are reported in Tables 3 and 4 respectively.

3.3.4.1. Multivariate Effects of Behavioural and Subjective Self-Relevance on Combined DVs.

Behavioural self-relevance, F(3, 219) = 4.47, p = .01, $\eta_p^2 = .06$ (medium-sized), and subjective self-relevance, F(3, 219) = 5.29, p = .002, $\eta_p^2 = .07$ (medium-sized), had significant main effects on the combined DVs. Diet, F(3, 219) = 88.16, p < .001, $\eta_p^2 = .55$ (large-sized), and gender, F(3, 219) = 2.91, p = .04, $\eta_p^2 = .04$ (small-sized), also had significant main effects on the combined DVs. I followed up these significant multivariate effects with univariate ANCOVAs on the DVs below.

3.3.4.2. H5: Univariate Effects of Behavioural and Subjective Self-Relevance on Post-Manipulation Dissonance.

Supporting H5: Behavioural self-relevance had a significant effect on postmanipulation dissonance, F(1, 221) = 12.51, p < .001, $\eta_p^2 = .05$ (small-to-medium-sized), whereby participants reported greater post-manipulation dissonance when behavioural selfrelevance was present than when behavioural self-relevance was absent. That is, participants felt greater discomfort when asked to imagine agreeing to consume the meat than when asked to imagine refusing to consume the meat. **Yet contradicting H5:** Subjective self-relevance had a significant effect on post-manipulation dissonance, F(1, 221) = 9.01, p = .003, $\eta_p^2 = .04$ (small-sized), in the opposite direction to predicted. Specifically, participants reported greater dissonance when subjective self-relevance was absent than when subjective self-relevance was present. That is, participants felt greater discomfort when asked to imagine disliking the meat than when asked to imagine liking the meat. There were also significant main effects of gender, F(1, 221) = 5.29, p = .02, $\eta_p^2 = .02$ (small-sized), and diet, F(1, 221) = 10.73, p =.001, $\eta_p^2 = .05$ (small-to-medium-sized), on post-manipulation dissonance, whereby women, B = -1.42, SE = .62, and non-meat consumers, B = -1.88, SE = .57, experienced significantly greater post-manipulation dissonance than men and meat consumers respectively.

3.3.4.3. Univariate Effects of Behavioural and Subjective Self-Relevance on Post-Evaluation Dissonance.

Neither behavioural self-relevance, F(1, 221) = .78, p = .38, $\eta_p^2 = .004$, nor subjective self-relevance, F(1, 221) = .83, p = .36, $\eta_p^2 = .004$, had a significant effect on post-evaluation dissonance.

3.3.4.4. Univariate Effects of Behavioural and Subjective Self-Relevance on Reported Weekly Meat Consumption.

Neither behavioural self-relevance, F(1, 221) = .3, p = .58, $\eta_p^2 = .001$, nor subjective self-relevance, F(1, 221) = 1.21, p = .27, $\eta_p^2 = .01$, had a significant effect on reported weekly meat consumption. There was therefore no evidence for under-reporting across behavioural and subjective self-relevance conditions. There was a significant effect of diet on reported weekly meat consumption, F(1, 221) = 257.58, p < .001, $\eta_p^2 = .54$ (large-sized), whereby, unsurprisingly, meat consumers (vs. non-meat consumers) reported greater weekly meat consumption, B = 4.19, SE = .26.

Table 3.

Descriptive Statistics for Effect of Behavioural Self-Relevance on Post-Manipulation Dissonance, Post-Evaluation Dissonance, and Reported Weekly Meat Consumption.

DV	Present	Present SD	Absent	Absent SD	
	Mean		Mean		
Post-Manipulation	10.63	4.18	8.78	4.08	
Dissonance**					
Post-Evaluation Dissonance	6.28	3.33	5.91	3.46	
Reported Weekly Meat	3.59	2.7	3.39	2.6	
Consumption					
Note: Present refers to behavioural self-relevance being present, whilst absent refers to					

behavioural self-relevance being absent. **p*<.05, ***p*≤.01, ****p*≤.001

Table 4.

Dissonance, I osi-Evaluation Dissonance, and Reported Weekly Meal Consumption.					
DV	Present	Present SD	Absent	Absent SD	
	Mean		Mean		
Post-Manipulation	9.04	4.18	10.26	4.19	
Dissonance***					
Post-Evaluation Dissonance	6.34	3.41	5.85	3.38	
Reported Weekly Meat	3.4	2.76	3.56	2.53	
Consumption					
<i>Note:</i> Present refers to subjective self-relevance being present, whilst absent refers to					
subjective self-relevance being absent. $*p < .05$. $**p < .01$. $***p < .001$					

Descriptive Statistics for Effect of Subjective Self-Relevance on Post-Manipulation Dissonance, Post-Evaluation Dissonance, and Reported Weekly Meat Consumption.

3.3.5. Testing H6-H9: Exploring Mediational Relationships Between Behavioural and

Subjective Self-Relevance, Dissonance, Warmth and Active Behavioural Intentions

Per motivated cognition theory, I hypothesised post-manipulation dissonance would mediate the relationship between behavioural (**a**) and subjective self-relevance (**b**) and postmanipulation warmth (**H6**), and that post-manipulation warmth would mediate the relationship between post-manipulation dissonance and post-evaluation dissonance (**H7**). Per the BIAS map, I predicted post-manipulation warmth would mediate a) the relationship between behavioural self-relevance and post-manipulation active harm and post-manipulation active help (**H8**), and b) the relationship between subjective self-relevance and postmanipulation active harm and post-manipulation active help (**H9**).

To test **H6** and **H8-H9**, I dummy coded behavioural self-relevance and subjective self-relevance into two dummy variables. That is, behavioural self-relevance was coded as absence (zero) and presence of behavioural self-relevance (one), whilst subjective self-relevance was coded as absence (zero) and presence of subjective self-relevance (one). Seven individual bootstrapped mediation analyses (10,000 bootstrapped samples) were run using Model 4 in PROCESS through SPSS (Hayes, 2022).¹² To control for Type I error from

¹²The causal steps approach (Baron & Kenny, 1986) is not utilised for any mediation analyses throughout this thesis due to multiple criticisms of this approach (Hayes, 2009; Pardo & Román, 2013; Zhao et al., 2010).

multiple mediation testing, I utilised an adjusted confidence interval of 99%. Significant mediations are interpreted through confidence intervals which do not contain zero per Hayes (2018).

Note that competence was not included in mediational hypotheses, as Study 1 found that behavioural and subjective self-relevance were associated only with warmth, and not with competence. Additionally, passive behavioural intentions were not included within mediational hypotheses or analyses, as passive behavioural intentions are usually linked to competence rather than warmth, and warmth does not always inform passive behavioural intentions (Sevillano & Fiske, 2016b). However, as post-manipulation warmth and postmanipulation competence were significantly correlated with each other within the current study as revealed through a Pearson correlation, r = .37, p < .001, and warmth and competence are conceptually related, post-manipulation competence was included as a parallel mediator within mediation analyses below which included post-manipulation warmth as a mediator (i.e., **H7-H9**). Inclusion of competence as a parallel mediator alongside warmth allows for testing of any mediational roles of warmth *whilst* controlling for competence (Hayes, 2018).

3.3.5.1. H6a: Mediation of Behavioural Self-Relevance on Post-Manipulation Warmth through Post-Manipulation Dissonance.

Not supporting H6a: There was no significant indirect effect of behavioural selfrelevance on post-manipulation warmth through post-manipulation dissonance, b = .16, SE = .1, 99% CI [-.06, .5]. Thus, post-manipulation dissonance did not significantly mediate the relationship between behavioural self-relevance and post-manipulation warmth.

3.3.5.2. H6b: Mediation of Subjective Self-Relevance on Post-Manipulation Warmth through Post-Manipulation Dissonance.

Not supporting H6b: There was no significant indirect effect of subjective selfrelevance on post-manipulation warmth through post-manipulation dissonance, b = -.09, SE =.07, 99% CI [-.34, .07]. Thus, post-manipulation dissonance did not significantly mediate the relationship between subjective self-relevance and post-manipulation warmth.

3.3.5.3. H7: Mediation of Post-Manipulation Dissonance on Post-Evaluation Dissonance through Post-Manipulation Warmth.

Not supporting H7: There was no significant indirect effect of post-manipulation dissonance on post-evaluation dissonance through post-manipulation warmth when controlling for post-manipulation competence, b = -.004, SE = .01, 99% CI [-.04, .03].¹³ Therefore, post-manipulation warmth did not significantly mediate the relationship between post-manipulation dissonance and post-evaluation dissonance.

3.3.5.4. H8a: Mediation of Behavioural Self-Relevance on Post-Manipulation Active Harm through Post-Manipulation Warmth.

Not supporting H8a: There was no significant indirect effect of behavioural selfrelevance on post-manipulation active harm through post-manipulation warmth when controlling for post-manipulation competence, b = .003, SE = .03, 99% CI [-.08, .1]. Therefore, post-manipulation warmth did not significantly mediate the relationship between behavioural self-relevance and post-manipulation active harm. Additionally, inspection of path *b* indicated post-manipulation warmth did not significantly predict active harm, b = -.07, SE = .05, p = .19, 99% CI [-.2, .07].

¹³The figures reported for the mediation analysis assessing if warmth mediates the relationship between postmanipulation dissonance and post-evaluation dissonance are fully standardised, as all variables within this mediation analysis are continuous. All other mediation analyses within this study, and throughout the rest of this thesis, are reported with non-standardised figures due to the inclusion of nominal IVs (Hayes, 2018).

3.3.5.5. H8b: Mediation of Behavioural Self-Relevance on Post-Manipulation Active Help through Post-Manipulation Warmth.

Not supporting H8b: There was no significant indirect effect of behavioural selfrelevance on post-manipulation active help through post-manipulation warmth when controlling for post-manipulation competence, b = -.03, SE = .24, 99% CI [-.66, .59]. Therefore, post-manipulation warmth did not significantly mediate the relationship between behavioural self-relevance and post-manipulation active help. Despite lack of mediation, inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater post-manipulation active help, b = .67, SE = .07, p < .001, 99% CI [.49, .85].

3.3.5.6. H9a: Mediation of Subjective Self-Relevance on Post-Manipulation Active Harm through Post-Manipulation Warmth.

Not supporting H9a: There was no significant indirect effect of subjective selfrelevance on post-manipulation active harm through post-manipulation warmth when controlling for post-manipulation competence, b = .05, SE = .04, 99% CI [-.03, .21]. Therefore, post-manipulation warmth did not significantly mediate the relationship between subjective self-relevance and post-manipulation active harm. Additionally, inspection of path b indicated post-manipulation warmth did not significantly predict post-manipulation active harm, b = -.08, SE = .05, p = .14, 99% CI [-.21, .06].

3.3.5.7. H9b: Mediation of Subjective Self-Relevance on Post-Manipulation Active Help through Post-Manipulation Warmth .

Not supporting H9b: There was no significant indirect effect of subjective selfrelevance on post-manipulation active help through post-manipulation warmth when controlling for post-manipulation competence, b = -.46, SE = .24, 99% CI [-1.12, .17]. Therefore, post-manipulation warmth did not significantly mediate the relationship between subjective self-relevance and post-manipulation active help. Despite lack of mediation, inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater post-manipulation active help, b = .68, SE = .07, p < .001, 99% CI [.5, .86].

3.4. Discussion

3.4.1. Discussion of Findings

3.4.1.1. H1-H5: Effects of Behavioural and Subjective Self-Relevance on Warmth, Active Harm, Active Help and Post-Manipulation Dissonance.

Neither behavioural nor subjective self-relevance had a multivariate effect on the combined DVs. Whilst effects of behavioural or subjective self-relevance on competence, passive help, and passive harm were not hypothesised, these null findings may not support H1 and H2a/b for warmth, H3a and H4a for active harm, and H3b and H4b for active help. That is, the null findings may not support hypotheses whereby subjectively self-relevant animals (vs. not) were expected to be deemed less warm (H1), and subject to greater active harm (H3a) and less active help (H3b), whilst behaviourally self-relevant animals (vs. not) were expected to be deemed less (H2a) or more warm (H2b), and, if deemed less warm, subject to greater active harm (H4a) and less active help (H4b). These null findings may therefore not support previous literature (Loughnan et al., 2010; Piazza & Loughnan, 2016), and may not align with Study 1 findings regarding the relationships between behavioural and subjective self-relevance and warmth. Overall, the null findings regarding behavioural and subjective self-relevance may arise from lack of effectiveness of the text manipulations in manipulating behavioural or subjective self-relevance. That is, the texts may have failed to affect perceptions of animals or behavioural intentions towards them due to an inability of the texts to elicit behavioural and subjective self-relevance. I therefore conducted a follow-up study to test effectiveness of the current self-relevance texts in manipulating behavioural and subjective self-relevance (Section 4.2.).

Partially supporting **H5**, and aligning with motivated cognition literature (Loughnan et al., 2010, 2014; Piazza & Loughnan, 2016), behavioural self-relevance (vs. none) increased post-manipulation dissonance (small-to-medium-sized). That is, imagining agreeing to consume the meat caused greater discomfort than imagining refusing to consume the meat. However, partly not supporting **H5** and this previous literature, subjective self-relevance (vs. none) decreased post-manipulation dissonance (small-sized) and was thus in the opposite direction to predicted. That is, imagining disliking the meat caused more discomfort than imagining liking the meat. This finding may have arisen from the dissonance scale measuring general discomfort instead of actual dissonance. That is, the scenario may have elicited discomfort due to participants imagining disliking a food they had just imagined consuming (Section 3.4.2. provides further discussion). The subjective self-relevance effect on postmanipulation dissonance was small-sized only, so may have lower practical significance.

3.4.1.2. H6-H9: Mediational Relationships Between Behavioural and Subjective Self-Relevance, Dissonance, Warmth and Active Behavioural Intentions.

Findings may not support the mediational hypotheses (**H6-H9**). That is, postmanipulation dissonance did not mediate relationships between behavioural self-relevance and post-manipulation warmth, nor between subjective self-relevance and post-manipulation warmth (**H6**). Post-manipulation warmth did not mediate the relationship between postmanipulation dissonance and post-evaluation dissonance (**H7**). Post-manipulation warmth did not mediate relationships between behavioural self-relevance and post-manipulation active help, nor between behavioural self-relevance and post-manipulation active hsp, post-manipulation warmth did not mediate relationships between subjective selfrelevance and post-manipulation active help, nor between subjective selfrelevance and post-manipulation active help, nor between subjective selfrelevance and post-manipulation active help, nor between subjective self-relevance and postmanipulation active harm (**H9**). Lack of support for **H8** and **H9** may reflect broader nonsignificant effects of behavioural and subjective self-relevance on perceptions of animals and behavioural intentions towards them (e.g., as identified within the main MANCOVA). Meanwhile, lack of support for H6 and H7 may indicate that motivated cognition is not evidenced. Specifically, lack of support for H6 may indicate that dissonance does not motivate reduced perceptions of animals' warmth after behavioural and subjective selfrelevance are elicited. Furthermore, lack of support for H7 may indicate people do not intentionally dehumanise animals following behavioural and subjective self-relevance by decreasing the animals' warmth to alleviate dissonance. These findings may not support previous literature regarding the 'meat paradox' and motivated cognition (Bilewicz et al., 2011; Piazza & Loughnan, 2016). Combined with null effects of behavioural and subjective self-relevance on all SCM/BIAS map DVs, and equivocal effects of behavioural and subjective self-relevance on dissonance, these findings may not support motivated cognition theory as applied to animals (Piazza & Loughnan, 2016). However, it is unclear if motivated cognition is genuinely not evidenced or whether motivated cognition could not be effectively tested here due to limitations with the behavioural and subjective self-relevance manipulations and measures used (e.g., for dissonance; Section 3.4.2.).

3.4.1.3. Exploring Associations Between Warmth and Active Behavioural Intentions.

Despite lack of significant mediations, greater warmth was associated with greater active help within the behavioural and subjective self-relevance mediation analyses, aligning with the BIAS map (Sevillano & Fiske, 2016b). However, not supporting the BIAS map, warmth was not associated with active harm within either the behavioural or subjective selfrelevance mediation analyses. Whilst the consistent relationship between warmth and active help supports the BIAS map, the lack of relationship between warmth and active harm may indicate active harm is not always informed by warmth. This finding is surprising based on previous literature exploring the SCM/BIAS map with animals, which has consistently found negative effects of warmth on active harm (Sevillano & Fiske, 2016b, 2019).

3.4.1.4. Unexpected Effects of Time on Warmth, Active Help, and Passive Harm.

There were unexpected effects of time on warmth, active help, and passive harm (small-sized). Specifically, regardless of the content of the text read by participants (e.g., behavioural and subjective self-relevance), simply reading about the animal increased warmth and active help, and decreased passive harm, towards the animal from pre- to postmanipulation. Thus, this finding indicates reading information even about imagining consuming an animal may counterintuitively improve the animal's warmth, increase active help, and decrease passive harm towards the animal. This finding therefore stands in contrast to extensive previous literature finding that people struggle with the 'meat paradox', whereby they use various strategies to avoid thinking about the harm they cause to animals they consume (see Gradidge et al., 2021a), and that reminders of this harm (e.g., consumption) elicit moral disengagement from animals (Graça et al., 2014). Instead, these findings indicate reading about the scenario, regardless of behavioural or subjective self-relevance, elicited moral engagement in warmth and active help, and a reduction in passive harm. Furthermore, passive harm may have reduced due to the nature of the passive harm items, whereby participants were less able to 'ignore' the animal after reading the manipulation. These findings are small-sized only, so may have lower practical significance. Additionally, the effect on passive harm should be interpreted with caution due to inadequate scale reliability.

3.4.1.5. Effects of Behavioural and Subjective Self-Relevance on Reported Meat Consumption.

Neither behavioural self-relevance nor subjective self-relevance had effects on reported meat consumption. Thus, neither behavioural self-relevance nor subjective self-

relevance caused under-reporting of meat consumption. This finding does not, however, elucidate if there was under-reporting present across all conditions (e.g., Rothgerber, 2019).

3.4.2. Limitations of the Current Study and Directions for Future Research

The current study has some limitations, which are addressed in subsequent pilot studies (Pilot Studies 1a, 2a, 3a, 4a; Sections 4.2. to 4.5.) and/or Study 3: a) lack of manipulation checks, b) possible measurement of general discomfort instead of dissonance, and c) possible elicitation of food neophobia.

Firstly, the current study assumed effectiveness of the texts in manipulating behavioural self-relevance and subjective self-relevance, instead of testing this effectiveness explicitly. Thus, null effects on most DVs may arise from a lack of effectiveness of the behavioural and subjective self-relevance manipulations rather than from genuine lack of effects of behavioural and subjective self-relevance. Therefore, to test this possible explanation, a follow-up study (Section 4.2.) aims to test if the behavioural and subjective self-relevance text manipulations used within the current study fail to effectively manipulate behavioural and subjective self-relevance.

Secondly, the dissonance scale may have measured general discomfort instead of dissonance. For example, the scale constitutes the items 'uncomfortable', 'uneasy', and 'bothered', and so theoretically could have measured participants' (non-dissonance-related) discomfort, possibly explaining why the main effect of subjective self-relevance on post-manipulation dissonance was in the opposite direction to predicted. That is, imagining disliking the meat may have caused greater post-manipulation dissonance simply because imagining consuming a food they dislike was uncomfortable for participants. Thus, the dissonance measure may have been measuring broader general discomfort instead of moral discomfort. Whilst the Dissonance Affect Questionnaire was again included in Study 3 to

maintain consistency across studies, the variable of 'dissonance' is hereby renamed discomfort to reflect the questionnaire's possible measurement of discomfort instead of dissonance.

Thirdly, related to the above limitations, the current behavioural and subjective selfrelevance manipulation may have elicited *food neophobia* (anxiety and hesitancy about consuming new foods; Damsbo-Svendsen et al., 2017), which applies especially to novel types of meat (Çınar et al., 2021). Thus, the manipulation, especially within the behavioural self-relevance and lack of subjective self-relevance condition (imagining consuming and disliking the meat), may have inadvertently reduced willingness to imagine the situation due to a reluctance to engage with novel meats. To better address this limitation, I implement an alternative behavioural self-relevance manipulation in Study 3 (piloted in Pilot Studies 1a, 2a, 3a, 4a, Sections 4.2. to 4.5.) using the common dish of spaghetti bolognese instead of meat specifically, which may reduce food neophobia, as food neophobia is especially sensitive to novel meats.

3.4.3. Conclusion

Overall, the current experiment indicates behavioural and subjective self-relevance may not affect an animal's warmth and competence nor behavioural intentions towards the animal. Therefore, a follow-up study will be conducted (Follow-Up/Pilot Study 1; Section 4.2.) to test if the behavioural and subjective self-relevance manipulations utilised within the current study are failing to manipulate behavioural and subjective self-relevance. If so, this failure in the manipulations may explain the null results, as opposed to non-significant effects of behavioural and subjective self-relevance themselves.

Regarding the mediation analyses findings, although greater warmth was consistently associated with greater active help in mediation analyses, warmth was not associated with

active harm. Thus, the current experiment provides some support for the BIAS map, but also indicates warmth and active harm may be more separable than originally theorised. That is, unlike what would be expected following previous SCM/BIAS map research with animals (Sevillano & Fiske, 2016b), greater warmth may not always link to decreased active harm.

Additionally, whilst behavioural self-relevance did increase post-manipulation dissonance like originally hypothesised, the experiment does not provide support for the full motivated cognition theory. Furthermore, subjective self-relevance had the opposite effect to behavioural self-relevance on post-manipulation dissonance, whereby imagining disliking the meat caused greater dissonance, indicating either that behavioural and subjective selfrelevance have differential, opposite effects, or that the dissonance scale used here has limitations (e.g., by measuring general discomfort instead of dissonance). There was no evidence for under-reporting of meat consumption within any specific behavioural or subjective self-relevance condition.

Overall, the current experiment indicates behavioural and subjective self-relevance, at least as manipulated here, may not inform peoples' perceptions of animals (warmth, competence), nor behavioural intentions towards them (active help, passive help, active harm, passive harm). The current study has limitations regarding lack of manipulation checks, the dissonance measure, and possible elicitation of food neophobia. Follow-Up/Pilot Study 1 (Section 4.2.) therefore acts as a follow-up to the current study by testing if the behavioural and subjective self-relevance manipulations used within the current study fail to effectively manipulate behavioural and subjective self-relevance, and whether this possible failed manipulation therefore explains the current study's null results.

Chapter 4. Pilot Studies and Study 3 Exploring Behavioural and Subjective Self-Relevance

4.1. Introduction

Following null effects of the behavioural and subjective self-relevance manipulations in Study 2, the current chapter presents four pilot studies (Pilot Studies 1a, 2a, 3a, 4a) to a) test if these null effects may be arising from ineffectiveness of the original behavioural and subjective self-relevance texts in manipulating behavioural and subjective self-relevance (Follow-Up/Pilot Study 1a) and b) develop and pilot alternative behavioural and subjective self-relevance manipulations which do successfully manipulate behavioural and subjective self-relevance (Pilot Studies 2a, 3a, 4a). An alternative and successfully piloted behavioural and subjective self-relevance manipulation is then implemented within Study 3 in this chapter, to again test if behavioural and subjective self-relevance are causes of pet speciesism. These studies are now reported in turn below.

4.2. Follow-Up/Pilot Study 1a to Test Behavioural and Subjective Self-Relevance Manipulations

4.2.1. Introduction and Extension from Study 2

As behavioural and subjective self-relevance manipulations revealed null effects on SCM/BIAS map variables in Study 2, the current pilot study (Pilot Study 1a) therefore aims to determine if these null effects may be arising from failure of the texts to manipulate behavioural and subjective self-relevance. That is, the current pilot study aims to test if the behavioural and subjective self-relevance manipulations used within Study 2 are (in)effective at manipulating their intended variables, whereby behavioural and subjective self-relevance are again manipulated through text manipulations regarding a 'target animal' previously unknown to the participant. Specifically, participants are presented with an imaginary scenario where they are offered meat from an unknown animal, and they are asked to imagine

either that they like/dislike this meat (manipulation of subjective self-relevance) and that they agree/refuse to consume this meat (manipulation of behavioural self-relevance). To check effectiveness of the behavioural self-relevance manipulation (manipulation check), the item *'How many days a week would you be willing to eat products made from tree*

kangaroos/fossas/tamanduas? ' was included to measure behavioural self-relevance. To check effectiveness of the subjective self-relevance manipulation (manipulation check), the first two items from the Product Involvement Scale were included to measure subjective self-relevance. No hypotheses are made for this follow-up/pilot study, as this pilot study is instead exploratory to determine (in)effectiveness of the behavioural and subjective self-relevance manipulations.

Note that this pilot study utilises the same participant sample to also pilot text manipulations for familiarity, similarity, and pet status, but only information relevant to the behavioural and subjective self-relevance manipulations is reported within the current chapter (see Section 5.2. for pilot study information about the similarity and familiarity manipulations [Pilot Study 1b], and Section 7.2. for pilot study information about the pet status manipulation [Pilot Study 1c]).

4.2.2. Methods

4.2.2.1. Participants.

Sixty-eight participants were recruited via social media and Sona as a volunteer sample. Thirteen participants were excluded for providing partial data. As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 4.2.3. for further discussion). Exclusions left a total sample size of 55 participants (81.8% women, 18.2% men; M_{age} =23.38, SD_{age} =5.68, age range: 18-48), which exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Participants within each condition are shown in

Table 5. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For dietary group: 69.1% (meat consumer), 12.7% (reducetarian), 7.3% (vegetarian), and 5.5% each (flexitarian, pescatarian). For nationality: 54.5% (British), 5.5% (Romanian), 3.6% each (German, Indian, Lithuanian, no response, Portuguese, Spanish), and 18.4% other. For ethnicity: 76.1% (White), 12.6% (Asian), 3.6% each (mixed, would rather not say), and 4.1% other. For country of residence ($M_{duration}$ =17.91, $SD_{duration}$ =10.12, range: 1-48 years): 81.8% (United Kingdom), 3.6% (Germany), and 14.6% other. For religion: 29.1% (atheism), 21.8% (Christianity), 18.2% (agnosticism), 14.5% (would rather not say), 5.5% (no response), 3.6% each (Buddhism, Islam), and 3.7% other.

Table 5.

Participants Within Each Condition for Pilot Study 1a.

Behavioural Self-Relevance	Subjective Self-Relevance	Number of				
Condition	Condition	Participants				
Present	Present	14				
Present	Absent	12				
Absent	Present	15				
Absent	Absent	14				

4.2.2.2. Design.

The current follow-up/pilot study follows a 2(subjective self-relevance: present vs. absent) x 2(behavioural self-relevance: present vs. absent) between-subjects ANCOVA design, with measured behavioural and subjective self-relevance as the DVs. Behavioural and subjective self-relevance were included as DVs to test effectiveness of the manipulations. Gender was included as a covariate (see Section 4.2.3. for further discussion).

4.2.2.3. Materials.

4.2.2.3.1. Behavioural and Subjective Self-Relevance Manipulations.

Behavioural and subjective self-relevance were manipulated as in Study 2. That is, through an imagined scenario about the participant's target animal as follows: '*Imagine that*

you like/dislike tree kangaroo meat. You have been offered tree kangaroo meat again. Imagine you agree/refuse to consume the tree kangaroo meat.'. Subjective self-relevance is manipulated through use of the word 'like' (subjective self-relevance) or 'dislike' (lack of subjective self-relevance). Behavioural self-relevance is manipulated through use of the word 'agree' (behavioural self-relevance) or 'refuse' (lack of behavioural self-relevance).

4.2.2.3.2. Measured Behavioural Self-Relevance.

Like Study 1, measured behavioural self-relevance was measured through a single item developed by the researcher '*How many days a week would you be willing to eat products made from tree kangaroos/fossas/tamanduas*? (0-7 days). A higher score indicates greater behavioural self-relevance.

4.2.2.3.3. Measured Subjective Self-Relevance (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009).

Like Study 1, measured subjective self-relevance was measured using an adapted version of the Product Involvement Scale (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009), whereby participants indicated their level of agreement with statements regarding products from the target animal on a Likert scale from one (*'strongly disagree'*) to seven (*'strongly agree'*). This scale was adapted by a) utilising a Likert scale instead of a semantic differential to reduce cognitive effort and ease scale completion (Friborg et al., 2006), and b) referring to products from the target animal specifically. A sample item is *'I am very interested in products made from tree kangaroos/fossas/tamanduas'*. Due to inadequate loading and communality of the third item, alongside poor correlations with the first two items, found in Study 1, only the first two statements from the Product Involvement Scale were utilised within the current study to measure subjective self-relevance. Therefore, both items were summed to create a two-item subjective self-relevance index, with higher scores indicating greater subjective self-relevance of the target animal (i.e., greater involvement with

products from the target animal). Neither item is reverse-scored. Reliability was inadequate within this study per the Spearman-Brown coefficient (r_{sb} =.68). As this scale only contains two items, I did not run a PAFA for this scale, and I was unable to remove an item to improve reliability. Instead, I retained both items in the scale, and findings using this scale should be interpreted with caution due to inadequate reliability.

4.2.2.3.4. Attention and Memory Checks.

Attention paid to and memory for the text manipulations was tested through five attention and memory check items developed by the researcher as follows: '*Was the text you just read descriptive text or did it ask you to imagine a scenario*?', '*What animal did the text refer to*?', '*Did the text ask you to imagine that you were or were not offered tree kangaroo/fossa/tamandua meat*?', '*Did the text that you just read ask you to imagine that you like or do not like tree kangaroo/fossa/tamandua meat*?', and '*Did the text that you just read ask you to imagine that you agreed or refused to consume tree kangaroo/fossa/tamandua meat again*?'. These checks aimed to test that participants paid attention to and remembered the texts. Correct answers for each item received a score of one, whilst incorrect answers for each item received a score of zero. All answers were then added together, for a highest possible score of five. Inspection of these checks revealed that the vast majority of the sample (>75%) had perfect scores, indicating sufficient attention and memory for these manipulations.

4.2.2.3.5. Vividness of the Self-Relevance Manipulation.

Perceived vividness of the imagined self-relevance manipulation was measured, as imagined scenarios which are viewed as more vivid have greater impacts on behavioural intentions (Husnu & Crisp, 2010). Vividness was measured through a six-item semantic differential vividness measure from Husnu and Crisp (2010; α =.94), whereby participants answer how much they perceive their imagined scenario as being '*faint vs. vivid*', '*fuzzy vs.* clear', 'dim vs. bright', 'vague vs. sharp', 'dull vs. lively', and 'simple vs. detailed'. Reliability was adequate for this scale within this study, ω =.93. Therefore, items were summed together to create a total vividness score. Higher scores indicate greater vividness. No items are reverse-scored. The majority of the sample (>50%) perceived the self-relevance manipulations to be vivid.

4.2.2.3.6. Ease of Imaginability of the Self-Relevance Manipulation.

Perceived ease of imaginability of the imagined self-relevance manipulation was measured in order to ensure participants were able to imagine the scenarios without difficulty. Ease of imaginability was measured through a single item '*How easy or difficult was it for you to imagine the previous scenario in the text you have just read*?' on a -50 (extremely difficult) to 50 (extremely easy) visual analogue scale, informed by previous literature (e.g., Black & Barnes, 2020; Broemer & Diehl, 2004). A higher score indicates greater ease of imaginability. The majority of the sample (>50%) perceived the self-relevance manipulations to be easy to imagine.

4.2.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in September and October 2020 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos (alongside a photograph of the animal), followed by a fossa and tamandua if participants recognised any previous animal. Animal species names were used throughout the study. The study either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants completed the three sections (self-relevance, familiarity/similarity, and pet status) of the pilot study in a randomised order.

Here, only the self-relevance section is described (see Section 5.2. for description of the familiarity/similarity section, and Section 7.2. for description of the pet status section). For the self-relevance section, participants were randomly assigned to read one of the four behavioural and subjective self-relevance text manipulations described in Section 4.2.2.3.1. regarding their target animal. After reading their self-relevance text, participants answered the five attention and memory checks, indicated their perceived vividness and ease of imaginability of the imagined scenario, and then completed the measures for behavioural and subjective self-relevance (manipulation checks). Finally, participants provided demographics before being debriefed. One participant reported technical difficulties, but their response was complete and maintained within analyses.

4.2.3. Analyses

Two 2(subjective self-relevance: present vs. absent) x 2(behavioural self-relevance: present vs. absent) between-subjects ANCOVAs, including gender as a covariate, were conducted on measured behavioural and subjective self-relevance respectively, to test if the manipulations managed or failed to effectively manipulate behavioural and subjective selfrelevance. Note that behavioural and subjective self-relevance were both included as IVs within these analyses, rather than separate one-way ANCOVAs (one each for subjective and behavioural self-relevance), to test for and therefore ensure that the behavioural and subjective self-relevance manipulations do not interact. Whilst the DVs should theoretically be conceptually related (i.e., different forms of self-relevance), they were not significantly correlated within the current study, r = .13, p = .34, and therefore separate ANCOVAs are conducted rather than a MANCOVA.

Gender was again included as a covariate within these ANCOVAs. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases

power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured behavioural/subjective self-relevance (the outcome variables) to be statistically controlled for.

4.2.3.1. Statistical Assumptions for Manipulation Checks: Exploring Effects of Behavioural and Subjective Self-Relevance Manipulations on Measured Behavioural and Subjective Self-Relevance (2x2 ANCOVAs).

Both DVs failed normality tests across most conditions, ps < .05. Measured subjective self-relevance had acceptable skewness, whilst measured behavioural self-relevance only had acceptable skewness in the manipulation whereby behavioural self-relevance was absent and subjective self-relevance was present. Measured subjective self-relevance had non-extreme univariate outliers, but these did not cause excessive skewness. Running a follow-up univariate ANCOVA including and excluding univariate outliers for measured subjective self-relevance did change one major finding (Footnote 14). I report the measured subjective self-relevance ANCOVA including outliers to reflect the unamended dataset. Excessive skewness, multiple univariate outliers, and floor effects in measured behavioural selfrelevance meant assumptions for the ANCOVA on this DV were violated. However, I proceeded with this ANCOVA due to robustness of ANCOVA to non-normality and lack of non-parametric alternatives. Results from this ANCOVA should be interpreted with caution. Homogeneity of variances was present for both DVs, ps > .05.

4.2.3.2. Manipulation Checks: Exploring Effects of Behavioural and Subjective Self-Relevance Manipulations on Measured Behavioural and Subjective Self-Relevance.

Two 2(behavioural self-relevance: present vs. absent) x 2(subjective self-relevance: present vs. absent) between-subjects ANCOVAs, including gender as a covariate, were run on measured behavioural and subjective self-relevance to test manipulation checks.

4.2.3.2.1. Univariate Effects of Behavioural and Subjective Self-Relevance Manipulations on Measured Behavioural Self-Relevance.

The ANCOVA revealed that the behavioural self-relevance manipulation did not manipulate measured behavioural self-relevance, either as a main effect, F(1, 50) = .22, p = .64, $\eta_p^2 = .004$, or as an interaction with the subjective self-relevance manipulation, F(1, 50) = .12, p = .73, $\eta_p^2 = .002$.

4.4.2.2. Univariate Effects of Behavioural and Subjective Self-Relevance Manipulations on Measured Subjective Self-Relevance.¹⁴

The ANCOVA revealed that the subjective self-relevance manipulation did not manipulate measured subjective self-relevance, either as a main effect, F(1, 50) = .07, p = .8, $\eta_p^2 = .001$, or as an interaction with the behavioural self-relevance manipulation, F(1, 50) = 1.12, p = .3, $\eta_p^2 = .02$.

¹⁴When excluding outliers, the main effect of manipulated subjective self-relevance on measured subjective self-relevance remained non-significant, F(1, 47) = .75, p = .39, $\eta_p^2 = .02$. However, the interaction between the behavioural and subjective self-relevance manipulations on measured subjective self-relevance became significant, F(1, 47) = 5.33, p = .03, $\eta_p^2 = .1$ (medium-to-large-sized). *Post hoc* pairwise comparisons revealed that, when manipulated subjective self-relevance was present only, measured subjective self-relevance was greater when manipulated behavioural self-relevance was present (M=2.26, SE=.63), p = .02. Additionally, when manipulated behavioural self-relevance was present (M=2.26, SE=.63), p = .02. Additionally, when manipulated subjective self-relevance was greater when manipulated subjective self-relevance was present (M=2.26, SE=.63), p = .02. Additionally, when manipulated subjective self-relevance was greater when manipulated subjective self-relevance was present (M=2.26, SE=.63), p = .02. Additionally, when manipulated subjective self-relevance was greater when manipulated subje

4.2.4. Discussion and Conclusion

The current follow-up/pilot study found that the manipulations from Study 2 may not be effective in manipulating behavioural and subjective self-relevance, suggesting that null effects from Study 2 may be arising from ineffectiveness of the manipulations. These manipulations may have failed due to food neophobia: Specifically, the manipulations may have inadvertently reduced willingness to imagine the situation due to a reluctance to engage with novel meats (e.g., Çınar et al., 2021; Damsbo-Svendsen et al., 2017). To overcome this limitation, Pilot Studies 2a, 3a and 4a (Sections 4.3. to 4.5.) test an alternative self-relevance manipulation which utilises an imagined restaurant scenario instead, and uses the common dish of spaghetti bolognese instead of meat specifically as a way to potentially reduce food neophobia.

4.3. Pilot Study 2a to Test Amended Behavioural and Subjective Self-Relevance Manipulations

4.3.1. Introduction

Due to the previous behavioural and subjective self-relevance manipulations failing to effectively manipulate behavioural and subjective self-relevance, the current pilot study aims to test and pilot alternative behavioural and subjective self-relevance manipulations developed by the current researcher. Specifically, behavioural and subjective self-relevance are now manipulated through an imagined restaurant scenario whereby participants imagine consuming a meat dish, which is either described favourably (subjective self-relevance) or unfavourably (lack of subjective self-relevance) and is later revealed to contain meat from the participant's target animal (manipulation of behavioural self-relevance). These manipulations are designed to overcome any issues with food neophobia from the previous manipulations (e.g., lack of willingness to try new forms of meat; Çınar et al., 2021; Damsbo-Svendsen et al., 2017) by a) referring to the dish of spaghetti bolognese instead of meat directly, and b)
revealing what animal the meat comes from only at the end of the manipulation (as opposed to the original manipulations which stated the source of the meat straightaway). As with Study 2, an unknown animal (tree kangaroo) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of familiarity. To check effectiveness of the behavioural self-relevance manipulation (manipulation check), the single item '*How many days a week would you be willing to eat meat from this species*?' is included to measure behavioural self-relevance. To check effectiveness of the subjective self-relevance manipulation (manipulation check), the single item '*How much do you like the spaghetti bolognese presented in this scenario*?' is included to measure subjective self-relevance. Designed to reduce food neophobia (Çınar et al., 2021; Damsbo-Svendsen et al., 2017), I hypothesise that:

H1: Imagining consuming a meat dish will significantly increase measured behavioural self-relevance post-manipulation (compared to pre-manipulation).

H2: The meat dish being described favourably (vs. unfavourably) will significantly increase measured subjective self-relevance.

Note that this pilot study utilises the same participant sample to also pilot a manipulation for familiarity, but only information relevant to the behavioural and subjective self-relevance manipulations is reported within the current chapter (see Section 6.2. for pilot study information about the familiarity manipulation [Pilot Study 2b]).

4.3.2. Methods

4.3.2.1. Participants.

Sixty participants were recruited via social media and Sona as a volunteer sample. Nineteen participants were excluded for providing partial data, and 11 participants were excluded for not consuming meat, meaning that all remaining participants were meat

consumers only. As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 4.3.3. for further discussion). Exclusions left a total sample size of 30 participants (50% women, 50% men; M_{age} =25.7, SD_{age} =6.87, age range: 18-50), which exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Participants within each condition are shown in Table 6. Demographics of the sample are reported as follows in percentages. For nationality: 46.7% (United Kingdom/British), and 3.3% each (Australian, Canadian, Croatian, dual Dutch/Norwegian, Finnish, French, German, Greek, Israeli, Italian, Japanese, Lithuanian, Malaysian, Spanish, Taiwanese, United States). For ethnicity: 63.2% (White), 19.9% (Asian), 10% (Black), and 6.6% (mixed). For country of residence ($M_{duration}$ =16.44, $SD_{duration}$ =9.98, range: 9 months to 35 years): 63.3% (United Kingdom), 6.7% (Malaysia), and 3.3% each (Croatia, Finland, France, Germany, Greece, Hong Kong, Israel, no response, and Norway). For religion: 33.3% (atheism), 23.3% (Christianity), 13.3% each (agnosticism, would rather not say), 6.7% (no response), and 3.3% each (Hinduism, Islam, Judaism).

Table 6.

Participants Within Each Condition for Pilot Study 2a.

Subjective Self-Relevance Condition	Number of Participants
Present	15
Absent	15

4.3.2.2. Design.

The current pilot study follows an ANCOVA design, with a one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA on measured behavioural self-relevance, and a one-way (subjective self-relevance: present vs. absent) between-subjects ANCOVA on measured subjective self-relevance. Behavioural and subjective self-relevance were included as DVs to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 4.3.3. for further discussion).

4.3.2.3. Materials.

4.3.2.3.1. Target Animal and Animal Photograph.

The current pilot study employed the same photograph of the tree kangaroo as Study 2, whereby this image was sourced from https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035. Like Study 2, this photograph was included to enable easier visualisation of the imagined scenario. However, unlike Study 2, this photograph was also used here to ensure participants knew what their target animal was, whilst avoiding informing participants of the name of the species. For instance, the name 'tree kangaroo' may have impacted participants' perceptions of the animal due to the names' association with kangaroos, or because seemingly neutral names influence perceptions of stimuli (Köhler, 1947; Maurer et al., 2006; Ramachandran & Hubbard, 2001). Thus, the current pilot study improved upon Study 2 to further reduce preconceptions and bias by withholding the species name.

4.3.2.3.2. Behavioural and Subjective Self-Relevance Manipulations.

As imagined scenarios with animals have previously been utilised to inform perceptions of animals (Auger & Amiot, 2019b), behavioural self-relevance was manipulated via an imagined restaurant scenario, employing images and text developed by the researcher in order to enhance realism of the scenario, whereby participants imagined consuming a meat dish in a fictional restaurant. Specifically, participants read the following text to set the scene: 'You will now be asked to take part in an imaginary scenario. Please engage with the text and photographs provided. You enter the below restaurant. When inside the restaurant, you are greeted by the waitress in this picture. The waitress takes you to your table below. When sat at your table, you read the menu. After some consideration, you eventually decide on ordering a spaghetti bolognese. After a short wait, your spaghetti bolognese is brought to you and you start to eat.' Images accompanying this text can be seen in Appendix 6. As the manipulation of subjective self-relevance, participants then read the text: 'Whilst eating the spaghetti bolognese, you remark to yourself how flavoursome/flavourless the dish is. The dish is perfectly seasoned/under-seasoned, and the ingredients [do not] blend well', whereby the text stating that the dish was 'flavoursome', 'perfectly seasoned' and that 'the ingredients blend well' is the subjective self-relevance condition, and the text stating that the dish was 'flavourless', 'under-seasoned' and that 'the ingredients do not blend well' is the lack of subjective self-relevance condition. To reveal to participants that the spaghetti bolognese dish actually contained meat from the participants' target animal (i.e., tree kangaroo), as a way to use salience of imagining having consumed the animal as a (pre- vs. post-) manipulation of behavioural self-relevance, participants read the following text: 'After you finish your meal, you go to pay the bill. The waitress asks you if you enjoyed your meal. She then reveals to you that the meat in your spaghetti bolognese came from the species you saw in the first photograph (pictured again below)'.

4.3.2.3.3. Measured Behavioural Self-Relevance (Manipulation Check).

As a scale to measure behavioural self-relevance has not yet been created, behavioural self-relevance was measured via the single item developed by the researcher: *'How many days a week would you be willing to eat meat from this species?'* from zero to seven days per week. A higher score indicates greater behavioural self-relevance.

4.3.2.3.4. Measured Subjective Self-Relevance (Manipulation Check).

As a scale to measure subjective self-relevance has not yet been developed, subjective self-relevance was measured via the single item developed by the researcher: '*How much do you like the spaghetti bolognese presented in this scenario*?' on a visual analogue from zero ('*not at all*') to 100 ('*very much*'). A higher score indicates greater subjective self-relevance.

4.3.2.3.5. Attention Check.

Attention paid to and memory for the manipulation was tested through an attention check developed by the researcher: '*What dish did the restaurant scenario refer to*?', with the correct answer being '*spaghetti bolognese*'. This check aimed to test that participants paid attention to and remembered the text. No participants were removed for failing this attention check.

4.3.2.3.6. Vividness of the Self-Relevance Manipulation.

Perceived vividness of the imagined self-relevance manipulation was measured, as imagined scenarios which are viewed as more vivid have greater impacts on behavioural intentions (Husnu & Crisp, 2010). As with Pilot Study 1a, vividness was measured through a six-item semantic differential vividness measure from Husnu and Crisp (2010; α =.94), whereby participants answer how much they perceive their imagined scenario as being '*faint vs. vivid*', '*fuzzy vs. clear*', '*dim vs. bright*', '*vague vs. sharp*', '*dull vs. lively*', and '*simple vs. detailed*'. Reliability was adequate for this scale within this study, ω =.86. Therefore, items were summed together to create a total vividness score. Higher scores indicate greater vividness. No items are reverse-scored. The majority of the sample (>50%) perceived the self-relevance manipulation to be vivid.

4.3.2.3.7. Ease of Imaginability of the Self-Relevance Manipulation.

Perceived ease of imaginability of the imagined self-relevance manipulation was measured in order to ensure participants were able to imagine the imagined scenario without difficulty. As with Pilot Study 1a, ease of imaginability was measured through a single item *'How easy or difficult was it for you to imagine the restaurant scenario?'* on a -50 (extremely difficult) to 50 (extremely easy) visual analogue scale, informed by previous literature (e.g., Black & Barnes, 2020; Broemer & Diehl, 2004). A higher score indicates greater ease of

imaginability. The majority of the sample (>50%) perceived the self-relevance manipulation to be easy to imagine.

4.3.2.3.8. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be?*' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

4.3.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in July 2021 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial behavioural self-relevance. Participants then completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the self-relevance section is described (see Section 6.2. for description of the familiarity section). For the self-relevance section, all participants first read the imagined restaurant scenario described in Section 4.3.2.3.2. and were randomly allocated

to read either the subjective self-relevance text or lack of subjective self-relevance text nested within this scenario. After reading their subjective self-relevance text, participants then rated how much they liked the dish (measure of subjective self-relevance; manipulation check), before reading that the dish contained meat from their target animal. After reading this text, participants again indicated their behavioural self-relevance of the animal (manipulation check), stated how vivid and easy to imagine the scenario was, and answered the attention check. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Two participants reported technical difficulties, but their responses were complete and maintained within analyses.

4.3.3. Analyses

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was conducted on measured behavioural self-relevance, to test if the manipulation effectively manipulates behavioural self-relevance. Additionally, a one-way (subjective self-relevance: present vs. absent) between-subjects ANCOVA, including gender and neutrality as covariates, was conducted on measured subjective self-relevance, to test if the manipulation effectively manipulates subjective self-relevance. Note that the behavioural and subjective self-relevance manipulations could not be combined into one ANCOVA as IVs, as the subjective selfrelevance manipulation is nested within the behavioural self-relevance manipulation.

Gender was again included as a covariate within these ANCOVAs. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007),

including gender as a covariate allows for any theoretically-informed relationships between gender and measured behavioural and subjective self-relevance (the outcome variables) to be statistically controlled for. Neutrality was also included within analyses as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between neutrality and behavioural and subjective self-relevance (Tabachnick & Fidell, 2014).

4.3.3.1. Statistical Assumptions to Test H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance (One-Way ANCOVA).

Measured behavioural self-relevance failed normality both pre- and postmanipulation, p < .05. Excessive skewness, multiple extreme univariate outliers, and floor effects in measured behavioural self-relevance meant assumptions for the ANCOVA were violated. Running the ANCOVA including and excluding the univariate outliers was not possible due to the floor effects. However, I proceeded with this ANCOVA due to robustness of ANCOVA to non-normality, and I report this ANCOVA including outliers to reflect the unamended dataset. Results from this ANCOVA should be interpreted with caution.

4.3.3.2. Statistical Assumptions to Test H2: Exploring Effects of Subjective Self-Relevance Manipulation on Measured Subjective Self-Relevance (One-Way ANCOVA).

The normality assumption for measured subjective self-relevance was met as assessed through a normality test across conditions, p > .05. Skewness was also acceptable across conditions. There were outliers in measured subjective self-relevance in the subjective selfrelevance condition. Running the analysis including and excluding these outliers did not change conclusions. I report this ANCOVA including outliers to reflect the unamended dataset. Homogeneity of variances was present, p = .35.

4.3.3.3. Testing H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance.

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was run on measured behavioural self-relevance to test **H1. Not supporting H1:** There was no significant main effect of manipulated behavioural self-relevance on measured behavioural self-relevance, F(1, 27) = .39, p = .54, $\eta_p^2 = .01$.

4.3.3.4. Testing H2: Exploring Effects of Subjective Self-Relevance Manipulation on Measured Subjective Self-Relevance.

A one-way (subjective self-relevance: present vs. absent) between-subjects ANCOVA, including gender and neutrality as covariates, was run on measured subjective self-relevance to test **H2. Supporting H2:** There was a significant main effect of manipulated subjective self-relevance on measured subjective self-relevance in the expected direction, $F(1, 26) = 47.92, p < .001, \eta_p^2 = .65$ (large-sized). That is, the subjective self-relevance manipulation caused greater measured subjective self-relevance (*M*=73.53, *SD*=23.57) than when subjective self-relevance was absent (*M*=21.93, *SD*=14.54).

4.3.4. Discussion and Conclusion

As expected, the subjective self-relevance condition led to greater subjective selfrelevance than the lack of subjective self-relevance condition, thereby supporting **H2**. Therefore, the current study demonstrates that the subjective self-relevance manipulation developed by the researcher is effective at manipulating subjective self-relevance. Informed by these findings, the subjective self-relevance manipulation is utilised within Study 3 to test causal effects of subjective self-relevance on pet speciesism. As the subjective self-relevance manipulation was found to be effective in manipulating subjective self-relevance within the current pilot study, this manipulation is unchanged within Study 3.

However, not supporting **H1**, the behavioural self-relevance manipulation had no effect on measured behavioural self-relevance. This lack of effectiveness may have arisen from floor effects in the behavioural self-relevance manipulation check measure, whereby most participants were unwilling to consume meat from the animal. Therefore, before applying the behavioural self-relevance manipulation to pet speciesism, Pilot Studies 3a and 4a re-test the behavioural self-relevance manipulation with amendments to the behavioural self-relevance manipulation with amendments to the behavioural self-relevance manipulation effectively increases behavioural self-relevance. Specifically, Pilot Study 3a adds the following statement before participants answer the behavioural self-relevance manipulation check item, as a way to ensure equal availability of the meat and equal harm caused to the animal compared to standard 'food' animals: '*Presuming meat from the species in the previous photograph is readily available in shops and does not cause any more harm than eating meat from animals such as cows*'.

4.4. Pilot Study 3a to Test Behavioural Self-Relevance Manipulation

4.4.1. Introduction

Due to the behavioural self-relevance manipulation failing to effectively manipulate behavioural self-relevance within Pilot Study 2a, the current pilot study aims to re-test this behavioural self-relevance manipulation with an amendment to the behavioural self-relevance manipulation check item, as a way to reduce floor effects in this manipulation check item. Therefore, behavioural self-relevance is again manipulated through an imagined restaurant scenario whereby participants imagine consuming a meat dish and is later revealed to contain meat from the participant's target animal (again, an unknown animal). To check effectiveness

of the behavioural self-relevance manipulation (manipulation check), the item '*How many days a week would you be willing to eat meat from this species?*' is again included to measure behavioural self-relevance. However, within this pilot study, the following statement is added before participants answer the behavioural self-relevance manipulation check item, as a way to ensure equal availability of the meat and equal harm caused to the animal compared to standard 'food' animals: '*Presuming meat from the species in the previous photograph is readily available in shops and does not cause any more harm than eating meat from animals such as cows'*. With this updated behavioural self-relevance manipulation check item, I hypothesise that:

H1: Imagining consuming a meat dish will significantly increase measured behavioural self-relevance post-manipulation (compared to pre-manipulation).

Note that this pilot study utilises the same participant sample to also pilot a manipulation for familiarity, but only information relevant to the behavioural self-relevance manipulation is reported within the current chapter (see Section 6.3. for pilot study information about the familiarity manipulation [Pilot Study 3b]).

4.4.2. Methods

4.4.2.1. Participants.

Fifty-five participants were recruited via social media and Sona as a volunteer sample. Eighteen participants were excluded for providing partial data, and seven participants were excluded for not consuming meat (meaning that all remaining participants were meat consumers only). As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 4.4.3. for further discussion). Exclusions left a total sample size of 30 participants (50% women, 50% men; $M_{age} = 25.3$, $SD_{age} = 6.13$, age range: 18-50), which

exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Demographics of the sample are reported as follows in percentages. For nationality: 16.7% (United States), 13.3% (British), 6.7% each (Chinese, Colombian, Filipino, French), and 3.3% each (Canadian, Croatian, Czech, Danish, Dutch, Estonian, Indian, Indonesian, Italian, Malaysian, Polish, Swedish, Taiwanese). For ethnicity: 53% (White), 26.5% (Asian), 10% (mixed), and 3.3% each (Black, Hispanic, Latin American). For country of residence ($M_{duration}$ =17.79, $SD_{duration}$ =11.79, range: 1 month to 50 years): 23.3% (United States), 20% (United Kingdom), 10% (France), 6.7% each (Germany, the Philippines), and 3.3% each (Canada, Croatia, Czech Republic, Denmark, Hungary, Indonesia, Malaysia, Poland, Sweden, the Netherlands). For religion: 40% (Christianity), 26.7% (atheism), 10% each (agnosticism, would rather not say), and 3.3% each (Buddhism, Hinduism, Judaism, no religion).

4.4.2.2. Design.

The current pilot study follows a one-way (behavioural self-relevance: pre- vs. postmanipulation) within-subjects ANCOVA design on measured behavioural self-relevance. Behavioural self-relevance was included as a DV to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 4.4.3. for further discussion).

4.4.2.3. Materials.

4.4.2.3.1. Target Animal and Animal Photograph.

The current pilot study employed the same photograph of the tree kangaroo as Study 2 and Pilot Study 2a, whereby this image was sourced from <u>https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035</u>. This photograph enables easier visualisation of the imagined scenario and reduces biased perceptions of the animal due to species name.

4.4.2.3.2. Behavioural Self-Relevance Manipulation.

Behavioural self-relevance was again manipulated via an imagined restaurant scenario, employing images and text developed by the researcher in order to enhance realism of the scenario, whereby participants imagined consuming a meat dish in a fictional restaurant. Specifically, participants read the following text to set the scene: 'You will now be asked to take part in an imaginary scenario. Please engage with the text and photographs provided. You enter the below restaurant. When inside the restaurant, you are greeted by the waitress in this picture. The waitress takes you to your table below. When sat at your table, you read the menu. After some consideration, you eventually decide on ordering a spaghetti bolognese. After a short wait, your spaghetti bolognese is brought to you and you start to eat.' Images accompanying this text can be seen in Appendix 6. As the manipulation of subjective self-relevance, participants then read the text: 'Whilst eating the spaghetti bolognese, you remark to yourself how flavoursome/flavourless the dish is. The dish is perfectly seasoned/under-seasoned, and the ingredients [do not] blend well', whereby the text stating that the dish was 'flavoursome', 'perfectly seasoned' and that 'the ingredients blend well' is the subjective self-relevance condition, and the text stating that the dish was 'flavourless', 'under-seasoned' and that 'the ingredients do not blend well' is the lack of subjective self-relevance condition. To reveal to participants that the spaghetti bolognese dish actually contained meat from the participants' target animal (i.e., tree kangaroo), as a way to use salience of imagining having consumed the animal as a (pre- vs. post-) manipulation of behavioural self-relevance, participants read the following text: 'After you finish your meal, you go to pay the bill. The waitress asks you if you enjoyed your meal. She then reveals to you that the meat in your spaghetti bolognese came from the species you saw in the first photograph (pictured again below)'.

4.4.2.3.3. Measured Behavioural Self-Relevance (Manipulation Check).

Behavioural self-relevance was again measured via the single item developed by the researcher: '*How many days a week would you be willing to eat meat from this species*?' from zero to seven days per week. However, the following statement was added before the behavioural self-relevance item to attempt to reduce floor effects, by reducing any reluctance to consume the meat from the animal by presenting the meat as akin to other meats: '*Presuming meat from the species in the previous photograph is readily available in shops and does not cause any more harm than eating meat from animals such as cows*'. A higher score indicates greater behavioural self-relevance.

4.4.2.3.4. Attention Check.

Attention paid to and memory for the manipulation was again tested through an attention check developed by the researcher: '*What dish did the restaurant scenario refer to*?', with the correct answer being '*spaghetti bolognese*'. This check aimed to test that participants paid attention to and remembered the text. No participants were removed for failing this attention check.

4.4.2.3.5. Vividness of the Self-Relevance Manipulation.

Perceived vividness of the imagined self-relevance manipulation was measured, as imagined scenarios which are viewed as more vivid have greater impacts on behavioural intentions (Husnu & Crisp, 2010). As with Pilot Studies 1a and 2a, vividness was measured through a six-item semantic differential vividness measure from Husnu and Crisp (2010; α =.94), whereby participants answer how much they perceive their imagined scenario as being '*faint vs. vivid*', '*fuzzy vs. clear*', '*dim vs. bright*', '*vague vs. sharp*', '*dull vs. lively*', and '*simple vs. detailed*'. Reliability was adequate for this scale within this study, ω =.92. Therefore, items were summed together to create a total vividness score. Higher scores indicate greater vividness. No items are reverse-scored. The majority of the sample (>50%) perceived the self-relevance manipulation to be vivid.

4.4.2.3.6. Ease of Imaginability of the Self-Relevance Manipulation.

Perceived ease of imaginability of the imagined self-relevance manipulation was measured in order to ensure participants were able to imagine the imagined scenario without difficulty. As with Pilot Studies 1a and 2a, ease of imaginability was measured through a single item '*How easy or difficult was it for you to imagine the restaurant scenario?*' on a -50 (extremely difficult) to 50 (extremely easy) visual analogue scale, informed by previous literature (e.g., Black & Barnes, 2020; Broemer & Diehl, 2004). A higher score indicates greater ease of imaginability. The majority of the sample (>50%) perceived the self-relevance manipulation to be easy to imagine.

4.4.2.3.7. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be?*' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

4.4.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in September 2021 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial behavioural self-relevance. Participants then completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the self-relevance section is described (see Section 6.3. for description of the familiarity section). For the self-relevance section, all participants first read the imagined restaurant scenario described in Section 4.4.2.3.2. including reading that the dish contained meat from their target animal. After reading this text, participants again indicated their behavioural self-relevance of the animal (manipulation check), stated how vivid and easy to imagine the scenario was, and answered the attention check. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Three participants reported technical difficulties, but their responses were complete and maintained within analyses.

4.4.3. Analyses

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was conducted on measured behavioural self-relevance, to test if the manipulation effectively manipulates behavioural self-relevance. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013;

Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured behavioural self-relevance (the outcome variable) to be statistically controlled for. Neutrality was also included within analyses as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between neutrality and behavioural self-relevance (Tabachnick & Fidell, 2014).

4.4.3.1. Statistical Assumptions to Test H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance (One-Way ANCOVA).

Measured behavioural self-relevance failed normality both pre- and postmanipulation, p < .05, but skewness was acceptable. There were outliers in measured behavioural self-relevance both pre- and post-manipulation. Running the analysis including and excluding these outliers did not change conclusions. I report this ANCOVA including outliers to reflect the unamended dataset.

4.4.3.2. Testing H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance.

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was run on measured behavioural self-relevance to test **H1. Not supporting H1:** There was no significant main effect of manipulated behavioural self-relevance on measured behavioural self-relevance, F(1, 27) = -.04, p = .85, $\eta_p^2 = .001$.

4.4.4. Discussion and Conclusion

Not supporting H1, the behavioural self-relevance manipulation again had no effect on measured behavioural self-relevance, although this time floor effects in the behavioural self-relevance manipulation check were not present. This lack of effectiveness may have arisen from an insufficiently large and underpowered sample size. Additionally, even though floor effects were not present in the behavioural self-relevance manipulation check item, the mean score for this item was still low both pre-manipulation, M = 1.07, and postmanipulation, M = 1.27. Therefore, before applying the behavioural self-relevance manipulation to pet speciesism, Pilot Study 4a re-tests the behavioural self-relevance manipulation with a further amendment to the behavioural self-relevance manipulation check item, whereby participants are now asked 'How willing would you be to include meat from this species within your diet?' on a 0-100 visual analogue scale, rather than how many days per week they would be willing to consume meat from the animal. This re-scaling and change in wording of the item should enhance spread of scores, and therefore better enable detection of whether the behavioural self-relevance manipulation is effective or not. Additionally, Pilot Study 4a tests the behavioural self-relevance manipulation with a larger sample size in line with a G*Power power analysis, in case current null effects are due to an underpowered analysis.

4.5. Pilot Study 4a to Test Behavioural Self-Relevance Manipulation

4.5.1. Introduction

Due to the behavioural self-relevance manipulation again failing to effectively manipulate behavioural self-relevance within Pilot Study 3a, the current pilot study aims to re-test this behavioural self-relevance manipulation with an additional amendment to the behavioural self-relevance manipulation check item. The current pilot study also aims to test the behavioural self-relevance manipulation with a larger sample size, in case the analysis is

underpowered. Therefore, behavioural self-relevance is again manipulated through an imagined restaurant scenario whereby participants imagine consuming a meat dish and is later revealed to contain meat from the participant's target animal (again, an unknown animal). To check effectiveness of the behavioural self-relevance manipulation (manipulation check), the item 'How willing would you be to include meat from this species within your diet?' on a 0-100 visual analogue scale is included to measure behavioural self-relevance, which has been amended from days per week in Pilot Studies 2a and 3a to more general willingness within the current pilot study as a way to enhance spread of scores, and therefore better enable detection of whether the behavioural self-relevance manipulation is effective or not. Additionally, as with Pilot Study 3a, the following statement is included before participants answer the behavioural self-relevance manipulation check item, as a way to reduce floor effects by ensuring equal availability of the meat and equal harm caused to the animal compared to standard 'food' animals: 'Presuming meat from the species in the previous photograph is readily available in shops and does not cause any more harm than eating meat from animals such as cows'. With this updated behavioural self-relevance manipulation check item and a larger sample size, I hypothesise that:

H1: Imagining consuming a meat dish will significantly increase measured behavioural self-relevance post-manipulation (compared to pre-manipulation).

Note that this pilot study utilises the same participant sample to also pilot a manipulation for familiarity, but only information relevant to the behavioural self-relevance manipulation is reported within the current chapter (see Section 6.4. for pilot study information about the familiarity manipulation [Pilot Study 4b]).

4.5.2. Methods

4.5.2.1. Participants.

One-hundred-and-sixty participants were recruited via Prolific as a volunteer sample. Quota sampling was partially used through Prolific's 'balance by sex'¹⁵ option to obtain approximately equal numbers of men and women. Thirty participants were excluded for: recognising the tree kangaroo (n = 16), taking part in a previous study within this project (n = 16) 7), withdrawing from the study (n = 5), and not consuming meat (n = 2, meaning that all remaining participants were meat consumers only). As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 4.5.3. for further discussion). Exclusions left a total sample size of 130 participants (51.5% women, 48.5% men; $M_{age} = 24.78$, $SD_{age} = 7.13$, age range: 18-56), which exceeded the minimum required sample size of 128 per a G*Power a priori power analysis for a ANCOVA analysis (medium effect size f=.25, two groups, two covariates, one numerator df, α =.05, power = .8). Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For nationality: 35.4% (South African), 15.4% (Portuguese), 9.2% (Polish), 5.4% (Italian), 4.6% (Zimbabwean), 3.8% (Hungarian), 3.1% each (British, Mexican), 2.3% (Spanish), and 17.7% other. For ethnicity: 49.5% (White), 34.6% (Black), 6.2% (mixed), 3.8% each (Asian, would rather not say), and 2.1% other. For country of residence (M_{duration}=22.54, SD_{duration}=8.81, range: 3 months to 56 years): 42.3% (South Africa), 15.4% (Portugal), 8.5% (Poland), 6.9% (Italy), 6.1% (United Kingdom), 4.6% (Hungary), 3.1% (Mexico), 2.3% (Spain), and 10.8% other. For religion: 63.8%

¹⁵Prolific does not provide an option to 'balance by gender'.

(Christianity), 19.2% (atheism), 8.5% (agnosticism), 3.8% (would rather not say), and 4.7% other.

4.5.2.2. Design.

The current pilot study follows a one-way (behavioural self-relevance: pre- vs. postmanipulation) within-subjects ANCOVA design on measured behavioural self-relevance. Behavioural self-relevance was included as a DV to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 4.5.3. for further discussion).

4.5.2.3. Materials.

4.5.2.3.1. Target Animal and Animal Photograph.

The current pilot study employed the same photograph of the tree kangaroo as Study 2 and Pilot Studies 2a and 3a, whereby this image was sourced from <u>https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035</u>. This photograph enables easier visualisation of the imagined scenario and reduces biased

perceptions of the animal due to species name.

4.5.2.3.2. Behavioural Self-Relevance Manipulation.

Behavioural self-relevance was again manipulated via an imagined restaurant scenario, employing images and text developed by the researcher in order to enhance realism of the scenario, whereby participants imagined consuming a meat dish in a fictional restaurant. Specifically, participants read the following text to set the scene: '*You will now be asked to take part in an imaginary scenario. Please engage with the text and photographs provided. You enter the below restaurant. When inside the restaurant, you are greeted by the waitress in this picture. The waitress takes you to your table below. When sat at your table, you read the menu. After some consideration, you eventually decide on ordering a spaghetti bolognese. After a short wait, your spaghetti bolognese is brought to you and you start to* *eat.* 'Images accompanying this text can be seen in Appendix 6. As the manipulation of subjective self-relevance, participants then read the text: 'Whilst eating the spaghetti bolognese, you remark to yourself how flavoursome/flavourless the dish is. The dish is perfectly seasoned/under-seasoned, and the ingredients [do not] blend well', whereby the text stating that the dish was 'flavoursome', 'perfectly seasoned' and that 'the ingredients blend well' is the subjective self-relevance condition, and the text stating that the dish was 'flavourless', 'under-seasoned' and that 'the ingredients do not blend well' is the lack of subjective self-relevance condition. To reveal to participants that the spaghetti bolognese dish actually contained meat from the participants' target animal (i.e., tree kangaroo), as a way to use salience of imagining having consumed the animal as a (pre- vs. post-) manipulation of behavioural self-relevance, participants read the following text: 'After you finish your meal, you go to pay the bill. The waitress asks you if you enjoyed your meal. She then reveals to you that the meat in your spaghetti bolognese came from the species you saw in the first photograph (pictured again below)'.

4.5.2.3.3. Measured Behavioural Self-Relevance (Manipulation Check).

Behavioural self-relevance was measured via the single item developed by the researcher: 'How willing would you be to include meat from this species within your diet?' on a zero ('not at all willing') to 100 ('extremely willing') visual analogue scale. The following statement was also included before the behavioural self-relevance item to reduce floor effects, by reducing any reluctance to consume the meat from the animal by presenting the meat as akin to other meats: 'Presuming meat from the species in the previous photograph is readily available in shops and does not cause any more harm than eating meat from animals such as cows'. A higher score indicates greater behavioural self-relevance.

4.5.2.3.4. Attention Check.

Attention paid to and memory for the manipulation was again tested through an attention check developed by the researcher: '*What dish did the restaurant scenario refer to*?', with the correct answer being '*spaghetti bolognese*'. This check aimed to test that participants paid attention to and remembered the text. No participants were removed for failing this attention check.

4.5.2.3.5. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be?*' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

4.5.2.4. Procedure

Participants were recruited via Prolific. All participants had to be 18+ meat consumers who had not taken part in previous studies within this thesis. The experiment was conducted in November 2021 as a voluntary, open survey open to participants who registered for the study via Prolific. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further. Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial behavioural self-relevance. Participants then completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the self-relevance section is described (see Section 6.4. for description of the familiarity section). For the self-relevance section, all participants first read the imagined restaurant scenario described in Section 4.5.2.3.2. including reading that the dish contained meat from their target animal. After reading this text, participants again indicated their behavioural self-relevance of the animal (manipulation check) and answered the attention check. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Two participants reported technical difficulties, but their responses were complete and maintained within analyses.

4.5.3. Analyses

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was conducted on measured behavioural self-relevance, to test if the manipulation effectively manipulates behavioural self-relevance. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured behavioural self-relevance (the outcome variable) to be statistically controlled for. Neutrality was also included within analyses as a covariate to

account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between neutrality and behavioural self-relevance (Tabachnick & Fidell, 2014).

4.5.3.1. Statistical Assumptions to Test H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance (One-Way ANCOVA).

Measured behavioural self-relevance failed normality both pre- and postmanipulation, p < .05, but skewness was acceptable. There were no outliers in measured behavioural self-relevance.

4.5.3.2. Testing H1: Exploring Effects of Behavioural Self-Relevance Manipulation on Measured Behavioural Self-Relevance.

A one-way (behavioural self-relevance: pre-manipulation vs. post-manipulation) within-subjects ANCOVA, including gender and neutrality as covariates, was run on measured behavioural self-relevance to test **H1. Supporting H1:** There was a significant main effect of manipulated behavioural self-relevance on measured behavioural self-relevance, F(1, 127) = 4.9, p = .03, $\eta_p^2 = .04$ (small-sized). That is, the behavioural self-relevance manipulation caused greater measured behavioural self-relevance (M=34.76, SD=28.56) compared to pre-manipulation (M=28.53, SD=27.18).

4.5.4. Discussion and Conclusion

Supporting **H1**, the behavioural self-relevance manipulation now has the expected effect on measured behavioural self-relevance, whereby measured behavioural self-relevance was greater post-manipulation compared to pre-manipulation, meaning that this behavioural

self-relevance manipulation has now been found to effectively manipulate behavioural selfrelevance. Therefore, informed by these findings, this behavioural self-relevance manipulation is utilised within Study 3 to test causal effects of behavioural self-relevance on pet speciesism. As this manipulation was found to be effective in manipulating behavioural self-relevance here, this manipulation is unchanged within Study 3. However, findings across pilot studies indicate that the previous failed behavioural self-relevance manipulations in Pilot Studies 2a and 3a may have arisen from underpowered analyses and/or issues with the measured behavioural self-relevance is again included as a manipulation check in Study 3, but is measured via the willingness version of the item used within the current pilot study rather than the days per week item used within Pilot Studies 2a and 3a.

4.6. Study 3: Applying Self-Relevance to Dogs and Pigs

4.6.1. Introduction and Extension from Studies 1-2 and Pilot Studies

Study 2 found that neither type of self-relevance affected warmth (not aligning with Study 1), competence (in line with Study 1), or active and passive behavioural intentions. However, null effects of behavioural and subjective self-relevance may have arisen from the failure of the behavioural and subjective self-relevance manipulations to manipulate measured behavioural and subjective self-relevance (e.g., Follow-Up/Pilot Study 1).

Following successful amendment of the subjective and behavioural self-relevance manipulations from Pilot Study 2a and 4a respectively, I employ these behavioural and subjective self-relevance manipulations within the current study to again assess if behavioural and subjective self-relevance causally affect warmth, competence, active help, passive help, active harm, and passive harm, to see if results align or not with motivated cognition theory and literature (Gradidge et al., 2021a; Loughnan et al., 2010, 2014; Piazza & Loughnan, 2016; Rothgerber, 2020). That is, to investigate if behavioural and subjective self-relevance motivate negative perceptions of (i.e., lower warmth and competence) and negative behavioural intentions towards (i.e., greater harm and lower help) a self-relevant animal, in line with motivated cognition theory.

The amended behavioural and subjective self-relevance manipulations within the current study are identical to Pilot Study 4a, whereby participants are asked to imagine a restaurant scenario whereby they consume a meat dish, which is either described favourably (subjective self-relevance) or unfavourably (lack of subjective self-relevance) and is later revealed to contain meat from the participant's target animal (manipulation of behavioural self-relevance). Like Study 2, the current study again tests causal effects of behavioural and subjective self-relevance with an unknown target animal (tree kangaroo) in order to enable unmoderated causal effects of behavioural and subjective self-relevance to be determined. Extending beyond Study 2, the current study also tests if the Study 2 null findings apply to dogs and pigs, thereby applying behavioural and subjective self-relevance to pet speciesism specifically.

Due to Study 2 null effects of behavioural and subjective self-relevance, the current study does not make specific hypotheses about effects of behavioural and subjective self-relevance on perceptions of animals (warmth, competence) or behavioural intentions (active help, active harm, passive help, passive harm), whereby warmth, competence, and behavioural intentions are again measured within the current study using the SCM/BIAS map subscales from Sevillano and Fiske (2016b). Instead, the current study represents an exploratory study to elucidate Study 2 results, i.e., to explore if null results are again found with now amended and effective behavioural and subjective self-relevance manipulations (ruling out the explanation of null results arising from ineffective manipulations), or to explore if findings are now instead significant and in line with motivated cognition theory.

Additionally, unlike Study 2, the full motivated cognition model is not tested within the current study (i.e., whether behavioural and subjective self-relevance enhance dissonance, which in turn would be expected to motivate negative perceptions of the self-relevant animal; Gradidge et al., 2021a), as the Dissonance Affect Questionnaire may be measuring general discomfort instead of dissonance. To maintain continuity across studies, I still implement the Dissonance Affect Questionnaire for the current study, but the variable is included for exploratory purposes and is now referred to as 'discomfort' throughout the chapter. As in Study 2, reported weekly meat consumption is included here for exploratory reasons in case of motivated cognition (i.e., deliberate under-reporting of meat consumption to alleviate discomfort; Rothgerber, 2019), and is again measured through a single item asking participants how much meat they consume per week.

Species is mostly included within this study as an exploratory variable. However, previous pet speciesism research (Bilewicz et al., 2011; Caviola & Capraro, 2020; Gradidge et al., 2021b), SCM/BIAS map research (Sevillano & Fiske, 2016a, 2016b, 2019), and Study 1 findings indicate that dogs are viewed more positively than other animals (e.g., pigs), whilst pigs are viewed more negatively than other animals (e.g., dogs). As such, I hypothesise that:

H1: Dogs will be deemed significantly warmer (**a**) and more competent (**b**) than tree kangaroos across self-relevance conditions, who will in turn be deemed significantly warmer and more competent than pigs.

Furthermore, due to evidence of pet speciesism in behavioural intentions towards animals as well as perceptions of them (Gradidge et al., 2021a), alongside active and passive behavioural intentions towards an animal being informed by the animal's perceived warmth and competence (Sevillano & Fiske, 2016; Sevillano & Fiske, 2019), I predict that

differences in warmth/competence perceptions of species will extend to behavioural intentions, such that:

H2: Dogs will be subject to significantly more active (**a**) and passive help (**b**), and significantly less active (**c**) and passive harm (**d**), than tree kangaroos across self-relevance conditions, who will in turn be subject to significantly more active and passive help, and significantly less active and passive harm, than pigs.

Finally, as previous BIAS map research both with humans (Cuddy et al., 2007) and with animals (Sevillano & Fiske, 2019) indicates that warmth typically informs active behavioural intentions whilst competence typically informs passive behavioural intentions, I hypothesise that:

H3: Warmth will significantly mediate relationships between species and active help and active harm. That is, dogs will be deemed significantly warmer than tree kangaroos and pigs, significantly increasing active help (**a**) and decreasing active harm (**b**).

H4: Competence will significantly mediate relationships between species and passive help and passive harm. That is, dogs will be deemed significantly more competent than tree kangaroos and pigs, significantly increasing passive help (**a**) and decreasing passive harm (**b**).

4.6.2. Methods

4.6.2.1. Participants.

Two-hundred-and-eighty-three participants were recruited via Prolific as a volunteer sample. Quota sampling was partially used through Prolific's 'balance by sex'¹⁶ option to

¹⁶Prolific does not provide an option to 'balance by gender'.

obtain approximately equal numbers of men and women. Sixty-eight participants were excluded for: withdrawing their response (n=20), stating their culture and/or religion does not typically view dogs as pets and/or pigs as food¹⁷ (n=10), recognising the tree kangaroo (n=9), participating in a previous experiment (n=9), providing partial data (n=7), not consuming meat (n=5), failing the attention and memory check (n=3), or being under 18 (n=2). As all analyses included gender as a covariate (see Section 4.6.3. for further discussion), three nonbinary participants were excluded to enable gender to be dummy coded into women (coded as zero; n=107) and men (coded as one; n=108). Exclusions left a sample of 215 participants $(50.2\% \text{ men}, 49.8\% \text{ women}; M_{age}=26.27, SD_{age}=8.28, age range=18-65)$, which exceeded the minimum required sample size of 211 per a G*Power a priori power analysis for a MANOVA analysis ('repeated measures, within-between interaction', medium effect size f=.25, six groups, two measurements, α =.05, power = .8). Participants within each condition are shown in Table 7. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For nationality: 20.5% (South African), 18.6% (Portuguese), 15.8% (Polish), 6.1% (Italian), 5.6% (Greek), 5.2% (British), 4.7% (Mexican), 2.3% (Dutch), and 21.2% other. For ethnicity: 66.9% (White), 19.1% (Black), 9.1% (mixed), 2.8% (Asian), and 2.1% other. For country of residence ($M_{duration}$ =24.07, $SD_{duration}$ =9.5, range: less than one month to 65 years): 24.2% (South Africa), 17.2% (Portugal), 16.3% (Poland), 7.4% (Italy), 6.1% (United Kingdom), 5.1% (Greece), 4.7% (Mexico), and 19% other. For religion: 50.7% (Christianity), 20.9% (atheism), 15.3% (agnosticism), 7.4% (would rather not say), and 5.7% other.

¹⁷These participants were excluded as cultural differences in perceptions of dogs and pigs may impact participant responses (Gradidge et al., 2021a).

1 anterpants // think Eden Condition for Study C.		
Species	Subjective Self-Relevance Condition	Number of Participants
Dog	Present	41
Pig	Present	31
Tree Kangaroo	Present	35
Dog	Absent	31
Pig	Absent	37
Tree Kangaroo	Absent	40

Table 7.Participants Within Each Condition for Study 3.

4.6.2.2. Design.

This experiment follows a 3(species: tree kangaroo vs. dog vs. pig; between-subjects) x 2(behavioural self-relevance/time: pre- vs. post-manipulation; within-subjects) x 2(subjective self-relevance: present vs. absent; between-subjects) mixed MANCOVA design, with warmth, competence, active help, passive help, active harm, passive harm, reported weekly meat consumption, and post-manipulation discomfort as the DVs. Behavioural and subjective self-relevance were included as DVs to test effectiveness of the manipulations. Gender and neutrality were included as covariates (see Section 4.6.3. for further discussion).

4.6.2.3. Materials.

4.6.2.3.1. Target Animals and Animal Photographs.

As in Study 2, the current study employed the novel animal paradigm. However, the only unknown animal included within this study was the tree kangaroo, with participants within the tree kangaroo condition being asked if they recognised the animal and excluded from the study if so. The current study employed the same photograph of the tree kangaroo as Study 2, whereby this image was sourced from https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035. Dogs and pigs were also included as target animals to test if behavioural and subjective self-relevance cause pet speciesism specifically. This study therefore implemented individual photographs of a dog and a pig, whereby the animal was directly facing the camera and was included as the central focus of the

photograph. These photographs were sourced from

https://www.flickr.com/photos/dave_see/8523607444 (dog) and

https://www.pexels.com/photo/nature-animals-pig-alp-rona-63285/ (pig). Like Study 2, these photographs were included to enable easier visualisation of the imagined scenario. However, unlike Study 2, these photographs were also used here to ensure participants knew what their target animal was, whilst avoiding informing participants of the name of the species. For instance, the name 'tree kangaroo' may have impacted participants' perceptions of the animal due to the names' association with kangaroos, or because seemingly neutral names influence perceptions of stimuli (Köhler, 1947; Maurer et al., 2006; Ramachandran & Hubbard, 2001). Thus, the current study improved upon Study 2 to further reduce preconceptions and bias by withholding the species name.

4.6.2.3.2. Behavioural and Subjective Self-Relevance Text Manipulations.

As imagined scenarios with animals have previously been utilised to inform perceptions of animals (Auger & Amiot, 2019b), behavioural self-relevance was manipulated via an imagined restaurant scenario, employing images and text developed and successfully piloted by the researcher as described in Pilot Studies 2a, 3a, and 4a (Sections 4.2. to 4.5.), whereby participants imagined consuming a meat dish in a fictional restaurant. Specifically, participants read the following text to set the scene: '*You will now be asked to take part in an imaginary scenario. Please engage with the text and photographs provided. You enter the below restaurant.* When inside the restaurant, you are greeted by the waitress in this picture. The waitress takes you to your table below. When sat at your table, you read the menu. After some consideration, you eventually decide on ordering a spaghetti bolognese. After a short wait, your spaghetti bolognese is brought to you and you start to eat.' Images accompanying this text can be seen in Appendix 6. As the manipulation of subjective self-relevance, participants then read the text: 'Whilst eating the spaghetti bolognese, you remark to yourself

how flavoursome/flavourless the dish is. The dish is perfectly seasoned/under-seasoned, and the ingredients [do not] blend well', whereby the text stating that the dish was 'flavoursome', 'perfectly seasoned' and that 'the ingredients blend well' is the subjective self-relevance condition, and the text stating that the dish was 'flavourless', 'under-seasoned' and that 'the ingredients do not blend well' is the lack of subjective self-relevance condition. To reveal to participants that the spaghetti bolognese dish actually contained meat from the participants' target animal (i.e., tree kangaroo), as a way to use salience of imagining having consumed the animal as a (pre- vs. post-) manipulation of behavioural self-relevance, participants read the following text: 'After you finish your meal, you go to pay the bill. The waitress asks you if you enjoyed your meal. She then reveals to you that the meat in your spaghetti bolognese came from the species you saw in the first photograph (pictured again below)'.

4.6.2.3.3. Warmth and Competence (Sevillano & Fiske, 2016b).

Warmth and competence were measured with the same items from Studies 1-2: *'warm'*, *'well-intentioned'*, *'friendly'* (warmth), and *'competent'*, *'skillful'*, *'intelligent'* (competence; Sevillano & Fiske, 2016b). However, participants were asked how much they viewed *'the species in the previous photograph'* as possessing these qualities, instead of previous studies where the animals' species name was used, to reduce bias in perceptions of the animal. Items were re-scaled from a five-point Likert scale to a seven-point semantic differential from -3 to 3, with original items (e.g., *'warm'*) located at the positive, right side of the scale, and their opposites (e.g., *'cold'*) located at the negative, left side. This re-scaling was implemented to incorporate a zero midpoint, and to more easily assess if baseline perceptions of animals are neutral (zero), positive, or negative.

Due to these adaptations, I ran separate PAFAs (Appendices 2c-2d) with direct oblimin rotation to test factorial validity, alongside assessing inter-item correlations to test reliability and communality values to test common variance, on the warmth and competence subscales. The PAFAs revealed adequate loading of all items onto the warmth and competence subscales, pre- and post-manipulation (\geq .75), supporting factorial validity, alongside adequate communalities pre- and post-manipulation (\geq .56). Additionally, inter-item correlations to assess reliability were adequate for the warmth subscale across timepoints and the competence subscale pre-manipulation, $.6 \le rs \le .79$. At post-manipulation only, interitem correlations between the first and second competence items, r=.84, and the first and third competence items, r=.83, were high (whereby high inter-item correlation is $rs \ge .8$; Field, 2018), whilst the inter-item correlation between the second and third competence items was adequate, r=.79. Whilst high inter-item correlations indicate redundancy (i.e., where two items constitute almost identical measurements and are therefore 'redundant'), and cause difficulty with determining how much each item contributes individually to the factor (Field et al., 2012), I decided to retain the first competence item within the competence subscale across timepoints, despite this item's high inter-item correlations, due to adequate inter-item correlations for this item pre-manipulation. Items within each subscale were summed to provide warmth and competence scores. Reliability was adequate within this study for warmth (pre-manipulation ω =.83, 95% BCa CI [.77, .87]; post-manipulation ω =.88, 95% BCa CI [.85, .91]) and competence (pre-manipulation ω =.88, 95% BCa CI [.85, .91]; postmanipulation ω=.93, 95% BCa CI [.9, .95]).

4.6.2.3.4. Behavioural Intentions Towards the Animal (Sevillano & Fiske, 2016b).

Behavioural intentions were measured with the same items from Study 2: 'support', 'help', 'behave friendly towards', 'interact with' the animal (active help), 'sustain', 'conserve' the animal (passive help), 'kill', 'injure', 'exterminate', 'trap', 'reject' the animal (active harm), and 'let [the animal] die off', 'ignore' the animal (passive harm; Sevillano & Fiske, 2016b). However, animals were described as 'the species in the previous photograph' instead of using their species name. These items were re-scaled, akin to the warmth and competence subscales, into a seven-point semantic differential from -3 to 3. Original items (e.g., '*support*') were placed on the positive, right side of the scale, and their opposites (e.g., '*not support*') placed on the negative, left side.

Due to these adaptations, I ran separate PAFAs (Appendices 2e-2f) with direct oblimin rotation to test factorial validity, alongside assessing inter-item correlations to test reliability and communality values to test common variance, on the active help and active harm subscales. As the passive help and passive harm subscales only contain two items, I did not run PAFAs for these subscales.

The PAFAs for the active help and active harm subscales revealed adequate loading of all items onto each subscale, pre- and post-manipulation (\geq .46), supporting factorial validity, alongside adequate communalities pre- and post-manipulation for all items in the active help subscale (\geq .51) and for items one to four in the active harm subscale (\geq .41). Communality was adequate (though not ideal) for the fifth active harm item pre-manipulation (.21) and post-manipulation (.28; Child, 2006). Inter-item correlations to assess reliability were adequate for the active help subscale pre-manipulation, $.53 \le rs \le .62$. At postmanipulation only, the inter-item correlation between the first and second active help items, *r*=.81, was high, whilst all other inter-item correlations were adequate, $.62 \le rs \le .67$. Whilst high inter-item correlations indicate redundancy (i.e., where two items constitute almost identical measurements and are therefore 'redundant'), and cause difficulty with determining how much each item contributes individually to the factor (Field et al., 2012), I retained the first and second active help items within the active help subscale, due to the first and second items' adequate inter-item correlations with other items and their adequate inter-item correlations pre-manipulation. The first four items of the active harm subscale also had adequate inter-item correlations across timepoints, $.44 \le rs \le .68$. Whilst the fifth item in the active harm subscale had an inadequate inter-item correlation with the first item pre-

manipulation, r=.28, and post-manipulation, r=.25, this fifth item had adequate inter-item correlations with the three other active harm items across timepoints, $.3 \le rs \le .49$. Therefore, I retained the fifth active harm item within the active harm subscale.

Items within each subscale were therefore summed to provide active help, passive help, active harm, and passive harm scores. Reliability was adequate within this study across timepoints for active help (pre-manipulation ω =.84, 95% BCa CI [.79, .87]; postmanipulation ω =.88, 95% BCa CI [.83, .92]), and active harm (pre-manipulation ω =.82, 95% BCa CI [.74, .87]; post-manipulation ω =.83, 95% BCa CI [.78, .88]). Reliability was also adequate for passive help at post-manipulation (r_{sb} =.73) but was inadequate for passive help at pre-manipulation (r_{sb} =.66) and was inadequate for passive harm at both timepoints (premanipulation r_{sb} =.63; post-manipulation r_{sb} =.69). As the passive harm and passive help subscales contain two items, I did not run PAFAs for these subscales, and I was unable to remove an item to improve reliability. Instead, I retained both items in the subscales, and findings using these subscales should be interpreted with caution due to inadequate reliability at one or both timepoints.

4.6.2.3.5. Discomfort (Harmon-Jones, 2000).

Discomfort was measured with the Dissonance Affect Questionnaire from Study 2, whereby items were '*uncomfortable*', '*uneasy*', and '*bothered*' (Harmon-Jones, 2000) on a Likert scale from one ('*not at all*') to five ('*extremely*'). Items were summed to create a total discomfort score. Reliability was adequate within this study across timepoints (premanipulation ω =.88, 95% BCa CI [.82, .91]; post-manipulation ω =.96, 95% BCa CI [.94, .97]).
4.6.2.3.6. Reported Weekly Meat Consumption.

Reported weekly meat consumption was measured as in Study 2 with the single item developed by the researcher: '*How many days a week do you consume meat*?' from zero to seven days per week.

4.6.2.3.7. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be*?' on a visual analogue scale from zero to 100. Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

4.6.2.3.8. Behavioural Self-Relevance (Manipulation Check).

As a scale to measure behavioural self-relevance has not yet been created, behavioural self-relevance was measured via the single item developed by the researcher: '*How willing would you be to include meat from this species within your diet*?' on a visual analogue scale from zero to 100, whereby '*this species*' refers to the participants' target animal. This item is identical to that used within Pilot Study 4a, as the amended behavioural self-relevance manipulation was effective in manipulating behavioural self-relevance when measured utilising this item. A higher score indicates greater behavioural self-relevance.

4.6.2.3.9. Subjective Self-Relevance (Manipulation Check).

As a scale to measure subjective self-relevance has not yet been developed, subjective self-relevance was measured via the single item developed by the researcher: '*How much do*

you like the spaghetti bolognese presented in this scenario?' on a visual analogue from zero to 100. As the amended subjective self-relevance manipulation was effective in manipulating subjective self-relevance when subjective self-relevance was measured utilising this item (Pilot Study 2a), this item was implemented within this study without amendments to measure subjective self-relevance. A higher score indicates greater subjective self-relevance.

4.6.2.3.10. Culture and Religion Questions.

To reduce results being biased by cultural and/or religious differences in how dogs and pigs are traditionally viewed, participants were asked '*Does your main culture and/or religion typically view pigs as food*?' and '*Does your main culture and/or religion typically view dogs as pets*?' in randomised order. Participants could answer 'yes', 'no' or 'other'. Ten participants were excluded for answering '*no*' to at least one question. No participants answered 'other'.

4.6.2.4. Procedure

Participants were recruited via Prolific (Appendix 7). All participants had to be meat consumers who had not taken part in previous studies within this thesis. The experiment was conducted in November 2021 as a voluntary survey open to participants who registered for the study via Prolific. After providing informed consent, participants provided demographic information, before being randomised into one of three species conditions (tree kangaroo vs. dog vs. pig; 'target animal'). Participants were first presented with a photograph of their target animal. For the tree kangaroo condition, participants were asked if they recognised the animal and, if so, excluded from the experiment. Animal species names were not used during the study. Participants then indicated their perceived neutrality towards their target animal, behavioural self-relevance of the animal, the animal's warmth and competence, behavioural intentions towards the animal, discomfort, and reported weekly meat consumption (all scales and items within scales presented in randomised order).

Participants subsequently read the imagined restaurant scenario, whereby they were randomly allocated into one of the two subjective self-relevance conditions. Participants indicated how much they liked the dish described in the scenario before it was revealed the dish supposedly contained meat from their target animal, accompanied with the target animal's photograph to remind participants of their animal. Following the reveal, participants again rated their behavioural self-relevance towards the target animal, discomfort, reported weekly meat consumption, the animal's warmth and competence, and behavioural intentions towards the animal (all scales and items within scales presented in randomised order), alongside completing an attention and memory check item. Specifically, to assess if participants paid attention to and remembered the imagined restaurant scenario, participants were asked 'What dish did the restaurant scenario refer to?' [Options: 'spaghetti bolognese', 'lasagne', 'salad', 'noodles', 'meatballs']. Three participants did not give the correct answer ('spaghetti bolognese'), and so were excluded. Finally, participants indicated if their culture and/or religion typically views dogs as pets and pigs as food, before being debriefed and redirected back to Prolific. One participant reported technical difficulties, but this response was complete and thus maintained within analyses. Participants received £1 via Prolific as reimbursement for full responses. Participants who provided partial data or who recognised the tree kangaroo were reimbursed a proportion of the £1 based either on their percentage of completion (partial data) or time spent completing the survey (recognition), whilst adhering to Prolific's minimum hourly rate.

4.6.3. Analyses

A 3(species: tree kangaroo vs. dog vs. pig) x 2(behavioural self-relevance/time: prevs post-manipulation) x 2(subjective self-relevance: present vs. absent) mixed MANCOVA was conducted on warmth, competence, active help, passive help, active harm, passive harm, reported weekly meat consumption, discomfort, and behavioural self-relevance. Warmth,

competence, active help, passive help, active harm, and passive harm are all conceptually related (i.e., all from the SCM/BIAS map and subscales of the same overall scales). Additionally, many of these DVs were found to be statistically related through moderate correlations, $rs \ge .3$. Thus, MANCOVA was deemed appropriate for this analysis.

As subjective self-relevance is nested within behavioural self-relevance and not measured pre-manipulation, a separate 3(species: tree kangaroo vs. dog vs. pig) x 2(subjective self-relevance: present vs. absent) between-subjects ANCOVA was conducted on subjective self-relevance.

Gender was again included as a covariate within the MANCOVA and ANCOVA. Diet was not included as a covariate within this study, as the sample only contained meat consumers. Again, although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and warmth/competence perceptions of animals and behavioural intentions towards them (the outcome variables) to be statistically controlled for. Neutrality was also included within analyses as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between neutrality and perceptions of/behavioural intentions towards animals (Tabachnick & Fidell, 2014).

4.6.3.1. Statistical Assumptions to Test H1-H2, Exploratory Analyses and Manipulation Check: Exploring Effects of Behavioural and Subjective Self-Relevance Across Species on SCM/BIAS Map Variables, Meat Consumption, Discomfort, and Measured Behavioural Self-Relevance (3x2x2 MANCOVA)

All DVs failed normality tests, p < .05. However, skewness was acceptable for reported meat consumption, passive help, and post-manipulation discomfort. Skewness was also acceptable for warmth and competence across most conditions, except at postmanipulation in the lack of subjective self-relevance condition with the dog as the target animal. Excessive skewness within that condition arose in warmth and competence due to one univariate outlier, and removal of this univariate outlier led to warmth and competence having acceptable skewness across conditions. I therefore maintained warmth and competence within the MANCOVA, and as discussed below, I also ran the MANCOVA including and excluding outliers, which did not alter most multivariate conclusions. Behavioural self-relevance, active harm, and passive harm all had excessive skewness, multiple extreme univariate outliers, and floor effects in the dog conditions, whilst active help had excessive skewness, extreme outliers, and ceiling effects when the dog was the target animal and when there was a lack of subjective self-relevance. Due to excessive skewness, extreme univariate outliers, and floor or ceiling effects in, active harm, passive harm, behavioural self-relevance, and active help, I also ran a MANCOVA excluding active harm, passive harm, behavioural self-relevance, and active help as DVs, which did not change most multivariate conclusions. One conclusion did change, whereby there was no longer a significant behavioural self-relevance and subjective self-relevance interaction on the combined DVs when excluding active harm, passive harm, behavioural self-relevance, and active help as DVs. However, this change likely arose from the exclusion of behavioural selfrelevance, as the follow-up ANCOVAs indicate a significant univariate behavioural selfrelevance and subjective self-relevance interaction on behavioural self-relevance and no other DVs. I therefore report the MANCOVA including active harm, passive harm, behavioural self-relevance, and active help as DVs.

Besides active harm, passive harm, behavioural self-relevance, and active help, reported meat consumption had no univariate outliers, whilst passive help and discomfort had non-extreme univariate outliers, which did not cause excessive skewness. Warmth and competence had non-extreme univariate outliers which did not cause excessive skewness in all conditions, except the post-manipulation lack of subjective self-relevance condition when the target animal was a dog, whereby one extreme univariate outlier caused excessive skewness in warmth and competence in this condition only. There were multivariate outliers for nine DVs, critical values ≥ 27.88 , p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in active harm, passive harm, and behavioural self-relevance, and ceiling effects in active help. However, running the MANCOVA including and excluding univariate and multivariate outliers without active harm, passive harm, behavioural self-relevance, and active help as DVs indicated only one multivariate conclusion changed (Footnote 19), and I therefore report the MANCOVA including univariate and multivariate outliers.

There was mostly no multicollinearity, with most $rs \le .85$. However, there was a high correlation between post-manipulation active help and passive harm, r=.91, in the lack of subjective self-relevance condition with the dog as the target animal, indicating multicollinearity between these DVs within this condition. However, correlations between these variables were acceptable (< .9) pre-manipulation and/or when the target animal was instead a pig or a tree kangaroo, so I maintained both variables separately within the MANCOVA. Linear relationships were present between some DVs across conditions, but not all DVs had linear relationships. Running follow-up univariate ANCOVAs including and excluding univariate outliers for passive help, discomfort, and warmth did not change conclusions. Excluding univariate outliers changed one major finding for competence (Footnote 21). I report this ANCOVA including outliers to reflect the unamended dataset. Excessive skewness, multiple univariate outliers, and floor or ceiling effects in active harm, passive harm, behavioural self-relevance, and active help meant assumptions for the ANCOVAs on these four DVs were violated. However, I proceeded with these ANCOVAs due to robustness of ANCOVA to nonnormality, and lack of non-parametric alternatives. Results from these ANCOVAs should be interpreted with caution.

Most DVs (warmth, pre-manipulation active help, active harm, passive harm, discomfort, reported meat consumption, behavioural self-relevance) failed homogeneity of variances, ps < .05. Homogeneity of covariances was not present, p < .001.

4.6.3.2. Statistical Assumptions for Manipulation Check: Effect of Subjective Self-Relevance Across Species on Measured Subjective Self-Relevance (3x2 ANCOVA)

Subjective self-relevance failed the normality test, p < .05, and subjective selfrelevance had outliers and excessive skewness. However, including and excluding outliers for the ANCOVA did not change main conclusions (differing findings reported in Footnote 18). I therefore report the subjective self-relevance ANCOVA including outliers to reflect the unaltered dataset. Homogeneity of variances was not present, p < .05.

4.6.3.3. Subjective Self-Relevance Manipulation Check: Effect of Subjective Self-Relevance Across Species on Measured Subjective Self-Relevance¹⁸

A 3(species: tree kangaroo vs. dog vs. pig) x 2(subjective self-relevance: present vs. absent) between-subjects ANCOVA, including gender and neutrality as covariates, was conducted on subjective self-relevance to test effectiveness of the subjective self-relevance text in manipulating measured subjective self-relevance. A 3x2 between-subjects ANCOVA was used for this manipulation check instead of a one-way (subjective self-relevance: present vs. absent) between-subjects ANCOVA to check consistent effectiveness of the subjective self-relevance manipulation across species conditions (see Table A4 in Appendix 4 for full inferential statistics). The ANCOVA revealed a significant effect of the subjective selfrelevance manipulation on measured subjective self-relevance, F(1, 207) = 261.89, p < .001, $\eta_p^2 = .56$ (large-sized), in the expected direction. Specifically, participants demonstrated greater measured subjective self-relevance when subjective self-relevance was present in the manipulation (M=84.2, SD=20.86) than when subjective self-relevance was absent in the manipulation (M=30.2, SD=26.76). Thus, the subjective self-relevance manipulation was effective at manipulating its intended variable of subjective self-relevance. Importantly, there was no significant interaction between the subjective self-relevance manipulation and species on measured subjective self-relevance, F(2, 207) = 1.48, p = .23, $\eta_p^2 = .01$, indicating the subjective self-relevance manipulation is effective across species.

¹⁸When excluding outliers, the interaction between the subjective self-relevance manipulation and species on measured subjective self-relevance became significant, F(2, 191) = 3.64, p = .03, $\eta_p^2 = .04$ (small-to-medium-sized). *Post hoc* pairwise comparisons revealed that, when subjective self-relevance was absent only, tree kangaroos were rated with significantly greater measured subjective self-relevance (M=35.06, SE=2.92) than dogs (M=21.31, SE=3.5), p = .01, or pigs (M=23.78, SE=3.11), p = .03. There were no significant pairwise comparisons when subjective self-relevance was present, ps > .05.

4.6.3.4. Testing H1-H2, Exploratory Analyses and Manipulation Check: Exploring Effects of Behavioural and Subjective Self-Relevance Across Species on SCM/BIAS Map Variables, Meat Consumption, Discomfort, and Measured Behavioural Self-Relevance.

A 3(species: dog vs. pig vs. tree kangaroo) x 2(subjective self-relevance: present vs. absent) x 2(behavioural self-relevance/time: pre- vs. post-manipulation) mixed MANCOVA, including gender and neutrality as covariates, with warmth, competence, active help, passive help, active harm, passive harm, reported meat consumption, post-manipulation discomfort, and behavioural self-relevance as DVs was run to: a) test effectiveness of the behavioural self-relevance text in manipulating measured behavioural self-relevance, b) test **H1-H2**, and c) run exploratory analyses to test causal effects of behavioural and subjective self-relevance on SCM/BIAS map variables. Behavioural self-relevance was included in this 3x2x2 mixed MANCOVA instead of a separate one-way (behavioural self-relevance: pre- vs. post-manipulation) within-subjects ANCOVA to check consistent effectiveness of the behavioural self-relevance manipulation across species and subjective self-relevance conditions (see Table A5 in Appendix 4 for full inferential statistics). Descriptive statistics for significant discussed main effects or for *post hoc* pairwise comparisons for significant interaction effects are reported in Tables 8-11.

4.6.3.4.1. Multivariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on the Combined DVs.¹⁹

There was a significant two-way interaction between behavioural self-relevance/time and subjective self-relevance on the combined DVs, F(9, 199) = 1.97, p = .045, $\eta_p^2 = .08$ (medium-sized). There was also a significant two-way interaction between behavioural self-

¹⁹Excluding univariate and multivariate outliers with warmth, competence, passive help, reported meat consumption, and post-manipulation discomfort as DVs led to an interaction between species and subjective self-relevance on the combined DVs becoming significant, F(10, 254) = 2.43, p = 01, $\eta_p 2 = .09$ (medium-sized).

relevance/time and species on the combined DVs, F(18, 400) = 4.76, p < .001, $\eta_p^2 = .18$ (large-sized), and a significant two-way interaction between behavioural self-relevance/time and gender on the combined DVs, F(9, 199) = 2.45, p = .01, $\eta_p^2 = .1$ (medium-sized). Beyond interactions, there were significant main effects of behavioural self-relevance/time, F(9, 199) = 6.05, p < .001, $\eta_p^2 = .22$ (large-sized), species, F(18, 400) = 13.14, p < .001, $\eta_p^2 = .37$ (large-sized), and gender, F(9, 199) = 2.96, p = .003, $\eta_p^2 = .12$ (medium-to-large-sized), on the combined DVs. All other multivariate interactions and main effects were non-significant, ps > .05. I followed up significant multivariate effects with univariate ANCOVAs on the DVs below.

4.6.3.4.2. Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Measured Behavioural Self-Relevance (Manipulation Check).

To be deemed effective, the behavioural self-relevance manipulation would be expected to increase measured behavioural self-relevance from pre- to post-manipulation, without differing across species or subjective self-relevance conditions. However, there was an unexpected significant interaction between the behavioural self-relevance manipulation and species on measured behavioural self-relevance, F(2, 207) = 3.54, p = .03, $\eta_p^2 = .03$ (small-sized), thus indicating effects of the behavioural self-relevance manipulation on measured behavioural self-relevance differ across species. *Post hoc* pairwise comparisons revealed that, for pigs only, measured behavioural self-relevance significantly decreased from pre-manipulation to post-manipulation, p = .001, d = .36 (small-sized). No such change occurred for dogs, p = .66, or tree kangaroos, p = .18. Thus, the behavioural self-relevance manipulation was not effective in manipulating measured behavioural self-relevance for dogs or tree kangaroos, and the manipulation affected measured behavioural self-relevance in the opposite direction to that expected for pigs.²⁰

Table 8.

Descriptive Statistics for In	nteraction Effect o	of Behavioural	Self-Relevance	and Species on
Measured Behavioural Sel	f-Relevance.			

Behavioural Self-	Dog	Dog	Pig	Pig	Tree	Tree
Relevance Condition	Mean	SD	Mean	SD	Kangaroo	Kangaroo
					Mean	SD
Pre-Manipulation	7.29	17.61	75.87	25.63	27.55	28.58
Post-Manipulation	8.19	19.63	69.1	29.08	24.45	27.5

In addition, there was a significant two-way interaction between the behavioural self-relevance manipulation and the subjective self-relevance manipulation on measured behavioural self-relevance, F(1, 207) = 12.17, p < .001, $\eta_p^2 = .06$ (medium-sized). *Post hoc* pairwise comparisons showed that measured behavioural self-relevance significantly decreased from pre-manipulation to post-manipulation in the condition whereby subjective self-relevance was absent, p < .001, d = .38 (small-sized), whilst measured behavioural self-relevance did not change from pre- to post-manipulation in the condition whereby subjective self-relevance was present, p = .46. Thus, effects of the behavioural self-relevance manipulation differ across subjective self-relevance conditions, whereby measured behavioural self-relevance decreased in the condition without subjective self-relevance and did not change in the condition with subjective self-relevance. These findings therefore indicate the behavioural self-relevance manipulation either had no effect (when subjective self-relevance was present) or had an effect in the opposite direction to predicted, whereby measured behavioural self-relevance manipulation (when

²⁰There was also a significant main effect of species on measured behavioural self-relevance, F(2, 207) = 134.66, p < .001, $\eta_p^2 = .57$ (large-sized). However, this effect was qualified by the higher-order two-way interaction between species and behavioural self-relevance as reported in the main text.

subjective self-relevance was absent). These results therefore indicate the behavioural selfrelevance manipulation was not effective at manipulating measured behavioural selfrelevance.

Beyond the manipulation check, *post hoc* pairwise comparisons for the two-way interaction between the behavioural self-relevance manipulation and subjective self-relevance manipulation also revealed that, post-manipulation only, the condition whereby subjective self-relevance was present (i.e., imagine liking the meat) significantly increased measured behavioural self-relevance compared to the condition whereby subjective self-relevance was absent (i.e., imagine disliking the meat), p = .03, d = .11 (small-sized). That is, participants were more willing to include meat from the animal within their diets when participants were asked to imagine liking (vs. disliking) the meat dish.

Table 9.

Descriptive Statistics for Interaction Effect of Behavioural Self-Relevance and Subjective Self-Relevance on Measured Behavioural Self-Relevance.

Subjective Self-	Pre-	Pre-	Post-	Post-
Relevance	Manipulation	Manipulation	Manipulation	Manipulation
Condition	Mean	SD	Mean	SD
Present	33.45	38.46	35.2	38.17
Absent	38.62	36.31	31.08	33.92

Whilst there was a significant effect of the behavioural self-relevance manipulation on measured behavioural self-relevance, F(1, 207) = 4.85, p = .03, $\eta_p^2 = .02$ (small-sized), this effect was in the opposite direction to expected if the manipulation were effective. That is, measured behavioural self-relevance was lower post-manipulation (*M*=33.13, *SD*=36.07) than pre-manipulation (*M*=36.05, *SD*=37.39). Additionally, the above interactions indicate that the behavioural self-relevance manipulation either a) fails to have any effect on measured behavioural self-relevance (when the target animal is a dog or tree kangaroo, and/or in the condition whereby subjective self-relevance is present), or b) unexpectedly decreases

measured behavioural self-relevance from pre- to post-manipulation (when the target animal is a pig, and/or in the condition whereby subjective self-relevance is absent).

4.6.3.4.3. H1a: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Warmth.

There was no significant interaction between behavioural self-relevance/time and subjective self-relevance on warmth, F(1, 207) = 1.95, p = .16, $\eta_p^2 = .01$. The behavioural self-relevance manipulation also had no significant main effect on warmth, F(1, 207) = .34, p = .56, $\eta_p^2 = .002$. **Partially consistent with H1a:** There was a significant main effect of species on warmth, F(2, 207) = 34, p < .001, $\eta_p^2 = .25$ (large-sized). *Post hoc* pairwise comparisons revealed dogs were deemed significantly warmer than pigs, d = 1.21 (large-sized), and tree kangaroos d = 1.19 (large-sized), ps < .001. However, **partially not supporting H1a**, there was no significant difference between pigs' and tree kangaroos' warmth, p = 1.

4.6.3.4.4. H1b: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Competence.²¹

In line with Studies 1-2, there were no significant effects of the behavioural selfrelevance manipulation, F(1, 207) = .42, p = .52, $\eta_p^2 = .002$, or the subjective self-relevance manipulation interacting with behavioural self-relevance/time, F(1, 207) = 1.59, p = .21, $\eta_p^2 = .01$, on competence. Aligning with H1b: Species had a significant main effect across timepoints on competence, F(2, 207) = 27.54, p < .001, $\eta_p^2 = .21$ (large-sized). Post hoc pairwise comparisons revealed dogs were deemed significantly more competent than tree kangaroos, d = .82 (large-sized), and pigs, d = 1.24 (large-sized), ps < .001. Additionally, tree

²¹When excluding univariate outliers, the behavioural self-relevance manipulation had a significant main effect on competence, F(1, 203) = 3.99, p = .47, $\eta_p^2 = .02$ (small-sized): Animals were deemed more competent post-manipulation (*M*=3.65, *SD*=3.96) than pre-manipulation (*M*=3.29, *SD*=3.83).

kangaroos were deemed significantly more competent than pigs, p = .04, d = .42 (small-sized). There was also a significant main effect of gender on competence, F(1, 207) = 8.98, p = .003, $\eta_p^2 = .04$ (small-sized), whereby women viewed animals as significantly more competent than men did pre-manipulation, B = -1.45, SE = .49, p = .004, and post-manipulation, B = -1.49, SE = .52, p = .005.

4.6.3.4.5. H2a: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Active Help.

There were no significant effects of the behavioural self-relevance manipulation, F(1, 207) = .06, p = .8, $\eta_p^2 < .001$, or the subjective self-relevance manipulation interacting with behavioural self-relevance/time, F(1, 207) = 2.83, p = .09, $\eta_p^2 = .01$, on active help. **Partially agreeing with H2a:** There was a significant effect of species on active help, F(2, 207) = 19,48, p < .001, $\eta_p^2 = .16$ (large-sized). *Post hoc* pairwise comparisons show people are significantly more willing to actively help dogs than pigs, d = .95 (large-sized), or tree kangaroos, d = .86 (large-sized), ps < .001. **Partially not supporting H2a:** There was no significant difference in active help for pigs and tree kangaroos, p = 1.

4.6.3.4.6. H2b: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Passive Help.

In line with Study 2, there was no significant effects of the behavioural self-relevance manipulation, F(1, 207) = .71, p = .4, $\eta_p^2 = .003$, or the subjective self-relevance manipulation interacting with behavioural self-relevance/time, F(1, 207) = .78, p = .38, $\eta_p^2 = .004$, on passive help. **Partially consistent with H2b:** There was a significant effect of species on passive help, F(2, 207) = 7.11, p = .001, $\eta_p^2 = .06$ (medium-sized). *Post hoc* pairwise comparisons show people are significantly more willing to passively help dogs than pigs, d = .59 (medium-sized), p = .002, or tree kangaroos, d = .51 (medium-sized), p = .01.

Partially not supporting H2b: There was no significant difference in passive help for pigs and tree kangaroos, p = 1.

4.6.3.4.7. H2c: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Active Harm.

There were no significant effects of the behavioural self-relevance manipulation, F(1, 207) = .14, p = .71, $\eta_p^2 = .001$, or the subjective self-relevance manipulation interacting with behavioural self-relevance/time, F(1, 207) = .46, p = .5, $\eta_p^2 = .002$, on active harm. Agreeing with H2c: There was a significant effect of species on active harm, F(2, 207) = 31.12, p < .001, $\eta_p^2 = .23$ (large-sized). *Post hoc* pairwise comparisons show people are significantly less willing to actively harm dogs than to harm pigs, d = 1.34 (large-sized), or tree kangaroos, d = .65 (medium-sized), ps < .001. People are also significantly more willing to actively harm

4.6.3.4.8. H2d: Univariate Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Passive Harm.

In line with Study 2, there was no significant interaction between the subjective selfrelevance manipulation and behavioural self-relevance/time on passive harm, F(1, 207) = .01, p = .94, $\eta_p^2 < .001$. However, the behavioural self-relevance manipulation did have a significant main effect on passive harm, F(1, 207) = 5.16, p = .02, $\eta_p^2 = .02$ (small-sized). That is, passive harm was lower post-manipulation (M=-3.84, SD=2.6) than pre-manipulation (M=-3.37, SD=2.74). Agreeing with H2d: There was a significant effect of species on passive harm, F(2, 207) = 25.46, p < .001, $\eta_p^2 = .2$ (large-sized). *Post hoc* pairwise comparisons show people are significantly less willing to passively harm dogs than pigs, d =1.2 (large-sized), or tree kangaroos, d = .74 (medium-sized), ps < .001. People are also significantly more willing to passively harm pigs than to harm tree kangaroos, d = .47 (smallsized), p = .02. There was also a significant interaction between behavioural selfrelevance/time and gender on passive harm, F(1, 207) = 3.97, p = .048, $\eta_p^2 = .02$ (small-sized), whereby, post-manipulation only, men were significantly more willing to passively harm animals than women were, B = .86, SE = .32, p = .01. This effect was not significant pre-manipulation, B = .36, SE = .34, p = .29.

Table 10.

Descriptive Statistics for Effect of Species on Warmth, Competence, Active Help, Active Harm, Passive Help, Passive Harm.

DV	Dog	Dog	Pig	Pig	Tree Kangaroo	Tree Kangaroo
	Mean	SE	Mean	SE	Mean	SE
Warmth	7.21	.38	3.32	.39	3.38	.37
Competence	5.72	.43	1.28	.43	2.77	.41
Active Help	9.31	.54	4.97	.55	5.38	.52
Active Harm	-13.71	.52	-7.85	.53	-10.87	.5
Passive Help	3.76	.32	2.16	.33	2.38	.31
Passive Harm	-5.04	.27	-2.34	.27	-3.39	.26

4.6.3.4.9. Univariate Effects of Behavioural Self-Relevance, Subjective Self-

Relevance and Species on Reported Weekly Meat Consumption.

There were no significant effects on reported weekly meat consumption (see Table A5 in Appendix 4). There was therefore no evidence for under-reporting across behavioural self-relevance, subjective self-relevance, or species conditions.

4.6.3.4.10. Univariate Effects of Behavioural Self-Relevance, Subjective Self-

Relevance and Species on Discomfort.

There was a significant two-way interaction between behavioural self-relevance/time and species on discomfort, F(2, 207) = 38.28, p < .001, $\eta_p^2 = .27$ (large-sized). Post hoc pairwise comparisons indicate participants felt significantly greater discomfort postmanipulation when the scenario involved a dog, d = 1.45 (large-sized), or tree kangaroo, d =.96 (large-sized), compared to pre-manipulation, ps < .001. When the scenario involved a pig, discomfort did not significantly differ from pre-manipulation to post-manipulation, p = .3. The subjective self-relevance manipulation did not significantly interact with behavioural self-relevance/time, F(1, 207) = .81, p = .37, $\eta_p^2 = .004$, on discomfort. The behavioural self-relevance manipulation also had a significant interaction with gender on discomfort, F(1, 207) = 13.66, p < .001, $\eta_p^2 = .06$ (medium-sized), whereby women felt more discomfort than men pre-manipulation, B = -.75, SE = .35, p = .03, and post-manipulation, B = -2.37, SE = .47, p < .001.

Table 11.

Descriptive Statistics for Interaction Effect of Behavioural Self-Relevance and Species on Discomfort.

215001190111						
Behavioural Self-	Dog	Dog	Pig	Pig	Tree	Tree
Relevance Condition	Mean	SD	Mean	SD	Kangaroo	Kangaroo SD
					Mean	
Pre-Manipulation	5.13	.31	4.97	.31	4.88	.3
Post-Manipulation	10.28	.41	5.37	.41	7.77	.39

4.6.3.5. Testing H3-H4: Exploring Mediational Relationships Between Species and Active and Passive Behavioural Intentions Through Warmth/Competence

Per the BIAS map, I predicted warmth would mediate the relationship between species and active help (H3a) and active harm (H3b), and that competence would mediate the relationship between species and passive help (H4a) and passive harm (H4b), across timepoints. To test H3-H4, I used indicator coding due to the multi-categorical IV of species (Hayes & Preacher, 2014), creating two dummy variables for species (X1 and X2). Dog was utilised as the reference category (zero), whilst pig was coded as one on X1 and zero on X2, and tree kangaroo was coded as one on X2 and zero on X1. Thus, X1 represents comparisons between pigs (one) vs. non-pigs (zero), whilst X2 represents comparisons between tree kangaroos (one) vs. non-tree kangaroos (zero). Eight individual boot-strapped mediation analyses (10,000 bootstrap samples) were run using Model 4 in PROCESS through SPSS (Hayes, 2022) to assess mediational relationships for the four behavioural intentions pre- and post-manipulation. To control for Type I error from multiple mediation testing, I utilised an adjusted confidence interval of 99%. Significant mediations are interpreted through confidence intervals which do not contain zero per Hayes (2018).

As warmth and competence were significantly correlated with each other at both timepoints within the current study as revealed through Pearson correlations, premanipulation r = .61, p < .001, post-manipulation r = .71, p < .001, and warmth and competence are conceptually related, warmth and competence were included as parallel mediators within the below mediation analyses. Inclusion of warmth and competence as parallel mediators allows for testing of any mediational roles of warmth for active behavioural intentions *whilst* controlling for competence, and any mediational roles of competence for passive behavioural intentions *whilst* controlling for warmth (Hayes, 2018).

4.6.3.5.1. H3a: Mediation of Species on Active Help through Warmth.

Agreeing with H3a: There were significant indirect effects of species both on premanipulation active help through pre-manipulation warmth when controlling for premanipulation competence, and on post-manipulation active help through post-manipulation warmth when controlling for post-manipulation competence. Thus, warmth significantly mediated the relationship between species and active help across timepoints (Figures 2-3). The model explained 45.1% (R^2 = .451) and 54% (R^2 = .54) of variance in active help premanipulation and post-manipulation respectively. Additionally, inspection of path *b* indicated greater warmth significantly predicted greater active help across timepoints (Figures 2-3).



Figure 2. Pre- and post-manipulation mediational models of the species X1 variable (pig vs. non-pig) on active help through warmth when controlling for competence. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. ***p<.001



Omnibus tests of direct effect of species (X1 and X2) on active help through warmth and competence: T1: F(2, 210) = 1.51, p = .22, $R^2 = .01$ T2: F(2, 210) = .9, p = .41, $R^2 = .004$

Figure 3. Pre- and post-manipulation mediational models of the species X2 variable (tree kangaroo vs. non-tree kangaroo) on active help through warmth when controlling for competence. *Note:* T1 refers to premanipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. Statistics for the omnibus test of the overall direct effects of species (X1 and X2 combined) on active help through warmth and competence are reported beneath the figure.***p<.001

4.6.3.5.2. H3b: Mediation of Species on Active Harm through Warmth.

Agreeing with H3b: There were significant indirect effects of species both on premanipulation active harm through pre-manipulation warmth when controlling for premanipulation competence, and on post-manipulation active harm through post-manipulation warmth when controlling for post-manipulation competence. Thus, warmth significantly mediated the relationship between species and active harm across timepoints (Figures 4-5). The model explained 35.8% (R^2 = .358) and 41.3% (R^2 = .413) of variance in active harm premanipulation and post-manipulation respectively. Additionally, inspection of path *b* indicated greater warmth significantly predicted less active harm across timepoints (Figures 4-5).



Figure 4. Pre- and post-manipulation mediational models of the species X1 variable (pig vs. non-pig) on active harm through warmth when controlling for competence. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. ***p<.001



Omnibus tests of direct effect of species (X1 and X2) on active harm through warmth and competence: T1: F(2, 210) = 11.98, p < .001, $R^2 = .07$ T2: F(2, 210) = 10.63, p < .001, $R^2 = .06$

Figure 5. Pre- and post-manipulation mediational models of the species X2 variable (tree kangaroo vs. non-tree kangaroo) on active harm through warmth when controlling for competence. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. Statistics for the omnibus test of the overall direct effects of species (X1 and X2 combined) on active harm through warmth and competence are reported beneath the figure. ***p<.001

4.6.3.5.3. H4a: Mediation of Species on Passive Help through Competence.

Partially consistent with H4a: There was a significant indirect effect of species on pre-manipulation passive help through pre-manipulation competence when controlling for pre-manipulation warmth (Figures 6-7). However, **not supporting H4a**, the indirect effect of species on post-manipulation passive help through post-manipulation competence when controlling for post-manipulation warmth was not significant (Figures 6-7). Thus, competence significantly mediated the relationship between species and passive help premanipulation, yet not post-manipulation (Figures 6-7). The model explained 20.2% ($R^2 =$

.202) of variance in pre-manipulation passive help. Inspection of path *b* indicated greater competence significantly predicted greater passive help pre-manipulation, yet not post-manipulation (Figures 6-7).



Figure 6. Pre- and post-manipulation mediational models of the species X1 variable (pig vs. non-pig) on passive help through competence when controlling for warmth. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. ***p < .001, $**p \leq .01$



Omnibus tests of direct effect of species (X1 and X2) on passive help through competence and warmth: T1: F(2, 210) = .19, p = .83, $R^2 = .002$ T2: F(2, 210) = .2, p = .82, $R^2 = .002$

Figure 7. Pre- and post-manipulation mediational models of the species X2 variable (tree kangaroo vs. non-tree kangaroo) on passive help through competence when controlling for warmth. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. Statistics for the omnibus test of the overall direct effects of species (X1 and X2 combined) on passive help through competence and warmth are reported beneath the figure. ***p<.001, **p≤.01

4.6.3.5.4. H4b: Mediation of Species on Passive Harm through Competence.

Not supporting H4b: There were no significant indirect effects of species on passive harm through competence when controlling for warmth, either pre-manipulation for X1, b = .3, SE = .24, 99% CI [-.25, .99], or X2, b = .19, SE = .15, 99% CI [-.16, .6], or post-manipulation for X1, b = .56, SE = .28, 99% CI [-.14, 1.37], or X2, b = .37, SE = .18, 99% CI [-.09, .9]. Thus, competence did not significantly mediate the relationship between species and passive harm at either timepoint. Additionally, inspection of path *b* indicated that greater competence significantly predicted less passive harm post-manipulation, b = ..12, SE = .05, *p*

= .01, 99% CI [-.24, .001], but not pre-manipulation, *b* = -.07, *SE* = .05, *p* = .17, 99% CI [-.19, .06].

4.6.4. Discussion

4.6.4.1. Discussion of Findings.

4.6.4.1.1. H1-H2: Effects of Species on SCM/BIAS Map Variables (Evidence for Pet Speciesism).

Mostly supporting **H1-H2** and previous pet speciesism literature (Bilewicz et al., 2011; Caviola & Capraro, 2020; Gradidge et al., 2021b; Sevillano & Fiske, 2016b), dogs were deemed warmer and more competent than pigs and tree kangaroos (large-sized) and were subject to greater active help (large-sized), greater passive help (medium-sized), and less active and passive harm (medium- or large-sized) than the other animals. Additionally, supporting **H1-H2**, tree kangaroos were deemed more competent (small-sized) and subject to less active harm (medium-sized) and less passive harm (small-sized) than pigs. However, not supporting **H1-H2** and previous literature (Sevillano & Fiske, 2016b), there were no significant differences between pigs' and tree kangaroos' warmth nor active and passive help towards the animals. Most of these findings align with or are larger than previously found medium-sized effects of species on other perceptions of animals (Gradidge et al., 2021b), although the small-sized effects may have lower practical significance. Additionally, the passive harm and passive help findings should be interpreted with caution due to inadequate reliability.

Combined, the across-species findings suggest an overarching preference for dogs across various perceptions and behavioural intentions, and this preference applies over 'food' animals like pigs and unknown animals like tree kangaroos. Thus, pet speciesism may be partly motivated by prejudice in favour of dogs across all dimensions. Pet speciesism may also be motivated by prejudice against 'food' animals like pigs, at least in the domains of

competence, active harm, and passive harm. Indeed, people can deny and/or disregard the cognitive capacities of 'food' animals like pigs (Bastian et al., 2012a; Bilewicz et al., 2011), whereas people do not deny and/or disregard these capacities in unfamiliar, fictional, or non-'food' animals (Higgs et al., 2020; Piazza & Loughnan, 2016). Thus, these previous findings regarding denial and/or disregarding of 'food' animals' cognitive capacities could explain why pigs were viewed within this study as less competent than tree kangaroos. This difference in tree kangaroos' and pigs' competence may in turn explain the differences in passive harm towards the two species found here. For example, people are more willing to passively harm 'prey' animals like pigs (due to these animals' lower competence) than animals deemed more competent (Sevillano & Fiske, 2016b). Thus, as tree kangaroos were deemed more competent than pigs within this study, this greater competence may explain why participants showed less passive harm towards tree kangaroos than towards pigs. However, although competence typically informs passive harm and passive help, there was no differences in passive help towards tree kangaroos and pigs. This finding therefore demonstrates occasional separability of perceptions of animals per the SCM (e.g., competence) and behavioural intentions per the BIAS map (e.g., passive help).

The lack of difference between tree kangaroos' and pigs' warmth may not support previous research, whereby people typically deny the positive qualities of typical 'food' animals like pigs (Bastian et al., 2012a; see Gradidge et al., 2021a). Thus, pigs, as typical 'food' animals, would be expected to be deemed less warm than non-'food' animals like tree kangaroos. However, Sevillano and Fiske (2016b) found that pigs and other 'food' animals ('prey' animals per Sevillano & Fiske, 2016b) were still deemed moderately high in warmth, although not as high in warmth as companion animals like dogs. Thus, as perceptions of tree kangaroos have not been tested before, tree kangaroos (and other unknown animals) may fall closer to the 'prey' (vs. 'companion') cluster on the warmth dimension. Participants also did

not differ in their active help towards tree kangaroos and pigs. Whilst a lack of differences between tree kangaroos and pigs on warmth and active help was unexpected, and does not support **H1-H2**, these null effects are at least in line with each other and the BIAS map, whereby warmth informs active help (Cuddy et al., 2007). That is, as tree kangaroos and pigs did not differ on warmth, it is in line with the BIAS map that tree kangaroos and pigs also did not differ on active help.

Conversely, tree kangaroos and pigs did differ on active harm. Thus, this finding again demonstrates occasional possible separability of perceptions of animals per the SCM (e.g., warmth) and behavioural intentions per the BIAS map (e.g., active harm). Whilst not yet explicitly tested, this finding may again be due to pigs' status as 'food', whereby people may be more willing to actively harm animals that are already harmed through food production than to harm animals which are not already being harmed.

4.6.4.1.2. Effects of Behavioural and Subjective Self-Relevance on SCM/BIAS Map Variables.

The current study found mostly null effects of self-relevance on perceptions of, and behavioural intentions towards, animals. Such null effects were expected for competence and passive help, whilst a null effect of subjective self-relevance on passive harm was also expected. These findings are consistent with Study 2 and the BIAS map, whereby greater competence informs greater passive help and less passive harm. That is, lack of an effect of behavioural and subjective self-relevance on competence and its associated passive behavioural intentions (passive help, passive harm) is consistent with the BIAS map. However, standing in contrast to the BIAS map, there was an unexpected significant smallsized effect of behavioural self-relevance on passive harm, whereby passive harm was lower after (vs. before) the manipulation (although this passive harm finding should be interpreted with caution due to floor effects and inadequate scale reliability). In other words, after

participants had imagined consuming meat from the animal (compared to before), they were less likely to want to passively harm the animal, regardless of whether they liked the meat or which species the meat came from. As behavioural self-relevance was measured within this study as a within-subjects variable from pre- to post-manipulation, and due to the nature of the passive harm items (*'ignore'*, *'let them die off'*), the reduction in passive harm may be arising simply from reading more information about the animal, rather than due to salience of behavioural self-relevance specifically. That is, participants may be less willing to *'ignore'* the animal after reading the behavioural self-relevance manipulation simply because the manipulation made the animal salient, and less willing to *'let [the animal] die off'* as the animal's possible role in food production is salient. Therefore, participants may be more invested in stopping the species from becoming extinct due to the species' utility for food production.

Whilst there was this significant effect of behavioural self-relevance on passive harm, this finding was in the opposite direction to expected, which is contradictory to previous motivated cognition literature, whereby behavioural self-relevance would theoretically increase harm (Loughnan et al., 2014; Piazza & Loughnan, 2016). However, this finding does agree with Study 2, whereby passive harm also reduced from pre- to post-manipulation. This effect was small-sized across Studies 2-3, so may have lower practical significance.

In line with Study 2, behavioural and subjective self-relevance did not affect warmth, active help, or active harm (although the active harm and active help ANCOVAs should be interpreted with caution due to floor and ceiling effects respectively). Lack of relationships between behavioural and subjective self-relevance and warmth contrast to Study 1 and may not support previous literature on self-relevance and motivated cognition (Gradidge et al., 2021a; Loughnan et al., 2010, 2014; Piazza & Loughnan, 2016; Rothgerber, 2020), which would expect people to view animals as less warm when they are behaviourally and/or

subjectively self-relevant (due to motivated cognition). However, Studies 2-3 provide no evidence for these hypothesised effects of behavioural and subjective self-relevance, and instead may indicate that behavioural and subjective self-relevance, as manipulated and measured across Studies 2-3, do not cause pet speciesism.

4.6.4.1.3. H3-H4: Mediational Relationships Between Species, Warmth/Competence and Active and Passive Behavioural Intentions.

The mediation hypothesis for warmth (H3) was supported, whereby warmth significantly mediated relationships between species and active help and active harm. That is, the animal being a pig or a tree kangaroo (vs. not) decreased the animals' warmth, which in turn decreased active help and increased active harm. However, mediation hypotheses for competence (H4) were mostly not supported. Specifically, pre-manipulation competence significantly mediated the relationship between species and pre-manipulation passive help, but competence did not significantly mediate relationships between species and postmanipulation passive help, nor between species and passive harm across timepoints. Therefore, the animal being a pig or a tree kangaroo (vs. not) decreased the animals' premanipulation competence, which in turn decreased pre-manipulation passive help, but this finding did not extend to post-manipulation passive help nor to passive harm across timepoints. As dogs were included as the reference category for these mediation analyses, mediational findings are not presented for perceptions of dogs. The current mediational findings fully corroborate the BIAS map as applied to animals for warmth (Sevillano & Fiske, 2016), as warmth predicted greater active help and lower active harm across timepoints, but not always for competence, as competence predicted greater passive help only at pre-manipulation (with no significant relationship post-manipulation) and lower passive harm only at post-manipulation (with no significant relationship pre-manipulation).

4.6.4.1.4. Effects of Behavioural Self-Relevance, Subjective Self-Relevance and Species on Reported Meat Consumption.

Not supporting prior research (Rothgerber, 2019), but in line with Study 2, there was no evidence for motivated under-reporting of meat consumption within any behavioural and subjective self-relevance or species condition. However, under-reporting present across all conditions could not be investigated within this study due to lack of implicit meat consumption measures (e.g., food diaries; Gradidge et al., 2021a) to corroborate the explicit meat consumption measure used here. Instead, future research should compare reported meat consumption with actual meat consumption (e.g., through food diaries) to detect underreporting.

4.6.4.1.5. Effect of Behavioural Self-Relevance on Discomfort.

There was a significant large-sized effect of behavioural self-relevance on discomfort differing across species, with participants feeling greater discomfort post-manipulation (vs. pre-manipulation) after exposure to dogs and tree kangaroos, but not pigs. This species difference may arise from normalisation of pigs as food and hence no discomfort due to the everyday reality and accessibility of pig meat, unlike the lack of normalisation of dogs and tree kangaroos as food with the current sample.

4.6.4.2. Limitations of the Current Study and Directions for Future Research.

The current study has some limitations to be addressed in future research: a) assumed neutrality of the tree kangaroo, and b) lack of testing of the full motivated cognition model.

Firstly, drawing on the 'novel animal paradigm', the study assumes neutrality of the 'unknown' animal (the tree kangaroo), whereby the tree kangaroo would be deemed less positive than dogs and more positive than pigs. However, bias was not explicitly tested within the current study beyond a self-report neutrality item, which may itself be subject to self-

report bias. Whilst efforts to reduce confounding variables and bias were implemented (e.g., the tree kangaroo matching dogs and pigs on evident characteristics like being a mammal, withholding the species name, and including neutrality as a covariate within analyses), participants may still have been biased by the photograph of the animal, or may even be more broadly biased in their perceptions of unknown mammalian animals. For example, one-tailed one-sample *t*-tests on the current data indicate all animals, even the tree kangaroo, were deemed significantly warmer and more competent than zero, ps < .05. Participants were also significantly more willing to actively and passively help, and less willing to actively and passively harm, all animals compared to zero, ps < .05. These effects were all present premanipulation. These biased preconceptions of the animals, even the 'neutral' tree kangaroo, mean unmoderated and unbiased causal effects of behavioural and subjective self-relevance on perceptions of and behavioural intentions towards animals are difficult to assess. Future research will therefore need to directly account for and include (implicit) bias as a covariate within analyses, such as through an Implicit Association Test, whereby photographs of different animals have to be paired with 'good' vs. 'bad' descriptors.

A second limitation of the current study is its lack of testing of the full motivated cognition model due to inadequate measures of dissonance. Valid and reliable measures must be developed to measure meat-related dissonance and other motivated cognition variables, like moral engagement (Gradidge et al., 2021a). These measures would ideally measure dissonance and moral engagement implicitly to avoid biased self-report. For example, the current study assumes positive perceptions of animals as a proxy for moral engagement, whereby greater warmth indicates moral engagement, and lower warmth indicates moral disengagement. However, moral engagement was not measured directly, so these assumptions cannot be explicitly tested. Whilst a reliable and valid 'Moral Disengagement in Meat Questionnaire' has been developed (Graça et al., 2016), future research should aim to

develop an equivalent 'Moral Engagement in Meat' scale to aid full testing of the motivated cognition model. Future research could also measure dissonance with alternative measures that are not vulnerable to issues with self-report, like skin conductance response (Harmon-Jones et al., 1996), and brain imaging (Izuma et al., 2010; Izuma & Murayama, 2019).

4.6.4.3. Conclusion for the Current Study.

To summarise, the current study mostly corroborates previous pet speciesism literature by finding that dogs are perceived as warmer and more competent than pigs and tree kangaroos, and subject to more positive behavioural intentions. However, participants unexpectedly did not view tree kangaroos and pigs as differing on warmth, and participants did not differ in active and passive help towards tree kangaroos and pigs. The current study also found null effects of behavioural and subjective self-relevance on warmth, competence, active help, active harm, and passive help, alongside an unexpected effect of behavioural selfrelevance on passive harm for all animals, not supporting previous self-relevance literature.

Finally, the study replicated applicability of the SCM/BIAS map across species for warmth, as demonstrated by significant mediations between species and behavioural intentions through warmth. However, the study did not always replicate applicability of the SCM/BIAS map across species for competence, as mediations between competence and postmanipulation passive help, and between competence and passive harm across timepoints, were not significant. The current study has limitations regarding biases in perceptions of animals, even unknown animals, and lack of testing of the full motivated cognition model. Future research should investigate and include implicit bias as a covariate and test the motivated cognition model in full.

Overall, the current study combines with Study 2 to suggest that behavioural and subjective self-relevance mostly may not inform perceptions of and behavioural intentions

towards animals. Additionally, whilst there was an effect of behavioural self-relevance on passive harm, this effect was in the opposite direction to predicted. These findings therefore may not support motivated cognition literature, and thus indicate behavioural and subjective self-relevance may not explain why people view pigs negatively (i.e., pet speciesism).

To conclude, Studies 2-3 are the first to causally test the roles of behavioural and subjective self-relevance with dogs and pigs, and uniquely may not support established motivated cognition literature by indicating neither behavioural nor subjective self-relevance cause pet speciesism.

4.7. Overall Discussion

Through four follow-up and pilot studies, it was established that a) the original behavioural and subjective self-relevance manipulations utilised within Study 2 were ineffective at manipulating behavioural and subjective self-relevance, which may explain null findings from Study 2, and b) an alternative 'imagined restaurant scenario' was effective at manipulating behavioural and subjective self-relevance. As such, this alternative selfrelevance manipulation was implemented within Study 3, to again test if behavioural and subjective self-relevance are causes of pet speciesism. Overall, Study 3 found that behavioural and subjective self-relevance mostly may not inform perceptions of and behavioural intentions towards animals, and thus may not be causes of pet speciesism.

Chapter 5. Pilot Study and Study 4 Exploring Familiarity and Similarity

5.1. Introduction

The current chapter first presents a pilot study (Pilot Study 1b) to develop and pilot familiarity and similarity manipulations, before then implementing these manipulations within Study 4 in this chapter, to test if familiarity and similarity are causes of pet speciesism. These studies are now reported in turn below.

5.2. Pilot Study 1b to Test Familiarity and Similarity Manipulations

5.2.1. Introduction

As Study 1 found correlational relationships between familiarity, similarity, and dogs'/pigs' warmth and/or competence, I aimed to test if these relationships are causal, and therefore if familiarity and similarity can explain pet speciesism. However, before testing for these causal roles, I first needed to develop and pilot familiarity and similarity manipulations which effectively manipulate familiarity and similarity respectively. As such, the current pilot study (Pilot Study 1b) therefore aims to test if familiarity and similarity manipulations developed by the current researcher are effective at manipulating their intended variables. Familiarity is manipulated through either imagined contact with the participant's target animal (familiar condition) or no text (unfamiliar condition), and this manipulation has been informed by imagined contact literature with animals (e.g., Auger & Amiot, 2019b; Cerrato & Forestell, 2022). Similarity is manipulated through fictional newspaper text describing the target animal as either being similar or dissimilar to humans across four traits (behaviour, level of intelligence, sociality, and emotionality), and this manipulation has been informed by previous text manipulations used within anthrozoological literature to elicit similarity (Horne et al., 2021; Leach et al., 2021; Piazza & Loughnan, 2016). As with Studies 2-3, an unknown animal (either tree kangaroo, fossa, or tamandua) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of familiarity. To

check effectiveness of the familiarity manipulation (manipulation check), the item '*How familiar do you perceive tree kangaroos/fossas/tamanduas to be to you?*' is included to measure familiarity. To check effectiveness of the similarity manipulation (manipulation check), the item '*How similar do you perceive tree kangaroos/fossas/tamanduas to be to you?* is included to measure similarity. Informed by previous imagined contact manipulations with animals (Auger & Amiot, 2019b; Cerrato & Forestell, 2022) and previous text manipulations of similarity for animals (Horne et al., 2021; Leach et al., 2021; Piazza & Loughnan, 2016), I hypothesise that:

H1: Imagined contact with an animal (vs. not) will significantly increase the animal's measured familiarity.

H2: Stated similarity (vs. dissimilarity) of an animal to humans will significantly increase the animal's measured similarity.

Note that this pilot study utilises the same participant sample to also follow-up/pilot text manipulations for behavioural self-relevance, subjective self-relevance, and pet status, but only information relevant to the familiarity and similarity manipulations is reported within the current chapter (see Section 4.2. for pilot study information about the behavioural and subjective self-relevance manipulations [Pilot Study 1a], and Section 7.2. for pilot study information about the pet status manipulation [Pilot Study 1c]).

5.2.2. Methods

5.2.2.1. Participants.

Sixty-eight participants were recruited via social media and Sona as a volunteer sample. Thirteen participants were excluded for providing partial data. As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 5.2.3. for further

discussion). Exclusions left a total sample size of 55 participants (81.8% women, 18.2% men;
M_{age} =23.38, SD_{age} =5.68, age range: 18-48), which exceeds the recommend 12 participants per
condition for pilot studies (Julious, 2005). Participants within each condition are shown in
Table 12. Demographics of the sample are reported as follows in percentages, with
demographic categories which make up less than 2% of the sample being collapsed into
'other'. For dietary group: 69.1% (meat consumer), 12.7% (reducetarian), 7.3% (vegetarian),
and 5.5% each (flexitarian, pescatarian). For nationality: 54.5% (British), 5.5% (Romanian),
3.6% each (German, Indian, Lithuanian, no response, Portuguese, Spanish), and 18.4% other.
For ethnicity: 76.1% (White), 12.6% (Asian), 3.6% each (mixed, would rather not say), and
4.1% other. For country of residence ($M_{duration}$ =17.91, $SD_{duration}$ =10.12, range: 1-48 years):
81.8% (United Kingdom), 3.6% (Germany), and 14.6% other. For religion: 29.1% (atheism),
21.8% (Christianity), 18.2% (agnosticism), 14.5% (would rather not say), 5.5% (no
response), 3.6% each (Buddhism, Islam), and 3.7% other.

Table 12.

Participants Within Each Condition for Pilot Study I	Participants	Within	Each	Condition	for	Pilot	Study	1b.
--	--------------	--------	------	-----------	-----	-------	-------	-----

Familiarity Condition	Similarity Condition	Number of Participants
Familiar	Similar	13
Familiar	Dissimilar	16
Unfamiliar	Similar	13
Unfamiliar	Dissimilar	13

5.2.2.2. Design.

The current pilot study follows a 2(familiarity: familiar vs. unfamiliar) x 2(similarity: similar vs. dissimilar) between-subjects MANCOVA design, with measured familiarity and similarity as the DVs. Familiarity and similarity were included as DVs to test effectiveness of the manipulations. Gender was included as a covariate (see Section 5.2.3. for further discussion).
5.2.2.3. Materials.

5.2.2.3.1. Familiarity and Similarity Manipulations.

Familiarity was manipulated via an imagined contact scenario developed by the researcher. Specifically, participants were either asked to '*Please now imagine meeting and interacting with a tree kangaroo/fossa/tamandua* [dependent on target animal]' (familiar condition) or alternatively read no text (unfamiliar condition). Similarity was manipulated via the following texts developed by the researcher: '*Tree kangaroos/fossa/tamanduas* [dependent on target animal] *are reportedly similar/dissimilar* [manipulation of similarity condition] *to humans in many areas, including their behaviour, level of intelligence, sociality and emotionality*'. The familiarity and similarity manipulations were combined within this pilot study to assess if familiarity and similarity interact (e.g., from previous literature; Norton et al., 2007), whereby the similar or dissimilar condition only (participants in the unfamiliar condition saw no additional text).

5.2.2.3.2. Measured Familiarity.

Perceived familiarity was measured via the same single item from Study 1, except 'How familiar do you perceive the following animals to be to you?' was reworded to 'How familiar do you perceive tree kangaroos/fossas/tamanduas to be to you?' Additionally, the item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score indicates greater perceived familiarity.

5.2.2.3.3. Measured Similarity.

Perceived similarity of the animal to humans was measured using the same single item from Study 1, except '*How similar do you perceive the following animals to be to humans?*' was reworded to '*How similar do you perceive tree kangaroos/fossas/tamanduas* *to be to humans?* '. Like the perceived familiarity manipulation check, this perceived similarity to humans item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score indicates greater perceived similarity to humans.

5.2.2.3.4. Attention and Memory Checks.

Attention paid to and memory for the text manipulations was tested through four attention and memory check items developed by the researcher as follows: '*Did the text that you just read ask you to imagine a scenario*?', '*What animal did the text refer to*?', '*Did the text that you just read ask you to imagine contact with a tree kangaroo/fossa/tamandua*?', and '*Did the text that you just read state that tree kangaroos/fossa/tamanduas are similar or dissimilar to humans*?'. These checks aimed to test that participants paid attention to and remembered the texts. Correct answers for each item received a score of one, whilst incorrect answers for each item received a score of zero. All answers were then added together, for a highest possible score of four. Inspection of these checks revealed that the vast majority of the sample (>75%) had perfect scores, indicating sufficient attention and memory for these manipulations.

5.2.2.3.5. Vividness of the Familiarity Manipulation.

Perceived vividness of the imagined familiarity manipulation was measured, as imagined scenarios which are viewed as more vivid have greater impacts on behavioural intentions (Husnu & Crisp, 2010). Vividness was measured through a six-item semantic differential vividness measure from Husnu and Crisp (2010; α =.94), whereby participants answer how much they perceive their imagined scenario as being '*faint vs. vivid*', '*fuzzy vs. clear*', '*dim vs. bright*', '*vague vs. sharp*', '*dull vs. lively*', and '*simple vs. detailed*'. Reliability was adequate for this scale within this study, ω =.89. Therefore, items were summed together to create a total vividness score. Higher scores indicate greater vividness.

No items are reverse-scored. The majority of the sample (>50%) perceived the familiarity manipulation to be vivid.

5.2.2.3.6. Ease of Imaginability of the Familiarity Manipulation.

Perceived ease of imaginability of the imagined familiarity manipulation was measured in order to ensure participants were able to imagine the imagined contact scenario without difficulty. Ease of imaginability was measured through a single item '*How easy or difficult was it for you to imagine the previous scenario in the text you have just read*?' on a -50 (extremely difficult) to 50 (extremely easy) visual analogue scale, informed by previous literature (e.g., Black & Barnes, 2020; Broemer & Diehl, 2004). A higher score indicates greater ease of imaginability. The majority of the sample (>50%) perceived the familiarity manipulation to be easy to imagine.

5.2.2.3.7. Believability of the Similarity Manipulation (Beltramini, 1982; Beltramini & Evans, 1985; Chang, 2011).

Perceived believability of the similarity text manipulation was measured in order to ensure the text was not simply disregarded. Believability was measured through a six-item semantic differential believability measure (Beltramini, 1982; Beltramini & Evans, 1985; α =.94 from Chang, 2011), whereby participants answer how much they perceive the text to be 'unbelievable vs. believable', 'untrustworthy vs. trustworthy', 'not credible vs. credible', 'unreasonable vs. reasonable', 'not convincing vs. convincing', and 'biased vs. unbiased'. Reliability was adequate for this scale within this study, ω =.89. Therefore, items were summed together to create a total believability score. Higher scores indicate greater believability. No items are reverse-scored. The majority of the sample (>50%) perceived the similarity manipulation to be believable.

5.2.2.4. Procedure

Participants were recruited via social media and Sona. The experiment was conducted in September and October 2020 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos (alongside a photograph of the animal), followed by a fossa and tamandua if participants recognised any previous animal. Animal species names were used throughout the study. The study either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants completed the three sections (selfrelevance, familiarity/similarity, and pet status) of the pilot study in a randomised order. Here, only the familiarity/similarity section is described (see Section 4.2. for description of the self-relevance section, and Section 7.2. for description of the pet status section). For the familiarity/similarity section, participants were randomly assigned to read one of the four familiarity and similarity text manipulations described in Section 5.2.2.3.1. regarding their target animal. After reading their familiarity/similarity text, participants answered the five attention and memory checks, completed the measures for familiarity and similarity (manipulation checks), and then indicated their perceived vividness and ease of imaginability of the imagined scenario, and perceived believability of the text. Finally, participants provided demographics before being debriefed. One participant reported technical difficulties, but their response was complete and maintained within analyses.

5.2.3. Analyses

A 2(familiarity: familiar vs. unfamiliar) x 2(similarity: similar vs. dissimilar) between-subjects MANCOVA, including gender as a covariate, was conducted on measured familiarity and similarity, to test if the manipulations effectively manipulate familiarity and similarity. As the DVs were conceptually related (i.e., previously found to interact, Norton et

al., 2007), and were moderately correlated within the current study, r > .3, MANCOVA was deemed appropriate for this analysis.

Gender was again included as a covariate within this MANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured familiarity/similarity (the outcome variables) to be statistically controlled for.

Descriptive statistics for *post hoc* pairwise comparisons for significant interaction effects are reported in Tables 13 and 14.

5.2.3.1. Statistical Assumptions to Test H1-H2: Exploring Effects of Familiarity and Similarity Manipulations on Measured Familiarity and Similarity (2x2 MANCOVA).

The DVs failed normality tests in some conditions, ps < .05. However, both DVs had acceptable skewness. Measured similarity had non-extreme univariate outliers in two conditions, which did not cause excessive skewness. There were no univariate outliers in measured familiarity. There was one multivariate outlier for two DVs, critical values ≥ 13.82 , p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers changed one minor multivariate conclusion (Footnote 22). I report the MANCOVA including univariate and multivariate outliers to reflect the unamended dataset. There was multicollinearity in the unfamiliar condition whereby the animal is described as similar, and in the unfamiliar condition whereby the animal is described as dissimilar, rs = .92. However, correlations between the DVs were acceptable (<.9) in the familiar conditions, $rs \le .64$, so I maintained both variables separately within a MANCOVA. Linear relationships were present across conditions. Running a follow-up univariate ANCOVA including and excluding univariate outliers did not change univariate conclusions. I report this ANCOVA including outliers to reflect the unamended dataset.

Homogeneity of variances was present for measured familiarity, p = .1, but not for measured similarity, p = .001. Homogeneity of covariances was not present, p < .001.

5.2.3.2. Testing H1-H2: Exploring Effects of Familiarity and Similarity Manipulations on Measured Familiarity and Similarity.

A 2(familiarity: familiar vs. unfamiliar) x 2(similarity: similar vs. dissimilar) between-subjects MANCOVA, including gender as a covariate, was run on measured familiarity and similarity to test **H1-H2**.

5.2.3.2.1. Multivariate Effects of Familiarity and Similarity Manipulations on the Combined DVs.²²

Not supporting H1: There was no significant main effect of familiarity on the combined DVs, F(2, 49) = .55, p = .58, $\eta_p^2 = .02$. However, there was a significant main effect of similarity, F(2, 49) = 8.52, p < .001, $\eta_p^2 = .26$ (large-sized), and a significant interaction between familiarity and similarity, F(2, 49) = 6, p = .01, $\eta_p^2 = .2$ (large-sized), on the combined DVs. I followed up these significant multivariate effects with univariate ANCOVAs on the DVs separately.

²²When excluding univariate and multivariate outliers, the interaction between familiarity and similarity on the combined DVs becomes non-significant, F(2, 37) = 2.9, p = .07, $\eta_p^2 = .14$.

5.2.3.2.2. H1: Univariate Effects of Familiarity and Similarity Manipulations on Measured Familiarity.

There was a significant interaction between familiarity and similarity on measured familiarity, F(1, 50) = 4.42, p = .04, $\eta_p^2 = .08$ (medium-sized). *Post hoc* pairwise comparisons revealed that, when the animal was described as being similar to humans only, the animal was deemed more familiar in the familiar condition than in the unfamiliar condition, p = .046. No such effect was present when the animal was described as being dissimilar to humans, p = .39. Additionally, in the familiar condition only, the animal was deemed more familiar when the animal was described as being similar to humans compared to when the animal was described as being dissimilar to humans, p = .002. No such effect was present in the unfamiliar condition, p = .78.

Table 13.

Descriptive Statistics for Interaction Effect of Familiarity and Similarity Manipulations on Measured Familiarity.

Condition	Familiar Mean	Familiar SE	Unfamiliar Mean	Unfamiliar SE
Similar	42.95	6.39	24.36	6.43
Dissimilar	14.21	5.85	21.82	6.43

There was also a significant main effect of similarity on measured familiarity, F(1, 50) = 6.3, p = .02, $\eta_p^2 = .11$ (medium-to-large-sized). However, this main effect is qualified by the above two-way interaction and is therefore not expanded upon further here.

5.2.3.2.3. H2: Univariate Effects of Familiarity and Similarity Manipulations on Measured Similarity.

Supporting H2: There was a significant main effect of similarity on measured similarity, F(1, 50) = 16.4, p < .001, $\eta_p^2 = .25$ (large-sized), whereby animals were deemed more similar when described as similar (M = 40.33, SE = 4.53) as opposed to described as dissimilar (M = 15.03, SE = 4.3). However, this main effect is qualified by a significant

interaction between familiarity and similarity on measured similarity, F(1, 50) = 11.54, p = .001, $\eta_p^2 = .19$ (large-sized). *Post hoc* pairwise comparisons revealed that, when the animal was described as being similar to humans only, the animal was deemed more similar in the familiar condition than in the unfamiliar condition, p = .003. No such effect was present when the animal was described as being dissimilar to humans, p = .11. Additionally, in the familiar condition only, the animal was deemed more similar when the animal was described as being dissimilar to humans, p = .11. Additionally, in the familiar condition only, the animal was deemed more similar when the animal was described as being dissimilar to humans, p = .001. No such effect was present in the unfamiliar condition, p = .65.

Table 14.

Descriptive Statistics for Interaction Effect of Familiarity and Similarity Manipulations on Measured Similarity.

Condition	Familiar Mean	Familiar SE	Unfamiliar Mean	Unfamiliar SE
Similar	54.37	6.4	26.3	6.45
Dissimilar	7.84	5.86	22.22	6.45

5.2.4. Discussion and Conclusion

As expected, the similar condition led to greater perceived similarity than the dissimilar condition, thereby supporting **H2**, and in line with previous literature utilising text manipulations for similarity (e.g., Horne et al., 2021; Leach et al., 2021; Piazza & Loughnan, 2016). Therefore, the current study demonstrates that the similarity manipulation developed by the researcher is effective at manipulating similarity. However, effects of this manipulation on measured similarity may vary depending on *familiarity*, indicating that similarity and familiarity must be separated within experimental manipulations to avoid interaction effects.

Furthermore, the familiarity manipulation was not found to be effective within the current pilot study, which does not support **H1** or previous imagined contact literature (Auger & Amiot, 2019b; Cerrato & Forestell, 2022). However, again, the familiarity and similarity

manipulations did interact on measured familiarity, thus further emphasising the importance of separating similarity and familiarity within manipulations.

Informed by these findings, the similarity and familiarity manipulations are separated and utilised within Study 4 to test causal effects of familiarity and similarity on pet speciesism. As the similarity manipulation was found to be effective in manipulating similarity here, this manipulation is mostly unchanged within Study 4. However, the manipulation has been slightly amended to attempt to enhance effectiveness of the manipulation. Specifically, legitimacy of the texts and appeal to authority are emphasised (Walton, 2010) by including the phrases '*imagine reading the following text in the* newspaper' and 'according to animal behaviour experts' at the beginning of the text within Study 4. As the familiarity manipulation was found to be ineffective in manipulating familiarity here, the manipulation has been amended for Study 4 in an attempt to make this manipulation effective. Firstly, participants elaborated on the imagined contact by writing a passage of text describing the imagined scenario for one minute, providing as much detail as possible. This descriptive task was included to enhance vividness and realism of the imagined scenario ('elaboration'; Husnu & Crisp, 2010). A similar elaboration task has been utilised before for imagined contact with animals (Auger & Amiot, 2019b). Secondly, the unfamiliar condition was amended from a control condition with no imagined scenario to an imagined scenario interacting with a football, to ensure comparison of like-for-like imagined scenarios (i.e., both scenarios require active involvement and occur outdoors).

5.3. Study 4: Causal Exploration of Familiarity and Similarity

5.3.1. Introduction and Extension from Study 1

Study 1 found that familiarity was significantly positively linked only to dogs' warmth and only pigs' competence. The differential relationships between familiarity and warmth vs. competence depending on species only partially align with previous research,

which would instead suggest positive relationships between familiarity and all perceptions of animals (Morris et al., 2012; Possidónio et al., 2019). Specifically, people have typically been found to prefer animals with which they are more familiar (e.g., through imagined contact; Auger & Amiot, 2019b), whereby research on familiarity with human outgroups indicates positive effects of familiarity through intergroup contact operate through reducing negative affect (e.g., intergroup anxiety; Pettigrew & Tropp, 2008), and increasing positive affect (Tausch & Hewstone, 2010), towards the outgroup. Therefore, imagined intergroup contact with an animal would be theorised to enhance positive affect, and reduce negative affect, towards the animal, thus improving the animal's perceived warmth and competence, and behavioural intentions towards the animal. Study 1 findings, whereby greater familiarity was associated only with dogs' greater warmth and pigs' greater competence, therefore indicate greater complexity when applying familiarity to the SCM, whereby the relationship between familiarity and warmth/competence perceptions of an animal may be moderated by the species of the animal.

The current study therefore extends previous literature and further investigates Study 1 findings by testing *causal* effects of familiarity on SCM variables (warmth/competence), whereby familiarity is manipulated through an imagined contact manipulation either with the participant's target animal (familiar condition) or an unrelated object (football; unfamiliar condition). This familiarity manipulation was piloted in Pilot Study 1b (Section 5.2.), and has been adapted slightly within the current study by changing the unfamiliar condition from no text within the pilot study to an alternative imagined scenario with a football in the current study. Additionally, to enhance vividness of the imagined contact, participants are now asked within the current study to write about this imagined scenario for one minute. As with Studies 2-3, an unknown animal (either tree kangaroo, fossa, or tamandua) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of

familiarity. Warmth and competence are again measured within the current study using the SCM subscales from Sevillano and Fiske (2016b). In line with the equivocal Study 1 findings discussed above, I predict that:

H1: Familiar (vs. unfamiliar) animals will be deemed significantly warmer (a) and/or more competent (b).

Unlike familiarity, Study 1 found that similarity was significantly positively associated with warmth and competence of dogs and pigs. These positive relationships between similarity and perceptions of animals align fully with previous literature (Leach et al., 2021), whereby people prefer and/or view more favourably animals which are described as or deemed more similar to humans (Batt, 2009; Kozachenko & Piazza, 2021; Leach et al., 2021; Possidónio et al., 2019). This favourable effect of similarity is theorised to arise from similarity enabling 'humanisation' of the similar animal, by encouraging inclusion of the animal within our moral circle, and thereby enhancing moral concern for the animal (Bastian et al., 2012): As such, previously found positive effects of similarity may extend to the SCM, whereby an animals' greater perceived similarity to humans would theoretically cause the animal's greater perceived warmth and competence. The current study therefore extends previous literature and Study 1 findings by testing causal effects of similarity on SCM variables (warmth/competence), whereby similarity is manipulated within the current study through fictional newspaper text describing the target animal as either being similar or dissimilar to humans across four traits (behaviour, level of intelligence, sociality and emotionality), piloted in Pilot Study 1b (Section 5.2.). Again, an unknown animal (either tree kangaroo, fossa, or tamandua) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of similarity. Informed by above discussed previous literature and Study 1 findings, it is hypothesised that:

H2: Purportedly similar (vs. dissimilar) animals will be deemed significantly warmer(a) and more competent (b).

Additionally, like Studies 2-3, the current study extends possible causal effects of familiarity and similarity from warmth and competence to include behavioural intentions (active help, active harm, passive help, passive harm), in line with the BIAS map, whereby behavioural intentions are again measured within the current study using the BIAS map subscales from Sevillano and Fiske (2016b). Specifically, the BIAS map states that warmth should inform greater active help and less active harm, whilst competence should inform greater passive harm (Cuddy et al., 2007). Thus, per the BIAS map, relationships between familiarity and similarity and warmth and/or competence from Study 1 should extend to behavioural intentions within this study. Therefore, I predict that differences in warmth/competence perceptions of the target animal arising from familiarity and similarity will extend to behavioural intentions, such that:

H3: If familiar (vs. unfamiliar) animals are deemed significantly warmer (i.e., H1a is evidenced), familiar (vs. unfamiliar) animals will be subject to significantly less active harm (a) and more active help (b). If familiar (vs. unfamiliar) animals are deemed significantly more competent (i.e., H1b is evidenced), familiar (vs. unfamiliar) animals will be subject to significantly less passive harm (c) and more passive help (d).

H4: If purportedly similar (vs. dissimilar) animals are deemed significantly warmer (i.e., H2a is evidenced), purportedly similar (vs. dissimilar) animals will be subject to significantly less active harm (a) and more active help (b). If purportedly similar (vs. dissimilar) animals are deemed significantly more competent (i.e., H2b is evidenced),

purportedly similar (vs. dissimilar) animals will be subject to significantly less passive harm (c) and more passive help (d).

Inclusion of measurements for behavioural intentions within this study also again enables applicability of the BIAS map to animals to be tested (Sevillano & Fiske, 2016b, 2019). That is, it can be tested directly if any effects of familiarity and similarity on active and passive behavioural intentions are mediated through warmth and competence respectively, in line with both general SCM and BIAS map literature (Cuddy et al., 2007) and SCM/BIAS map literature applied to animals specifically (Sevillano & Fiske, 2016b, 2019). As such, I hypothesise that:

H5: Post-manipulation warmth will significantly mediate relationships between familiarity and active harm, and familiarity and active help. That is, familiar (vs. unfamiliar) animals will be deemed significantly warmer post-manipulation, which will in turn significantly decrease post-manipulation active harm (**a**) and increase post-manipulation active help (**b**).

H6: Post-manipulation warmth will significantly mediate relationships between similarity and active harm, and similarity and active help. That is, animals portrayed as similar (vs. dissimilar) will be deemed significantly warmer post-manipulation, which will in turn significantly decrease post-manipulation active harm (**a**) and increase post-manipulation active help (**b**).

H7: Post-manipulation competence will significantly mediate relationships between familiarity and passive harm, and familiarity and passive help. That is, familiar (vs. unfamiliar) animals will be deemed significantly more competent post-manipulation, which will in turn significantly decrease post-manipulation passive harm (**a**) and increase post-manipulation passive help (**b**).

H8: Post-manipulation competence will significantly mediate relationships between similarity and passive harm, and similarity and passive help. That is, animals portrayed as similar (vs. dissimilar) will be deemed significantly more competent post-manipulation, which will in turn significantly decrease post-manipulation passive harm (**a**) and increase post-manipulation passive help (**b**).

5.3.2. Methods

5.3.2.1. Participants.

Three-hundred-and-forty-six participants were recruited via volunteer sampling. Qualtrics quota sampling was implemented to ensure approximately 50% men and 50% women, alongside representative numbers of people of different dietary identities in the sample in proportion to the population: approximately 86% meat consumers, 6% vegetarians, 5% pescatarians, and 3% vegans per the United Kingdom population (Johnson, 2022). People with a flexitarian diet did not have a specific quota, as numbers of flexitarians in the population have not been systematically measured.

One-hundred-and-forty-seven participants were excluded for: exceeding demographic quotas (n=99), partial data (n=37), or participant withdrawal (n=9). As all analyses included gender as a covariate (see Section 5.3.3. for further discussion), two non-binary participants were excluded to enable gender to be dummy coded into men (coded as zero; n=98) and women (coded as one; n=101). Exclusions left a sample of 199 participants (50.8% women, 49.2% men; $M_{age}=25.36$, $SD_{age}=8.43$, age range=18-69), which exceeded the minimum required sample size of 179 per a G*Power *a priori* power analysis for a MANOVA analysis ('repeated measures, within-between interaction', medium effect size f=.25, four groups, two measurements, $\alpha=.05$, power = .8).

One-hundred-and-ninety participants completed the study about tree kangaroos, whilst nine participants recognised the tree kangaroo and completed the study about fossas (n=8) or tamanduas (if they recognised the fossa; n=1). Participants within each condition are shown in Table 15. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. The percentage of participants for each diet approximately matches sizes within the general population. For dietary group: 84.4% (meat consumer), 5.5% (vegetarian), 5% (pescatarian), 2.5% (vegan), and 2.6% other. For nationality: 51.3% (United Kingdom/British), 12.6% (United States), 7% (Indian), 2% (Romanian), and 27.1% other. For ethnicity: 73.2% (White), 14.5% (Asian), 6.5% (mixed), 2.5% (Black), and 3.3% other. For country of residence ($M_{duration}$ =20.53, $SD_{duration}$ =10.8, range: six months to 68 years): 62.3% (United Kingdom), 12.6% (United States), 6% (India), and 19.1% other. For religion: 29.1% (Christianity), 24.1% (atheism), 14.6% (agnosticism), 12.6% (would rather not say), 6.5% (Hinduism), 4.1% each (Islam, no religion), 2.5% (Budchism), and 2.4% other.

Table 15.

Participants Within Each Condition for Study 4.

Condition	Number of Participants	
Similar	48	
Dissimilar	52	
Familiar	50	
Unfamiliar	49	

5.3.2.2. Design.

This experiment follows a 4(condition: familiar vs. unfamiliar vs. similar vs. dissimilar; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed MANCOVA design, with warmth, competence, active help, passive help, active harm, and passive harm as the DVs. Additionally, to test effectiveness of the texts in manipulating

measured familiarity or measured similarity (manipulation checks), a one-way (condition: familiar vs. unfamiliar) ANCOVA on measured familiarity, and a one-way (condition: similar vs. dissimilar) ANCOVA on three measurements of similarity, were also conducted. Gender and neutrality were included as covariates (see Section 5.3.3. for further discussion).

5.3.2.3. Materials.

5.3.2.3.1. Target Animals and Animal Photographs.

The current study again employed the novel animal paradigm by implementing the same photographs of the animals (tree kangaroo, fossa, tamandua) as Study 2, to enable easier visualisation of the imagined scenario, and avoid biased perceptions of the animal due to species name. The tree kangaroo photograph was sourced from https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035, the fossa photograph sourced from https://www.flickr.com/photos/mathiasappel/19504925051, and the tamandua photograph sourced from https://www.flickr.com/photos/mathiasappel/19504925051,

5.3.2.3.2. Familiarity Text Manipulations.

As imagined contact with animals has previously been utilised to inform perceptions of animals (Auger & Amiot, 2019b), familiarity was manipulated via an imagined contact scenario developed by the researcher (piloted in Pilot Study 1b). Specifically, participants were asked to either '*Please now imagine meeting and interacting with the animal in the photograph*', whereby the 'animal in the photograph' refers to the participant's unknown target animal (tree kangaroo, fossa, tamandua; familiar condition), or asked to '*Please now imagine interacting with a football*', whereby the football constitutes an unrelated object (unfamiliar condition). All participants were then asked to do the following description task: '*In the textbox below, describe this imagined scenario. Provide as much detail as possible and keep writing until at least 1 minute has passed.*' Nearly all participants (94% in familiar condition; 97.96% in unfamiliar condition) engaged with this description task within this

study. Participants were not excluded for not engaging with this task, as the task was intended as an additional visualisation aid for elaboration purposes only. Overall, this familiarity text manipulation was informed by previous imagined contact manipulations with animals (Auger & Amiot, 2019b).

5.3.2.3.3. Similarity Text Manipulations.

Similarity was manipulated via texts developed and successfully piloted by the researcher (piloted in Pilot Study 1b). Specifically, participants were asked to '*Imagine reading the following text in the newspaper*', with the following text stating that '*According to animal behaviour experts, the species in the photograph is reportedly similar [similar condition] / dissimilar [dissimilar condition] to humans in many ways, including in terms of behaviour, level of intelligence, sociality and emotionality.' Text manipulations have been used previously within anthrozoological literature to manipulate similarity (Horne et al., 2021; Leach et al., 2021; Piazza & Loughnan, 2016). Participants read the text for at least one minute to ensure sufficient attention.*

5.3.2.3.4. Warmth and Competence (Sevillano & Fiske, 2016b).

Warmth and competence were measured with the same items from Studies 1-3: '*warm*', '*well-intentioned*', '*friendly*' (warmth), and '*competent*', '*skillful*', '*intelligent*' (competence; Sevillano & Fiske, 2016b). The current study utilised the same semantic differential scaling from -3 to 3 as Study 3 and referred to '*the species within the previous photograph*'. Items within each subscale were summed to provide warmth and competence scores. Reliability was adequate within this study for warmth (pre-manipulation ω =.73; 95% BCa CI [.63, .8]; post-manipulation ω =.84; 95% BCa CI [.77, .88]) and competence (premanipulation ω =.79; 95% BCa CI [.72, .84]; post-manipulation ω =.91; 95% BCa CI [.88, .94]).

5.3.2.3.5. Behavioural Intentions Towards the Animal (Sevillano & Fiske, 2016b).

Behavioural intentions were measured with the same items from Studies 2-3: 'support', 'help', 'behave friendly towards', 'interact with' the animal (active help), 'sustain', 'conserve' the animal (passive help), 'kill', 'injure', 'exterminate', 'trap', 'reject' the animal (active harm), and 'let the species die off', 'ignore' the animal (passive harm; Sevillano & Fiske, 2016b). The current study utilised the same semantic differential scaling from -3 to 3 as Study 3 and referred to 'the species within the previous photograph'. Items within each subscale were summed to provide active help, passive help, active harm, and passive harm scores. Reliability was adequate within this study for active harm (premanipulation $\omega = .8$; 95% BCa CI [.69, .88]; post-manipulation $\omega = .9$; 95% BCa CI [.81, .94]), and active help (pre-manipulation ω =.71; 95% BCa CI [.62, .79]; post-manipulation ω =.77; 95% BCa CI [.68, .84]. Reliability was inadequate within this study across timepoints for passive help (pre-manipulation r_{sb} =.66; post-manipulation r_{sb} =.69), and passive harm (premanipulation r_{sb} =.57; post-manipulation r_{sb} =.46). As the passive help and passive harm subscales only contain two items, I did not run PAFAs for these subscales. Due to the inadequate reliability of these subscales, findings from theses subscales should be interpreted with caution.

5.3.2.3.6. Perceived Neutrality Towards the Animal.

Neutrality was measured via the same single item from Study 3, with a higher score indicating higher neutrality and less bias: '*How neutral do you perceive the species in the previous photograph to be*?' from zero to 100.

5.3.2.3.7. Perceived Familiarity (Manipulation Check).

Perceived familiarity was measured via the same single item from Study 1, except *'How familiar do you perceive the following animals to be to you?'* was reworded to *'How familiar do you perceive the species in the photograph to be to you?'* to avoid participant bias from hearing the species name. Additionally, the item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score indicates greater perceived familiarity.

5.3.2.3.8. Perceived Similarity to Humans (Manipulation Check).

Perceived similarity of the animal to humans was measured with the same single item from Study 1, except '*How similar do you perceive the following animals to be to humans?*' was reworded to '*How similar do you perceive the species in the photograph to be to humans?*' to avoid participant bias from hearing the species name. This single item was utilised as one of the manipulation checks for the similarity manipulation. Like the perceived familiarity manipulation check, this perceived similarity to humans item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score indicates greater perceived similarity to humans.

5.3.2.3.9. Perceived Shared Traits with Humans (Manipulation Check; Bastian et al., 2012a, Bilewicz et al., 2011, Rothgerber, 2014).

To validate the perceived similarity to humans manipulation check, I also utilised a scale which explicitly measures perceived shared traits of the target animal with humans (Rothgerber, 2014, derived from Bastian et al., 2012a, Bilewicz et al., 2011). The scale splits into two subscales (emotional and cognitive capacities) and asks participants to rate how much they view an animal as sharing the following capacities with humans: '*nostalgia*', '*happiness*', '*melancholy*', '*excitement*', '*guilt*', '*panic*' (emotional capacities), and '*self-control*', '*morality*', '*memory*', '*planning*' (cognitive capacities). Wording was adapted within this study to refer to '*the species in the photograph*' throughout. Participants answered on a Likert scale from one '*only humans have this capacity*' to seven '*the species in the photograph and humans have this capacity to the same degree*', with the midpoint at four '*the species in the photograph has this capacity to some degree*'. No items were reverse-

scored. Items within each subscale were summed to provide emotional and cognitive capacity scores, with higher scores on each subscale indicating higher levels of that variable. Reliability was adequate within this study for emotional capacities (ω =.73; 95% BCa CI [.5, .81]), but reliability was inadequate for cognitive capacities (ω =.63; 95% BCa CI [.47, .73]). I therefore ran a PAFA (Appendix 2g) with direct oblimin rotation to assess factorial validity of the cognitive capacities subscale, alongside assessing inter-item correlations to further test reliability, and communality values to test common variance. All four items adequately loaded onto one factor (\geq .47), indicating acceptable factorial validity, whilst the first item had adequate communality (.57), and the other three items had adequate (though not ideal) communality (\geq .22; Child, 2006). Additionally, whilst item three correlated inadequately with item two, *r*=.14, and item four, *r*=.25, item three correlated adequately with item one, *r*=.41. I therefore retained item three. All other items correlated adequately with each other, *r*s \geq .31, and were therefore also retained. However, findings with this scale should be interpreted with caution due to inadequate reliability.

5.3.2.4. Procedure.

Participants were recruited via avenues like social media and Sona (Appendix 8). All participants were required to not have taken part in previous studies within this thesis. The experiment was conducted in January and February 2021 as a voluntary, open survey. After providing informed consent, participants gave demographic information, with participants exceeding diet and/or gender quotas being automatically excluded from participating further by Qualtrics. Remaining eligible participants were presented with a photograph of a tree kangaroo and asked if they recognised the animal, followed by a fossa and then tamandua if participants recognised any previous animal. Animal species names were not used during the study. The experiment either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected to the end of the

experiment. After initial recognition, participants indicated their perceived neutrality towards their target animal, the animal's warmth and competence, and behavioural intentions towards the animal (all scales and items within scales presented in randomised order).

Participants were subsequently randomly assigned into one of four conditions (familiar vs. unfamiliar vs. similar vs. dissimilar) and read the text within their condition as described in Sections 5.3.2.3.2. and 5.3.2.3.3. After reading their assigned text, participants rated the animal's perceived familiarity, perceived similarity to humans, and perceived shared emotional and cognitive traits with humans, and again rated the animal's warmth and competence and behavioural intentions towards the animal (all scales and items within scales presented in randomised order). Finally, participants were debriefed. Twenty-one participants reported technical difficulties, but responses were complete and maintained within analyses. ARU undergraduate psychology students (n=33) received 0.25 Sona credits as reimbursement. All other participants could enter a prize draw to win of two £50 Amazon gift vouchers if they wished.

5.3.3. Analyses

A 4(condition: familiar vs. unfamiliar vs. similar vs. dissimilar; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed MANCOVA was conducted on warmth, competence, active help, passive help, active harm, and passive harm. As all of the DVs were conceptually related (from the SCM/BIAS map and subscales of the same overall scales), alongside many DVs being statistically related through moderate correlations, $rs \ge .3$, MANCOVA was deemed appropriate for this analysis. Additionally, a one-way (familiarity: familiar vs. unfamiliar) between-subjects ANCOVA was run on measured familiarity, whilst a one-way (similarity: similar vs. dissimilar) between-subjects MANCOVA was conducted on perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans. For this MANCOVA, all of the DVs were

conceptually related (i.e., all measuring similarity to humans in different ways), alongside perceived shared emotional traits with humans and perceived shared cognitive traits with humans constituting subscales on the same overall scale. Although the DVs were not moderately statistically related in the similar condition, $rs \le .23$, there were moderate statistical relationships in the dissimilar condition between perceived similarity to humans and perceived shared emotional traits with humans, r = .36, and between perceived shared emotional traits with humans and perceived shared cognitive traits with humans, r = .63. Therefore, MANCOVA was deemed appropriate for this analysis to account for these underlying relationships. Finally, like Study 3, one-tailed one-sample t-tests were conducted on warmth, competence, and behavioural intentions.

Gender was again included as a covariate within the analyses. Again, although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and warmth/competence perceptions of animals and behavioural intentions towards them (the outcome variables) to be statistically controlled for. Neutrality was also included within analyses as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between the tradity and perceptions of/behavioural intentions towards animals (Tabachnick & Fidell, 2014).

Descriptive statistics for significant discussed main effects or for *post hoc* pairwise comparisons for significant interaction effects are reported in Tables 16-20.

5.3.3.1. Statistical Assumptions to Test H1-H4: Exploring Effects of Familiarity/Similarity Condition on SCM/BIAS Map Variables (4x2 MANCOVA)

DVs frequently failed normality tests, ps < .05, but all DVs had acceptable skewness, except for active harm across conditions. Due to excessive skewness, multiple extreme univariate outliers, and floor effects in active harm, I also ran a MANCOVA excluding active harm as a DV, which did not change multivariate conclusions compared to when this variable was included as a DV. I therefore report the MANCOVA including active harm as a DV.

All DVs except active harm had univariate outliers which did not cause excessive skewness. There were multivariate outliers for six DVs, critical values \geq 22.46, p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in active harm. Running the MANCOVA including and excluding univariate and multivariate outliers without active harm as a DV indicated multivariate conclusions do not change. I therefore report the MANCOVA including univariate and multivariate outliers.

There was no multicollinearity, $rs \le .83$. Linear relationships between the DVs across conditions were mostly present. Running follow-up univariate ANCOVAs including and excluding univariate outliers for warmth, competence, and passive help did not change conclusions. Excluding outliers changed one main conclusion for active help (Footnote 23), and one minor finding for passive harm (Footnote 24). I report all analyses including outliers to reflect the unaltered dataset. Excessive skewness, multiple univariate outliers, and floor effects in active harm meant assumptions for the ANCOVA on this DV were violated.

However, I proceeded with this ANCOVA due to robustness of ANCOVA to non-normality, and lack of non-parametric alternatives. Results from this ANCOVA should be interpreted with caution.

Homogeneity of variances was present for warmth, pre-manipulation competence, pre-manipulation active help, passive help, passive harm, and pre-manipulation active harm, ps > .05. Homogeneity of variances was absent for post-manipulation competence, postmanipulation active help, and post-manipulation active harm, ps < .05. Homogeneity of covariances was not present, p < .001.

5.3.3.2. Statistical Assumptions for Familiarity Manipulation Check: Exploring Effect of Familiarity Manipulation on Measured Familiarity (One-Way ANCOVA)

Measured familiarity failed normality tests across conditions, ps < .05, but skewness was acceptable. There were no outliers in measured familiarity. Homogeneity of variances was present, p > .05.

5.3.3.3. Statistical Assumptions for Similarity Manipulation Checks: Exploring Effects of Similarity Manipulation on Measured Similarity (One-Way MANCOVA)

DVs sometimes failed normality tests, ps < .05, but all DVs had acceptable skewness. Perceived shared emotional and cognitive traits had no univariate outliers, whilst perceived similarity had one univariate outlier in the dissimilar condition only. There were no multivariate outliers for three DVs, critical values < 16.27, p < .001. Running the MANCOVA including and excluding univariate outliers did not change multivariate conclusions. I therefore report the MANCOVA including univariate outliers. There was no multicollinearity, $rs \le .63$. Whilst some linear relationships were present between some DVs in the dissimilar condition, not all DVs had linear relationships, especially in the similar condition. Running a follow-up univariate ANCOVA including and excluding the univariate outlier for perceived similarity to humans did not change conclusions. Homogeneity of variances was present for all DVs, ps > .05. Homogeneity of covariances was present, p > .001.

5.3.3.4. Familiarity Manipulation Check: Exploring Effect of Familiarity Manipulation on Measured Familiarity

A one-way(condition: familiar vs. unfamiliar) between-subjects ANCOVA, including gender and neutrality as covariates, was conducted on familiarity to test effectiveness of the familiarity text in manipulating measured familiarity. An ANCOVA was conducted for the familiarity manipulation check instead of an independent-samples *t*-test to enable inclusion of gender and neutrality as covariates. To be deemed effective, the familiarity manipulation would be expected to increase measured familiarity when asked to imagine interacting with the target animal (familiar condition), instead of interacting with a football (unfamiliar condition; see Table A6 in Appendix 4 for full inferential statistics). The ANCOVA did not find a significant effect of the familiarity manipulation on measured familiarity, F(1, 93) = 1.14, p = .29, $\eta_p^2 = .01$. Thus, the familiarity manipulation was not effective at manipulating its intended variable of measured familiarity. Beyond the manipulation check, there was also a significant main effect of neutrality on measured familiarity, F(1, 93) = 6.63, p = .01, $\eta_p^2 = .07$ (medium-sized), whereby greater neutrality was linked to significantly greater familiarity, B = .36, SE = .14.

5.3.3.5. Similarity Manipulation Checks: Exploring Effects of Similarity Manipulation on Measured Similarity

A one-way(condition: similar vs. dissimilar) between-subjects MANCOVA, including gender and neutrality as covariates, was conducted on three similarity measures (perceived similarity to humans, perceived shared emotional capacities with humans, and perceived shared cognitive capacities with humans), to test effectiveness of the similarity text in manipulating measured similarity. To be deemed effective, the similarity manipulation would be expected to increase all three measures of similarity when the target animal is described as similar to humans (similar condition), instead of dissimilar (dissimilar condition; see Table A7 in Appendix 4 for full inferential statistics).

5.3.3.5.1. Multivariate Effects of Similarity Manipulation on the Combined DVs.

The similarity manipulation had a significant effect on the combined DVs, F(3, 93) = 9.17, p < .001, $\eta_p^2 = .23$ (large-sized). In addition, neutrality had a significant effect on the combined DVs, F(3, 93) = 3.45, p = .02, $\eta_p^2 = .1$ (medium-sized). There was no significant effect of gender on the combined DVs, F(3, 93) = 1.18, p = .32, $\eta_p^2 = .04$. I followed up significant multivariate effects with univariate ANCOVAs on the DVs below.

5.3.3.5.2. Univariate Effects of Similarity Manipulation on Perceived Similarity to Humans.

There was a significant effect of the similarity manipulation on perceived similarity to humans, F(1, 95) = 19.46, p < .001, $\eta_p^2 = .17$ (large-sized), in the expected direction. That is, participants in the similar condition perceived the animal to be more similar to humans than participants in the dissimilar condition. Thus, the similarity text was effective at manipulating similarity when measured through perceived similarity to humans. Beyond the manipulation check, there was also a significant main effect of neutrality on perceived similarity to

humans, F(1, 95) = 7.95, p = .01, $\eta_p^2 = .08$ (medium-sized), whereby greater neutrality was linked to significantly greater perceived similarity to humans, B = .36, SE = .13.

5.3.3.5.3. Univariate Effects of Similarity Manipulation on Perceived Shared Emotional Traits with Humans.

There was a significant effect of the similarity manipulation on perceived shared emotional traits, F(1, 95) = 11.2, p = .001, $\eta_p^2 = .11$ (medium-sized), in the expected direction. Specifically, participants in the similar condition perceived the animals to share more emotional traits with humans than participants in the dissimilar condition. Thus, the similarity text was effective at manipulating similarity when measured through perceived shared emotional traits.

5.3.3.5.4. Univariate Effects of Similarity Manipulation on Perceived Shared Cognitive Traits with Humans.

There was a significant effect of the similarity manipulation on perceived shared cognitive traits, F(1, 95) = 13.66, p < .001, $\eta_p^2 = .13$ (medium-to-large-sized), in the expected direction. Specifically, participants in the similar condition perceived the animals to share more cognitive traits with humans than participants in the dissimilar condition. Thus, the similarity text was effective at manipulating similarity when measured through perceived shared cognitive traits. However, this finding should be interpreted with caution due to inadequate scale reliability.

Table 16.

Descriptive Statistics for Effect of Similarity Manipulation on Measured Similarity Variables.

Manipulated	Similarity	Similarity	Emotional	Emotional	Cognitive	Cognitive
Similarity	to	to	Traits	Traits SD	Traits	Traits SD
Condition	Humans	Humans	Mean		Mean	
	Mean	SD				
Similar	51.53	26.37	26.51	5.67	16.96	4.64
Dissimilar	30.35	25.44	22.04	7.1	13.56	4.7

5.3.3.6. Testing H1-H4: Exploring Effects of Familiarity/Similarity Condition on SCM/BIAS Map Variables.

A 4(condition: familiar vs. unfamiliar vs. similar vs. dissimilar) x 2(time: pre- vs. post-manipulation) mixed MANCOVA, including gender and neutrality as covariates, was run on warmth, competence, active harm, active help, passive harm, and passive help to test **H1-H4** (see Table A8 in Appendix 4 for full inferential statistics).

5.3.3.6.1. Multivariate Effects of Familiarity/Similarity Condition on the Combined DVs.

There was a significant interaction effect between time and condition on the combined DVs, F(18, 570) = 3.16, p < .001, $\eta_p^2 = .09$ (medium-sized). There was also a significant main effect of condition on the combined DVs, F(18, 570) = 1.71, p = .03, $\eta_p^2 = .05$ (small-to-medium-sized). All other multivariate interactions and main effects were non-significant, ps > .05. I followed up significant multivariate effects with univariate ANCOVAs on the DVs below.

5.3.3.6.2. Univariate Effects of Familiarity/Similarity Condition on Warmth.

There was a significant interaction between condition and time on warmth, F(3, 193)= 4.04, p = .01, $\eta_p^2 = .06$ (medium-sized). Agreeing with H2a: *Post hoc* pairwise comparisons revealed the similar condition significantly increased warmth from premanipulation to post-manipulation, p = .01, d = .3 (small-sized). Additionally, the dissimilar condition significantly decreased warmth from pre-manipulation to post-manipulation, p =.03, d = .31 (small-sized). Post-manipulation only, the similar condition caused more warmth than the dissimilar condition, p = .002, d = .73 (medium-sized), an effect not present premanipulation, p = 1. Not supporting H1a: *Post hoc* pairwise comparisons revealed no significant post-manipulation difference in warmth between the familiar and unfamiliar conditions, p = 1, nor any significant differences from pre- to post-manipulation after exposure to the familiar, p = .65, or unfamiliar, p = .22, conditions.

Table 17.

on Warmth.				
Condition	Pre-Manipulation	Pre-	Post-Manipulation	Post-
	Mean	Manipulation SE	Mean	Manipulation SE
Similar	3.36	.5	4.37	.51
Dissimilar	2.67	.48	1.81	.49
Familiar	3.74	.49	3.5	.5
Unfamiliar	3.4	.49	2.89	.5

Descriptive Statistics for Interaction Effect of Familiarity/Similarity Condition and Time on Warmth.

5.3.3.6.3. Univariate Effects of Familiarity/Similarity Condition on Competence.

There was a significant interaction between condition and time on competence, $F(3, 193) = 15.09, p < .001, \eta_p^2 = .19$ (large-sized). Agreeing with H2b: *Post hoc* pairwise comparisons revealed the similar condition significantly increased competence from premanipulation to post-manipulation, p < .001, d = .79 (medium-sized). Additionally, the dissimilar condition significantly decreased competence from premanipulation to post-manipulation, p = .001, d = .47 (small-sized). Post-manipulation only, participants in the similar condition perceived the animal to be more competent than participants in the dissimilar condition, p < .001, d = .98 (large-sized), an effect not present pre-manipulation, p = 1. Participants in the similar condition also viewed animals as being more competent postmanipulation than participants in the familiar, p = .02, d = .6 (medium-sized), and unfamiliar, p = .03, d = .58 (medium-sized), conditions. Not supporting H1a: *Post hoc* pairwise comparisons revealed no significant post-manipulation differences in competence between the familiar and unfamiliar conditions, p = 1, nor any significant differences in competence from pre- to post-manipulation after exposure to the familiar, p = .36, or unfamiliar, p = .36, conditions.

on competence.					
Condition	Pre-	Pre-	Post-	Post-	
	Manipulation	Manipulation	Manipulation	Manipulation	
	Mean	SE	Mean	SE	
Similar	2.67	.45	5.36	.55	
Dissimilar	3.1	.44	1.64	.53	
Familiar	2.68	.45	3.1	.54	
Unfamiliar	3.57	.45	3.15	.54	

 Table 18.

 Descriptive Statistics for Interaction Effect of Familiarity/Similarity Condition and Time on Competence

5.3.3.6.4. Univariate Effects of Familiarity/Similarity Condition on Active Harm.

Not supporting H3a and H4a: There was no significant interaction between condition and time on active harm, F(3, 193) = .11, p = .96, $\eta_p^2 = .002$.

5.3.3.6.5. Univariate Effects of Familiarity/Similarity Condition on Active Help.²³

There was a significant interaction between condition and time on active help, F(3, 193) = 5.15, p = .002, $\eta_p^2 = .07$ (medium-sized). **Partially consistent with H4b:** *Post hoc* pairwise comparisons found that participants in the similar condition expressed significantly greater active help post-manipulation compared to pre-manipulation, p < .001, d = .43 (small-sized). **Partially not supporting H4b:** Participants in the similar condition did not express significantly greater post-manipulation active help than participants in the dissimilar condition, p = .14. Participants in the similar condition did, however, show significantly greater post-manipulation active help than participants in the unfamiliar condition, p = .01, d = .67 (medium-sized), but not more than participants in the familiar condition, p = .23. Unlike above effects on warmth and competence, participants in the dissimilar condition did not significantly differ in active help from pre-manipulation to post-manipulation, p = .4. **Not supporting H3b:** *Post hoc* pairwise comparisons found no significant post-manipulation

²³When excluding univariate outliers, *post hoc* pairwise comparisons for the interaction between condition and time interaction on active help revealed a significant post-manipulation difference in active help between the similar (M=8.19, SE=.72) and dissimilar conditions (M=5.57, SE=.66), p = .049, d = .54 (medium-sized).

difference in active help between the familiar and unfamiliar conditions, p = .23, nor any significant differences from pre- to post-manipulation after exposure to the familiar, p = .55, or unfamiliar, p = .25, conditions.

Table 19.

Condition Pre-Pre-Post-Post-Manipulation Manipulation Manipulation Manipulation Mean SE Mean SE Similar .69 5.86 .65 7.78 Dissimilar 6 .63 .67 5.58 Familiar 6.9 .64 6.59 .68 Unfamiliar .65 4.55 .69 5.15

Descriptive Statistics for Interaction Effect of Familiarity/Similarity Condition and Time on Active Help.

5.3.3.6.6. Univariate Effects of Familiarity/Similarity Condition on Passive Harm.²⁴

Not supporting H3c and H4c: There was no significant interaction between

condition and time on passive harm, F(3, 193) = 1.25, p = .29, $\eta_p^2 = .02$.

5.3.3.6.7. Univariate Effects of Familiarity/Similarity Condition on Passive Help.

There was a significant interaction between condition and time on passive help, F(3, 193) = 4.38, p = .01, $\eta_p^2 = .06$ (medium-sized). **Partially consistent with H4d:** *Post hoc* pairwise comparisons found that participants in the similar condition showed significantly greater passive help post-manipulation compared to pre-manipulation, p = .01, d = .33 (small-sized). **Partially not supporting H4d:** Participants in the similar condition did not demonstrate significantly greater post-manipulation passive help than participants in the dissimilar condition, p = 1. Unlike above effects on warmth and competence, participants in the dissimilar condition did not significantly differ in passive help from pre-manipulation to post-manipulation, p = .37. **Not supporting H3d:** *Post hoc* pairwise comparisons found that,

²⁴When excluding univariate outliers, condition had a significant effect on passive harm, F(3, 181) = 5.08, p = .002, $\eta_p^2 = .08$ (medium-sized). However, this effect did not interact with time, F(3, 181) = 1.42, p = .24, $\eta_p^2 = .02$, indicating this finding may be due to randomly occurring differences between groups only.

for participants within the familiar condition, passive help significantly decreased from premanipulation to post-manipulation, p = .02, d = .33 (small-sized). That is, participants were less likely to passively help the animal after being exposed to the familiarity manipulation, which was the opposite direction to predicted. There was no significant difference in passive help from pre- to post-manipulation for participants within the unfamiliar condition, p = .84, nor any significant post-manipulation difference in passive help between participants in the familiar and unfamiliar conditions, p = 1.

Table 20.

Familiar

Unfamiliar

on Passive Help. Condition Pre-Pre-Post-Post-Manipulation Manipulation Manipulation Manipulation Mean SE Mean SE Similar 2.69 .4 3.54 .42 Dissimilar .39 .4 3.53 3.26

.4

.4

3.15

3.08

.41

.42

Descriptive Statistics for Interaction Effect of Familiarity/Similarity Condition and Time

5.3.3.7. Testing if Warmth/Competence Perceptions and Behavioural Intentions

Towards the Animal Differ from Zero.

3.9

3.14

Twelve one-tailed one-sample *t*-tests were conducted on warmth, competence, active help, active harm, passive help, and passive harm at pre- and post-manipulation to test if perceptions of, and behavioural intentions towards, animals significantly differ from zero. These one-sample *t*-tests revealed that all perceptions and behavioural intentions significantly differed from zero, ps < .001, whereby the animal was deemed warmer and more competent than zero across timepoints. Additionally, participants were more willing to actively and passively help, and less willing to actively and passively harm, the animal compared to zero. Like Study 3, these findings indicate a positivity bias towards the unknown animal.

5.3.3.8. Testing H5-H8: Exploring Mediational Relationships Between Familiarity/Similarity and Behavioural Intentions Through Warmth and Competence.

Per the BIAS map, I predicted post-manipulation warmth would mediate the relationship between familiarity and post-manipulation active harm (**a**) and post-manipulation active help (**b**; **H5**). I also hypothesised post-manipulation warmth would mediate the relationship between similarity and post-manipulation active harm (**a**) and post-manipulation active help (**b**; **H6**). I predicted post-manipulation competence would mediate the relationship between familiarity and post-manipulation passive harm (**a**) and post-manipulation passive help (**b**; **H7**). Finally, I hypothesised post-manipulation competence would mediate the relationship between similarity and post-manipulation passive harm (**a**) and post-manipulation passive help (**b**; **H7**). Finally, I hypothesised post-manipulation passive harm (**a**) and post-manipulation passive help (**b**; **H7**).

To test **H5-H8**, I dummy coded familiarity and similarity into two separate dummy variables, with familiarity coded as unfamiliar (zero) and familiar (one), and similarity coded as dissimilar (zero) and similar (one). Eight individual boot-strapped mediation analyses (10,000 bootstrap samples) were run using Model 4 in PROCESS through SPSS (Hayes, 2022) to assess mediational relationships for the four behavioural intentions with two dummy variables (familiarity and similarity). To control for Type I error from multiple mediation testing, I utilised an adjusted confidence interval of 99%. Significant mediations are interpreted through confidence intervals which do not contain zero per Hayes (2018).

As post-manipulation warmth and post-manipulation competence were significantly correlated with each other within the current study as revealed through a Pearson correlation, r = .69, p < .001, and warmth and competence are conceptually related, warmth and competence were included as parallel mediators within the below mediation analyses. Inclusion of warmth and competence as parallel mediators allows for testing of any

mediational roles of warmth for active behavioural intentions *whilst* controlling for competence, and any mediational roles of competence for passive behavioural intentions *whilst* controlling for warmth (Hayes, 2018).

5.3.3.8.1. H5a: Mediation of Familiarity on Active Harm through Warmth.

Not supporting H5a: There was no significant indirect effect of familiarity on postmanipulation active harm through post-manipulation warmth when controlling for postmanipulation competence, b = -.2, SE = .33, 99% CI [-1.34, .65]. That is, post-manipulation warmth did not significantly mediate the relationship between familiarity and postmanipulation active harm. Inspection of path *b* indicated post-manipulation warmth did not significantly predict post-manipulation active harm, b = -.41, SE = .21, p = .053, 99% CI [-.95, .14].

5.3.3.8.2. H5b: Mediation of Familiarity on Active Help through Warmth.

Not supporting H5b: There was no significant indirect effect of familiarity on postmanipulation active help through post-manipulation warmth when controlling for postmanipulation competence, b = .33, SE = .47, 99% CI [-1.07, 1.75]. That is, post-manipulation warmth did not significantly mediate the relationship between familiarity and postmanipulation active help. Inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater post-manipulation active help, b = .66, SE = .17, p < .001, 99% CI [.22, 1.11].

5.3.3.8.3. H6a: Mediation of Similarity on Active Harm through Warmth.

Not supporting H6a: There was no significant indirect effect of similarity on postmanipulation active harm through post-manipulation warmth when controlling for postmanipulation competence, b = -.27, SE = .36, 99% CI [-1.35, .72]. Thus, post-manipulation warmth did not significantly mediate the relationship between similarity and postmanipulation active harm. Inspection of path *b* indicated post-manipulation warmth did not significantly predict post-manipulation active harm, b = -.11, SE = .15, p = .45, 99% CI [-.5, .28].

5.3.3.8.4. H6b: Mediation of Similarity on Active Help through Warmth.

Agreeing with H6b: There was a significant indirect effect of similarity on postmanipulation active help through post-manipulation warmth when controlling for postmanipulation competence. Thus, post-manipulation warmth significantly mediated the relationship between similarity and post-manipulation active help (Figure 8). This model explained 32.9% of variance in post-manipulation active help ($R^2 = .329$). Additionally, inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater post-manipulation active help (Figure 8).



Figure 8. Mediational model of similarity on post-manipulation active help through postmanipulation warmth when controlling for post-manipulation competence. *Note:* Standard error is reported for indirect effect only. **p < .01 ***p < .001

5.3.3.8.5. H7a: Mediation of Familiarity on Passive Harm through Competence.

Not supporting H7a: There was no significant indirect effect of familiarity on postmanipulation passive harm via post-manipulation competence when controlling for postmanipulation warmth, b = .01, SE = .07, 99% CI [-.24, .28]. Therefore, post-manipulation competence did not significantly mediate the relationship between familiarity and postmanipulation passive harm. Inspection of path *b* indicated post-manipulation competence did not significantly predict post-manipulation passive harm, b = -.06, SE = .1, p = .54, 99% CI [-.31, .19].

5.3.3.8.6. H7b: Mediation of Familiarity on Passive Help through Competence.

Not supporting H7b: There was no significant indirect effect of familiarity on postmanipulation passive help via post-manipulation competence when controlling for postmanipulation warmth, b = -.06, SE = .24, 99% CI [-.81, .6]. Therefore, post-manipulation competence did not significantly mediate the relationship between familiarity and postmanipulation passive help. Inspection of path *b* indicated greater post-manipulation competence significantly predicted greater post-manipulation passive help, b = .33, SE = .1, *p* = .001, 99% CI [.07, .58].

5.3.3.8.7. H8a: Mediation of Similarity on Passive Harm through Competence.

Not supporting H8a: There was no significant indirect effect of similarity on postmanipulation passive harm via post-manipulation competence when controlling for postmanipulation warmth, b = -.08, SE = .22, 99% CI [-.77 .51]. Therefore, post-manipulation competence did not significantly mediate the relationship between similarity and postmanipulation passive harm. Inspection of path *b* indicated post-manipulation competence did not significantly predict post-manipulation passive harm, b = -.02, SE = .07, p = .74, 99% CI [-.2, .16].

5.3.3.8.8. H8b: Mediation of Similarity on Passive Help through Competence.

Not supporting H8b: There was no significant indirect effect of similarity on postmanipulation passive help via post-manipulation competence when controlling for post-
manipulation warmth, b = .31, SE = .32, 99% CI [-.53 1.24]. Thus, post-manipulation competence did not significantly mediate the relationship between similarity and postmanipulation passive help. Inspection of path *b* indicated post-manipulation competence did not significantly predict post-manipulation passive help, b = .09, SE = .09, p = .32, 99% CI [-.14, .31].

5.3.4. Discussion

5.3.4.1. Discussion of Findings.

5.3.4.1.1. H2 and H4: Effects of Similarity on SCM/BIAS Map Variables.

As expected, and aligning with previous literature (Batt, 2009; Westbury & Neumann, 2008), similarity was superior to the dissimilarity condition in causing greater postmanipulation warmth (medium-sized; H2a) and post-manipulation competence (large-sized; H2b), and similarity increased warmth (small-sized; H2a), competence (medium-sized; H2b), active help (small-sized; H4b), and passive help (small-sized; H4d) from pre- to postmanipulation. That is, animals described as similar (vs. dissimilar) were deemed warmer and more competent post-manipulation. Furthermore, animals merely being described as similar increased their warmth and competence, and active help and passive help towards them, from pre- to post-manipulation. Additionally, dissimilarity caused participants to view animals as lower in warmth and competence from pre- to post-manipulation (small-sized). Thus, within this study, similarity has two-pronged effects on the warmth and competence of unknown animals: A positive effect of similarity and a negative effect of dissimilarity. Whilst previous research has focussed on the benefits conferred by similarity (Batt, 2009; Westbury & Neumann, 2008), negative effects of dissimilarity on perceptions of animals have not yet been explored. Thus, the current study uniquely highlights this negative role of dissimilarity on perceptions of unknown animals (warmth, competence). Some effects are small-sized only, so may have lower practical significance.

Unlike for warmth and competence, similarity was not superior to the dissimilarity condition for post-manipulation active and passive help, not supporting H4b and H4d. Thus, although similarity increased active and passive help towards animals from pre- to postmanipulation, the similar condition did not increase post-manipulation active and passive help over the dissimilar condition. For active help, this finding may be due to univariate outliers, as excluding univariate outliers led to the similar condition causing greater postmanipulation active help than the dissimilar condition (medium-sized; Footnote 23). Alternatively, the effect of similarity on help intentions from pre- to post-manipulation may not have been strong enough alone to differentiate post-manipulation help intentions in the similar and dissimilar conditions. Indeed, the post-manipulation differences in warmth and competence between the similar (vs. dissimilar) conditions may have arisen from the combined positive effect of similarity and negative effect of dissimilarity on warmth and competence from pre- to post-manipulation. There was no such negative effect of dissimilarity on active and passive help from pre- to post-manipulation, which may thus explain the lack of post-manipulation differences between the similar (vs. dissimilar) condition on help intentions found here. Passive help findings should be interpreted with caution due to inadequate scale reliability.

The lack of effects of similarity on active and passive harm within this study indicate effects of similarity on warmth, competence, and help intentions may not extend to harm intentions towards unknown animals, not supporting **H4a** and **H4c** (although the null findings for active and passive harm should be interpreted with caution due to inadequate scale reliability [passive harm] or floor effects [active harm]). Whilst the finding for active harm may be due to floor effects (i.e., regardless of similarity, participants were averse to actively harming animals), similarity may also simply not inform harm intentions towards unknown animals. That is, harm intentions towards unknown animals may be typically extremely low

and robust to change, whilst perceptions of animals (warmth, competence) and help intentions towards them may be more pliable to change following similarity information. Thus, these null findings of similarity on harm intentions may indicate separability between SCM perceptions of unknown animals (e.g., warmth, competence) and BIAS map behavioural intentions towards them (e.g., active harm, passive harm).

5.3.4.1.2. H1 and H3: Effects of Familiarity on SCM/BIAS Map Variables.

Hypotheses for familiarity (H1a-H1b, H3a-H3d), which predicted positive effects of familiarity, may not be supported by current findings. Specifically, results were mostly null effects, alongside an unexpected effect of familiarity on passive help (small-sized) in the opposite direction to predicted. That is, familiarity unexpectedly reduced passive help from pre- to post-manipulation, contrary to H3d, whereby participants were less likely to passively help the animal after being exposed to the familiarity manipulation. However, this effect was only small-sized, so may have lower practical significance, and passive help findings should be interpreted with caution due to inadequate scale reliability. The null or unexpected findings do not support previous literature (Auger & Amiot, 2019b), which suggest familiarity through imagined contact would be beneficial in improving humans' relationship to animals, and therefore encourage positive behavioural intentions towards them.

The null findings observed in this study may be due to ineffectiveness of the familiarity manipulation (i.e., the text failed to manipulate familiarity), despite the new amendments implemented within this study following Pilot Study 1b. For instance, Auger and Amiot's (2019b) imagined contact scenario used positive wording (e.g., '*friendly contact with a dog*'). Conversely, the current study only used neutral wording ('*please now imagine meeting and interacting with the animal in the photograph*') to avoid confounding effects of positivity on perceptions of animals. However, this lack of positive wording may have led to

the text not successfully manipulating familiarity, whereby imagined contact may need to be positive to improve perceptions of animals (Auger & Amiot, 2019b).

The lack of positive wording may also explain the unexpected small-sized effect of the familiarity manipulation on passive help. That is, because imagined contact was not explicitly worded as friendly, participants may have viewed the contact as negative, especially as the animal is wild and unfamiliar, which may have in turn reduced passive help. Indeed, inspection of the texts provided within the description task accompanying the imagined contact scenario indicated many participants viewed the interaction as dangerous for themselves (i.e., the animal may be dangerous or carry disease) or the animal (i.e., the animal is timid and needs to be left alone). As perceived harmfulness of animals causes people to view them more negatively (Piazza et al., 2014), the imagined contact may have inadvertently caused participants to reflect on the animals' possible dangerous nature, and hence reduced passive help. Recognising the above limitations of the familiarity manipulation, future research should therefore implement positive wording and check effectiveness of this wording by asking participants to rate whether the imagined contact was positive, negative, or neutral, and emphasise the animal is not dangerous.

The familiarity manipulation may also have been ineffective in manipulating familiarity due to the manipulation check employing an explicit measure (within the participant's control and awareness, e.g., self-report), instead of an implicit measure (outside of the participant's control and awareness, e.g., reaction time). Recent research (Cerrato & Forestell, 2022) indicates imagined contact with a calf (vs. kitten or child) reduces implicit desire for beef and other types of meat (as measured via a forced-choice food task), but not explicit desire (as measured via self-reported willingness to consume and perceived pleasantness of the food). Thus, familiarity through imagined contact may not inform explicit judgements of warmth, competence, and most behavioural intentions, but may inform

implicit judgements. Implicit measures like the Implicit Association Test are effective when applied to warmth and competence (Carlsson & Björklund, 2010), and could therefore be implemented within future research to contrast effects of imagined contact with animals on explicit and implicit SCM/BIAS map measures.

Alternatively, familiarity could be manipulated via texts which do not implement imagined contact. For example, familiarity may be increased through the mere exposure effect (Zajonc, 1968), whereby mere exposure to a stimulus increases liking for that stimulus. As the mere exposure effect increases liking for multiple types of stimuli (Bornstein & D'Agostino, 1992), repeated exposure to an animal would theoretically increase liking, reflected in improved perceptions of and behavioural intentions towards the animal. I therefore test this alternative familiarity manipulation using the mere exposure effect in Pilot Studies 2b, 3b and 4b (Sections 6.2 to 6.4).

5.3.4.1.3. H5-H8: Mediational Relationships Between Familiarity/Similarity and Behavioural Intentions Through Warmth/Competence.

Mediation analyses supported only one mediational hypothesis (**H6b**). That is, as expected and aligning with previous literature on the BIAS map with animals (Sevillano & Fiske, 2016b), post-manipulation warmth mediated the relationship between similarity and post-manipulation active help (**H6b**), whereby similarity caused greater post-manipulation warmth, which in turn predicted greater post-manipulation active help. All other hypotheses (**H5, H6a, H7, H8**) may not be supported by the mediation analyses due to null indirect effects. For instance, warmth did not mediate the relationship between familiarity and postmanipulation active help and post-manipulation active harm (**H5**). There was also no significant mediation between similarity and post-manipulation active harm through postmanipulation warmth (**H6a**). Additionally, post-manipulation competence did not mediate the relationship between familiarity and post-manipulation passive help and post-manipulation

passive harm (H7). Finally, post-manipulation competence did not mediate the relationship between similarity and post-manipulation passive help and post-manipulation passive harm (H8). These findings therefore indicate a) warmth and competence do not mediate relationships between familiarity and behavioural intentions, b) warmth does not mediate relationships between similarity and harm intentions, and c) competence does not mediate relationships between similarity and behavioural intentions. These null mediations may be arising from lack of effects of familiarity on most DVs (due to the failed familiarity manipulation) and lack of effects of similarity on harm intentions found within the main analyses as discussed in Sections 5.3.4.1.1 and 5.3.4.1.2.

5.3.4.1.4. Extension of Warmth/Competence to Behavioural Intentions.

Despite the lack of indirect mediation effects discussed above, significant *b* pathways revealed that greater warmth consistently predicted greater active help within the familiarity and similarity mediation analyses, although warmth did not predict active harm in either the similarity or familiarity mediation analysis. Additionally, whilst greater competence did predict greater passive help within the familiarity mediation analysis, competence did not predict passive help in the similarity mediation analysis, and competence did not predict passive harm in either the similarity or familiarity or familiarity mediation analysis. Therefore, the BIAS map as applied to animals (Sevillano & Fiske, 2016b) is only partially supported, whereby warmth consistently predicts greater active help.

5.3.4.1.5. Warmth/Competence Perceptions and Behavioural Intentions Towards the Animal Differ from Zero.

Like Study 3, the unknown animal here was deemed significantly warmer and more competent than zero. Additionally, participants were significantly more willing to actively or passively help, and significantly less willing to actively or passively harm, the animal compared to zero. Thus, the current study combines with Study 3 to evidence a positivity bias

in perceptions of and behavioural intentions towards animals, including an animal unknown to participants.

5.3.4.2. Limitations of the Current Study and Directions for Future Research.

The current study has some limitations, which are addressed in subsequent pilot studies (Pilot Studies 2b, 3b and 4b; Sections 6.2. to 6.4.) and/or Study 5: a) lack of effectiveness of the familiarity manipulation, and b) lack of applicability to other animals.

Firstly, as the familiarity manipulation failed, causal effects of familiarity on perceptions of and behavioural intentions towards animals cannot be fully ascertained. Thus, any role of familiarity in causing pet speciesism is difficult to determine, as null effects may be due to familiarity genuinely not causing pet speciesism, or due to familiarity not being effectively manipulated. Due to this failed familiarity manipulation, I instead test an alternative familiarity intervention within Pilot Studies 2b, 3b and 4b, which implements repeated photographs of the target animal, in line with the mere exposure effect (Zajonc, 1968).

Secondly, the current study (like Studies 2 and 6) applies only to perceptions of (predominantly) tree kangaroos, alongside fossas and tamanduas. As perceptions of different species vary widely (López-Cepero et al., 2021, Sevillano & Fiske, 2016b), current findings are encouraging for possible interventions (e.g., a robust effect on active help), but may apply to perceptions of (mostly) tree kangaroos only. Thus, effects of similarity found here do not necessarily apply to perceptions of different species, including dogs and pigs. I therefore test causal effects of similarity on perceptions of dogs and pigs in Study 5, to determine if similarity causes pet speciesism.

5.3.4.3. Conclusion.

The current study uniquely demonstrates the causal role of similarity in informing perceptions of and behavioural intentions towards animals, whereby similarity of unknown animals increased their warmth and competence, and enhanced help intentions towards them (although the effect on passive help may not be robust due to inadequate scale reliability). As similarity had robust effects on warmth, competence and active help, similarity may play a role in causing pet speciesism (tested in Study 5). The study also deviates from some previous literature by demonstrating the limitations of imagined contact with animals. Imagined contact may have failed to elicit greater familiarity due to lack of positive wording for the imagined contact, presumed dangerousness of the animal, or due to use of explicit instead of implicit measures to check effectiveness of the text in manipulating familiarity. This failed imagined contact manipulation may then explain why familiarity did not inform perceptions of and behavioural intentions towards animals, or alternatively familiarity may simply not inform perceptions of and behavioural intentions towards animals. The current study therefore has inconclusive findings on whether familiarity plays a causal role in pet speciesism.

Overall, the current study only supports limited extension of the BIAS map from SCM (e.g., warmth informing active help, but not warmth informing active harm or competence informing passive behavioural intentions). This study has limitations including its measurement of familiarity and applicability to unknown animals only. Due to these limitations, I conduct subsequent pilot studies (Pilot Studies 2b, 3b and 4b; Sections 6.2. to 6.4.) to test an alternative manipulation of familiarity via the mere exposure effect, and I investigate causal effects of similarity on perceptions of and behavioural intentions towards dogs and pigs specifically (Study 5).

5.4. Overall Discussion

Through a pilot study, a similarity manipulation was developed and piloted which successfully manipulates similarity, whilst a familiarity manipulation was found to be unsuccessful in manipulating familiarity. As such, the similarity manipulation was implemented unchanged within Study 4, whilst the familiarity manipulation was implemented with changes within Study 4, to test if similarity and familiarity are causes of pet speciesism. Overall, Study 4 found that similarity causally informs perceptions of and help intentions towards animals, whilst familiarity failed to have any effects on perceptions of and behavioural intentions towards animals, perhaps due to the manipulation being unsuccessful in manipulating familiarity. These findings are extended within the next chapter through three further pilot studies to test an alternative familiarity manipulation, and through an additional main study testing causal effects of similarity on perceptions of and behavioural intentions towards dogs and pigs specifically (Study 5).

Chapter 6. Pilot Studies for Familiarity and Study 5 Exploring Similarity

6.1. Introduction

The current chapter first presents three pilot studies (Pilot Study 2b, 3b, 4b) to develop and pilot an alternative familiarity manipulation following the unsuccessful familiarity manipulation in Pilot Study 1b and Study 4. This chapter then applies similarity to dogs and pigs specifically within Study 5, to see if similarity is a cause of pet speciesism. These studies are now reported in turn below.

6.2. Pilot Study 2b to Test Amended Familiarity Manipulation

6.2.1. Introduction

Due to the previous familiarity manipulation failing to effectively manipulate familiarity, the current pilot study aims to test and pilot an alternative familiarity manipulation developed by the current researcher. Specifically, familiarity is now manipulated through the mere exposure effect (Zajonc, 1968) rather than imagined contact, whereby participants view nine photographs of the target animal (familiar condition) vs. nine photographs of trees which act as the control (unfamiliar condition). As with Studies 2-3, an unknown animal (tree kangaroo) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of familiarity. To check effectiveness of the familiarity manipulation (manipulation check), the item '*How familiar do you perceive the species in the previous photograph to be to you*?' is included to measure familiarity. In line with the mere exposure effect (Zajonc, 1968), I hypothesise that:

H1: Viewing photographs of an animal (vs. control) will significantly increase measured familiarity of the animal.

Note that this pilot study utilises the same participant sample to also pilot manipulations for behavioural and subjective self-relevance, but only information relevant to the familiarity manipulation is reported within the current chapter (see Section 4.3. for pilot study information about the behavioural and subjective self-relevance manipulations [Pilot Study 2a]).

6.2.2. Methods

6.2.2.1. Participants.

Sixty participants were recruited via social media and Sona as a volunteer sample. Nineteen participants were excluded for providing partial data, and 11 participants were excluded for not consuming meat, meaning that all remaining participants were meat consumers only. As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 6.2.3. for further discussion). Exclusions left a total sample size of 30 participants (50% women, 50% men; M_{age} =25.7, SD_{age} =6.87, age range: 18-50), which exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Participants within each condition are shown in Table 21. Demographics of the sample are reported as follows in percentages. For nationality: 46.7% (United Kingdom/British), and 3.3% each (Australian, Canadian, Croatian, dual Dutch/Norwegian, Finnish, French, German, Greek, Israeli, Italian, Japanese, Lithuanian, Malaysian, Spanish, Taiwanese, United States). For ethnicity: 63.2% (White), 19.9% (Asian), 10% (Black), and 6.6% (mixed). For country of residence (M_{duration}=16.44, SD_{duration}=9.98, range: 9 months to 35 years): 63.3% (United Kingdom), 6.7% (Malaysia), and 3.3% each (Croatia, Finland, France, Germany, Greece, Hong Kong, Israel, no response, and Norway). For religion: 33.3% (atheism), 23.3% (Christianity), 13.3% each (agnosticism, would rather not say), 6.7% (no response), and 3.3% each (Hinduism, Islam, Judaism).

Participants within Each Condition for Pilot Study 20.	
Familiarity Condition	Number of Participants
Familiar	15
Unfamiliar	15

Table 21.Participants Within Each Condition for Pilot Study 2b.

6.2.2.2. Design.

The current pilot study follows a 2(familiarity: familiar vs. unfamiliar; betweensubjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed ANCOVA design on measured familiarity. Familiarity was included as a DV to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 6.2.3. for further discussion).

6.2.2.3. Materials.

6.2.2.3.1. Target Animal and Initial Animal Photograph.

The current pilot study employed the same initial photograph of the tree kangaroo as Studies 2-4, whereby this image was sourced from <u>https://www.shutterstock.com/image-</u> <u>photo/close-shot-cute-tree-kangaroo-168261035</u>. This photograph enables easier visualisation of the imagined scenario and reduces biased perceptions of the animal due to species name.

6.2.2.3.2. Familiarity Manipulation.

Familiarity was manipulated through the mere exposure effect (Zajonc, 1968), whereby participants were first informed that '*you will now see several photographs over the following pages. Each photograph will automatically move to the next photograph after 1 second*', and then either viewed nine photographs of the target animal (familiar condition) or nine photographs of trees (unfamiliar condition; control). Trees were chosen as a matched control as they relate to nature and are living beings, like the target animal, yet will not increase familiarity with the target animal. Nine photographs were utilised in line with mere exposure effect literature, whereby the mere exposure effect starts to reduce after 10 or more exposures (Bornstein, 1989). Each photograph was shown for one second each, in line with mere exposure effect research (Bornstein, 1989; Montoya et al., 2017). After one second of viewing each image, the page automatically advanced.

6.2.2.3.3. Measured Familiarity (Manipulation Check).

Familiarity was measured via the single item developed by the researcher: '*How* familiar do you perceive the species in the previous photograph to be to you?' on a zero ('not familiar at all') to 100 ('*extremely familiar*') visual analogue scale. A higher score indicates greater familiarity.

6.2.2.3.4. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be?*' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

6.2.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in July 2021 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further. Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial familiarity with the animal. Participants then completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the familiarity section is described (see Section 4.3. for description of the self-relevance section). For the familiarity section, participants were randomly allocated to see nine images either of the target animal or of trees. After viewing the images, participants then again rated their familiarity with the animal. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Two participants reported technical difficulties, but their responses were complete and maintained within analyses.

6.2.3. Analyses

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was conducted on measured familiarity, to test if the manipulation effectively manipulates familiarity. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured familiarity (the outcome variable) to be statistically controlled for. Neutrality was also included within the analysis as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings

(Keen & Tiemeier, 2022), and increasing power of the analysis by statistically controlling for any possible relationship between neutrality and familiarity (Tabachnick & Fidell, 2014).

6.2.3.1. Statistical Assumptions to Test H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity (2x2 ANCOVA).

Measured familiarity mostly failed normality tests, p < .05, but skewness was acceptable across conditions. There were outliers in measured familiarity in all conditions except the post-manipulation unfamiliar condition. Running the analysis including and excluding these outliers did not change conclusions. I report this ANCOVA including outliers to reflect the unamended dataset. Homogeneity of variances was present, ps > .5.

6.2.3.2. Testing H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity.

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was run on measured familiarity to test **H1. Not supporting H1:** There was no significant main effect of manipulated familiarity on measured familiarity, F(1, 26) = 1.15, p = .29, $\eta_p^2 = .04$.

6.2.4. Discussion and Conclusion

Not supporting **H1**, the familiarity manipulation had no effect on measured familiarity. This lack of effectiveness may have arisen from technical issues in presentation of the images. Specifically, participants mentioned in comments at the end of the study that the images were not fully visible on the screen, which may therefore have undermined any mere exposure effect. Therefore, Pilot Study 3b re-tests the familiarity manipulation after ensuring the images are clearly and fully visible on each page to overcome this technical issue.

6.3. Pilot Study 3b to Test Amended Familiarity Manipulation

6.3.1. Introduction

Following null effects of the familiarity manipulation on measured familiarity in Pilot Study 2b, the current pilot study aims to re-test this familiarity manipulation after resolving a technical error and ensuring all images are clearly and fully visible on the screen. Therefore, familiarity is again manipulated through the mere exposure effect (Zajonc, 1968), whereby participants view nine photographs of the unknown target animal (familiar condition) vs. nine photographs of trees which act as the control (unfamiliar condition). However, this familiarity manipulation is amended slightly by displaying photographs at the very top of the screen to ensure easy viewing of the photographs, following participant feedback from Pilot Study 2b that the entirety of the photographs were difficult to view. To check effectiveness of the familiarity manipulation (manipulation check), the item '*How familiar do you perceive the species in the previous photograph to be to you*?' is again included to measure familiarity. In line with the mere exposure effect (Zajonc, 1968), I hypothesise that:

H1: Viewing photographs of an animal (vs. control) will significantly increase measured familiarity of the animal.

Note that this pilot study utilises the same participant sample to also pilot a manipulation for behavioural self-relevance, but only information relevant to the familiarity manipulation is reported within the current chapter (see Section 4.4 for pilot study information about the behavioural self-relevance manipulation [Pilot Study 3a]).

6.3.2. Methods

6.3.2.1 Participants.

Fifty-five participants were recruited via social media and Sona as a volunteer sample. Eighteen participants were excluded for providing partial data, and seven participants were

excluded for not consuming meat (meaning that all remaining participants were meat consumers only). As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 6.3.3. for further discussion). Exclusions left a total sample size of 30 participants (50% women, 50% men; $M_{age} = 25.3$, $SD_{age} = 6.13$, age range: 18-50), which exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Participants within each condition are shown in Table 22. Demographics of the sample are reported as follows in percentages. For nationality: 16.7% (United States), 13.3% (British), 6.7% each (Chinese, Colombian, Filipino, French), and 3.3% each (Canadian, Croatian, Czech, Danish, Dutch, Estonian, Indian, Indonesian, Italian, Malaysian, Polish, Swedish, Taiwanese). For ethnicity: 53% (White), 26.5% (Asian), 10% (mixed), and 3.3% each (Black, Hispanic, Latin American). For country of residence ($M_{duration}=17.79$, $SD_{duration}=11.79$, range: 1 month to 50 years): 23.3% (United States), 20% (United Kingdom), 10% (France), 6.7% each (Germany, the Philippines), and 3.3% each (Canada, Croatia, Czech Republic, Denmark, Hungary, Indonesia, Malaysia, Poland, Sweden, the Netherlands). For religion: 40% (Christianity), 26.7% (atheism), 10% each (agnosticism, would rather not say), and 3.3% each (Buddhism, Hinduism, Judaism, no religion).

Table 22.

Participants Within Each Condition for Pilot Study 3b.	
Familiarity Condition	Number of Participants
Familiar	15
Unfamiliar	15

6.3.2.2. Design.

The current pilot study follows a 2(familiarity: familiar vs. control; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed ANCOVA design on measured

familiarity. Familiarity was included as a DV to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 6.3.3. for further discussion).

6.3.2.3. Materials.

6.3.2.3.1. Target Animal and Animal Photograph.

The current pilot study employed the same photograph of the tree kangaroo as Study 2 and Pilot Study 2a, whereby this image was sourced from

<u>https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035</u>. This photograph enables easier visualisation of the imagined scenario and reduces biased perceptions of the animal due to species name.

6.3.2.3.2. Familiarity Manipulation.

Familiarity was again manipulated through the mere exposure effect (Zajonc, 1968), whereby participants were first informed that '*you will now see several photographs over the following pages. Each photograph will automatically move to the next photograph after 1 second*', and then either viewed nine photographs of the target animal (familiar condition) or nine photographs of trees (unfamiliar condition; control). After one second of viewing each image, the page automatically advanced.

6.3.2.3.3. Measured Familiarity (Manipulation Check).

Familiarity was again measured via the single item developed by the researcher: 'How familiar do you perceive the species in the previous photograph to be to you?' on a zero ('not familiar at all') to 100 ('extremely familiar') visual analogue scale. A higher score indicates greater familiarity.

6.3.2.3.4. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be*?' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

6.3.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in September 2021 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial familiarity. Participants then completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the familiarity section is described (see Section 4.4. for description of the self-relevance section). For the familiarity section, participants were randomly allocated to see nine images either of the target animal or of trees. After viewing the images, participants then again rated their familiarity with the animal. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Three participants reported technical difficulties, but their responses were complete and maintained within analyses.

6.3.3. Analyses

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was conducted on measured familiarity, to test if the manipulation effectively manipulates familiarity. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured familiarity (the outcome variable) to be statistically controlled for. Neutrality was also included within the analysis as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analysis by statistically controlling for any possible relationship between neutrality and familiarity (Tabachnick & Fidell, 2014).

6.3.3.1. Statistical Assumptions to Test H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity (2x2 ANCOVA).

Measured familiarity failed normality tests at pre-manipulation, ps < .05, whilst the normality assumption was met for measured familiarity at post-manipulation, ps > .05. However, skewness was acceptable across all conditions. There were outliers in measured familiarity. Running the analysis including and excluding these outliers did not change conclusions. I report this ANCOVA including outliers to reflect the unamended dataset. Homogeneity of variances was not present, ps < .5.

6.3.3.2. Testing H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity.

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was run on measured familiarity to test **H1. Not supporting H1:** There was no significant main effect of manipulated familiarity on measured familiarity, F(1, 26) = .02, p =.9, $\eta_p^2 = .001$.

6.3.4. Discussion and Conclusion

Not supporting **H1**, the familiarity manipulation again had no effect on measured familiarity, indicating that the lack of effectiveness is not arising from technical issues in presentation of the images. Lack of effectiveness of the familiarity manipulation could instead be due to participants' confusion about exactly what is meant by 'familiar' within the familiarity manipulation check item, and/or could be due to an underpowered sample. Pilot Study 4b therefore re-tests the familiarity manipulation with a larger and sufficiently powered sample per a G*Power power analysis, and includes the following definition of familiarity for participants to read: '*By familiar, we mean a feeling of acquaintance with the species, e.g. based on seeing it and/or learning about its habit and environment, even if it is from this very experiment for the first time. Higher scores mean a feeling of higher familiarity and lower scores mean a feeling of lower familiarity.'*

6.4. Pilot Study 4b to Test Amended Familiarity Manipulation

6.4.1. Introduction

Following null effects of the familiarity manipulation on measured familiarity in Pilot Study 3b, the current pilot study aims to re-test this familiarity manipulation with a larger sample size in case of underpowered analyses, and with an amendment to the familiarity manipulation check item. Therefore, familiarity is again manipulated through the mere exposure effect (Zajonc, 1968), whereby participants view nine photographs of the unknown target animal (familiar condition) vs. nine photographs of trees which act as the control (unfamiliar condition). To check effectiveness of the familiarity manipulation (manipulation check), the item '*How familiar do you perceive the species in the previous photograph to be to you?*' is again included to measure familiarity. However, the following definition for familiarity is also included alongside this manipulation check item, to ensure that participants are clear on what is being meant by 'familiar': '*By familiar, we mean a feeling of acquaintance with the species, e.g. based on seeing it and/or learning about its habit and environment, even if it is from this very experiment for the first time. Higher scores mean a feeling of higher familiarity and lower scores mean a feeling of lower familiarity.* 'In line with the mere exposure effect (Zajonc, 1968), I hypothesise that:

H1: Viewing photographs of an animal (vs. control) will significantly increase measured familiarity of the animal.

Note that this pilot study utilises the same participant sample to also pilot a manipulation for behavioural self-relevance, but only information relevant to the familiarity manipulation is reported within the current chapter (see Section 4.5. for pilot study information about the behavioural self-relevance manipulation [Pilot Study 4a]).

6.4.2. Methods

6.4.2.1. Participants.

One-hundred-and-sixty participants were recruited via Prolific as a volunteer sample. Quota sampling was partially used through Prolific's 'balance by sex'²⁵ option to obtain approximately equal numbers of men and women. Thirty participants were excluded for:

²⁵Prolific does not provide an option to 'balance by gender'.

recognising the tree kangaroo (n = 16), taking part in a previous study within this project (n = 16) 7), withdrawing from the study (n = 5), and not consuming meat (n = 2, meaning that all remaining participants were meat consumers only). As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 6.4.3. for further discussion). Exclusions left a total sample size of 130 participants (51.5% women, 48.5% men; $M_{age} = 24.78$, $SD_{age} = 7.13$, age range: 18-56), which exceeded the minimum required sample size of 128 per a G*Power *a priori* power analysis for a ANCOVA analysis (medium effect size f=.25, two groups, two covariates, one numerator df, α =.05, power = .8). Participants within each condition are shown in Table 23. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For nationality: 35.4% (South African), 15.4% (Portuguese), 9.2% (Polish), 5.4% (Italian), 4.6% (Zimbabwean), 3.8% (Hungarian), 3.1% each (British, Mexican), 2.3% (Spanish), and 17.7% other. For ethnicity: 49.5% (White), 34.6% (Black), 6.2% (mixed), 3.8% each (Asian, would rather not say), and 2.1% other. For country of residence (M_{duration}=22.54, SD_{duration}=8.81, range: 3 months to 56 years): 42.3% (South Africa), 15.4% (Portugal), 8.5% (Poland), 6.9% (Italy), 6.1% (United Kingdom), 4.6% (Hungary), 3.1% (Mexico), 2.3% (Spain), and 10.8% other. For religion: 63.8% (Christianity), 19.2% (atheism), 8.5% (agnosticism), 3.8% (would rather not say), and 4.7% other.

Table 23.

Participants Within Each Condition for Pilot Study 4b.

Familiarity Condition	Number of Participants
Familiar	65
Unfamiliar	65

6.4.2.2. Design.

The current pilot study follows a 2(familiarity: familiar vs. control; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed ANCOVA design on measured familiarity. Familiarity was included as a DV to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 6.4.3. for further discussion).

6.4.2.3. Materials.

6.4.2.3.1. Target Animal and Animal Photograph.

The current pilot study employed the same photograph of the tree kangaroo as Study 2 and previous pilot studies, whereby this image was sourced from <u>https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035</u>. This photograph enables easier visualisation of the imagined scenario and reduces biased perceptions of the animal due to species name.

6.4.2.3.2. Familiarity Manipulation.

Familiarity was again manipulated through the mere exposure effect (Zajonc, 1968), whereby participants were first informed that '*you will now see several photographs over the following pages. Each photograph will automatically move to the next photograph after 1 second*', and then either viewed nine photographs of the target animal (familiar condition) or nine photographs of trees (unfamiliar condition; control). After one second of viewing each image, the page automatically advanced.

6.4.2.3.3. Measured Familiarity (Manipulation Check).

Familiarity was again measured via the single item developed by the researcher: 'How familiar do you perceive the species in the previous photograph to be to you?' on a zero ('not familiar at all') to 100 ('extremely familiar') visual analogue scale. However, the following definition of what is meant by 'familiar' was included within the current pilot study alongside

the familiarity manipulation check item, to ensure that participants are clear on what is being meant by 'familiar': '*By familiar, we mean a feeling of acquaintance with the species, e.g. based on seeing it and/or learning about its habits and environment, even if it is from this very experiment for the first time. Higher scores mean a feeling of higher familiarity and lower scores mean a feeling of lower familiarity.*' A higher score indicates greater familiarity.

6.4.2.3.4. Perceived Neutrality Towards the Animal.

Perceived neutrality was included as a covariate to account for bias and preconceptions in perceptions of the target animal. Due to lack of a pre-existing scale to measure perceived neutrality towards the animal, perceived neutrality was measured via a single item developed by the researcher: '*How neutral do you perceive the species in the previous photograph to be?*' on a visual analogue scale from zero ('*not neutral at all*') to 100 ('*most definitely neutrality*'). Neutrality was defined as '*By neutral, we mean how non-biased you believe your opinions of the species are. Higher scores mean less bias, whilst lower scores mean more bias*'. A higher score indicates higher neutrality and less bias.

6.4.2.4. Procedure.

Participants were recruited via Prolific. All participants had to be 18+ meat consumers who had not taken part in previous studies within this thesis. The experiment was conducted in November 2021 as a voluntary, open survey open to participants who registered for the study via Prolific. After providing informed consent, participants were asked if they recognised tree kangaroos, alongside a photograph of the animal. The study either proceeded using the tree kangaroo as the target animal, or, if participants recognised the tree kangaroo, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants provided demographics and stated their perceived neutrality towards the animal and their initial familiarity. Participants then

completed the two sections (self-relevance, familiarity) of the pilot study in a randomised order. Here, only the familiarity section is described (see Section 4.5. for description of the self-relevance section). For the familiarity section, participants were randomly allocated to see nine images either of the target animal or of trees. After viewing the images, participants then again rated their familiarity with the animal. Finally, participants had the chance to report technical difficulties and provide any final comments about the study, before being debriefed. Two participants reported technical difficulties, but their responses were complete and maintained within analyses.

6.4.3. Analyses

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was conducted on measured familiarity, to test if the manipulation effectively manipulates familiarity. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured familiarity (the outcome variable) to be statistically controlled for. Neutrality was also included within the analysis as a covariate to account for bias in perceptions of animals. That is, by including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analysis by statistically controlling for any possible relationship between neutrality and familiarity (Tabachnick & Fidell, 2014).

6.4.3.1. Statistical Assumptions to Test H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity (2x2 ANCOVA).

Measured familiarity failed normality tests across most conditions, ps < .05, but skewness was acceptable across all conditions. There were outliers in measured familiarity at pre-manipulation. Running the analysis including and excluding these outliers did not change conclusions. I report this ANCOVA including outliers to reflect the unamended dataset. Homogeneity of variances was present, ps > .5.

6.4.3.2. Testing H1: Exploring Effects of Familiarity Manipulation on Measured Familiarity.

A 2(familiarity: familiar vs. unfamiliar; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed ANCOVA, including gender and neutrality as covariates, was run on measured familiarity to test **H1. Not supporting H1:** There was no significant main effect of manipulated familiarity on measured familiarity, F(1, 126) = .75, p = .39, $\eta_p^2 = .01$.

6.4.4. Discussion and Conclusion

Not supporting **H1**, the familiarity manipulation again had no effect on measured familiarity, indicating that the lack of effectiveness of the manipulation is not arising from underpowered analyses and/or participant confusion about what is meant by 'familiar'. Due to this familiarity manipulation failing across three pilot studies, despite amendments, further investigation of familiarity and its possible causal effects of familiarity on perceptions of dogs and pigs was not pursued and cannot be determined within this thesis. However, these failed familiarity manipulations do indicate that imagined contact and the mere exposure effect may not always effectively apply to animals, which does not support previous literature (e.g., Auger & Amiot, 2019b; Cerrato & Forestell, 2022; Zajonc, 1968). As such, the next

study (Study 5) only explores *similarity* as a possible cause of pet speciesism and does not explore familiarity.

6.5. Study 5: Applying Similarity to Dogs and Pigs

6.5.1. Introduction and Extension from Studies 1 and 4

Study 4 found that similarity informed warmth and competence of, and active and passive help towards, an unknown animal (tree kangaroo, fossa, tamandua). These effects of similarity mostly agreed with previous literature (Leach et al., 2021), whilst uniquely demonstrating how similarity does not inform harm intentions towards unknown animals. Specifically, these findings indicate that positive effects of similarity found in previous literature (e.g., Batt, 2009; Kozachenko & Piazza, 2021; Leach et al., 2021; Possidónio et al., 2019) extend to the SCM (warmth/competence perceptions) and partially extend to the BIAS map (help intentions) with unknown animals, whereby these positive effects of similarity are theorised to arise from similarity enabling 'humanisation' of the similar animal, by encouraging inclusion of the animal within our moral circle, and thereby enhancing moral concern for the animal (Bastian et al., 2012).

The current study aims to extend Study 4 findings applied to the SCM/BIAS map to pet speciesism specifically, by testing causal effects of similarity on warmth/competence perceptions of and behavioural intentions towards dogs and pigs specifically. This finding therefore directly tests if similarity is a cause of pet speciesism: As dogs are deemed more similar to humans than pigs (Study 1), if similarity contributes positively to perceptions of dogs' warmth and competence and help intentions towards them within this study, then similarity can be considered a cause of pet speciesism across these dimensions. That is, dogs' greater similarity (compared to pigs) would inform dogs' greater warmth and competence, and greater help intentions towards dogs, whilst pigs' lower similarity (compared to dogs) would inform pigs' lower warmth and competence, and lower help intentions towards pigs.

Similarity is again manipulated through fictional newspaper text describing the target animal as either being similar or dissimilar to humans across four traits (behaviour, level of intelligence, sociality, and emotionality), piloted in Pilot Study 1b (Section 5.2.). However, unlike Study 4, this study does not utilise unknown animals within the similarity manipulation as Study 4 already tested similarity with unknown animals. Warmth, competence, and behavioural intentions are again measured within the current study using the SCM/BIAS map subscales from Sevillano and Fiske (2016b). Following positive effects of similarity on warmth, competence and help intentions from Study 4, and in line with above discussed positive effects of similarity on perceptions of animals from previous research (e.g., Leach et al., 2021), I hypothesise that:

H1: Purportedly similar (vs. dissimilar) dogs will be deemed significantly warmer (a) and more competent (b).

H2: Similarity of dogs will significantly increase active (a) and passive (b) help intentions towards dogs from pre- to post-manipulation.

Note, however, that I do not formulate hypotheses for pigs, as people can disregard similarity information about pigs (Piazza & Loughnan, 2016). Specifically, animals deemed more (vs. less) intelligent are usually awarded higher moral status, yet this intelligence information is sometimes disregarded for pigs, whereby (high or low) intelligence of pigs is not used to inform their perceived moral status (Piazza & Loughnan, 2016). However, more recent research has found no evidence for people disregarding similarity information about 'food' animals including pigs (Leach et al., 2021), and thus it is currently unclear if and when similarity information about pigs and other 'food' animals is disregarded. As such, I do not include pigs within **H1-H2**, and instead any effects of similarity on pigs (including mediations) are investigated for exploratory purposes.

Alongside the above hypotheses, previous pet speciesism research (Bilewicz et al., 2011; Caviola & Capraro, 2020; Gradidge et al., 2021b), SCM/BIAS map research (Sevillano & Fiske, 2016a, 2016b, 2019), and Studies 1 and 3 findings indicate that dogs are viewed more positively than other animals (e.g., pigs), whilst pigs are viewed more negatively than other animals (e.g., pigs), whilst pigs are viewed more negatively than other animals (e.g., dogs). As such, I hypothesise that:

H3: Dogs will be deemed significantly warmer (a) and more competent (b) than pigs across similarity conditions and timepoints.

Furthermore, due to evidence of pet speciesism in behavioural intentions towards animals as well as perceptions of them (Gradidge et al., 2021a; Study 3 in this thesis), alongside active and passive behavioural intentions towards an animal being informed by the animal's perceived warmth and competence (Sevillano & Fiske, 2016; Sevillano & Fiske, 2019), I predict that differences in warmth/competence perceptions of species will extend to behavioural intentions, such that:

H4: Dogs will be subject to significantly more active (**a**) and passive help (**b**), and less active (**c**) and passive harm (**d**), than pigs across similarity conditions and timepoints.

Finally, inclusion of measurements for behavioural intentions within this study also again enables applicability of the BIAS map to animals to be tested (Sevillano & Fiske, 2016b, 2019). Firstly, it can be tested directly if any effects of *similarity* on active and passive behavioural intentions are mediated through warmth and competence respectively, in line with both general SCM and BIAS map literature (Cuddy et al., 2007) and SCM/BIAS map literature applied to animals specifically (Sevillano & Fiske, 2016b, 2019). I therefore hypothesise that:

H5: Post-manipulation warmth will significantly mediate the relationship between dogs' similarity and active help, per Study 4. That is, dogs portrayed as similar (vs. dissimilar) will be deemed significantly warmer post-manipulation, significantly increasing post-manipulation active help.

H6: Post-manipulation competence will significantly mediate the relationship between dogs' similarity and passive help, per Study 4. That is, dogs portrayed as similar (vs. dissimilar) will be deemed significantly more competent postmanipulation, significantly increasing post-manipulation passive help.

Secondly, it can be tested directly if any effects of *species* on active and passive behavioural intentions are mediated through warmth and competence respectively, in line with both general SCM and BIAS map literature (Cuddy et al., 2007) and SCM/BIAS map literature applied to animals specifically (Sevillano & Fiske, 2016b, 2019). I therefore predict that:

H7: Warmth will significantly mediate relationships between species and active help and active harm across timepoints. That is, dogs will be deemed significantly warmer than pigs, significantly increasing active help (**a**) and decreasing active harm (**b**).

H8: Competence will significantly mediate relationships between species and passive help and passive harm across timepoints. That is, dogs will be deemed significantly more competent than pigs across timepoints, significantly increasing passive help (a) and decreasing passive harm (b).

6.5.2. Methods

6.5.2.1. Participants.

Two-hundred-and-thirty participants were recruited via Prolific as a volunteer sample. Quota sampling was partially used through Prolific's 'balance by sex' option to obtain

approximately equal numbers of men and women. Fifty participants were excluded for: withdrawing their response (n=18), indicating their culture and/or religion does not typically view dogs as pets and/or pigs as food (n=16), participating in a previous experiment (n=6), failing the attention check (n=4), providing partial data only (n=3), or not consuming meat (n=3). All analyses included gender as a covariate (see Section 6.5.3. for further discussion), whereby gender was dummy coded into women (coded as zero; n=86) and men (coded as one; n=94). However, there were no participants of other genders and thus no further exclusions. Exclusions left a sample of 180 participants (52.2% men, 47.8% women; M_{age} =27.26, SD_{age} =9.09, age range=18-81), which exceeded the minimum required sample size of 179 per a G*Power a priori power analysis for a MANOVA analysis ('repeated measures, within-between interaction', medium effect size f=.25, four groups, two measurements, α =.05, power = .8). Participants within each condition are shown in Table 24. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For nationality: 19.4% (Portuguese), 17.2% (South African), 10% (Polish), 7.8% (Mexican), 6.1% (British), 5% (Chilean), 3.9% (Italian), 2.8% each (Greek, no response), 2.2% each (Estonian, Hungarian, United States), and 18.4% other. For ethnicity: 59.3% (White), 17.8% (Black), 9.8% (mixed), 4.4% (Hispanic/Latino), 3.3% each (Asian, would rather not say), and 2.1% other. For country of residence ($M_{duration}$ =23.97, $SD_{duration}$ =10.13, range: three months to 81 years): 21.1% (South Africa), 19.4% (Portugal), 9.4% (Poland), 8.3% (Mexico), 7.8% (United Kingdom), 5% (Chile), 3.9% (Italy), 2.8% each (Estonia, Greece, United States), 2.2% (Hungary), and 14.5% other. For religion: 47.8% (Christianity), 26.1% (atheism), 17.2% (agnosticism), 6.1% (would rather not say), and 2.8% other.

_1 anterpants within Each Condition for Study 5.		
Species	Similarity Condition	Number of Participants
Dog	Similar	44
Pig	Similar	43
Dog	Dissimilar	44
Pig	Dissimilar	49

Table 24.

 Participants Within Each Condition for Study 5.

6.5.2.2. Design.

This experiment follows a 2(similarity: similar vs. dissimilar; between-subjects) x 2(species: dog vs. pig; between-subjects) x 2(time: pre- vs. post-manipulation; withinsubjects) mixed MANCOVA design, with warmth, competence, active help, passive help, active harm, passive harm, perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans as the DVs. Measurements of similarity were included as DVs to test effectiveness of the similarity manipulation. Gender and neutrality were included as covariates (see Section 6.5.3. for further discussion).

6.5.2.3. Materials.

6.5.2.3.1. Animal Photographs.

The current study employed the same dog and pig photographs as Study 3, whereby the dog photograph was sourced from <u>https://www.flickr.com/photos/dave_see/8523607444</u> and the pig photograph was sourced from <u>https://www.pexels.com/photo/nature-animals-pig-alp-rona-63285/</u>. These photographs were included to enable easier visualisation of the imagined scenario.

6.5.2.3.2. Similarity Text Manipulations.

Similarity was manipulated via the same text manipulations from Study 4. Specifically, participants were asked to '*Imagine reading the following text in the newspaper*', with the following text stating that '*According to animal behaviour experts, the species in the photograph is reportedly similar [similar condition] / dissimilar [dissimilar for the species in the photograph is reportedly similar [similar condition] / dissimilar [dissimilar for the species in the photograph is reportedly similar [similar condition] / dissimilar [dissimilar for the species in the photograph is reported for the species in the species in the species in the photograph is reported for the species in the spe* condition] to humans in many ways, including in terms of behaviour, level of intelligence, sociality and emotionality'.

6.5.2.3.3. Warmth and Competence (Sevillano & Fiske, 2016b).

Warmth and competence were measured with the same items from Studies 1-4: '*warm*', '*well-intentioned*', '*friendly*' (warmth), and '*competent*', '*skillful*', '*intelligent*' (competence; Sevillano & Fiske, 2016b). Like Studies 3-4, these items were measured using a semantic differential from -3 to 3, with higher scores indicating greater warmth or competence. Items within each subscale were summed to provide warmth and competence scores. Reliability was adequate within this study for warmth (pre-manipulation ω =.83, 95% BCa CI [.78, .87]; post-manipulation ω =.89, 95% BCa CI [.84, .92]) and competence (premanipulation ω =.89, 95% BCa CI [.85, .92]; post-manipulation ω =.93, 95% BCa CI [.9, .95]) across timepoints.

6.5.2.3.4. Behavioural Intentions Towards the Animal (Sevillano & Fiske, 2016b).

Behavioural intentions were measured with the same items from Studies 2-4: *'support'*, *'help'*, *'behave friendly towards'*, *'interact with'* the animal (active help), *'sustain'*, *'conserve'* the animal (passive help), *'kill'*, *'injure'*, *'exterminate'*, *'trap'*, *'reject'* the animal (active harm), and *'let the species in the photograph die off'*, *'ignore'* the animal (passive harm; Sevillano & Fiske, 2016b). Like Studies 3-4, these items were measured using a semantic differential from -3 to 3 with higher scores indicating greater active help, active harm, passive help, or passive harm. Items within each subscale were summed to provide active help, passive help, active harm, and passive harm scores. Reliability was adequate within this study for active help (pre-manipulation ω =.88, 95% BCa CI [.83, .91]; postmanipulation ω =.91, 95% BCa CI [.87, .94]), and active harm (pre-manipulation ω =.77, 95% BCa CI [.68, .83]; post-manipulation ω =.84, 95% BCa CI [.77, .89]) across timepoints. Whilst reliability was adequate within this study for post-manipulation passive harm

 $(r_{sb}=.72)$, reliability was inadequate for pre-manipulation passive harm $(r_{sb}=.55)$ and inadequate for passive help across timepoints (pre-manipulation $r_{sb}=.66$; post-manipulation $r_{sb}=.62$). As the passive help and passive harm subscales contain two items, I did not run PAFAs for these subscales. Instead, I retained both items in the subscales, and findings using these subscales should be interpreted with caution due to inadequate reliability at one or both timepoints.

6.5.2.3.5. Perceived Neutrality Towards the Animal.

Neutrality was measured via the same single item utilised in Studies 3-4, with a higher score indicating higher neutrality and less bias: '*How neutral do you perceive the species in the previous photograph to be*?' from zero to 100.

6.5.2.3.6. Perceived Similarity to Humans (Manipulation Check).

Perceived similarity to humans was measured utilising the same single item from Studies 1 and 4 (*'How similar do you perceive the species in the photograph to be to humans?*'), measured on a 0-100 visual analogue scale. A higher score indicates greater perceived similarity to humans.

6.5.2.3.7. Perceived Shared Traits with Humans (Manipulation Check; Bastian et al., 2012a, Bilewicz et al., 2011, Rothgerber, 2014).

Perceived shared traits with humans was measured utilising the same scale from Study 4 (Bastian et al., 2012a; Bilewicz et al., 2011; Rothgerber, 2014) with the following items split into two subscales of emotional capacities and mental capacities: '*nostalgia*', '*happiness*', '*melancholy*', '*excitement*', '*guilt*', '*panic*' (emotional capacities), '*self-control*', '*morality*', '*memory*', '*planning*' (cognitive capacities). Participants answered on a Likert scale from one '*only humans have this capacity*' to seven '*the species in the photograph and humans have this capacity to the same degree*', with the midpoint at four '*the species in the* photograph has this capacity to some degree'. This measure was used as an additional measure of similarity to validate the perceived similarity to humans manipulation check. Items within each subscale were summed to provide emotional capacity and cognitive capacity scores, with higher scores on each subscale indicating greater shared emotional or cognitive traits with humans. Reliability was adequate within this study across timepoints: emotional capacities (pre-manipulation ω =.73, 95% BCa CI [.64, .8]; post-manipulation ω =.79, 95% BCa CI [.72, .84]).

6.5.2.4. Procedure.

Participants were recruited via Prolific (Appendix 9). All participants had to be meat consumers who had not taken part in previous studies within this thesis. The experiment was conducted in November and December 2021 as a voluntary survey open to participants who registered for the study via Prolific. After providing informed consent, participants gave demographic information, before being randomised into one of two species conditions (dog vs. pig; their target animal) and viewing a photograph of their target animal. Animal species names were not used during the study. Participants then indicated their perceived neutrality towards their target animal, the animal's warmth and competence, and behavioural intentions towards the animal, and the animal's perceived similarity to humans and perceived shared emotional and cognitive traits with humans (all scales and items within scales presented in randomised order).

Participants were subsequently randomly assigned into one of the two similarity conditions (similar vs. dissimilar) and read the text within their condition. After reading their assigned text, participants were reminded of their target animal through again seeing the photograph of the animal, before again rating the animal's perceived similarity to humans and perceived shared emotional and cognitive traits with humans, the animal's warmth and
competence, and behavioural intentions towards the animal (all scales and items within scales presented in randomised order), alongside completing an attention check within the perceived shared emotional and cognitive traits with humans scale. Specifically, to assess whether participants were paying sufficient attention to the measures and completing them correctly, participants were asked to '*Please select the answer '5'*. *This is to assess your attention*'. Four participants were excluded for failing this attention check (i.e., not providing the correct answer of five). Finally, participants indicated if their culture and/or religion typically views dogs as pets and pigs as food, before being debriefed and redirected back to Prolific. No participants reported technical difficulties. Participants received £1 via Prolific as reimbursement for full responses. Participants who provided partial data were reimbursed a proportion of the £1 based on their percentage of completion, whilst adhering to Prolific's minimum hourly rate.

6.5.3. Analyses

A 2(similarity: similar vs. dissimilar; between-subjects) x 2(species: dog vs. pig; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed MANCOVA was conducted on warmth, competence, active help, passive help, active harm, passive harm, perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans. Warmth, competence, active help, passive help, active harm, and passive harm were all conceptually related (i.e., from the SCM/BIAS map and subscales of the same overall scales), whilst perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans are conceptually related as they all measure similarity to humans albeit it in different ways. Additionally, perceived shared emotional traits with humans and perceived shared cognitive traits with humans are subscales of the same overall scale. Many of the DVs were statistically

related through moderate correlations, $rs \ge .3$, including between SCM/BIAS map DVs and similarity DVs. Thus, MANCOVA was deemed appropriate for this analysis.

Gender was again included as a covariate within the analyses. Again, although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and warmth/competence perceptions of animals and behavioural intentions towards them (the outcome variables) to be statistically controlled for. Neutrality was also included within analyses as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between between neutrality and perceptions of/behavioural intentions towards animals (Tabachnick & Fidell, 2014).

Descriptive statistics for significant discussed main effects or for *post hoc* pairwise comparisons for significant interaction effects are reported in Tables 25-32.

6.5.3.1. Statistical Assumptions to Test H1-H4 and for Similarity Manipulation Checks: Exploring Effects of Manipulated Similarity on SCM/BIAS Map Variables and Measured Similarity (2x2x2 MANCOVA).

All DVs failed normality tests, ps < .05, in at least one condition, but competence, active help, passive help, perceived similarity to humans, perceived shared emotional traits

with humans, and perceived shared cognitive traits with humans had acceptable skewness. Warmth had acceptable skewness in the pig conditions, but excessive skewness in the dog similar condition at pre-manipulation only and the dog dissimilar condition at postmanipulation only. Excessive skewness within those conditions arose in warmth due to one extreme univariate outlier in the dog similar condition at pre-manipulation, and one nonextreme univariate outlier in the dog dissimilar condition at post-manipulation. Removal of these two univariate outliers led to warmth having acceptable skewness across all conditions. Whilst comparing the MANCOVA including and excluding outliers was not possible (see below), running the MANCOVA including and excluding these two univariate outliers for warmth did not change multivariate conclusions. I therefore maintained warmth within the MANCOVA. Active harm and passive harm had excessive skewness, multiple univariate outliers, and floor effects in some of the dog conditions. Due to excessive skewness, multiple univariate outliers, and floor effects in active harm and passive harm, I also ran a MANCOVA excluding active harm and passive harm as DVs, which did not change multivariate conclusions. I therefore report the MANCOVA including active harm and passive harm as DVs.

Besides warmth, active harm, and passive harm, perceived similarity to humans had no univariate outliers, whilst all other DVs had non-extreme univariate outliers which did not cause excessive skewness. There were multivariate outliers for nine DVs, critical values \geq 27.88, p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in active harm and passive harm. Running the MANCOVA including univariate and multivariate outliers without active harm and passive harm as DVs was also not possible, as exclusion of univariate and multivariate outliers led to homogenous scores in most DVs due to exclusion

of most of the dataset. I therefore report the MANCOVA including univariate and multivariate outliers.

There was no multicollinearity, $rs \le .8$. Linear relationships between the DVs were mostly present across conditions. Running follow-up univariate ANCOVAs including and excluding univariate outliers for warmth, passive help, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans did not change conclusions. Excluding univariate outliers changed one minor conclusion each for competence (Footnote 26) and active help (Footnote 27). I report these ANCOVAs including outliers to reflect the unamended dataset. Excessive skewness, multiple univariate outliers, and floor or ceiling effects in active harm and passive harm meant assumptions for the ANCOVAs on these two DVs were violated. However, I proceeded with these ANCOVAs due to robustness of ANCOVA to non-normality, and lack of non-parametric alternatives. Results from these ANCOVAs should be interpreted with caution.

Homogeneity of variances was present for pre-manipulation warmth, postmanipulation perceived shared emotional traits with humans, and for competence, passive help, perceived similarity to humans, and perceived shared cognitive traits with humans across timepoints, $p_s > .05$. Homogeneity of variances was not present for post-manipulation warmth, pre-manipulation perceived shared emotional traits with humans, and for active help, active harm, and passive harm across timepoints, $p_s < .05$. Homogeneity of covariances was not present, p < .001.

6.5.3.2. Testing H1-H4 and Similarity Manipulation Checks: Exploring Effects of Manipulated Similarity on SCM/BIAS Map Variables and Measured Similarity.

A 2(similarity: similar vs. dissimilar) x 2(species: dog vs. pig) x 2(time: pre- vs. postmanipulation) mixed MANCOVA, including gender and neutrality as covariates, was run on warmth, competence, active help, passive help, active harm, passive harm, perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans to: a) test effectiveness of the similarity text in manipulating measured similarity, and b) test **H1-H4**. Perceived similarity to humans, perceived shared emotional traits with humans, and perceived shared cognitive traits with humans were included in this MANCOVA instead of a separate 2(similarity: similar vs. dissimilar) x 2(time: pre- vs. post-manipulation) within-subjects MANCOVA to check consistent effectiveness of the similarity manipulation across species conditions (see Table A9 in Appendix 4 for full inferential statistics).

6.5.3.2.1. Multivariate Effects of Similarity and Species on Combined DVs.

There was a significant two-way interaction between time and similarity on the combined DVs, F(9, 166) = 5.71, p < .001, $\eta_p^2 = .24$ (large-sized). There were also significant main effects of similarity, F(9, 166) = 2.84, p = .004, $\eta_p^2 = .13$ (large-sized), and species, F(9, 166) = 15.22, p < .001, $\eta_p^2 = .45$ (large-sized), on the combined DVs. All other multivariate interactions and main effects were non-significant, ps > .05. I followed up significant multivariate effects with univariate ANCOVAs on the DVs below.

6.5.3.2.2. Manipulation Check: Univariate Effects of Manipulated Similarity and Species on Perceived Similarity to Humans.

To be deemed effective, the similarity manipulation would be expected to increase perceived similarity to humans from pre- to post-manipulation, and the similar condition should cause greater post-manipulation perceived similarity to humans than the dissimilar condition. Indeed, there was a significant interaction between time and similarity condition on perceived similarity to humans, F(1, 174) = 17.33, p < .001, $\eta_p^2 = .09$ (medium-sized). *Post hoc* pairwise comparisons revealed that, post-manipulation only, the similar condition significantly increased perceived similarity to humans more than the dissimilar condition, p = 0.000 (medium-sized).

.01, d = .42 (small-sized). As would be expected, this effect was not present premanipulation, p = .83. Additionally, the similar condition increased perceived similarity to humans from pre-manipulation to post-manipulation, p < .001, d = .54 (medium-sized). Conversely, in the dissimilar condition, perceived similarity to humans did not significantly differ from pre-manipulation to post-manipulation, p = .85. This interaction therefore suggests the similar text was effective in manipulating perceived similarity to humans from pre- to post-manipulation, and as compared to the dissimilar text. However, the dissimilar text was not effective in reducing perceived similarity to humans.

Table 25.

Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Perceived Similarity to Humans.

Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	39.36	2.75	49.87	2.81
Dissimilar	39.18	2.66	38.82	2.72

Beyond the manipulation check, there was a significant main effect of species on perceived similarity to humans, F(1, 174) = 6.49, p = .01, $\eta_p^2 = .04$ (small-sized), whereby dogs were deemed more similar to humans than pigs were. These findings suggest perceived similarity to humans was higher for dogs (vs. pigs) across similarity conditions and timepoints.

6.5.3.2.3. Manipulation Check: Univariate Effects of Manipulated Similarity and Species on Perceived Shared Emotional Traits with Humans.

To be deemed effective, the similarity manipulation would be expected to increase perceived shared emotional traits from pre- to post-manipulation, and to cause greater postmanipulation perceived shared emotional traits than the dissimilar condition. Indeed, there was a significant interaction between time and similarity condition on perceived shared emotional traits, F(1, 174) = 29.75, p < .001, $\eta_p^2 = .15$ (large-sized). Post hoc pairwise comparisons found that, post-manipulation only, participants in the similar condition perceived significantly higher shared emotional traits between the animal and humans compared to participants in the dissimilar condition, p < .001, d = .76 (medium-sized). As would be expected, this effect was not present pre-manipulation, p = .18. Additionally, participants in the similar condition perceived significantly higher shared emotional traits between the animal and humans post-manipulation compared to pre-manipulation, p < .001, d = .46 (small-sized). Unlike when similarity was measured via perceived similarity to humans, participants in the dissimilar condition perceived significantly fewer shared emotional traits between the animal and humans post-manipulation compared to pre-manipulation, p < .001, d = .44 (medium-sized). This interaction therefore evidences effectiveness of the similar text in manipulating perceived shared emotional traits with humans from pre- to post-manipulation, and as compared to the dissimilar text. It also suggests that, unlike when similarity is measured as perceived similarity to humans, the dissimilar condition decreases perceived shared emotional traits with humans from pre- to post-manipulation. Alongside this two-way interaction between time and similarity, there was a main effect of similarity on perceived shared emotional traits with humans, F(1, 174) = 13.26, p < .001, $\eta_p^2 = .07$ (medium-sized). However, this main effect is qualified by the above two-way interaction with time and is therefore not expanded upon further here.

Table 26.

Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Perceived Shared Emotional Traits with Humans.

Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	27.5	.6	29.59	.7
Dissimilar	26.36	.58	24.59	.68

Beyond the manipulation check, there was a significant main effect of species on perceived shared emotional traits with humans, F(1, 174) = 45.36, p < .001, $\eta_p^2 = .21$ (large-sized), whereby dogs were viewed as sharing significantly more emotional traits with humans than pigs were. These findings suggest perceived shared emotional traits with humans were higher for dogs (vs. pigs) across similarity conditions and timepoints.

6.5.3.2.4. Manipulation Check: Univariate Effects of Manipulated Similarity and Species on Perceived Shared Cognitive Traits with Humans.

To be deemed effective, the similarity manipulation would be expected to increase perceived shared cognitive traits with humans from pre- to post-manipulation, and to cause greater post-manipulation perceived shared cognitive traits with humans than the dissimilar condition. Indeed, there was a significant interaction between time and similarity condition on perceived shared cognitive traits with humans, F(1, 174) = 40.27, p < .001, $\eta_p^2 = .19$ (large-sized). Post hoc pairwise comparisons found that, post-manipulation only, participants in the similar condition perceived significantly more shared cognitive traits between the animal and humans compared to the dissimilar condition, p = .002, d = .47 (small-sized). As would be expected, this effect was not present pre-manipulation, p = .22. Additionally, participants in the similar condition perceived significantly more shared cognitive traits between the animal and humans post-manipulation compared to pre-manipulation, p < .001, d = .54 (medium-sized). Unlike when similarity was measured via perceived similarity to humans, participants in the dissimilar condition perceived significantly fewer shared cognitive traits between the animal and humans post-manipulation compared to premanipulation, p = .001, d = .52 (medium-sized). This interaction therefore evidences effectiveness of the similar text in manipulating perceived shared cognitive traits with humans from pre- to post-manipulation, and as compared to the dissimilar text. It also suggests that, unlike when similarity is measured as perceived similarity to humans, the

dissimilar condition decreases perceived shared cognitive traits with humans from pre- to post-manipulation.

Table 27.

Shared Cognitive Traits with Humans.					
Condition	Pre-	Pre-	Post-	Post-	
	Manipulation	Manipulation	Manipulation	Manipulation	
	Mean	SE	Mean	SE	
Similar	12.51	.44	14.49	.53	
Dissimilar	13.27	.43	12.18	.51	

Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Perceived Shared Cognitive Traits with Humans.

Beyond the manipulation check, there was a significant main effect of species on perceived shared cognitive traits with humans, F(1, 174) = 31.68, p < .001, $\eta_p^2 = .15$ (large-sized), whereby dogs were viewed as sharing significantly more cognitive traits with humans than pigs were. These findings suggest perceived shared cognitive traits with humans were higher for dogs (vs. pigs) across similarity conditions and timepoints.

6.5.3.2.5. H1a and H3a: Univariate Effects of Similarity and Species on Warmth.

There was a significant two-way interaction between time and similarity on warmth, $F(1, 174) = 8.02, p = .01, \eta_p^2 = .04$ (small-sized). Agreeing with H1a: Post hoc pairwise comparisons revealed the similar condition caused significantly greater post-manipulation warmth than the dissimilar condition, p < .001, d = .62 (medium-sized). As would be expected, there was no significant difference in warmth between the two similarity conditions pre-manipulation, p = .06. Additionally, the similar condition significantly increased warmth from pre-manipulation to post-manipulation, p = .02, d = .29 (small-sized). Conversely, warmth did not significantly differ from pre-manipulation to post-manipulation in the dissimilar condition, p = .12. Due to the absence of an interaction with species, these findings extend H1a by demonstrating applicability of similarity to pigs' warmth as well as dogs.

Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	5.62	.3	6.33	.34
Dissimilar	4.82	.29	4.39	.33

 Table 28.

 Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Warmth.

Corroborating H3a: There was a significant main effect of species on warmth, F(1, 174) = 72.84, p < .001, $\eta_p^2 = .3$ (large-sized), whereby dogs were deemed warmer than pigs.

6.5.3.2.6. H1b and H3b: Univariate Effects of Similarity and Species on

Competence.²⁶

There was a significant two-way interaction between time and similarity on competence, F(1, 174) = 23.41, p < .001, $\eta_p^2 = .12$ (large-sized). Agreeing with H1b: *Post hoc* pairwise comparisons revealed the similar condition caused significantly greater postmanipulation competence than the dissimilar condition, p < .001, d = .56 (medium-sized). As would be expected, there was no significant difference in competence between the two similarity conditions pre-manipulation, p = .9. Additionally, the similar condition significantly increased competence from pre-manipulation to post-manipulation, p = .003, d =.37 (small-sized). Additionally, the dissimilar condition significantly decreased competence from pre-manipulation to post-manipulation, p < .001, d = .5 (medium-sized). Due to the absence of an interaction with species, these findings extend H1b by demonstrating applicability of similarity to pigs' competence as well as dogs. Alongside this two-way interaction between time and similarity, there was a main effect of similarity on competence, F(1, 174) = 4.39, p = .04, $\eta_p^2 = .03$ (small-sized). However, this main effect is qualified by the above two-way interaction with time and is therefore not expanded upon further here.

²⁶When excluding univariate outliers, the main effect of similarity on competence was no longer significant, F(1, 172) = 3.45, p = .07, $\eta_p^2 = .02$.

Competence.				
Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	2.85	.38	3.86	.43
Dissimilar	2.92	.37	1.65	.41

 Table 29.

 Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Competence.

Supporting H3b: There was a significant main effect of species on competence, F(1, 174) = 90.12, p < .001, $\eta_p^2 = .34$ (large-sized), in the predicted direction whereby dogs were deemed more competent than pigs.

6.5.3.2.7. H2a and H4a: Univariate Effects of Similarity and Species on Active Help.²⁷

There was a significant two-way interaction between time and similarity on active help, F(1, 174) = 4.98, p = .03, $\eta_p^2 = .03$ (small-sized). *Post hoc* pairwise comparisons revealed that, post-manipulation only, the similar condition significantly increased active help compared to the dissimilar condition, p < .001, d = .52 (medium-sized). As expected, this effect was not present pre-manipulation, p = .08. **Agreeing with H2a:** The similar condition increased active help from pre-manipulation to post-manipulation, p = .02, d = .3 (smallsized). Conversely, the dissimilar condition did not cause significant differences in active help from pre-manipulation to post-manipulation, p = .42. Due to the absence of an interaction with species, these findings extend **H2a** by demonstrating applicability of similarity to active help towards pigs as well as dogs. Alongside this two-way interaction between time and similarity, there was a main effect of similarity on active help, F(1, 174) =

²⁷When excluding univariate outliers, the main effect of similarity on active help was no longer significant, F(1, 163) = 2.38, p = .13, $\eta_p^2 = .01$

7.64, p = .01, $\eta_p^2 = .04$ (small-sized). However, this main effect is qualified by the above twoway interaction with time and is therefore not expanded upon further here.

Table 30.

Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Active Help.

Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	7.33	.54	8.18	.51
Dissimilar	5.99	.52	5.71	.49

Corroborating H4a: There was a significant main effect of species on active help, $F(1, 174) = 55.63, p < .001, \eta_p^2 = .24$ (large-sized), whereby dogs were subject to greater active help than pigs.

6.5.3.2.8. H2b and H4b: Univariate Effects of Similarity and Species on Passive Help.

Not supporting H2b: There were no significant effects of similarity on passive help either as a main effect, F(1, 174) = .25, p = .62, $\eta_p^2 = .001$, or as a two-way interaction with time, F(1, 174) = 3.03, p = .08, $\eta_p^2 = .02$. Agreeing with H4b: There was a significant main effect of species on passive help, F(1, 174) = 25.59, p < .001, $\eta_p^2 = .13$ (large-sized), whereby dogs were subject to greater passive help than pigs .

6.5.3.2.9. H4c: Univariate Effects of Similarity and Species on Active Harm.

There was a significant two-way interaction between time and similarity on active harm, F(1, 174) = 7.81, p = .01, $\eta_p^2 = .04$ (small-sized). Post hoc pairwise comparisons revealed that, post-manipulation only, the similar condition significantly reduced active harm compared to the dissimilar condition, p = .01, d = .39 (small-sized). As would be expected, this effect was not present pre-manipulation, p = .31. Additionally, the similar condition decreased active harm from pre-manipulation to post-manipulation, p = .01, d = .32 (smallsized). Conversely, active harm did not significantly differ from pre-manipulation to postmanipulation in the dissimilar condition, p = .18.

Table 31.

Descriptive Statistics for Interaction Effect of Similarity Condition and Time on Active Harm.

Condition	Pre-	Pre-	Post-	Post-
	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Similar	-10.85	.46	-11.6	.49
Dissimilar	-10.19	.45	-9.81	.48

Agreeing with H4c: There was a significant main effect of species on active harm, $F(1, 174) = 77.33, p < .001, \eta_p^2 = .31$ (large-sized), whereby dogs were subject to less active harm than pigs.

6.5.3.2.10. H4d: Univariate Effects of Species and Similarity on Passive Harm.

There were no significant effects of similarity on passive harm either as a main effect, $F(1, 174) = 2.23, p = .14, \eta_p^2 = .01$, or as a two-way interaction with time, F(1, 174) = .02, p $= .88, \eta_p^2 < .001$. Aligning with H4d: There was a significant main effect of species on passive harm, $F(1, 174) = 47.21, p < .001, \eta_p^2 = .21$ (large-sized), in the predicted direction, whereby dogs were subject to less passive harm than pigs.

Table 32.

DV	Dog Mean	Dog SE	Pig Mean	Pig SE
Perceived	46.18	2.57	36.94	2.52
Similarity to				
Humans**				
Perceived	29.86	.6	24.15	.59
Shared				
Emotional				
Traits***	14.01	45	11.21	
Perceived	14.91	.45	11.31	.44
Snared				
Traits***				
Warmth***	7	.29	3.58	.28
Competence***	5.27	.37	.37	.36
Active Help***	9.38	.49	4.22	.48
Passive	3.7	.28	1.68	.28
Active	-13.42	.45	-7.81	.44
Harm***				
Passive	-4.65	.25	-2.27	.24
Harm***				
<i>Note</i> . ** <i>p</i> ≤.01, ***	<i>p</i> ≤.001			

Descriptive Statistics for Effect of Species on Similarity Measures, Warmth, Competence, Active Help, Passive Help, Active Harm, and Passive Harm.

6.5.3.3. Testing H5-H8: Exploring Mediational Relationships Between Similarity, Species and Behavioural Intentions Through Warmth/Competence.

Per the BIAS map, I predicted post-manipulation warmth would mediate the relationship between dogs' similarity and post-manipulation active help towards dogs (**H5**), whilst post-manipulation competence would mediate the relationship between dogs' similarity and post-manipulation passive help towards dogs (**H6**). I also hypothesised warmth would mediate relationships between species and active help (**a**) and active harm (**b**; **H7**) across timepoints, whilst competence would mediate relationships between species and active help (**a**) and passive harm (**b**; **H8**).

To test **H5-H6**, I dummy coded similarity into one dummy variable with similarity coded as dissimilar (zero) and similar (one). To test **H7-H8**, I dummy coded species into one dummy variable with dog coded as zero and pig coded as one. Ten individual boot-strapped mediation analyses (10,000 bootstrap samples) were run using Model 4 in PROCESS through SPSS (Hayes, 2022) to assess mediational relationships for the behavioural intentions (**H5-H6** help intentions post-manipulation only; **H7-H8** across timepoints for all behavioural intentions). Whilst **H5-H6** were originally hypothesised to apply to dogs only, the above MANCOVA indicates these hypotheses also apply to pigs, due to effects of similarity on warmth, competence, active help, and active harm across species. Thus, I instead tested here if post-manipulation warmth mediates the relationship between similarity and post-manipulation active help (**H5**), and if post-manipulation competence mediates the relationship between similarity and post-manipulation passive help (**H6**), across species. To control for Type I error from multiple mediation testing, I utilised an adjusted confidence interval of 99%. Significant mediations are interpreted through confidence intervals which do not contain zero per Hayes (2018).

As warmth and competence were significantly correlated with each other at both timepoints within the current study as revealed through Pearson correlations, premanipulation r = .6, p < .001, post-manipulation r = .7, p < .001, and warmth and competence are conceptually related, warmth and competence were included as parallel mediators within the below mediation analyses. Inclusion of warmth and competence as parallel mediators allows for testing of any mediational roles of warmth for active behavioural intentions *whilst* controlling for competence, and any mediational roles of competence for passive behavioural intentions *whilst* controlling for warmth (Hayes, 2018).

6.5.3.3.1. H5: Mediation of Similarity on Post-Manipulation Active Help through Post-Manipulation Warmth.

Agreeing with H5: There was a significant indirect effect of similarity on postmanipulation active help through post-manipulation warmth when controlling for postmanipulation competence. Thus, post-manipulation warmth significantly mediated the relationship between similarity and post-manipulation active help across species (Figure 9). This model explained 43.2% of variance in active help ($R^2 = .432$). Additionally, inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater postmanipulation active help (Figure 9).



Figure 9. Mediational model of similarity on post-manipulation active help through postmanipulation warmth when controlling for post-manipulation competence. *Note:* Standard error is reported for indirect effect only. ***p<.001

6.5.3.3.2. H6: Mediation of Similarity on Post-Manipulation Passive Help through Post-Manipulation Competence.

Supporting H6: There was a significant indirect effect of similarity on postmanipulation passive help via post-manipulation competence when controlling for postmanipulation warmth. Thus, post-manipulation competence significantly mediated the relationship between similarity and post-manipulation passive help across species (Figure 10). This model explained 26.8% of variance in passive help ($R^2 = .268$). Additionally, inspection of path *b* indicated greater post-manipulation competence significantly predicted greater post-manipulation passive help (Figure 10).



Figure 10. Mediational model of similarity on post-manipulation passive help through postmanipulation competence when controlling for post-manipulation warmth. *Note:* Standard error is reported for indirect effect only. ***p<.001

6.5.3.3.3. H7a: Mediation of Species on Active Help through Warmth.

Aligning with H7a: There was a significant indirect effect of species on active help through warmth pre-manipulation and post-manipulation when controlling for competence. Thus, warmth significantly mediated the relationship between species and active help across timepoints (Figure 11). The model explained 42.6% (R^2 = .426) and 45.5% (R^2 = .455) of variance in active help for pre- and post-manipulation respectively. Additionally, inspection of path *b* indicated greater warmth significantly predicted greater active help (Figure 11).



Figure 11. Pre- and post-manipulation mediational models of species on active help through warmth when controlling for competence. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. $**p \le .01$ ***p < .001

6.5.3.3.4. H7b: Mediation of Species on Active Harm through Warmth.

Supporting H7b: There was a significant indirect effect of species on active harm through warmth pre-manipulation and post-manipulation when controlling for competence. Thus, warmth significantly mediated the relationship between species and active harm across timepoints (Figure 12). The model explained 49.2% (R^2 = .492) and 47% (R^2 = .47) of variance in active harm for pre- and post-manipulation respectively. Additionally, inspection of path *b* indicated greater warmth significantly predicted less active harm (Figure 12).



Figure 12. Pre- and post-manipulation mediational models of species on active harm through warmth when controlling for competence. *Note:* T1 refers to pre-manipulation and T2 refers to post-manipulation. Standard error is reported for indirect effects only. $**p \le .01$, ***p < .001

6.5.3.3.5. H8a: Mediation of Species on Passive Help through Competence.

Not supporting H8a: There was no significant indirect effect of species on passive help through competence when controlling for warmth, either pre-manipulation, b = -.65, SE = .34, 99% CI [-1.56, .22], or post-manipulation, b = -.81, SE = .33, 99% CI [-1.62, .13]. Thus, competence did not significantly mediate the relationship between species and passive help at either timepoint. However, inspection of path *b* indicated greater competence did significantly predict greater passive help both pre-manipulation, b = .13, SE = .06, p = .045, 99% CI [-.04, .29], and post-manipulation, b = .17, SE = .06, p = .003, 99% CI [.02, .32].

6.5.3.3.6. H8b: Mediation of Species on Passive Harm through Competence.

Not supporting H8b: There was no significant indirect effect of species on passive harm through competence when controlling for warmth, either pre-manipulation, b = .76, SE = .35, 99% CI [-.2, 1.66], or post-manipulation, b = .56, SE = .23, 99% CI [-.03, 1.23]. Thus,

competence did not significantly mediate the relationship between species and passive harm at either timepoint. However, inspection of path *b* indicated greater competence did significantly predict lower passive harm, both pre-manipulation, b = -.15, SE = .05, p = .004, 99% CI [-.29, -.02], and post-manipulation, b = -.12, SE = .05, p = .02, 99% CI [-.25, .01].

6.5.4. Discussion

6.5.4.1. Discussion of Findings.

6.5.4.1.1. H1-H2: Effects of Similarity on SCM/BIAS Map Variables.

As expected, and agreeing with previous literature (Batt, 2009; Kozachenko & Piazza, 2021; Westbury & Neumann, 2008) and Study 4, the similar condition was superior to the dissimilar condition in causing increased post-manipulation warmth and post-manipulation competence across species (medium-sized; H1a-H1b), and similarity increased warmth and competence of, and active help towards, dogs and pigs from pre- to post-manipulation (smallsized; H1a-H2a). Pre- to post-manipulation effects are small-sized only, so may have lower practical significance. Although Studies 4-5 found no effect of the dissimilar condition on active help from pre- to post-manipulation, the similar condition within this study was unlike Study 4 in being superior to the dissimilar condition in causing increased post-manipulation active help across species (medium-sized). These findings not only support H1a-H2a, but also extend original hypotheses from dogs only to pigs. That is, despite positive similarity information (e.g., intelligence) about pigs sometimes being disregarded when people form their perceptions of pigs (Piazza & Loughnan, 2016), the current study does not find any disregarding of similarity information about pigs in warmth, competence, or active help. This lack of disregarding similarity information aligns with more recent research (Leach et al., 2021).

There were no significant effects of similarity on passive help (not supporting **H2b**) or on passive harm. That is, unlike Study 4, similarity did not increase passive help from pre-

to post-manipulation, and, like Study 4, the similar condition did not cause greater postmanipulation passive help compared to the dissimilar condition. Additionally, like Study 4, there were no effects of similarity on passive harm either from pre- to post-manipulation, or at post-manipulation only when contrasting with the dissimilar condition. However, findings for passive help and passive harm should be interpreted with caution due to inadequate scale reliability for both variables, and due to floor effects, excessive skewness, and outliers within passive harm.

Presuming the pre- to post-manipulation effect of similarity on passive help found within Study 4 is robust (e.g., if it can be replicated with an adequately reliable passive help scale), then there is a discrepancy in findings across Studies 4-5, as the current study did not find any pre- to post-manipulation effect of similarity on passive help. Such a discrepancy may arise from a combination of the target animal being unknown in Study 4 and the nature of the passive help items (*'sustain', 'conserve'*). Specifically, the passive help items address support for conservation behaviours, and support for conservation is informed by an animal's endangered status (Tisdell et al., 2007). Thus, describing the unknown animal in Study 4 as similar to humans, combined with a lack of awareness from participants on how endangered the animal is or is not, may have increased participants' willingness to passively help (sustain, conserve) the animal from pre- to post-manipulation. Conversely, as participants are aware dogs and pigs are not endangered (and so do not need conserving), similarity information may have not informed passive help towards them.

Also deviating from Study 4, there were significant effects of similarity on active harm within the study (although these findings should be interpreted with caution due to significant floor effects, excessive skewness, and outliers in active harm). Specifically, the current study found that the similar condition was superior to the dissimilar condition in decreasing post-manipulation active harm, whilst similarity also decreased active harm from pre- to post-manipulation (small-sized). These effects are small-sized only, so may have lower practical significance. However, supporting the BIAS map, these effects on active harm agree with the effects of similarity on warmth, as warmth predicts lower active harm. Thus, similarity positively impacts dogs' and pigs' warmth as expected, and negatively impacts active harm towards dogs and pigs, in alignment with the BIAS map. However, combined with Study 4 findings, this effect of similarity on active harm indicates similarity only decreases active harm towards known animals (i.e., dog and pigs; Study 5), and not unknown animals (i.e., tree kangaroos; Study 4).

6.5.4.1.2. Effects of Dissimilarity on SCM/BIAS Map Variables.

Warmth, active help, and passive help did not significantly differ from pre- to postmanipulation in the dissimilar condition. Whilst these findings are consistent with Study 4 regarding help intentions, these findings also indicate the negative effect of dissimilarity on warmth for unknown animals from Study 4 may not be evidenced with dogs or pigs. Unlike warmth, dissimilarity information did decrease dogs' and pigs' competence from pre- to postmanipulation within this study (medium-sized), aligning with Study 4. That is, participants found both dogs and pigs less competent after being informed these animals are dissimilar to humans. Thus, dissimilarity informs known and unknown animals' competence across studies.

Considering Study 4 and the current study results together, dogs' and pigs' (but not unknown animals') warmth is stable and not easily decreased by dissimilarity information, whilst dogs', pigs', and unknown animals' competence is decreased by dissimilarity information. Conversely, pigs', dogs', and unknown animals' warmth and competence are increased by similarity information. These findings therefore uniquely indicate that a) warmth and competence of unknown animals are informed by (dis)similarity information (perhaps due to lack of prior information and hence full reliance on [dis]similarity information), b)

perceptions of dogs' and pigs' competence decrease following dissimilarity information and increase following similarity information, and c) perceptions of dogs' and pigs' warmth are robust to change following dissimilarity information, but do improve following similarity information. That is, whilst dogs' and pigs' warmth can increase, dogs' and pigs' warmth may already be informed by participants' robust preconceptions of these animals, and thus these animals' warmth does not decrease following dissimilarity information.

Regarding harm intentions, there was no effect of dissimilarity information on active or passive harm from pre- to post-manipulation. These findings are in line with Study 4 null effects of dissimilarity on harm intentions.

6.5.4.1.3. H3-H4: Effect of Species on SCM/BIAS Map Variables (Pet Speciesism).

Aligning with **H3-H4**, previous literature (Caviola & Capraro, 2019; Gradidge et al., 2021b) and Studies 1 and 3, pet speciesism was evidenced across all DVs. That is, dogs were deemed significantly warmer and more competent, and subject to significantly more active and passive help, and less active and passive harm, than pigs (large-sized). Thus, the current study replicates previous evidence of pet speciesism and, alongside Studies 1 and 3, extends findings to the SCM/BIAS map specifically.

6.5.4.1.4. H5-H8: Exploring Mediational Relationships Between Similarity, Species and Behavioural Intentions Through Warmth/Competence

Mediation analyses supported **H5-H7**, but not **H8**, therefore providing some support for previous SCM/BIAS map literature as applied to animals (Sevillano & Fiske, 2016b). Specifically, warmth mediated relationships between similarity and post-manipulation active help, species and active help, and species and active harm. That is, aligning with Study 4, the similar (vs. dissimilar) condition caused greater post-manipulation warmth, which in turn predicted greater post-manipulation active help. Additionally, dogs were deemed warmer

than pigs, which in turn informed greater active help and less active harm towards them. Conversely, post-manipulation competence mediated the relationship between similarity and post-manipulation passive help. Therefore, agreeing with Study 4, the similar (vs. dissimilar) condition caused greater post-manipulation competence, which in turn predicted greater postmanipulation passive help. However, competence did not mediate the relationships between species and passive help, and species and passive harm, which does not support the BIAS map (Sevillano & Fiske, 2016b).

Inspection of path *bs* revealed that greater warmth was consistently associated with greater active help across the similarity and species mediation analyses, whilst greater warmth was consistently linked to lower active harm across the species mediation analyses. Greater competence was also consistently associated with greater passive help across the similarity and species mediation analyses, whilst greater competence was consistently linked to lower passive harm across the species mediation analyses, despite null mediational relationships between species and passive behavioural intentions through competence. The mediation analyses here therefore fully support the BIAS map applied to animals in regard to warmth, whilst only partially supporting the BIAS map applied to animals in regards to competence.

6.5.4.2. Limitations of the Current Study and Directions for Future Research.

The current study has some limitations to be addressed in future research: a) imprecision in the similarity manipulation, and b) lack of consideration of mediators.

Firstly, the text developed and used within this study and Study 4 stated species are similar or dissimilar to humans on four different attributes: behaviour, sociality, intelligence, and emotionality. It is therefore unclear a) whether effects of the similarity manipulation on DVs are arising from the animal simply being described as 'similar' or 'dissimilar' to

humans, or from one or more of the four attributes specifically, and b), if arising from the specific attributes, exactly which attribute/s are contributing to effects of similarity. For example, differing mental characteristics of animals inform separate aspects of mind attribution and differentially affect moral judgements (e.g., moral acceptability of consuming the animal; Leach et al., 2021). Thus, the four attributes may feasibly have differing effects on perceptions of, and behavioural intentions towards, animals. Future research should test whether effects of similarity are arising from 'similar' vs. 'dissimilar' labelling or from specific attributes/s, by manipulating whether a target animal is presented as 'similar' or 'dissimilar' to humans across the four attributes individually and without mention of any attributes.

Secondly, the study is limited in explaining exactly how similarity causes pet speciesism and would benefit from further research exploring possible mediators of the effects of similarity on perceptions of animals. One such possible set of mediators is emotions. That is, the SCM/BIAS map operate through emotions like pity, contempt, pride, and envy with perceptions of human groups (Fiske et al., 2002; Cuddy et al., 2007; Fiske, 2018), whereby combinations of warmth and competence inform emotions, which in turn inform behavioural intentions. However, emotions towards animals sometimes differ from emotions felt towards humans (Sevillano & Fiske, 2016b). For instance, 'threatening-awe' animals (high in competence, low in warmth) are typically viewed with awe instead of, as would be expected for perceptions of humans, envy. Meanwhile, 'prey' animals (moderate in warmth, low in competence) are viewed with indifference instead of, as would be expected for perceptions of humans, pity (Sevillano & Fiske, 2016b). Future research should therefore explicitly measure the animal-applicable emotions (fondness, awe, indifference, and contempt; Sevillano & Fiske, 2016b), and test if they mediate the relationships between (dis)similarity and perceptions of, and behavioural intentions towards, animals found here.

This research would a) further elucidate the process of how and why similarity causes pet speciesism, b) enable stronger similarity interventions to be developed which could reduce pet speciesism (e.g., by developing similarity interventions which elicit stronger positive emotions and thereby more positive behavioural intentions; also see Neves et al., 2022), and c) contribute to emerging literature on emotions towards animals in the context of the SCM/BIAS map. For example, similarity of an animal to humans may enhance active help towards the animal not only due to increased warmth, but also through the emotion of fondness, whereby greater fondness mediates the relationship between greater warmth (following similarity information) and greater active help.

6.5.4.3. Conclusion.

Using the SCM/BIAS map, the current study determined causal effects of similarity on perceptions of, and behavioural intentions towards, dogs and pigs for the first time, evidencing that similarity information contributes to positive perceptions of animals' warmth and competence, alongside increasing active help and decreasing active harm towards animals. Further, the current study found that dissimilarity decreases animals' competence, and again demonstrates that dogs are deemed more similar to humans than pigs are.

This study has limitations regarding imprecision in the similarity manipulation, and lack of consideration of emotions as mediators. Future research should therefore determine exactly which components of the similarity manipulation cause similarity's positive effects, and measure and test mediational roles of emotions in the relationships between perceptions of animals and behavioural intentions.

Overall, the similarity studies (Studies 4-5) indicate (dis)similarity informs the pet speciesism gaps in warmth, competence, active help, and active harm. Specifically, similarity causes pet speciesism in these domains in two main ways. Firstly, dogs' (vs. pigs') increased

similarity informs dogs' greater warmth and competence, alongside informing greater active help and lower active harm towards dogs as compared to pigs. Secondly, pigs' (vs. dogs') lower similarity informs pigs' lower warmth and competence, alongside informing lower active help and greater active harm towards pigs compared to dogs. Conversely, dissimilarity contributes to pet speciesism through decreasing competence, whereby pigs' greater perceived dissimilarity to humans informs pigs' (vs. dogs') lower competence. Unlike all other SCM/BIAS map dimensions, similarity may be unable to explain pet speciesism gaps in passive help and passive harm, due to null effects with known animals in this study.

To conclude, the current similarity studies are the first to causally test the role of similarity to humans in perceptions of and behavioural intentions towards animals, and the findings expand and add complexity to previous literature. Similarity consistently causes pet speciesism in warmth, competence, and active help, whereby dogs' greater perceived similarity to humans contributes to their greater warmth and competence and greater active help towards them, whilst pigs' lower perceived similarity to humans contributes to their lower warmth and competence and lower active help towards them.

6.6. Overall Discussion

Through three pilot studies, an alternative familiarity manipulation based on the mere exposure effect failed to effectively manipulate familiarity. As such, familiarity was not explored any further and it can therefore not be ascertained if familiarity is a cause of pet speciesism. However, *similarity* was applied to pet speciesism specifically within Study 5 as the similarity manipulation used within Study 4 had been effective in manipulating similarity. Overall, Study 5 found that similarity is a cause of pet speciesism in the dimensions of warmth, competence, and active help.

Chapter 7. Pilot Study and Study 6 Exploring Pet Status

7.1. Introduction

The current chapter first presents a pilot study (Pilot Study 1c) to develop and pilot a pet status manipulation, before then implementing this manipulation within Study 6 in this chapter, to test if pet status is a cause of pet speciesism. These studies are now reported in turn below.

7.2. Pilot Study 1c to Test Pet Status Manipulation

7.2.1. Introduction

As Study 1 found correlational relationships between pet status and dogs' and pigs' warmth and competence, I aimed to test if these relationships are causal, and therefore if pet status can explain pet speciesism. However, before testing for this causal role, I first needed to develop and pilot a pet status manipulation which effectively manipulates pet status. As such, the current pilot study (Pilot Study 1c) therefore aims to test if a pet status manipulation developed by the current researcher is effective at manipulating its intended variable. Pet status is manipulated through text describing a target animal as either being 'typically kept as a pet animal' (pet status condition) or 'typically not kept as a pet animal' (lack of pet status condition), taking a similar approach to previous categorisation manipulations which have utilised simple labels like 'edible'/'not edible' or 'food'/'not food' (e.g., Bastian et al., 2012a; Bilewicz et al., 2016; Bratanova et al., 2011). As with Studies 2-4, an unknown animal (either tree kangaroo, fossa, or tamandua) is utilised as the target animal, as an adapted version of the novel animal paradigm to assess unmoderated effects of familiarity. To check effectiveness of the pet status manipulation (manipulation check), the item 'How much do you perceive the species in the photograph to be a 'pet' animal (an animal that is kept within a household as a companion)?' is included to measure pet status. Utilising a similar approach

to previous categorisation manipulations (Bastian et al., 2012a; Bilewicz et al., 2016; Bratanova et al., 2011), I hypothesise that:

H1: Stating that the animal is typically kept as a pet animal (vs. not) in a text manipulation will significantly increase the animal's pet status.

Note that this pilot study utilises the same participant sample to also follow-up/pilot text manipulations for behavioural self-relevance, subjective self-relevance, familiarity, and similarity, but only information relevant to the pet status manipulation is reported within the current chapter (see Section 4.2. for pilot study information about the behavioural and subjective self-relevance manipulations [Pilot Study 1a], and Section 5.2. for pilot study information about the familiarity and similarity manipulations [Pilot Study 1b]).

7.2.2. Methods

7.2.2.1. Participants.

Sixty-eight participants were recruited via social media and Sona as a volunteer sample. Thirteen participants were excluded for providing partial data. As all participants stated they were either a man or woman, no participants had to be excluded to enable gender to be dummy coded for inclusion of gender as a covariate (see Section 7.2.3. for further discussion). Exclusions left a total sample size of 55 participants (81.8% women, 18.2% men; M_{age} =23.38, SD_{age} =5.68, age range: 18-48), which exceeds the recommend 12 participants per condition for pilot studies (Julious, 2005). Participants within each condition are shown in Table 33. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. For dietary group: 69.1% (meat consumer), 12.7% (reducetarian), 7.3% (vegetarian), and 5.5% each (flexitarian, pescatarian). For nationality: 54.5% (British), 5.5% (Romanian), 3.6% each (German, Indian, Lithuanian, no response, Portuguese, Spanish), and 18.4% other. For ethnicity: 76.1% (White), 12.6% (Asian), 3.6% each (mixed, would rather not say), and 4.1% other. For country of residence ($M_{duration}$ =17.91, $SD_{duration}$ =10.12, range: 1-48 years): 81.8% (United Kingdom), 3.6% (Germany), and 14.6% other. For religion: 29.1% (atheism), 21.8% (Christianity), 18.2% (agnosticism), 14.5% (would rather not say), 5.5% (no response), 3.6% each (Buddhism, Islam), and 3.7% other.

Table 33.

Participants Within Each Condition for Pilot Study 1c.			
Pet Status Condition	Number of Participants		
Pet	29		
Non-Pet	26		

7.2.2.2. Design.

The current pilot study follows a one-way (pet status: pet vs. non-pet) betweensubjects ANCOVA design, with measured pet status as the DV. Pet status was included as a DV to test effectiveness of the manipulation. Gender was included as a covariate (see Section 7.2.3. for further discussion).

7.2.2.3. Materials.

7.2.2.3.1. Pet Status Manipulation.

Pet status was manipulated via texts developed by the researcher, whereby participants read that '*tree kangaroos/fossas/tamanduas* [dependent on target animal] *are typically [pet status condition] / not typically [lack of pet status condition] kept as a pet animal.*"

7.2.2.3.2. Measured Pet Status.

Perceived pet status was measured via the same single item from Study 1, except 'How much do you perceive the following animals to be a 'pet' animal (an animal that is kept within a household as a companion)?' was reworded to 'How much do you perceive tree *kangaroos/fossas/tamanduas to be a 'pet' animal (an animal that is kept within a household as a companion)?'*. Additionally, this item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score on measured pet status indicates greater perceived pet status.

7.2.2.3.3. Attention and Memory Checks.

Attention paid to and memory for the text manipulations was tested through five attention and memory check items developed by the researcher as follows: '*What animal did the text that you just read refer to*?', '*What type of animal did the text that you just read mention*?', '*Did the text that you just read state that tree kangaroos/fossas/tamanduas are or are not typically considered to be pet animals*?', '*Did the text that you just read mention a specific culture*?', and '*Did the text that you just read mention pest animals*?'. These checks aimed to test that participants paid attention to and remembered the texts. Correct answers for each item received a score of one, whilst incorrect answers for each item received a score of these checks revealed that the vast majority of the sample (>75%) had perfect scores, indicating sufficient attention and memory for these manipulations.

7.2.2.3.4. Believability of the Pet Status Manipulation (Beltramini, 1982; Beltramini & Evans, 1985; Chang, 2011).

Perceived believability of the pet status text manipulation was measured in order to ensure the text was not simply disregarded. Believability was measured through a six-item semantic differential believability measure (Beltramini, 1982; Beltramini & Evans, 1985; α =.94 from Chang, 2011), whereby participants answer how much they perceive the text to be 'unbelievable vs. believable', 'untrustworthy vs. trustworthy', 'not credible vs. credible', 'unreasonable vs. reasonable', 'not convincing vs. convincing', and 'biased vs. unbiased'. Reliability was adequate for this scale within this study, ω =.89. Therefore, items were summed together to create a total believability score. Higher scores indicate greater believability. No items are reverse-scored. The majority of the sample (>50%) perceived the pet status manipulation to be believable.

7.2.2.4. Procedure.

Participants were recruited via social media and Sona. The experiment was conducted in September and October 2020 as a voluntary, open survey. After providing informed consent, participants were asked if they recognised tree kangaroos (alongside a photograph of the animal), followed by a fossa and tamandua if participants recognised any previous animal. Animal species names were used throughout the study. The study either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected towards the end of the study and did not participate further.

Following initial recognition stage, participants completed the three sections (selfrelevance, familiarity/similarity, and pet status) of the pilot study in a randomised order. Here, only the pet status section is described (see Section 4.2. for description of the selfrelevance section, and Section 5.2. for description of the familiarity and similarity section). For the pet status section, participants were randomly assigned to read one of the two pet status text manipulations described in Section 7.2.2.3.1. regarding their target animal. After reading their pet status text, participants answered the five attention and memory checks, completed the measures for pet status (manipulation check), and then indicated their perceived believability of the text. Finally, participants provided demographics before being debriefed. One participant reported technical difficulties, but their response was complete and maintained within analyses.

7.2.3. Analyses

A one-way (pet status: pet vs. non-pet) between-subjects ANCOVA, including gender as a covariate, was conducted on measured pet status, to test if the manipulation effectively manipulates pet status. Gender was again included as a covariate within this ANCOVA. Although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analysis by controlling for and partialling out any relationship between gender and the outcome variable (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and measured pet status (the outcome variable) to be statistically controlled for.

7.2.3.1. Statistical Assumptions to Test H1: Exploring Effects of Pet Status Manipulation on Measured Pet Status (One-Way ANCOVA).

Measured pet status failed normality in the lack of pet status condition, p < .05, whilst the normality assumption was met as assessed through a normality test in the pet status condition, p > .05. Skewness was acceptable in the pet status condition, but was extreme in the lack of pet status condition. Additionally, measured pet status had some extreme univariate outliers in the lack of pet status condition, and no univariate outliers in the pet status condition. Running the ANCOVA including and excluding the univariate outliers made skewness acceptable in the lack of pet status condition and did not change univariate conclusions. I report this ANCOVA including outliers to reflect the unamended dataset. Homogeneity of variances was not present, p = .01.

7.2.3.2. Testing H1: Exploring Effects of Pet Status Manipulations on Measured Pet Status.

A one-way (pet status: pet vs. non-pet) between-subjects ANCOVA, including gender as a covariate, was run on measured pet status to test **H1. Supporting H1:** There was a significant main effect of manipulated pet status on measured pet status, F(1, 52) = 13.73, p < .001, $\eta_p^2 = .21$ (large-sized). That is, the pet status condition resulted in participants viewing the animal as having greater pet status (*M*=42.31, *SD*=34.51) compared to participants in the no pet status condition (*M*=10.23, *SD*=21.17).

7.2.4. Discussion and Conclusion

As expected, the pet status condition led to greater perceived pet status than the lack of pet status condition, thereby supporting **H1**. Therefore, the current study demonstrates that the pet status manipulation developed by the researcher is effective at manipulating pet status. Informed by these findings, the pet status manipulation is utilised within Study 6 to test causal effects of pet status on pet speciesism. As the pet status manipulation was found to be effective in manipulating pet status within the current pilot study, this manipulation is mostly unchanged within Study 6. However, the manipulation has been slightly amended to attempt to enhance effectiveness of the manipulation. Specifically, in order to enhance legitimacy of the texts and appeal to authority (Walton, 2010), the phrases '*imagine reading the following sentence in the newspaper*' and '*according to experts*' are included at the beginning of the text within Study 6. Additionally, participants are asked to imagine they read this text in the newspaper as a way to make the texts seem more realistic (e.g., akin to Leach et al., 2021).

7.3. Study 6: Causal Exploration of Pet Status

7.3.1. Introduction and Extension from Study 1

Study 1 found that pet status (how much an animal is deemed to be a companion animal) had a significant positive relationship with dogs' and pigs' warmth and competence. These findings agree with previous literature which has found that people view pet animals more positively than non-pet animals (Signal et al., 2018; Taylor & Signal, 2009), whereby the simple labelling of an animal as a 'pet' engenders positive perceptions of the animal. These findings are underpinned by 'categorisation' theory, whereby the mere categorisation of an animal as a 'pet' informs how people perceive the animal (Bratanova et al., 2011).

The current study aims to extend Study 1 findings by exploring if pet status causes animals' warmth and competence, through causally manipulating the pet status of an unknown animal, whereby pet status is manipulated by text piloted in Pilot Study 1c (Section 7.2.) which describes a target animal as either being '*typically kept as a pet animal*' (pet status condition) or '*typically not kept as a pet animal*' (lack of pet status condition). As with Studies 2-4, the target animal is an unknown animal, which is again utilised as an adapted version of the novel animal paradigm to assess unmoderated effects of pet status. Warmth and competence are again measured within the current study using the SCM subscales from Sevillano and Fiske (2016b). In line with Study 1 findings whereby greater pet status was associated with greater warmth and competence, and in line with previous literature evidencing the positive effects of pet status (Signal et al., 2018; Taylor & Signal, 2009), I therefore hypothesise that:

H1: Animals portrayed as pet animals (vs. not) will be deemed significantly warmer(a) and more competent (b).

Like Studies 2-4, the current study also extends possible effects of pet status from warmth and competence to behavioural intentions. Specifically, warmth should inform greater active help and less active harm, whilst competence should inform greater passive help and less passive harm (Cuddy et al., 2007). Thus, per the BIAS map, relationships between pet status and warmth and competence from Study 1 should extend to causal effects on active and passive behavioural intentions within this study, whereby behavioural intentions are again measured within the current study using the BIAS map subscales from Sevillano and Fiske (2016b). As such, I predict that:

H2: If animals portrayed as pet animals (vs. not) are deemed significantly warmer (i.e., H1a is evidenced), 'pet' (vs. non-'pet') animals will be subject to significantly less active harm (a) and more active help (b). If animals portrayed as pet animals (vs. not) are deemed significantly more competent (i.e., H1b is evidenced), 'pet' (vs. non-'pet') animals will be subject to significantly less passive harm (c) and more passive help (d).

Finally, inclusion of measurements for behavioural intentions within this study again enables BIAS map applicability to animals to be tested (Sevillano & Fiske, 2016b, 2019). That is, this study can determine directly if any effects of pet status on active and passive behavioural intentions are mediated through warmth and competence respectively, in line with both general SCM and BIAS map literature (Cuddy et al., 2007) and SCM/BIAS map literature applied to animals specifically (Sevillano & Fiske, 2016b, 2019). Therefore, I hypothesise that:

H3: Post-manipulation warmth will significantly mediate relationships between pet status and active harm and active help. That is, animals portrayed as pet animals (vs. not) will be deemed significantly warmer post-manipulation, significantly increasing post-manipulation active help (**a**) and decreasing post-manipulation active harm (**b**).

H4: Post-manipulation competence will significantly mediate relationships between pet status and passive harm and passive help. That is, animals portrayed as pet animals (vs. not) will be deemed significantly more competent post-manipulation,
significantly increasing post-manipulation passive help (\mathbf{a}) and decreasing postmanipulation passive harm (\mathbf{b}) .

7.3.2. Methods

7.3.2.1. Participants.

One-hundred-and-ninety-six participants were recruited via volunteer sampling either via social media and Sona, or via Prolific. Qualtrics quota sampling was implemented to ensure approximately 50% men and 50% women, alongside representative numbers of people of different dietary identities in the sample in proportion to the population: approximately 86% meat consumers, 6% vegetarians, 5% pescatarians, and 3% vegans in the United Kingdom population (Johnson, 2022). As with Study 4, people with a flexitarian diet did not have a specific quota but were still included within the sample. For participants recruited via Prolific, quota sampling was partially used through Prolific's 'balance by sex' option to obtain approximately equal numbers of men and women. Sixty-eight participants were excluded for: exceeding quotas (n=41), providing partial data (n=23), withdrawing their data (n=1), or recognising all animals (n=1). As all analyses included gender as a covariate (see Section 7.3.3. for further discussion), two non-binary participants were excluded to enable gender to be dummy coded into women (coded as zero; n=64) and men (coded as one; n=64). Exclusions left a sample of 128 participants (50% women, 50% men; M_{age} =24.46, $SD_{age}=6.01$, age range=18-54), which met the minimum required sample size of 128 per a G*Power a priori power analysis for a MANOVA analysis ('repeated measures, withinbetween interaction', medium effect size f=.25, two groups, two measurements, $\alpha=.05$, power = .8).

One-hundred-and-twenty participants completed the study about tree kangaroos, whilst eight participants recognised the tree kangaroo and completed the study about fossas (n=7) or tamanduas (if they recognised the fossa; n=1). Participants within each condition are

shown in Table 34. Demographics of the sample are reported as follows in percentages, with demographic categories which make up less than 2% of the sample being collapsed into 'other'. The percentage of participants for each diet approximately matches sizes within the general population. For dietary group: 85.9% (meat consumer), 6.3% (vegetarian), 4.7% (pescatarian), 2.3% (vegan), and .8% other. For nationality: 32.9% (United Kingdom/British), 7.8% (Polish), 5.5% (Canadian), 3.9% (Malaysian), 3.1% each (Italian, Mexican, Nigerian, Portuguese, South African, United States), 2.3% each (Chinese, German, Greek, Latvian), and 22.1% other. For ethnicity: 61.1% (White), 14.8% (Asian), 9.4% (mixed), 7.9% (Black), 3.1% each (Arab, Hispanic/Latino), and .6% other. For country of residence (*M*_{duration}=18.5, *SD*_{duration}=9.6, range: one to 48 years): 46.5% (United Kingdom), 6.3% (United States), 5.5% each (Canada, Poland), 4.7% (Malaysia), 3.1% each (Portugal, South Africa), 2.3% each (Australia, Germany, Mexico, the Netherlands), and 15.8% other. For religion: 37.6% (Christianity), 26.6% (atheism), 10.2% (agnosticism), 8.6% (would rather not say), 4.7% each (Buddhism, Islam), 3.1% (no response), 2.3% (no religion), and 2.2% other.

Table 34.

Participants	Within	Each	Condition	for	Study 6.

Pet Status Condition		Number of Participants		
Present		66		
Absent		62		

7.3.2.2. Design.

This experiment follows a 2(pet status: pet vs. non-pet; between-subjects) x 2(time: pre- vs. post-manipulation; within-subjects) mixed MANCOVA design, with warmth, competence, active help, passive help, active harm, and passive harm as the DVs. Pet status was included as a DV in a separate ANCOVA to test effectiveness of the manipulation. Gender and neutrality were included as covariates (see Section 7.3.3. for further discussion).

7.3.2.3. Materials.

7.3.2.3.1. Target Animals and Animal Photographs.

The current study again employed the novel animal paradigm by implementing the same photographs of the animals (tree kangaroo, fossa, tamandua) from Studies 2 and 4, whereby the tree kangaroo photograph was sourced from

https://www.shutterstock.com/image-photo/close-shot-cute-tree-kangaroo-168261035, the fossa photograph sourced from https://www.flickr.com/photos/mathiasappel/19504925051, and the tamandua photograph sourced from https://tinyurl.com/2jazjz3s. These photographs enable easier visualisation of the imagined scenario and reduce biased perceptions of the animal due to species name.

7.3.2.3.2. Pet Status Text Manipulations.

Pet status was manipulated via texts developed and successfully piloted by the researcher (Pilot Study 1c; Section 7.2.). Specifically, participants were asked to '*Imagine reading the following sentence in the newspaper*', and then read: "*According to experts, the species in the photograph is typically [pet status condition] / not typically [lack of pet status condition] kept as a pet animal.*" This pet status manipulation was amended slightly from Pilot Study 1c, in order to enhance legitimacy of the texts and appeal to authority (Walton, 2010) by including the phrases '*imagine reading the following sentence in the newspaper*' and '*according to experts*' at the beginning of the text within the current study. Additionally, participants were asked to imagine they read this text in the newspaper as a way to make the texts seem more realistic. For instance, previous research has successfully manipulated perceptions of animals by presenting information in a scientific article format (Leach et al., 2021). Participants read the text for at least one minute to ensure sufficient attention.

7.3.2.3.3. Warmth and Competence (Sevillano & Fiske, 2016b).

Warmth and competence were measured with the same items from Studies 1-5: '*warm*', '*well-intentioned*', '*friendly*' (warmth), and '*competent*', '*skillful*', '*intelligent*' (competence; Sevillano & Fiske, 2016b). Like Studies 3-5, these items were measured using a semantic differential from -3 to 3, with higher scores indicating greater warmth or competence. Items within each subscale were summed to provide warmth and competence scores. Reliability was adequate within this study for warmth (pre-manipulation ω =.76, 95% BCa CI [.67, .82]; post-manipulation ω =.86, 95% BCa CI [.8, .91) and competence (premanipulation: ω =.81, 95% BCa CI [.71, .87]; post-manipulation: ω =.89, 95% BCa CI [.84, .93]) across timepoints.

7.3.2.3.4. Behavioural Intentions Towards the Animal (Sevillano & Fiske, 2016b).

Behavioural intentions were measured with the same items from Studies 2-4: *'support', 'help', 'behave friendly towards', 'interact with'* the animal (active help), *'sustain', 'conserve'* the animal (passive help), *'kill', 'injure', 'exterminate', 'trap', 'reject'* the animal (active harm), and *'let the species die off', 'ignore'* the animal (passive harm; Sevillano & Fiske, 2016b). Like Studies 3-5, these items were measured using a semantic differential from -3 to 3, with higher scores indicating greater active help, active harm, passive help, or passive harm. Items within each subscale were summed to provide active help, passive help, active harm, and passive harm scores. Reliability was adequate within this study for passive help (pre-manipulation r_{sb} =.74; post-manipulation ω =.86, 95% BCa CI [.76, .91]) across timepoints. Whilst reliability was adequate for active help postmanipulation (ω =.76, 95% BCa CI [.66, .84]), reliability was inadequate for active help premanipulation (ω =.67, 95% BCa CI [.54, .76]), and for passive harm across timepoints (premanipulation: r_{sb} =.57; post-manipulation: r_{sb} =.48). As the passive harm subscale contains

two items, I did not run a PAFA for this subscale. Due to inadequate reliability of the active help subscale at pre-manipulation, I ran a PAFA (Appendix 2h) on this subscale at premanipulation only with direct oblimin rotation to test factorial validity, alongside assessing inter-item correlations to test reliability and communality values to test common variance. The PAFA revealed adequate loading of all items onto the active help subscale (\geq .5), supporting factorial validity, alongside adequate communality for the first item (.41), and adequate (though not ideal) communalities for all other items (\geq .25; Child, 2006). Except for an inadequate inter-item correlation between item three and item four, *r*=.29, all other interitem correlations to assess reliability were adequate, .31 \leq *r*s \leq .43. I therefore decided to retain the third and fourth items due to their adequate inter-item correlations with all other times, and due to adequate reliability of the active help subscale post-manipulation. Due to inadequate reliability of the passive harm and active help subscales at one or both timepoints, findings from these subscales should be interpreted with caution.

7.3.2.3.5. Perceived Neutrality Towards the Animal.

Neutrality was measured via the same single item utilised in Studies 3-5, with a higher score indicating higher neutrality and less bias: '*How neutral do you perceive the species in the previous photograph to be*?' from zero to 100.

7.3.2.3.6. Perceived Pet Status (Manipulation Check).

Perceived pet status was measured via the same single item from Study 1, except 'How much do you perceive the following animals to be a 'pet' animal (an animal that is kept within a household as a companion)?' was reworded to 'How much do you perceive the species in the photograph to be a 'pet' animal (an animal that is kept within a household as a companion)?' to avoid participant bias from hearing the species name. Additionally, this item was re-scaled from a one to five Likert scale used in Study 1 to a zero to 100 visual analogue scale here to ensure parametric data. A higher score on measured pet status indicates greater perceived pet status.

7.3.2.4. Procedure.

Participants were recruited via avenues like social media and Sona (*n*=95), or via Prolific (*n*=33; Appendix 10). All participants were required to not have taken part in previous studies within this thesis. The experiment was conducted in March 2021 and November 2021 as a voluntary, open survey. After providing informed consent, participants gave demographic information, with participants exceeding diet and/or gender quotas being automatically excluded from participating further by Qualtrics. Remaining eligible participants were presented with a photograph of a tree kangaroo and asked if they recognised the animal, followed by a fossa and then tamandua if participants recognised any previous animal. Animal species names were not used during the study. The experiment either proceeded using the unrecognised animal as the target animal, or, if participants recognised all three animals, they were redirected towards the end of the experiment and did not participate further. One participant recognised all animals and was therefore excluded from the study. After initial recognition, participants indicated their perceived neutrality towards their target animal, the animal's warmth and competence, and behavioural intentions towards the animal (all scales and items within scales presented in randomised order).

Participants were subsequently randomly assigned into one of the two pet status conditions (pet status vs. lack of pet status) and read the text within their condition as described in Section 7.3.2.3.2. After reading their assigned text, participants rated the animal's perceived pet status, followed by again rating the animal's warmth and competence and behavioural intentions towards the animal (all scales and items within scales presented in randomised order), before being debriefed. Eight participants reported technical difficulties, but responses were complete and maintained within analyses. ARU undergraduate

psychology students (n=2) received 0.25 Sona credits as reimbursement. All other participants not recruited via Prolific could enter a prize draw to win of two £50 Amazon gift vouchers if they wished. For participants recruited via Prolific, participants received £1 via Prolific as reimbursement for full responses. Participants who provided partial data or who recognised all animals were reimbursed a proportion of the £1 based either on their percentage of completion (partial data) or time spent completing the survey (recognition), whilst adhering to Prolific's minimum hourly rate.

7.3.3. Analyses

A 2(pet status: pet vs. non-pet; between-subjects) x 2(time: pre- vs. postmanipulation; within-subjects) mixed MANCOVA was conducted on warmth, competence, active help, passive help, active harm, and passive harm. As all of the DVs were conceptually related (i.e., from the SCM/BIAS map and subscales of the same overall scales), alongside many DVs being statistically related through moderate correlations, $rs \ge .3$, MANCOVA was deemed appropriate for this analysis. Additionally, a one-way (pet status: pet vs. non-pet) between-subjects ANCOVA was run on measured pet status.

Gender was again included as a covariate within the analyses. Again, although the current study randomly allocates participants to conditions, and therefore gender is approximately evenly balanced across conditions, inclusion of this covariate still increases power of the analyses by controlling for and partialling out any relationship between gender and the outcome variables (Tabachnick & Fidell, 2014). Therefore, as gender has previously been found to impact perceptions of animals (e.g., Apostol et al., 2013; Herzog, 2007), including gender as a covariate allows for any theoretically-informed relationships between gender and warmth/competence perceptions of animals and behavioural intentions towards them (the outcome variables) to be statistically controlled for. Neutrality was also included within analyses as a covariate to account for bias in perceptions of animals. That is, by

including neutrality as a covariate, the statistical model is then adjusted to account for this variable, thereby reducing variability and bias in the findings (Keen & Tiemeier, 2022), and increasing power of the analyses by statistically controlling for any possible relationship between neutrality and perceptions of/behavioural intentions towards animals (Tabachnick & Fidell, 2014).

Descriptive statistics for significant discussed main effects or for *post hoc* pairwise comparisons for significant interaction effects are reported in Tables 35-37.

Finally, one-tailed one-sample *t*-tests were conducted on warmth, competence, and behavioural intentions (like Studies 3-4).

7.3.3.1. Statistical Assumptions to Test H1-H2: Exploring Effects of Pet Status on SCM/BIAS Map Variables (2x2 MANCOVA).

The DVs frequently failed normality tests, ps < .05, but all DVs, except active harm, had acceptable skewness. Due to excessive skewness, multiple extreme univariate outliers, and floor effects in active harm, I also ran a MANCOVA excluding active harm as a DV, which changed one minor multivariate conclusion (Footnote 28). I therefore report the MANCOVA including active harm as a DV.

All DVs except active harm had univariate outliers which did not cause excessive skewness. There were multivariate outliers for six DVs, critical values ≥ 22.46 , p < .001. Running the MANCOVA including and excluding univariate and multivariate outliers was not possible due to floor effects in active harm. Running the MANCOVA including and excluding univariate and multivariate outliers without active harm as a DV indicated multivariate conclusions did change (Footnote 29). However, I report the MANCOVA including univariate and multivariate outliers to reflect the unamended dataset, and also because the dataset excluding univariate and multivariate and multivariate outliers was underpowered

(*n*=100). There was no multicollinearity, $rs \le .71$. Linear relationships between the DVs were mostly present across all conditions.

Running analyses including and excluding outliers did not change conclusions for competence and active help. Excluding outliers changed one minor conclusion for warmth (Footnote 30) and passive help (Footnote 32), and one main conclusion for passive harm (Footnote 31). I report all analyses including outliers to reflect the unaltered dataset. Excessive skewness, multiple univariate outliers, and floor effects in active harm meant assumptions for the ANCOVA on this DV were violated. However, I proceeded with this ANCOVA due to robustness of ANCOVA to non-normality, and lack of non-parametric alternatives. Results from this ANCOVA should be interpreted with caution.

Homogeneity of variances was present for all DVs across timepoints, ps > .05, except for passive help at pre-manipulation, p < .05. Homogeneity of covariances was present, p = .01.

7.3.3.2. Statistical Assumptions for Pet Status Manipulation Check: Exploring Effect of Manipulated Pet Status on Measured Pet Status (One-Way ANCOVA).

Measured pet status failed normality tests, ps < .05, but skewness was acceptable. There was one outlier in measured pet status in the lack of pet status condition. Running the analysis including and excluding this outlier did not change conclusions. Homogeneity of variances was present, > .05.

7.3.3.3. Manipulation Check: Exploring Effect of Manipulated Pet Status on Measured Pet Status.

A one-way between-subjects ANCOVA, including gender and neutrality as covariates, was conducted on pet status to test effectiveness of the pet status text in manipulating measured pet status. An ANCOVA was conducted for this manipulation check instead of an independent-samples *t*-test to enable inclusion of gender and neutrality as covariates (see Table A10 in Appendix 4 for full inferential statistics). There was a significant effect of the pet status manipulation on measured pet status, F(1, 123) = 5.06, p =.03, $\eta_p^2 = .04$ (small-sized), in the expected direction. Specifically, participants in the pet status condition reported greater perceived pet status of the animal (*M*=34.21, *SD*=29.55) compared to participants in the no pet status condition (*M*=21.36, *SD*=23.62). Thus, the pet status text was effective at manipulating pet status. Beyond the manipulation check, there was a significant main effect of neutrality on measured pet status, F(1, 123) = 6.28, p = .01, $\eta_p^2 =$.05 (small-to-medium-sized), whereby greater neutrality was linked to greater perceived pet status, B = .27, SE = .11.

7.3.3.4. Testing H1-H2: Exploring Effects of Pet Status on SCM/BIAS Map Variables.

A 2(pet status: pet vs. non-pet) x 2(time: pre- vs. post-manipulation) mixed MANCOVA, including gender and neutrality as covariates, was run on warmth, competence, active harm, active help, passive harm, and passive help to test **H1-H2** (see Table A11 in Appendix 4 for full inferential statistics).

7.3.3.4.1. Multivariate Effects of Pet Status on Combined DVs.^{28, 29}

There was a significant interaction effect between time and condition on the combined DVs, F(6, 119) = 2.46, p = .03, $\eta_p^2 = .11$ (medium-sized). There was also a significant main

²⁸When excluding active harm from the MANCOVA, an interaction between time and neutrality on the combined DVs became significant, F(5, 120) = 2.31, p = .048, $\eta p = .09$ (medium-sized), whereby neutrality had a significant multivariate effect on the combined DVs pre-manipulation, B = .04, SE = .01, p = .004, but not post-manipulation, B = .03, SE = .02, p = .07.

²⁹When excluding univariate and multivariate outliers, the interaction effect between time and condition on the combined DVs became non-significant, F(5, 92) = 2.23, p = .06, $\eta_p^2 = .11$. Additionally, an interaction effect between time and neutrality on the combined DVs became significant, F(5, 92) = 3.02, p = .01, $\eta_p^2 = .14$ (large-sized), and a main effect of time on the combined DVs also became significant, F(5, 92) = 2.86, p = .02, $\eta_p^2 = .13$ (medium-to-large-sized). However, the sample size excluding all univariate and multivariate outliers was highly underpowered (*n*=100), so these results should be interpreted with caution.

effect of neutrality on the combined DVs, F(6, 119) = 2.6, p = .02, $\eta_p^2 = .12$ (medium-tolarge-sized). All other multivariate interactions and main effects were non-significant, ps >.05. I followed up significant multivariate effects with univariate ANCOVAs on the DVs below.

7.3.3.4.2. H1a: Univariate Effects of Pet Status on Warmth.³⁰

There was a significant interaction between time and pet status on warmth, F(1, 124)= 10.03, p = .002, $\eta_p^2 = .08$ (medium-sized). Agreeing with H1a: *Post hoc* pairwise comparisons revealed that, post-manipulation only, the presence of pet status increased warmth compared to when pet status was absent, p = .03, d = .4 (small-sized). As would be expected, pet status conditions did not significantly differ at pre-manipulation, p = .54. Additionally, the presence of pet status significantly increased warmth from pre-manipulation to post-manipulation, p = .01, d = .34 (small-sized), whereas, when pet status was absent, warmth did not significantly differ from pre-manipulation to post-manipulation, p = .09. There was also a significant main effect of neutrality on warmth, F(1, 124) = 7.09, p = .01, $\eta_p^2 = .05$ (small-to-medium-sized), whereby greater neutrality was linked to greater warmth pre-manipulation, B = .04, SE = .01, p = .004, but not post-manipulation, B = .03, SE = .02, p = .07.

Descriptive Statistics for Interaction Effect of Pet Status Condition and Time on Warmth.				
Pet Status	Pre-	Pre-	Post-	Post-
Condition	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Present	3.09	.45	4.28	.48
Absent	3.49	.46	2.73	.5

Table 35.

³⁰When excluding univariate outliers, the significant main effect of neutrality on warmth remained significant, F(1, 121) = 12.52, p < .001, $\eta_p^2 = .09$ (medium-sized). As when including outliers, greater neutrality was linked to greater warmth pre-manipulation, B = .05, SE = .01, p = .0002, but this effect also became significant post-manipulation, B = .04, SE = .02, p = .02.

7.3.3.4.3. H1b: Univariate Effects of Pet Status on Competence.

Not supporting H1b: There was not a significant interaction between time and pet status on competence, F(1, 124) = 1, p = .32, $\eta_p^2 = .01$. There was, however, a significant main effect of neutrality on competence, F(1, 124) = 5.07, p = .03, $\eta_p^2 = .04$ (small-sized), whereby greater neutrality was linked to greater competence post-manipulation, B = .04, SE =.01, p = .01, but not pre-manipulation, B = .02, SE = .01, p = .11.

7.3.3.4.4. H2a: Univariate Effects of Pet Status on Active Harm.

There was a significant interaction between time and pet status on active harm, F(1,124) = 4.89, p = .03, $\eta_p^2 = .04$ (small-sized). Partially consistent with H2a: Post hoc pairwise comparisons found that the presence of pet status significantly decreased active harm from pre-manipulation to post-manipulation, p = .04, d = .25 (small-sized). When pet status was absent, active harm did not significantly differ from pre-manipulation to postmanipulation, p = .28. Not supporting H2a: Post hoc pairwise comparisons revealed the pet status conditions did not significantly differ post-manipulation in active harm, p = .14. As would be expected, the pet status conditions did not significantly differ pre-manipulation in active harm, p = .69.

Table 36.

Present

Absent

Manipulation Manıpulatıon Manıpulatıon Manıpulatıon Condition

SE

.61

.63

Mean

-12.47

-11.2

SE .59

.61

Descriptive Sta	tistics for Interaction	on Effect of Pet Stat	us Condition and T	ime on Active
Harm.				
Pet Status	Pre-	Pre-	Post-	Post-
Condition	Manipulation	Manipulation	Manipulation	Manipulation

7.3.3.4.5. H2b: Univariate Effects of Pet Status on Active Help.

Mean

-11.41

-11.76

Not supporting H2b: There was not a significant interaction between time and pet status on active help, $F(1, 124) = 2.69, p = .1, \eta_p^2 = .02$.

7.3.3.4.6. H2c: Univariate Effects of Pet Status on Passive Harm.³¹

There was a significant interaction between time and pet status on passive harm, F(1, 124) = 4.43, p = .04, $\eta_p^2 = .03$ (small-sized). Agreeing with H2c: Post hoc pairwise comparisons found that, post-manipulation only, the presence of pet status significantly decreased passive harm compared to when pet status was absent, p = .03, d = .4 (small-sized). As would be expected, there was no significant difference in passive harm between the pet status conditions pre-manipulation, p = .88. Additionally, the presence of pet status significantly decreased passive harm from pre-manipulation to post-manipulation, p = .03, d = .28 (small-sized). When pet status was absent, passive harm did not significantly differ from pre-manipulation to post-manipulation, p = .41.

Table 37.

Descriptive Statistics for Interaction Effect of Pet Status Condition and Time on Passive Harm.

Pet Status	Pre-	Pre-	Post-	Post-
Condition	Manipulation	Manipulation	Manipulation	Manipulation
	Mean	SE	Mean	SE
Present	-3.41	.34	-4.06	.3
Absent	-3.34	.35	-3.08	.31

7.3.3.4.6. H2d: Univariate Effects of Pet Status on Passive Help.³²

Not supporting H2d, there was no significant interaction between time and pet status on passive help, F(1, 124) = 1.39, p = .24, $\eta_p^2 = .01$. There was, however, a significant main effect of neutrality on passive help, F(1, 124) = 7.08, p = .01, $\eta_p^2 = .05$ (small-to-medium-

³¹When excluding univariate outliers, the significant interaction between time and pet status on passive harm remained significant, F(1, 121) = 7.24, p = .01, $\eta_p^2 = .06$ (medium-sized), and most *post hoc* pairwise comparisons did not change. However, the presence of pet status no longer significantly reduced postmanipulation passive harm compared to when pet status was absent, p = .06.

³²When excluding univariate outliers, the significant main effect of neutrality on passive help remained significant, F(1, 122) = 4.8, p = .03, $\eta_p^2 = .04$. However, greater neutrality was now linked to greater passive help only at pre-manipulation, B = .03, SE = .01, p = 01, and was no longer significant post-manipulation, B = .02, SE = .01, p = .13.

sized), whereby greater neutrality was linked to greater passive help pre-manipulation, B = .04, SE = .01, p = .004, and post-manipulation, B = .03, SE = .01, p = .048.

7.3.3.5. Testing if Warmth/Competence Perceptions and Behavioural Intentions Towards the Animal Differ from Zero.

Twelve one-tailed one-sample *t*-tests were conducted on warmth, competence, active help, active harm, passive help, and passive harm at pre- and post-manipulation to test if perceptions of and behavioural intentions towards animals significantly differ from zero. These *t*-tests revealed all perceptions and behavioural intentions significantly differed from zero, ps < .001, whereby the unknown animal was deemed significantly warmer and more competent than zero across timepoints. Additionally, participants were significantly more willing to actively and passively help, and less willing to actively and passively harm, the animal compared to zero. Like Studies 3-4, these findings indicate a positivity bias towards the unknown animal.

7.3.3.6. Testing H3-H4: Exploring Mediational Relationships Between Pet Status and Behavioural Intentions Through Warmth/Competence

Per the BIAS map, I predicted warmth would mediate the relationship between pet status and post-manipulation active help (**a**) and post-manipulation active harm (**b; H3**). I also hypothesised competence would mediate the relationship between pet status and post-manipulation passive help (**a**) and post-manipulation passive harm (**b; H4**).

To test **H3-H4**, I dummy coded pet status into one dummy variable with lack of pet status coded as zero and pet status coded as one. Four individual boot-strapped mediation analyses (10,000 bootstrap samples) were run using Model 4 in PROCESS through SPSS (Hayes, 2022) to assess mediational relationships for the four behavioural intentions. To control for Type I error from multiple mediation testing, I utilised an adjusted confidence

interval of 99%. Significant mediations are interpreted through confidence intervals which do not contain zero per Hayes (2018).

As post-manipulation warmth and post-manipulation competence were significantly correlated with each other within the current study as revealed through a Pearson correlation, r = .58, p < .001, and warmth and competence are conceptually related, warmth and competence were included as parallel mediators within the below mediation analyses. Inclusion of warmth and competence as parallel mediators allows for testing of any mediational roles of warmth for active behavioural intentions *whilst* controlling for competence, and any mediational roles of competence for passive behavioural intentions *whilst* controlling for warmth (Hayes, 2018).

7.3.3.6.1. H3a: Mediation of Pet Status on Active Help through Warmth.

Not supporting H3a: There was no significant indirect effect of pet status on postmanipulation active help via post-manipulation warmth when controlling for postmanipulation competence, b = .99, SE = .45, 99% CI [-.007, 2.37]. Thus, post-manipulation warmth did not significantly mediate the relationship between pet status and postmanipulation active help. Despite lack of mediation, inspection of path *b* indicated greater post-manipulation warmth significantly predicted greater post-manipulation active help, b =.54, SE = .12, p < .001, 99% CI [.22, .87].

7.3.3.6.2. H3b: Mediation of Pet Status on Active Harm through Warmth.

Supporting H3b: There was a significant indirect effect of pet status on postmanipulation active harm via post-manipulation warmth when controlling for postmanipulation competence (Figure 13). Thus, post-manipulation warmth significantly mediated the relationship between pet status and post-manipulation active harm. This model explained 15.5% of variance in active harm ($R^2 = .155$). Additionally, inspection of path *b* indicated greater warmth significantly predicted less active harm (Figure 13).



Figure 13. Mediational model of pet status on post-manipulation active harm through post-manipulation warmth when controlling for post-manipulation competence. *Note:* Standard error is reported for indirect effect only. $**p \le .01$

7.3.3.6.3. H4a: Mediation of Pet Status on Passive Help through Competence.

Not supporting H4a: There was no significant indirect effect of pet status on postmanipulation passive help via post-manipulation competence when controlling for postmanipulation warmth, b = .03, SE = .19, 99% CI [-.45, .6]. Therefore, post-manipulation competence did not significantly mediate the relationship between pet status and postmanipulation passive help. Despite lack of mediation, inspection of path *b* indicated greater competence significantly predicted greater passive help, b = .29, SE = .09, p = .002, 99% CI [.05, .54].

7.3.3.6.4. H4b: Mediation of Pet Status on Passive Harm through Competence.

Not supporting H4b: There was no significant indirect effect of pet status on postmanipulation passive harm via post-manipulation competence when controlling for postmanipulation warmth, b = -.002, SE = .04, 99% CI [-.13, .16]. Therefore, post-manipulation competence did not significantly mediate the relationship between pet status and postmanipulation passive harm. Additionally, inspection of path *b* indicated competence did not significantly predict passive harm, b = -.01, SE = .07, p = .86, 99% CI [-.2, .17].

7.3.4. Discussion

7.3.4.1. Discussion of Findings.

7.3.4.1.1. H1-H2: Effects of Pet Status on SCM/BIAS Map Variables.

As predicted, and agreeing with previous literature (Signal et al., 2018), animals described as pets were deemed warmer (H1a), from pre- to post-manipulation, and as compared to animals described as not being pets (small-sized). Additionally, a lack of pet status did not impact warmth from pre- to post-manipulation, indicating pet status confers a benefit in improving animals' warmth, whilst the absence of pet status may not confer a disadvantage. However, the pet status effect on warmth is small-sized only, so may have lower practical significance.

Conversely, not supporting **H1b** and previous literature indicating positive effects of pet status (Signal et al., 2018), pet status did not significantly affect competence. Previous literature on the role of pet status is limited and has only investigated non-causal relationships between pet status, generalised attitudes towards animals (e.g., the Animal Attitude Scale; Herzog et al., 1991), and empathy (Hazel et al., 2011; Signal et al., 2018), without yet investigating competence or related variables. Thus, pet status may simply not inform perceptions of animals' competence specifically, and the lack of effect of pet status on competence may thereby represent a unique finding of the current study. Hypothetically, pet status may impact warmth and not competence as pet status may be linked to the assumption that pets are domesticated, and therefore friendly by nature (e.g., domestication can be linked to pro-sociality; Hare et al., 2002). This friendliness (good intent) would reflect in higher

warmth. Conversely, assumed domestication may not necessarily be linked to an increased ability for an animal to act on intentions, and may therefore not impact upon competence.

Not supporting **H2b** and **H2d**, there were no effects of pet status on active and passive help (although findings for active help should be interpreted with caution due to inadequate reliability). As warmth typically informs active help, the null effect on active help is unexpected considering the significant effect of pet status on warmth. Thus, the positive effect of pet status on warmth does not appear to translate into a (positive) effect of pet status on active help, thereby not supporting the BIAS map (Sevillano & Fiske, 2016b), and possibly indicating separability of SCM perceptions of animals (e.g., warmth) and BIAS map behavioural intentions (e.g., active help) in the context of pet status. Pet status may not have informed help intentions as a) pet animals may be assumed to receive help from their guardians, whilst b) non-pet animals may be assumed to not require human help. Thus, participants' perceptions of how much the animal requires help, combined with whether the animal is already receiving help from other people, may be impacting how willing participants are to provide help (see Section 7.3.4.2. for further discussion).

Regarding harm intentions, pet status caused less active harm and passive harm from pre- to post-manipulation (small-sized), supporting **H2a** and **H2c**. However, pet status was only superior to a lack of pet status in decreasing post-manipulation passive harm (small-sized; agreeing with **H2c**), whereas post-manipulation active harm did not differ across pet status conditions (partially not supporting **H2a**). A lack of pet status had no effect on active and passive harm from pre- to post-manipulation. Overall, therefore, pet status decreases passive harm from pre- to post-manipulation and compared to a lack of pet status, whilst pet status only decreases active harm from pre- to post-manipulation. However, both effects should be interpreted with caution due to floor effects (active harm) and inadequate scale reliability (passive harm). If these findings are robust, the lack of difference between the pet

status conditions in post-manipulation active harm may be due to floor effects in active harm. That is, participants were highly unwilling to actively harm the animal, regardless of whether the animal was a pet or not. However, it is still unclear why there was a pre- to postmanipulation reduction in active harm within the pet status condition. It may be that learning that an animal is a pet made participants even more unwilling to actively harm the animal at post-manipulation compared to pre-manipulation. The lack of floor effects in passive harm may explain why post-manipulation passive harm was lower in the pet status condition (vs. lack of pet status). These effects on harm intentions were all small-sized, so may have lower practical significance.

7.3.4.1.2. H3-H4: Exploring Mediational Relationships Between Pet Status and Behavioural Intentions Through Warmth/Competence.

As expected, and aligning with previous BIAS map literature with animals (Sevillano & Fiske, 2016b), post-manipulation warmth mediated the relationship between pet status and post-manipulation active harm (H3b). This mediation occurred in the expected direction, whereby pet status (vs. lack of pet status) increased warmth, which in turn decreased active harm. Unlike for active harm, warmth did not mediate the relationship between pet status and active help (H3a), whilst competence did not mediate the relationships between pet status and passive help (H4a), nor between pet status and passive harm (H4b). These findings mostly do not support the BIAS map (Sevillano & Fiske, 2016b) in the context of pet status, indicating warmth mediates the relationship between pet status and active harm only but not active help, and competence does not mediate between pet status and passive behavioural intentions.

7.3.4.1.3. Exploring Associations Between Warmth, Competence and Behavioural Intentions.

Despite mostly a lack of significant mediations, inspection of path *b*s from the mediation analyses demonstrated greater warmth predicted greater active help and lower

active harm, whilst greater competence predicted greater passive help. However, competence did not predict passive harm. Therefore, these path *b* findings provide some support for the BIAS map, but not full support (e.g., competence may not inform passive harm).

7.3.4.1.4. Warmth/Competence Perceptions and Behavioural Intentions Towards the Animal Differ from Zero.

Like Studies 3-4, the unknown animal here was deemed significantly warmer and more competent than zero. Additionally, participants were significantly more willing to actively or passively help, and significantly less willing to actively or passively harm, the animal compared to zero. Thus, the current study combines with Studies 3-4 to evidence a positivity bias in perceptions of and behavioural intentions towards animals.

7.3.4.2. Limitations of the Current Study and Directions for Future Research.

The current study has some limitations to be addressed in future research: a) lack of measurement of possible mediators, and b) effects of the pet status label occurring due to aligning with the animal's guardian rather than with the animal themselves.

Firstly, as considered within Section 7.3.4.1.1., the pet status label may have affected warmth and active help due to assumed domestication and tameness acting as mediators of the effect. That is, pet status may have prompted participants to assume the pet animal is domesticated, and therefore warmer by temperament than a non-pet (wild) animal. For instance, domestication of animals is often associated in part with the animals' increased prosociality towards humans (Hare et al., 2002).³³ Conversely, a non-pet (wild) animal may be correctly or incorrectly perceived as dangerous to approach, and therefore deemed less warm. Indeed, the descriptions from participants' imagined contact with the unknown animal in Study 4 indicated many participants viewed the unknown animal as potentially dangerous,

³³Although domestication and pro-sociality do not equate to the same process (Losey, 2021).

and harmful animals are perceived as having lower moral standing than non-harmful animals (Piazza et al., 2014). Therefore, warmth may have increased not due merely to the pet status label, but instead due to implicit assumptions about an animal's harmless or harmful nature based on the animal's presumed domestication as a pet (vs. not). Whilst this effect does not undermine the pet status effect, it does introduce two additional possible mediators (presumed domestication, perceived harmfulness) not measured within this study. Future research should therefore measure perceived domestication and harmfulness and test effects of these variables on warmth. Research should also test if merely labelling an animal as 'domesticated' (vs. not) impacts the animal's warmth.

Secondly, the pet status label may elicit greater warmth and fewer harm intentions not due to improved perceptions of the animal themselves, but due to perceptions of and implied association with the animal's presumed guardian. For instance, the pet animal may be deemed warm only as an extension of being associated with their human guardian, rather than being perceived as warm in their own right. For example, dogs have been found to be viewed more as people when they are associated with humans, especially women (Chambers et al., 2020). Thus, an animals' implied association with humans (e.g., a guardian) due to being a pet may be improving the animal's warmth within the current study, rather than pet status enhancing the animal's warmth in and of itself. Such effects may also extend to harm intentions, whereby an animals' implied association with humans due to being a pet may reduce harm intentions towards them, due to the animals' increased perceived personhood. Additionally, people may be less willing to harm a pet (vs. non-pet) animal in case this harm would be hurtful to the animal's guardian, and not necessarily because the harm would be hurtful to the animal themselves. Future research could measure salience of the human guardian across pet status conditions through asking participants to what extent they thought of a human guardian after reading the pet status condition (vs. no pet status). Theoretically,

the human guardian should be more salient in the pet status (vs. not) condition, and this salience should be tested as a mediator between pet status and warmth and harm intentions.

7.3.4.3. Conclusion.

The current study highlights the role of pet status in informing people's perceptions of animals' warmth, but not animals' competence. Thus, as dogs are deemed pet animals more than pigs (Study 1) and pet animals are deemed warmer than non-pet animals (the current study), pet status could possibly explain why dogs are deemed warmer than pigs, whilst possibly failing to explain the dog vs. pig competence gap. However, this suggestion would need to be explicitly tested in future research by applying pet status back to dogs and pigs specifically. Alongside warmth, the study uniquely demonstrates how pet status may decrease active and passive harm, from pre- to post-manipulation or (passive harm only) compared to a lack of pet status. However, the effects of pet status on harm intentions found here may be unstable due to floor effects and inadequate scale reliability. Unexpectedly, pet status had no significant effects on active and passive help towards animals and therefore may be unable to explain the pet speciesism gap in active and passive help for dogs and pigs (Studies 3 and 5). Finally, the study supports a mediational relationship in line with the BIAS map for warmth and active help, nor for competence and passive behavioural intentions for warmth and active help, nor for

As the current effects of pet status on harm intentions must be interpreted with caution (due to floor effects or inadequate scale reliability) and are therefore not necessarily robust, pet status was not tested causally in an additional experiment with dogs and pigs. That is, this thesis only tests for causal effects on SCM/BIAS map perceptions of dogs and pigs with variables that have either robust causal effects on behavioural intentions towards unknown animals, or failed manipulation checks by implementing amended manipulations (Section 1.4.). Instead, future research could implement the same pet status manipulation used within

this study for unknown animals but utilise more subtle measures of harm intentions (to avoid floor effects), to test if the effects of pet status on harm intentions found here are robust. For example, whilst participants are extremely unwilling to '*kill*' or '*exterminate*' an animal, they may be more willing to engage in less violent active harm behaviours like '*shout at*' an animal or, due to prevalent animal-use behaviours (Gradidge et al., 2021a), be more willing to actively harm an animal for human benefit (e.g., '*use [an animal] for research*', '*eat*'). Measurement of less violent and/or prevalent harm behaviours may reduce or eliminate floor effects on the harm intentions subscales and enable effects of pet status on harm intentions to be tested for robustness.

The current study has limitations including lack of measurement of possible mediators, and positive pet status effects on warmth and active help possibly arising from the animals' association with a human guardian. Future research should test possible mediators (perceived domestication, harmfulness), and determine the salience of human guardians when considering pet status. To conclude, the current study causally manipulated pet status for the first time, uniquely revealing an effect of pet status on an unknown animal's warmth yet finding null effects on competence and help intentions. Additionally, whilst pet status informed active and passive harm towards an unknown animal, issues with the measurements for harm intentions (floor effects, inadequate reliability) mean these conclusions must be interpreted with caution. The current study uniquely demonstrates possible strengths (e.g., warmth) and limitations (e.g., competence, behavioural intentions) of pet status in explaining perceptions of animals.

7.4. Overall Discussion

Through a pilot study, a pet status manipulation was developed and piloted which successfully manipulates pet status. As such, the pet status manipulation was implemented within Study 6, to test if pet status may be a cause of pet speciesism. Overall, Study 6 found

that pet status causally informs animals' warmth, but not competence, and may also inform harm intentions, but not help intentions, towards animals.

Chapter 8. Discussion and Conclusions

8.1. Discussion of all Findings

This thesis has a) extended pet speciesism literature to consistently evidence pet speciesism through the SCM/BIAS map, b) provided some evidence for the BIAS map as an extension of the SCM, partly supporting Sevillano and Fiske (2016a, 2016b, 2019), and c) uniquely explored extrinsic causes of pet speciesism. All effects have been found with the SCM/BIAS map as the psychological framework, whereby pet speciesism has been operationalised through an animals' warmth (perceived good intent) and competence (perceived ability to enact intent), alongside active help (intentional and effortful aid), passive help (aid which requires minimal effort), active harm (intentional and effortful harm), and passive harm (harmful and exclusionary behaviour which requires minimal effort) towards the animal. I discuss pet speciesism effects, extension of the BIAS map, and causes of pet speciesism individually below.

8.1.1. Effects of Species on SCM/BIAS Map Variables (Evidence for Pet Speciesism)

Extending previous pet speciesism literature (Bilewicz et al., 2011; Caviola & Capraro, 2020; Gradidge et al., 2021b), the current project has consistently evidenced pet speciesism effects: Dogs are deemed warmer and more competent than pigs (Studies 1, 3 and 5), and subject to greater active and passive help and less active and passive harm than pigs (Studies 3 and 5), with effects being mostly of medium size or larger. The current project also uniquely compared perceptions of dogs vs. pigs with an unknown animal (tree kangaroo), finding that tree kangaroos are perceived as unlike pigs on some dimensions (competence, harm intentions; small-sized for competence and passive harm; medium-sized for active harm; Study 3), and like pigs on other dimensions (warmth, help intentions; Study 3). Whilst tree kangaroos themselves are not an ideal 'unbiased' metric (e.g., a positivity bias in Studies 3-4 and 6), these findings may indicate pet speciesism arises from especially positive

perceptions of and behavioural intentions towards dogs across all dimensions, combined with more negative perceptions of pigs' competence and greater harm intentions towards pigs. That is, dogs are seemingly perceived as particularly warm and competent, and subject to greater help and lower harm intentions, compared to other animals, whilst pigs are seemingly perceived as less competent and subject to greater harm intentions compared to other animals. This suggestion would mostly agree with previous research regarding perceptions of animals (Sevillano & Fiske, 2016b), whereby dogs, alongside other 'companion' animals, are deemed especially warm compared to all other categories of animals, and more competent compared to most other categories of animals (except 'predators'). In contrast, pigs, alongside other 'prey' animals, are deemed moderately warm, yet low in competence, compared to most other categories of animals (except 'pests'; Sevillano & Fiske, 2016b).

As all animals, including pigs, were viewed with some level of positivity bias (Study 3), the above findings indicate that: a) dogs, pigs, and tree kangaroos are viewed with positivity bias (e.g., a general positivity bias) across SCM/BIAS map dimensions, b) dogs are viewed with an extreme positivity bias across SCM/BIAS map dimensions specific to their species, and c) a general positivity bias is still present for pigs but is diminished in competence and harm intentions. That is, whilst pigs are deemed more competent and subject to less harm intentions than zero, they are still deemed less competent and subject to greater harm intentions than dogs and tree kangaroos.

Due to the limited range of species used within the current project, it is unclear if this general positivity bias extends to all animals or to mammals only. Such a positivity bias may not extend to animals like mosquitoes and flies, which are typically viewed with negative valence (Possidónio et al., 2019), or to animals like sharks which are deemed more competent than the scale midpoint (e.g., possible positivity bias for competence), but less warm than the scale midpoint (e.g., possible negativity bias for warmth; Neves et al., 2022). Thus, positivity

and negativity biases may differ across SCM/BIAS map dimensions and depending upon species. Future research needs to test across which types of animals (e.g., non-mammals like insects, reptiles and amphibians), and which dimensions (e.g., warmth vs. competence), the general positivity bias found here extends.

8.1.2. Exploring Associations Between Warmth, Competence and Behavioural Intentions (Evidence for the BIAS map as an Extension of SCM)

The project has partially replicated and extended the SCM/BIAS map to animals in the context of pet speciesism (Sevillano & Fiske, 2016b, 2019). For instance, greater warmth was consistently associated with greater active help across all studies (Studies 2-6), and greater warmth was also associated with lower active harm in the species mediation analyses (Studies 3 and 5), and the pet status mediation analysis (Study 6). Yet warmth was not associated with active harm in the behavioural and subjective self-relevance mediation analyses (Study 2) and familiarity and similarity mediation analyses (Study 4), indicating the relationship between warmth and active harm differs across samples and is unstable.

The relationships between competence and passive behavioural intentions also differ across studies and across timepoints: Greater competence was linked to greater passive help within the Study 3 species mediation analysis pre-manipulation only, the familiarity mediation analysis (Study 4), the Study 5 species and similarity mediation analyses, and the pet status mediation analysis (Study 6). Yet competence was not linked to passive help at post-manipulation within the Study 3 species mediation analysis, nor within the Study 4 similarity mediation analysis. Greater competence was linked to lower passive harm within the Study 3 species mediation analysis post-manipulation only, and within the Study 5 species mediation analysis. Yet competence was not linked to passive harm at premanipulation within the Study 3 species mediation analysis, within the similarity and familiarity mediation analyses (Study 4), nor within the pet status mediation analysis (Study

6). These findings indicate the relationships between competence and passive behavioural intentions differ across samples, may vary depending on exposure to a manipulation (e.g., pre- vs. post-manipulation), and are therefore unstable. Overall, whilst this project partly supports the SCM/BIAS map as applied to animals (e.g., the relationship between warmth and active help; Sevillano & Fiske, 2016b, 2019), it also uniquely may not support previous SCM/BIAS map research by finding that the relationships between warmth and active harm, and between competence and passive behavioural intentions, do not always replicate.

8.1.3. Exploring Possible Causes of Pet Speciesism: Behavioural Self-Relevance, Subjective Self-Relevance, Familiarity, Similarity, and Pet Status

Most importantly, this thesis has uniquely explored extrinsic causes of pet speciesism, which were informed by social psychological (Pettigrew & Tropp, 2006), anthrozoological (Possidónio et al., 2019), and speciesism (Loughnan et al., 2010) literature bases, and subsequently identified as significant regressors of perceptions of dogs and pigs (Study 1). These possible extrinsic causes were: behavioural and subjective self-relevance (behavioural and psychological investment in meat consumption respectively), familiarity (quantity and/or perceived quality of contact), similarity (perceived shared attributes of an animal to humans), and pet status (an animals' societal status as a companion animal or not). I discuss findings for these possible causes below.

8.1.3.1. Behavioural and Subjective Self-Relevance as Possible Causes of Pet Speciesism.

Surprisingly, despite being the most widely evidenced hypothesised causes of pet speciesism from previous literature (Gradidge et al., 2021a; Loughnan et al., 2010; Piazza & Loughnan, 2016), effects of behavioural and subjective self-relevance on perceptions of and behavioural intentions towards animals are mostly not found throughout this thesis. That is, behavioural and subjective self-relevance did not causally affect warmth, competence, active help, active harm, and passive help (Studies 2-3). Passive harm did reduce after (vs. before) reading the behavioural self-relevance manipulation in Study 3 (small-sized). However, as behavioural self-relevance was manipulated within Study 3 as a within-subjects pre- to post-manipulation variable, the effect on passive harm may be due either to the salience of behavioural self-relevance, or due to reading text about the animal regardless of behavioural self-relevance. Additionally, this finding on passive harm should be interpreted with caution due to floor effects, whereby most participants were unwilling to passively harm the animal, and inadequate scale reliability. Finally, even if the reduced passive harm in Study 3 is due to behavioural self-relevance, this finding cannot explain why people show more passive harm towards pigs than dogs. That is, although pigs are behaviourally self-relevant whilst dogs are not, behavioural self-relevance in Study 3 led to reduced (not increased) passive harm.

Overall, the contributing roles of behavioural and subjective self-relevance to pet speciesism are limited, whereby behavioural and subjective self-relevance may not contribute to an animal's lower warmth, lower competence, or more negative behavioural intentions towards the animal. Therefore, although pigs are typically behaviourally and subjectively self-relevant whilst dogs are not, behavioural and subjective self-relevance may not explain why pigs are deemed less warm and competent and subject to more negative behavioural intentions than dogs. This thesis thus does not support self-relevance literature (Gradidge et al., 2021a; Loughnan et al., 2010; Piazza & Loughnan, 2016), by finding no evidence that behavioural and subjective self-relevance cause pet speciesism.

8.1.3.2. Similarity as a Possible Cause of Pet Speciesism.

Unlike behavioural and subjective self-relevance, similarity represents a robust cause of pet speciesism, agreeing with previous literature (Batt, 2009; Kozachenko & Piazza, 2021). Specifically, the current project has consistently identified that similarity causes pet speciesism in warmth and competence, whereby similarity increases animals' warmth and competence from pre- to post-manipulation (small-sized for warmth across Studies 4-5 and competence in Study 5, medium-sized for competence in Study 4), and as compared to dissimilarity information (medium-sized for warmth across Studies 4-5 and competence in Study 5, large-sized for competence in Study 4). Thus, when applied to dogs and pigs specifically, dogs are deemed warmer and more competent than pigs as a) dogs are deemed more similar to humans than pigs are (Studies 1 and 5), and b) similar animals are deemed warmer and more competent than dissimilar animals (Studies 1 and 4 -5).

Effects of similarity sometimes extend to active help, whereby similarity consistently increases active help from pre- to post-manipulation (small-sized across Studies 4-5). However, despite this pre- to post-manipulation increase, similarity is superior to dissimilarity information only in informing active help towards known animals (dogs and pigs; medium-sized ; Study 5), and may not inform active help towards unknown animals (i.e., tree kangaroos; Study 4). As similarity (vs. dissimilarity) causes greater post-manipulation active help with dogs and pigs specifically (Study 5), this finding helps explains the pet speciesism gap in active help. That is, as a) dogs are deemed more similar to humans than pigs are (Studies 1 and 5), and b) similar (vs. dissimilar) animals are subject to more active help for dogs than pigs. Additionally, this effect of similarity on active help is of medium size (when compared to dissimilarity information), indicating similarity may have practical significance for interventions by improving active help.

Whilst similarity clearly explains the pet speciesism gaps in warmth, competence and active help, similarity had non-significant effects on passive harm across Studies 4-5, meaning similarity may not explain why pigs are subject to greater passive harm than dogs (although passive harm findings should be interpreted with caution due to inadequate scale reliability across Studies 4-5). Additionally, similarity had differing effects on passive help

and active harm, thereby complicating the role of similarity in informing dogs' and pigs' passive help and active harm. I discuss these differing effects below.

Firstly, whilst similarity increased passive help towards an unknown animal from preto post-manipulation (Study 4; small-sized), similarity did not increase passive help towards known animals (dogs and pigs) from pre- to post-manipulation (Study 5). Thus, whilst similarity explains increased passive help towards unknown animals, similarity may not explain the pet speciesism gap in passive help, as similarity was not found to cause greater passive help for dogs or pigs. Additionally, the significant effect from Study 4 may not be robust, as there was inadequate scale reliability in passive help across Studies 4-5.

Secondly, similarity informs active harm towards known animals (dogs and pigs; Study 5), but may not inform active harm towards unknown animals (i.e., tree kangaroos; Study 4). Specifically, similarity was not found to affect active harm in Study 4 with unknown animals, whilst similarity decreased active harm towards dogs and pigs in Study 5 from pre- to post-manipulation (small-sized), and as compared to dissimilarity information (small-sized). Thus, similarity seemingly informs active harm towards known species only, and, regardless of the known species (dog or pig), active harm decreases following similarity information. Similarity may therefore explain the pet speciesism gap in active harm. Specifically, as a) dogs are deemed more similar to humans than pigs are (Studies 1 and 5), and b) similar animals are subject to lower active harm (Study 5), dogs' greater perceived similarity to humans may contribute to lower active harm towards dogs compared to pigs. Thus, perceived similarity of dogs (but not pigs) to humans may act as a protective factor against active harm towards dogs and may explain why dogs are subject to less active harm than pigs. However, findings of similarity on active harm may not be robust and should be interpreted with caution, as active harm suffered from floor effects across Studies 4-5.

Alongside effects of similarity, dissimilarity had pre- to post-manipulation effects on warmth and/or competence depending on whether the animal was known or unknown. Specifically, dissimilarity reduced an unknown animal's warmth and competence from pre- to post-manipulation (small-sized; Study 4), yet only reduced a known animal's competence from pre- to post-manipulation (medium-sized; Study 5), and not their warmth. As dissimilarity had no pre- to post-manipulation effect on known animals' warmth, this finding indicates dissimilarity may not explain pet speciesism differences in warmth. However, as dissimilarity had a pre- to post-manipulation effect on known animals' competence, this finding suggests dissimilarity explains pet speciesism differences in competence. That is, as a) pigs are deemed more dissimilar to humans than dogs are (Studies 1 and 5), and b) dissimilarity to humans explains why pigs are deemed less competent than dogs. Thus, dissimilarity meaningfully explains the pet speciesism gap in competence.

8.1.3.3. Familiarity as a Possible Cause of Pet Speciesism.

Familiarity was unexpectedly difficult to manipulate, despite previously successful imagined contact manipulations with animals (Auger & Amiot, 2019b), and evidence for the mere exposure effect (e.g., Stafford & Grimes, 2012). The familiarity manipulation via the mere exposure effect in this thesis may have failed to elicit familiarity due to its reliance on absolute exposure (presenting nine photographs of the animal), instead of relative exposure (presenting nine animal photographs alongside a smaller number of photographs of alternative stimuli). For example, Mrkva and Van Boven (2020) indicate relative exposure is more effective than absolute exposure, as relative exposure enhances the salience of stimuli, whilst absolute exposure enables mere exposure to stimuli only and does not necessarily enhance salience. That is, salience arises due to stimuli being presented more frequently than other stimuli, and salience is by nature comparative and dependent upon the presentation of

alternative stimuli alongside target stimuli (Mrkva & Van Boven, 2020). Overall, Study 4's predominantly null effects of familiarity may therefore be due to a) failure in the familiarity manipulations to elicit familiarity, indicating future research should incorporate non-target stimuli within the familiarity manipulation, or b) genuine lack of effects of familiarity on perceptions of animals.

The effect of familiarity on passive help, whereby familiarity decreased passive help from pre- to post-manipulation (small-sized; Study 4), may arise from assumptions about the unknown animal acting as moderators, like assumed harmfulness (although this effect should be interpreted with caution due to inadequate scale reliability). Overall, whilst any contributing role of familiarity to pet speciesism cannot be determined from this thesis due to failed familiarity manipulations, the project has uniquely highlighted how imagined contact and mere exposure effect do not always extend to animals, which does not support previous literature (Auger & Amiot, 2019b).

8.1.3.4. Pet Status as a Possible Cause of Pet Speciesism.

Whilst pet status increases an animals' warmth from pre- to post-manipulation and as compared to a lack of pet status (small-sized), effects of pet status on active and passive harm may be unstable due to limitations with the measures (e.g., unreliable scale; floor effects; Study 6). That is, whilst pet status decreased active and passive harm from pre- to post-manipulation (small-sized) and decreased post-manipulation passive harm as compared to a lack of pet status (small-sized), these findings may not be robust due to the inadequate measures utilised for these variables. Due to this lack of robust causal effects on behavioural intentions, a subsequent study to test effects of pet status on perceptions of, and behavioural intentions towards, dogs and pigs was not conducted within this thesis (Section 7.3.4.3.).

Overall, the current project uniquely highlights limitations of previous pet status and categorisation literature (Signal et al., 2018; Taylor & Signal, 2009), which has previously been correlational, and has focussed on perceptions of and/or feelings towards animals, by showing the possible strengths (e.g., warmth) and limitations (e.g., competence, behavioural intentions) of pet status in explaining perceptions of animals.

8.1.3.5. Summary of Findings: Explaining Pet Speciesism Through Possible Causes.

To summarise, the current project uniquely contributes to explaining the pet speciesism gap in warmth, whereby dogs are deemed warmer than pigs, as dogs are deemed a) more similar to humans, and b) more as pet animals than pigs. Similarity and pet status of dogs in turn increases dogs' warmth, thus contributing to the pet speciesism gap in warmth. Neither behavioural nor subjective self-relevance contribute to this pet speciesism gap in warmth, whilst effects of familiarity on warmth are as yet undetermined due to failed familiarity manipulations.

The current project also uniquely contributes to explaining the pet speciesism gap in competence, whereby dogs are deemed more similar to humans than dogs are, and pigs are deemed more dissimilar to humans than dogs are. Similarity of dogs and dissimilarity of pigs in turn increases dogs' competence and decreases pigs' competence, contributing to the pet speciesism gap in competence and explaining why dogs are deemed more competent than pigs. There was a lack of effects of all other extrinsic variables (behavioural self-relevance, subjective self-relevance, familiarity, pet status) on competence.

Extending to behavioural intentions, the current project contributes to explaining pet speciesism gaps in active help and active harm through similarity. Specifically, as dogs are perceived as more similar to humans than pigs are, this greater perceived similarity of dogs explains why people are more willing to actively help, and less willing to actively harm, dogs than pigs. Additionally, the pet speciesism gaps in active harm and passive harm may be explained by pet status, whereby pet status reduces active and passive harm, but this finding has not yet been tested directly with dogs and pigs and may not be robust due to floor effects and unreliable scales within Study 6 in this thesis. Behavioural self-relevance, subjective selfrelevance, and familiarity do not appear to meaningfully contribute to understanding the pet speciesism gaps in any behavioural intentions. Meanwhile, similarity seems to be unable to explain the pet speciesism gaps in passive help and passive harm, and pet status may not explain the pet speciesism gaps in active help and passive help.

Overall, the current project highlights that the factor which explains pet speciesism gaps across most SCM/BIAS map dimensions is (dis)similarity, contributing to dogs' greater warmth, dogs' greater competence, pigs' lower competence, greater active help towards dogs, and lower active harm towards dogs. Together, this thesis indicates pet speciesism gaps are informed by extrinsic factors as follows: a) warmth and active harm are informed by similarity and possibly pet status, b) competence is informed by (dis)similarity, c) active help is informed by similarity, and d) passive harm is possibly informed by pet status. Passive help is the only pet speciesism gap which has not yet been found to be informed by extrinsic factors covered within this thesis, whilst other extrinsic factors within this thesis either do not seem to meaningfully contribute to pet speciesism (behavioural self-relevance, subjective self-relevance), or require an amended manipulation to causally test their contribution to pet speciesism (familiarity). Overall, the thesis uniquely contributes a broader understanding of extrinsic causes of pet speciesism, and, due to testing causal effects of extrinsic factors on pet speciesism for the first time, highlights challenges to be addressed in future research with both causal manipulations for extrinsic factors (e.g., familiarity), and measurement of SCM/BIAS map variables (e.g., less severe harm intentions).

8.1.4. Exploring Mediational Relationships Between Possible Causes/Species and Behavioural Intentions Through Warmth/Competence

After exploring possible extrinsic causes of pet speciesism, mediational analyses extended the BIAS map further to determine mediational relationships of warmth and competence between the possible extrinsic causes and active and passive behavioural intentions respectively. Overall, the BIAS map was not always supported in the mediations. For example, warmth did not significantly mediate the relationships between behavioural and subjective self-relevance and active help and active harm (Study 2), familiarity and active help or active harm (Study 4), similarity and active harm (Study 4), or pet status and active help (Study 6). In addition, competence did not significantly mediate the relationships between species and passive help either across timepoints (Study 5) or post-manipulation only (Study 3), species and passive harm (Studies 3 and 5), familiarity and passive help and passive harm (Study 4), similarity and passive harm (Study 4), similarity and passive help within Study 4, or pet status and passive help and passive harm (Study 6). However, warmth did significantly mediate the relationships between species and active help and active harm (Studies 3 and 5), similarity and active help (Studies 4-5), and pet status and active harm (Study 6). Competence significantly mediated the relationships between species and passive help pre-manipulation within Study 3 (but not Study 5), and similarity and passive help within Study 5 (but not Study 4). The non-significant mediations here may be arising from lack of effects of the IVs on the DVs and/or mediators, or due to differing relationships between warmth and active harm, and between competence and passive behavioural intentions.
8.2. Theoretical and Practical Implications

8.2.1. Theoretical Implications

This thesis has theoretical implications for multiple literature bases, including speciesism, human-animal interaction, and social psychological literature. For example, the project uniquely sheds light on similarity as an explanation for the pet speciesism gaps in warmth, competence, and active help. This finding a) supports previous human-animal interaction literature on the important role of similarity in informing perceptions of animals (Batt, 2009; Prguda & Neumann, 2014), b) uniquely extends this previous literature to the SCM/BIAS map specifically, and c) demonstrates people do not always disregard similarity information about pigs (unlike Piazza & Loughnan, 2016). Current findings may differ to Piazza and Loughnan (2016) due to variations in manipulations across studies, whereby this thesis manipulated overall similarity (vs. dissimilarity) to humans and referred to multiple attributes (behaviour, intelligence, sociality, emotionality), whereas Piazza and Loughnan (2016) manipulated intelligence only. As such, whether or not people disregard similarity information may depend upon the exact attribute being manipulated (e.g., intelligence), and may not occur with other attributes (e.g., behaviour), or when manipulating overall similarity.

The project also demonstrates that behavioural and subjective self-relevance may fail to explain pet speciesism gaps across SCM/BIAS map dimensions, which does not support extensive previous speciesism literature and motivated cognition theory (Gradidge et al., 2021a; Loughnan et al., 2010, 2014; Piazza & Loughnan, 2016). Such findings may differ due to the exact manipulations used: For example, Loughnan et al. (2010) manipulated (behavioural) self-relevance through actual meat consumption, whilst this thesis employed imagined scenarios. Thus, for behavioural self-relevance to affect perceptions of animals, it may need to be elicited through actual behaviour (e.g., meat consumption) rather than imagined behaviour. Alternatively, these differing findings may be arising from the

application of (behavioural) self-relevance to different animals, whereby this thesis manipulated the behavioural self-relevance of pigs, and Loughnan et al. (2010) manipulated the (behavioural) self-relevance of cows.

Surprisingly, the current project has revealed limitations of widely evidenced psychological measures to elicit familiarity, including imagined contact with animals (Auger & Amiot, 2019b), and the mere exposure effect (Mrkva & Van Boven, 2020; Newell & Shanks, 2007; Stafford & Grimes, 2012). These findings have implications for social psychological literature, like imagined contact theory (Crisp & Turner, 2009), by demonstrating that imagined contact with animals as manipulated within this thesis may not effectively elicit familiarity. By considering differences between how familiarity was manipulated within this thesis and previous literature (Auger & Amiot, 2019b; Cerrato & Forestell, 2022), imagined contact with animals may instead only elicit familiarity and inform perceptions of and/or behavioural intentions towards animals when a) imagined contact is described as positive (vs. neutral within this thesis; i.e., Auger & Amiot, 2019b), b) the animal is described as not being dangerous, and c) implicit measurements of prejudice (e.g., Implicit Association Test) are implemented in combination with, or instead of, explicit measures (i.e., Cerrato & Forestell, 2022).

In addition, the project has shown that previous literature on categorisation of animals (e.g., pet status; Signal et al., 2018; Taylor & Signal, 2009) may not extend to causal effects on some SCM/BIAS map variables. Where effects of pet status have been identified here, they are restricted to harm intentions and warmth. Thus, this thesis uniquely demonstrates how categorisation (e.g., pet status) informs perceptions of animals (e.g., warmth as found here or empathy as found within previous research; Signal et al., 2018), yet this effect on perceptions may not extend to help intentions and may therefore not sufficiently explain, or be able to change, real-world behaviour. Furthermore, the lack of significant relationships

between profit status and dogs' and pigs' warmth and competence from Study 1 may indicate an animal's categorisation as a commodity (vs. not) does not negatively impact upon perceptions of the animal, thus not supporting previous research comparing perceptions of 'food' vs. non-'food' animals (Bastian et al., 2012a).

Overall, the current project uniquely identifies and systematically explores extrinsic causes of pet speciesism, thus adding to emerging pet speciesism literature (Caviola & Capraro, 2020), expanding upon previous correlational research (Signal et al., 2018), and extending previous SCM/BIAS map research (Sevillano & Fiske, 2016b) by applying the SCM/BIAS map for the first time to explore pet speciesism.

8.2.2. Practical Implications

This thesis has practical implications for the development of interventions to reduce or prevent pet speciesism. For instance, the project indicates that, contrary to previous suggestions and research (Gradidge & Zawisza, 2019; Piazza & Loughnan, 2016), similarity is a strong contender for interventions targeting variables like pigs' warmth and competence and active help towards pigs. Thus, emphasising the similarity of pigs by demonstrating their human-like capabilities (e.g., intelligence, sociability) should enable pigs to be deemed warmer and more competent, and be subject to greater active help, both absolutely (pre- to post-manipulation) and relatively (decreasing the gap between dogs' and pigs' warmth, competence, and active help). Whilst not explicitly tested within this thesis, improving warmth, competence, and active help towards pigs may more broadly improve pigs' welfare through reducing pig meat consumption, and enhancing the public image of pigs and encouraging advocacy for pig welfare. Such advocacy is sorely needed considering regular pig welfare violations (e.g., tail docking; De Briyne et al., 2018), whilst (pig) meat consumption must urgently reduce considering the rising global climate and its associated disastrous phenomena (e.g., greenhouse gas emissions; Godfray et al., 2018). Reductions in

meat consumption, which could be achieved through interventions informed by this thesis, are thus a crucial part of the solution for slowing climate change (Springmann et al., 2018).

The current project also contributes practically to the development of interventions to help vulnerable animal groups beyond pigs, including 'pest' animals like rats and bats, endangered animals, and other animals used for meat production. For instance, highlighting similarities of vulnerable animals (e.g., 'pest', endangered, or 'food' animals) to humans may bolster support and increase advocacy for them by improving their warmth and competence. Indeed, recent research on shark conservation shows that, the warmer sharks are perceived as, the more willing people are to support their conservation (Neves et al., 2022). Such interventions are especially important considering the biodiversity crisis, whereby approximately 16,000 animal species are at threat of extinction (IUCN, 2021), alongside risks to ecological functions and whole ecosystems (Valiente-Banuet et al., 2015). This crisis has implications not only for animals themselves, due to threats to their survival, but also for human health like increased risk of zoonotic disease (Keesing & Ostfeld, 2021). Thus, psychological interventions to reduce the biodiversity crisis (e.g., emphasising animals' similarity to humans) are crucial to protect human and animal welfare. This thesis uniquely contributes to developing these interventions by demonstrating the importance of similarity in informing animals' warmth and competence, and active help towards them, alongside providing a simple but effective text manipulation of similarity to implement within interventions.

Overall, whilst emerging SCM/BIAS map literature has demonstrated how animals are perceived (Neves et al., 2022; Sevillano & Fiske, 2016a, 2016b, 2019), the current project extends upon this literature to uniquely explore why some animals (e.g., dogs) are deemed warmer and more competent, and are subject to greater help and lesser harm, whilst other animals (e.g., pigs) are not. Exploring why some animals are perceived more negatively than

others is imperative for developing interventions to improve perceptions of animals, as, by determining why animals are perceived as they are, causal factors can be removed or overcome. Thus, this thesis has identified causes of pet speciesism, such as (dis)similarity and possibly pet status, and developed causal manipulations of extrinsic factors, which provide future researchers with information and tools to develop research-driven interventions.

8.3. Limitations of the Project and Associated Directions for Future Research

The project has some limitations in scope which should be addressed in future research: a) lack of generalisability to other animals, b) lack of exploration of intrinsic factors, and c) possible culture boundedness. Additionally, the project has some methodological limitations regarding manipulations which should be resolved in future research: a) ineffectiveness of manipulations, perhaps due to the use of imagined scenarios, and b) issues with manipulation checks regarding their inconsistent use and transparency.

8.3.1. Limitations in Scope

Firstly, as dogs and pigs represent a good 'matched pair' due to their multiple ostensible similarities (Section 1.1.), the current project focussed only on dog vs. pig pet speciesism. Whilst this focus on dogs and pigs (instead of cats vs. pigs, or dogs vs. cows) enabled possible moderators (size, diet, domestication) to be eliminated, this spotlight on dogs and pigs means current findings do not necessarily generalise to perceptions of other animals. Future research should determine if pet speciesism and similarity as an extrinsic cause of pet speciesism generalise to other animals: For example, Sevillano and Fiske (2016b) identified multiple species in the 'prey' cluster (like pigs) that people may equally be prejudiced against, like cows, ducks, and rabbits. Sevillano and Fiske (2016b) identified multiple other species in the 'companion' cluster (like dogs) that people may equally be prejudiced towards, like horses, cats, and monkeys. Thus, a comparison of alternative forms of speciesism with animals from the 'companion' and 'prey' clusters (e.g., horses vs. cows,

which are relatively matched on diet, size, and social structure) would help determine if current findings apply more generally to 'companion' vs. 'prey' animals, or to dogs vs. pigs specifically. Additionally, such future research would determine if the extreme positivity bias found with dogs applies to other 'companion' animals, and if the reduced positivity bias found with pigs applies to other 'prey' animals.

Secondly, this thesis intentionally excluded and attempted to reduce effects of intrinsic factors (e.g., through the novel animal paradigm), so individual causal contributions of extrinsic variables to pet speciesism could be tested without the presence of moderators. However, measurement of both intrinsic and extrinsic factors would be beneficial to forming a complete view of pet (and other forms of) speciesism and its causal pathways. For instance, as discussed in Section 1.3, extensive research demonstrates the importance of intrinsic factors in informing peoples' perceptions of animals (e.g., appearance, Archer & Monton, 2011; behaviour; Pérez Fraga et al., 2021), whilst this thesis demonstrates the crucial role of perceived similarity (an extrinsic factor) in informing perceptions of animals. Future research should test the relative contributions of intrinsic and extrinsic factors to pet speciesism by measuring multiple intrinsic (e.g., appearance, responsiveness) and extrinsic factors (e.g., similarity), and utilising structural equation modelling to identify intrinsic and/or extrinsic pathways which determine peoples' perceptions of, and behavioural intentions towards, animals.

Finally, although all studies within this thesis intentionally recruited international, culturally heterogenous samples to maximise generalisability, some of the studies within the current project (Studies 3 and 5) are necessarily culture-bound by excluding participants who indicated their culture and/or religion does not view dogs as companions or pigs as 'food'. Whilst these exclusions were necessary to ensure results were not biased by cross-cultural differences, these exclusions mean that cross-cultural variability was reduced instead of

directly investigated. Therefore, it is unclear if people within cultures which do not keep dogs as companion animals and/or do not consume pigs as food view dogs and pigs in similar ways as within this thesis (i.e., if they exhibit dog vs. pig pet speciesism). Future research should directly compare participants from cultures who view dogs as companion animals and pigs as 'food' (vs. not) to test if dog vs. pig pet speciesism occurs across these cultures, or only in the former cultures, and thereby elucidate cultural factors which cause or moderate pet speciesism. Research may also more specifically explore if speciesism occurs against different animals by people in cultures which do not view dogs as companion animals and pigs as 'food', and if and how people within these cultures have similar contradictory relationships with animals (i.e., loving some yet consuming others; Gradidge et al., 2021a; Joy, 2010). Therefore, research needs to determine which animals people within different cultures are prejudiced against and towards, such as by asking people from different cultures to rate different animals on their warmth and competence (like Sevillano & Fiske, 2016b), and how people within different cultures are prejudiced against or towards animals, such as by exploring if people within different cultures have similar animal stereotypes (like Sevillano & Fiske, 2016b), and if extrinsic and intrinsic causes of speciesism apply across cultures.

Beyond Studies 3 and 5, the remaining studies (Studies 1-2, 4 and 6) are likely to be less culture-bound as they recruited international, culturally heterogenous samples, without exclusions based on whether participants' culture and/or religion does not view dogs as companion animals and pigs as 'food'. Additionally, Studies 2, 4 and 6 all explored perceptions of a novel animal only, and not perceptions of dogs and pigs, which means cultural bias and preconceptions in perceptions of the target animal should be reduced. However, due to utilising convenient and accessible volunteer samples, Studies 1-2, 4 and 6 still had an over-representation of participants from countries like the United Kingdom

(32.9% to 54.6% participants of United Kingdom/British nationality; 46.5% to 70.4% participants living within the United Kingdom), meaning findings from these studies may not generalise to all people. Future research should explore more specifically if culture moderates the effects of extrinsic factors on perceptions of animals by directly comparing participants from different cultures utilising quota sampling, to specifically determine if extrinsic factors (e.g., similarity) have the same effects on perceptions of animals cross-culturally.

8.3.2. Methodological Limitations

Firstly, manipulations were frequently found to be ineffective at manipulating their intended variable (i.e., behavioural self-relevance, subjective self-relevance, and familiarity), which limited the ability to explore any causal role of these variables in pet speciesism. This ineffectiveness of the self-relevance and familiarity manipulations may arise from the fact that these manipulations were imagined scenarios. Whilst previous research has indicated that imagined scenarios are effective with animals (Auger & Amiot, 2019), and imagined scenarios were chosen for ease of administration of the manipulations, these imagined components may have led the manipulations to suffer from lack of realism. This potential lack of realism may therefore explain why the imagined manipulations were largely found to be ineffective. Future research could therefore instead utilise real scenarios rather than imagined. For example, real contact with animals could be used to elicit familiarity and thus test any causal effects of familiarity on SCM/BIAS map variables. For self-relevance, participants could be asked to consume real meat (vs. not; manipulation of behavioural selfrelevance), akin to Loughnan et al. (2010), which is either prepared to have good taste (e.g., appropriately salted) or bad taste (e.g., overly salted; manipulation of subjective selfrelevance). Additionally, although the similarity and pet status manipulations were found to be effective in manipulating similarity and pet status respectively, aspects of these manipulations were also imagined (e.g., 'Imagine reading the following text/sentence in the

newspaper'), as well as being very short and simple, which may have reduced believability of the texts. Therefore, effects of the similarity and pet status manipulations may be enhanced further through adding additional text (e.g., on the specific [dis]similarities between humans and the animal) and presenting the manipulations within a news or scientific article format (as opposed to asking participants to imagine the text within an article), akin to Leach et al. (2021).

Secondly, manipulation checks were not always present (Study 2), and, when they were present (Studies 3-6), these checks were often basic, consisting of single items developed by the researcher and sometimes including the same wording as that used within the manipulation (e.g., 'How similar do you perceive the species in the photograph to be to humans?' for similarity; 'How much do you perceive the species in the photograph to be a 'pet' animal (an animal that is kept within a household as a companion)?' for pet status). Therefore, whilst the similarity and pet status manipulations were found to be effective when using these manipulation checks, these findings may arise from the fact that the manipulation and check utilised the same wording (i.e., both mention '[dis]similar' or the animal being a pet), rather than actual effectiveness of the manipulations. Whilst inclusion of alternative, validated subscales to measure effectiveness of the similarity manipulation (i.e., perceived shared emotional and cognitive traits with humans) overcomes this limitation to an extent, alternative scales to measure pet status have not yet been developed. Therefore, the pet status manipulation check used within this thesis may suffer from bias, rather than demonstrating true effectiveness of the manipulation in manipulating pet status. Additionally, although the manipulation checks for self-relevance and familiarity did not use the same wording as the manipulations, some of these manipulation checks again suffer from being oversimplified, single items developed by the researcher (e.g., 'How familiar do you perceive the species in the photograph to be to you?' for familiarity), without these measures having prior

validation. As such, due to lack of validation, these manipulation checks may not be truly measuring the underlying target construct (e.g., familiarity; Chester & Lasko, 2021). To overcome these limitations with the manipulation checks, future research should develop alternative, validated measures to check effectiveness of manipulations, such as scales which are developed and validated through factor analysis and use alternative wording to the manipulation, or implicit measures as a way to overcome self-report bias (e.g., like Cerrato & Forestell, 2022). Once alternative measures for the manipulation checks have been developed, the manipulations, especially those found to be effective at manipulating their intended variable here (i.e., similarity and pet status), can be re-tested for effectiveness with the new and validated manipulation check/s.

8.4. Overall Conclusions

The current project has a) consistently extended pet speciesism effects in the context of the SCM/BIAS map, whereby dogs are consistently viewed more positively than pigs, b) identified an overall positivity bias towards dogs, pigs and tree kangaroos, which is reduced for pigs and extreme for dogs, c) demonstrated some extension of the BIAS map from the SCM in line with previous SCM/BIAS map literature with animals, and d) explored extrinsic causes of pet speciesism using the SCM/BIAS map for the first time. Exploration of possible extrinsic causes of pet speciesism has demonstrated that a) similarity contributes to pet speciesism in warmth, competence, active help, and active harm, b) behavioural and subjective self-relevance surprisingly may not cause pet speciesism across any SCM/BIAS map dimensions, c) familiarity manipulations unexpectedly failed, and d) pet status informs unknown animals' warmth and, possibly, harm intentions towards unknown animals.

The project has limitations in scope due to a) the focus on dogs, pigs and unknown (mammalian) animals, b) exploration of extrinsic factors and not intrinsic factors, and c) possible culture-boundedness and questions regarding generalisability. Thus, to broaden the

scope of current findings, future research should test the results with a wider range of matched-pair species (e.g., horses vs. cows) and non-mammals, explore the relative contributions of extrinsic and intrinsic factors to pet speciesism, and compare perceptions of animals and pet speciesism across cultures. The project also has methodological limitations regarding manipulations (i.e., sometimes ineffective manipulations, use of imagined scenarios) and manipulation checks (i.e., not always present, transparent wording). Thus, future research should overcome these methodological limitations by utilising real-world scenarios rather than imagined, presenting the similarity and pet status manipulations in an actual newspaper articles format, and utilising validated scales or implicit measures for manipulation checks.

Overall, this thesis is the first to systematically test extrinsic factors which inform pet speciesism. The project has extensive theoretical implications for anthrozoological and social psychological literature by testing applicability of the SCM/BIAS map to pet speciesism and exploring extrinsic causes of pet speciesism for the first time, highlighting the importance of similarity in informing perceptions of animals, and revealing limitations of other extrinsic causes (behavioural self-relevance, subjective self-relevance, familiarity). The project has novel practical implications by acting as the crucial foundation for future research to test interventions (e.g., using similarity) to reduce and/or prevent pet speciesism, thereby possibly decreasing meat consumption and its widespread negative impacts on pig welfare and the planet. Such interventions, if effective, will encourage desperately required advocacy for vulnerable animals, and aid in protecting life on Earth by reducing climate change and its devastating consequences.

References

- Amiot, C. E., Sukhanova, K., & Bastian, B. (2020). Social identification with animals: Unpacking our psychological connection with other animals. *Journal of Personality* and Social Psychology, 118(5), 991. <u>https://doi.org/10.1037/pspi0000199</u>
- Apostol, L., Rebega, O. L., & Miclea, M. (2013). Psychological and socio-demographic predictors of attitudes toward animals. *Procedia-Social and Behavioral Sciences*, 78, 521-525. <u>https://doi.org/10.1016/j.sbspro.2013.04.343</u>
- Archer, J., & Monton, S. (2011). Preferences for infant facial features in pet dogs and cats. *Ethology*, 117(3), 217-226. <u>https://doi.org/10.1111/j.1439-0310.2010.01863.x</u>
- Ares, E., & Sutherland, N. (2019, Feb 19). Consumption of dog meat in the UK. UK Parliament. Retrieved from: <u>https://researchbriefings.files.parliament.uk/documents/CDP-2019-0045/CDP-20045/CDP-2019-0045/CDP-2</u>
- Arts, J. W., van der Staay, F. J., & Ekkel, E. D. (2009). Working and reference memory of pigs in the spatial holeboard discrimination task. *Behavioural Brain Research*, 205(1), 303-306. <u>https://doi.org/10.1016/j.bbr.2009.06.014</u>
- Auger, B., & Amiot, C. E. (2015). Can Pets Facilitate Identification with Other Animals? Testing the Pets as Ambassador Hypothesis. Poster presented at the 75th annual convention of the Canadian Psychological Association, Ottawa, ON.
- Auger, B., & Amiot, C. E. (2016). Are All Animals Equal? Testing How Contact with Different Animal Types is Associated with Identification with Animals. Poster presented at the 17th Annual Convention of the Society for Personality and Social Psychology, San Diego, CA.

- Auger, B., & Amiot, C. E. (2019a). Testing the roles of intergroup anxiety and inclusion of animals in the self as mechanisms that underpin the "pets as ambassadors" effect.
 Anthrozoös, 32(1), 5-21. <u>https://doi.org/10.1080/08927936.2019.1550277</u>
- Auger, B., & Amiot, C. E. (2019b). The impact of imagined contact in the realm of humananimal relations: Investigating a superordinate generalization effect involving both valued and devalued animals. *Journal of Experimental Social Psychology*, 85, 103872. <u>https://doi.org/10.1016/j.jesp.2019.103872</u>
- Backhaus, K., Erichson, B., Plinke, W., & Weiber, R. (2016). *Multivariate Analysemethoden*. Springer Berlin Heidelberg. <u>https://doi.org/10.1007/978-3-662-46076-4</u>
- Baron, R. M., & Kenny, D. A. (1986). The moderator-mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations. *Journal* of Personality and Social Psychology, 51(6), 1173.
- Bartlett, M. S. (1951). The effect of standardization on a χ 2 approximation in factor analysis. *Biometrika*, 38(3/4), 337-344. <u>https://doi.org/10.2307/2332580</u>
- Bastian, B., Loughnan, S., Haslam, N., & Radke, H. R. (2012a). Don't mind meat? The denial of mind to animals used for human consumption. *Personality and Social Psychology Bulletin, 38*(2), 247-256. <u>https://doi.org/10.1177/0146167211424291</u>
- Bastian, B., Costello, K., Loughnan, S., & Hodson, G. (2012b). When closing the humananimal divide expands moral concern: The importance of framing. *Social Psychological and Personality Science*, 3(4), 421-429.
 <u>https://doi.org/10.1177/19485506114251</u>

- Batt, S. (2009). Human attitudes towards animals in relation to species similarity to humans: a multivariate approach. *Bioscience Horizons*, 2(2), 180-190. <u>https://doi.org/10.1093/biohorizons/hzp021</u>
- Bedford, E. (2022, July 27). Leading pets ranked by household ownership in the United Kingdom (UK) in 2021/22. *Statista*. Retrieved from <u>https://www.statista.com/statistics/308218/leading-ten-pets-ranked-by-household-ownership-in-the-united-kingdom-uk/</u>
- Beltramini, R. F. (1982). Advertising perceived believability scale. *Proceedings of the Southwestern Marketing Association, 1*, 1-3.
- Beltramini, R. F., & Evans, K. R. (1985). Perceived believability of research results information in advertising. *Journal of Advertising*, *14*(3), 18-31.
- Berchin, I. I., Valduga, I. B., Garcia, J., & de Andrade, J. B. S. O. (2017). Climate change and forced migrations: An effort towards recognizing climate refugees. *Geoforum*, 84, 147-150. <u>https://doi.org/10.1016/j.geoforum.2017.06.022</u>
- Bilewicz, M., Imhoff, R., & Drogosz, M. (2011). The humanity of what we eat: Conceptions of human uniqueness among vegetarians and omnivores. *European Journal of Social Psychology*, 41(2), 201-209. <u>https://doi.org/10.1002/ejsp.766</u>
- Bilewicz, M., Michalak, J., & Kamińska, O. K. (2016). Facing the edible. The effects of edibility information on the neural encoding of animal faces. *Appetite*, 105, 542-548. <u>https://doi.org/10.1016/j.appet.2016.06.014</u>
- Bishop, P. A., & Herron, R. L. (2015). Use and misuse of the Likert item responses and other ordinal measures. *International Journal of Exercise Science*, 8(3), 297.

- Black, J. E., & Barnes, J. L. (2020). Morality and the imagination: Real-world moral beliefs interfere with imagining fictional content. *Philosophical Psychology*, 33(7), 1018-1044. https://doi.org/10.1080/09515089.2020.1775799
- Blanca, M. J., Alarcón, R., Arnau, J., Bono, R., & Bendayan, R. (2017). Non-normal data: Is ANOVA still a valid option? *Psicothema*, 29(4), 552–557. <u>https://doi.org/10.7334/psicothema2016.383</u>
- Bornstein, R. F., & D'Agostino, P. R. (1992). Stimulus recognition and the mere exposure effect. *Journal of Personality and Social Psychology*, *63*(4), 545.
- Bornstein, R. F., & D'Agostino, P. R. (1994). The attribution and discounting of perceptual fluency: Preliminary tests of a perceptual fluency/attributional model of the mere exposure effect. *Social Cognition*, 12(2), 103. https://doi.org/10.1521/soco.1994.12.2.103
- Bratanova, B., Loughnan, S., & Bastian, B. (2011). The effect of categorization as food on the perceived moral standing of animals. *Appetite*, 57(1), 193-196. <u>https://doi.org/10.1016/j.appet.2011.04.020</u>
- Bray, E. E., Gnanadesikan, G. E., Horschler, D. J., Levy, K. M., Kennedy, B. S., Famula, T. R., & MacLean, E. L. (2021). Early-emerging and highly heritable sensitivity to human communication in dogs. *Current Biology*, *31*(14), 3132-3136.
 https://doi.org/10.1016/j.cub.2021.04.055
- Bressler, R. D. (2021). The mortality cost of carbon. *Nature Communications, 12*(1), 1-12. https://doi.org/10.1038/s41467-021-24487-w

- Broemer, P., & Diehl, M. (2004). Romantic jealousy as a social comparison outcome: When similarity stings. *Journal of Experimental Social Psychology*, 40(3), 393-400. https://doi.org/10.1016/j.jesp.2003.08.002
- Broom, D. M., Sena, H., & Moynihan, K. L. (2009). Pigs learn what a mirror image represents and use it to obtain information. *Animal Behaviour*, 78(5), 1037-1041. <u>https://doi.org/10.1016/j.anbehav.2009.07.027</u>
- Byrne, D., Clore, G. L., & Smeaton, G. (1986). The attraction hypothesis: Do similar attitudes affect anything? *Journal of Personality and Social Psychology*, 51(6), 1167– 1170. <u>https://doi.org/10.1037/0022-3514.51.6.1167</u>
- Carlsson, R., & Björklund, F. (2010). Implicit stereotype content: Mixed stereotypes can be measured with the implicit association test. *Social Psychology*, 41(4), 213. <u>https://doi.org/10.1027/1864-9335/a000029</u>

Canadell, J. G., & Jackson, R. B. (2021). Ecosystem Collapse and Climate Change. Springer.

- Carr, E. W., Brady, T. F., & Winkielman, P. (2017). Are you smiling, or have I seen you before? Familiarity makes faces look happier. *Psychological Science*, 28(8), 1087-1102. <u>https://doi.org/10.1177/0956797617702003</u>
- Caviola, L., & Capraro, V. (2020). Liking but devaluing animals: Emotional and deliberative paths to speciesism. *Social Psychological and Personality Science*, 11(8), 1080-1088. <u>https://doi.org/10.1177/1948550619893959</u>
- Cembalo, L., Caracciolo, F., Lombardi, A., Del Giudice, T., Grunert, K. G., & Cicia, G. (2016). Determinants of individual attitudes toward animal welfare-friendly food products. *Journal of Agricultural and Environmental Ethics, 29*(2), 237-254.
 https://doi.org/10.1007/s10806-015-9598-z

- Cerrato, S., & Forestell, C. A. (2022). Meet your meat: The effect of imagined intergroup contact on wanting and liking of meat. *Appetite*, 168, 105656. <u>https://doi.org/10.1016/j.appet.2021.105656</u>
- Chambers, J., Quinlan, M. B., Evans, A., & Quinlan, R. J. (2020). Dog-human coevolution:
 Cross-cultural analysis of multiple hypotheses. *Journal of Ethnobiology*, 40(4), 414-433. <u>https://doi.org/10.2993/0278-0771-40.4.414</u>
- Chang, C. (2011). Feeling ambivalent about going green. *Journal of Advertising*, 40(4), 19-32.
- Child, D. (2006). The Essentials of Factor Analysis. A&C Black.
- Çınar, Ç., Karinen, A. K., & Tybur, J. M. (2021). The multidimensional nature of food neophobia. *Appetite*, 162, 105177. <u>https://doi.org/10.1016/j.appet.2021.105177</u>
- Claypool, H. M., Hugenberg, K., Housley, M. K., & Mackie, D. M. (2007). Familiar eyes are smiling: On the role of familiarity in the perception of facial affect. *European Journal* of Social Psychology, 37(5), 856-866. <u>https://doi.org/10.1002/ejsp.422</u>

Cohen, J. (1988). Statistical Power Analysis for the Behavioral Sciences (2nd ed.). Erlbaum.

- Cooper, J. J., Ashton, C., Bishop, S., West, R., Mills, D. S., & Young, R. J. (2003). Clever hounds: Social cognition in the domestic dog (Canis familiaris). *Applied Animal Behaviour Science*, 81(3), 229-244. <u>https://doi.org/10.1016/S0168-1591(02)00284-8</u>
- Cordoni, G., & Palagi, E. (2019). Back to the future: A glance over wolf social behavior to understand dog–human relationship. *Animals*, 9(11), 991. https://doi.org/10.3390/ani9110991

- Crisp, R. J., & Turner, R. N. (2009). Can imagined interactions produce positive perceptions?
 Reducing prejudice through simulated social contact. *American Psychologist*, 64(4),
 231. https://doi.org/10.1037/a0014718
- Csoltova, E., & Mehinagic, E. (2020). Where do we stand in the domestic dog (Canis familiaris) positive-emotion assessment: A state-of-the-art review and future directions. *Frontiers in Psychology*, 2131. <u>https://doi.org/10.3389/fpsyg.2020.02131</u>
- Cuddy, A. J., Fiske, S. T., & Glick, P. (2007). The BIAS map: Behaviors from intergroup affect and stereotypes. *Journal of Personality and Social Psychology*, 92(4), 631. <u>https://doi.org/10.1037/0022-3514.92.4.631</u>
- Curnutte, M. (2014). The big problem with mini-pigs. *National Geographic*. Retrieved from: <u>https://www.nationalgeographic.com/animals/article/140930-animals-culture-science-miniature-pigs-breeders-sanctuaries#:~:text=Since%201998%2C%20the%20number%20of,as%20many%20as%20a%20million.</u>
- Damsbo-Svendsen, M., Frøst, M. B., & Olsen, A. (2017). A review of instruments developed to measure food neophobia. *Appetite*, 113, 358-367. https://doi.org/10.1016/j.appet.2017.02.032
- Davis, M. H. (1980). A multidimensional approach to individual differences in empathy. JSAS Catalog of Selected Documents in Psychology, 10, 85.
- Davis, M. H. (1983). Measuring individual differences in empathy: Evidence for a multidimensional approach. *Journal of Personality and Social Psychology*, 44, 113–126.

- De Briyne, N., Berg, C., Blaha, T., Palzer, A., & Temple, D. (2018). Phasing out pig tail docking in the EU-present state, challenges and possibilities. *Porcine Health Management*, 4(1), 1-9. <u>https://doi.org/10.1186/s40813-018-0103-8</u>
- Demoulin, S., Leyens, J. P., Paladino, M. P., Rodriguez-Torres, R., Rodriguez-Perez, A., & Dovidio, J. (2004). Dimensions of "uniquely" and "non-uniquely" human emotions. *Cognition and Emotion*, 18(1), 71-96. <u>https://doi.org/10.1080/02699930244000444</u>
- DEFRA (2022, July 14). Monthly UK statistics on cattle, sheep and pig slaughter and meat production–statistics notice (data to June 2022). *DEFRA*. Retrieved from <u>https://www.gov.uk/government/statistics/cattle-sheep-and-pig-slaughter/monthly-uk-</u> <u>statistics-on-cattle-sheep-and-pig-slaughter-and-meat-production-statistics-notice-</u> <u>data-to-april-2022</u>
- DEFRA & APHA (2014, December 31). Keeping a pet pig or 'micropig'. UK Government. Retrieved from <u>https://www.gov.uk/guidance/keeping-a-pet-pig-or-micropig#register-as-pig-keeper</u>
- Dhont, K., Hodson, G., Costello, K., & MacInnis, C. C. (2014). Social dominance orientation connects prejudicial human–human and human–animal relations. *Personality and Individual Differences*, 61, 105-108. <u>https://doi.org/10.1016/j.paid.2013.12.020</u>
- Diamantopoulos, A., Florack, A., Halkias, G., & Palcu, J. (2017). Explicit versus implicit country stereotypes as predictors of product preferences: Insights from the stereotype content model. *Journal of International Business Studies*, 48(8), 1023-1036. <u>https://doi.org/10.1057/s41267-017-0085-9</u>
- Dunn, T. J., Baguley, T., & Brunsden, V. (2014). From alpha to omega: A practical solution to the pervasive problem of internal consistency estimation. *British Journal of Psychology*, 105(3), 399-412. <u>https://doi.org/10.1111/bjop.12046</u>

- Eisinga, R., Grotenhuis, M. T., & Pelzer, B. (2013). The reliability of a two-item scale:
 Pearson, Cronbach, or Spearman-Brown? *International Journal of Public Health*, 58(4), 637-642. <u>https://doi.org/10.1007/s00038-012-0416-3</u>
- Field, A., Miles, J., & Field, Z. (2012). Discovering Statistics Using R. Sage Publications Ltd.
- Field, A. (2018). Discovering Statistics Using IBM SPSS Statistics. Sage Publications Ltd.
- Fiske, S. T. (1998). Stereotyping, prejudice, and discrimination. In Gilbert, D. T., Fiske, S. T., & Lindzey, G. (Eds.), *The Handbook of Social Psychology* (pp. 357–411).McGraw-Hill.
- Fiske, S. T. (2018). Stereotype content: Warmth and competence endure. *Current Directions in Psychological Science*, 27(2), 67-73. <u>https://doi.org/10.1177/0963721417738825</u>
- Fiske, S. T., Xu, J., Cuddy, A. C., & Glick, P. (1999). (Dis) respecting versus (dis) liking: Status and interdependence predict ambivalent stereotypes of competence and warmth. *Journal of Social Issues*, 55(3), 473-489. <u>https://doi.org/10.1111/0022-</u> <u>4537.00128</u>
- Fiske, S. T., Cuddy, A. J., Glick, P., & Xu, J. (2002). A model of (often mixed) stereotype content: Competence and warmth respectively follow from perceived status and competition. *Journal of Personality and Social Psychology*, 82(6), 878. https://doi.org/10.1037/pspa0000163
- Flores, A. R., Haider-Markel, D. P., Lewis, D. C., Miller, P. R., Tadlock, B. L., & Taylor, J. K. (2018). Challenged expectations: Mere exposure effects on attitudes about transgender people and rights. *Political Psychology*, 39(1), 197-216. <u>https://doi.org/10.1111/pops.12402</u>

- Floyd, F. J., & Widaman, K. F. (1995). Factor analysis in the development and refinement of clinical assessment instruments. *Psychological Assessment*, 7(3), 286. https://doi.org/10.1037/1040-3590.7.3.286
- Friborg, O., Martinussen, M., & Rosenvinge, J. H. (2006). Likert-based vs. semantic differential-based scorings of positive psychological constructs: A psychometric comparison of two versions of a scale measuring resilience. *Personality and Individual Differences, 40*(5), 873-884. <u>https://doi.org/10.1016/j.paid.2005.08.015</u>
- Fukuzawa, M., & Igarashi, S. (2017). Mirror reflection or real image: Does past mirror experience influence a dog's use of a mirror? *Journal of Veterinary Behavior*, 22, 7-12. <u>https://doi.org/10.1016/j.jveb.2017.09.005</u>
- Garner, R. (1995). The politics of animal protection: A research agenda. *Society & Animals*, *3*(1), 43-60.
- Garner, R. (2008). The politics of animal rights. *British Politics*, *3*(1), 110-119. https://doi.org/10.1057/palgrave.bp.4200080
- Gaunt, R., Leyens, J. P., & Demoulin, S. (2002). Intergroup relations and the attribution of emotions: Control over memory for secondary emotions associated with the ingroup and outgroup. *Journal of Experimental Social Psychology*, 38(5), 508-514. <u>https://doi.org/10.1016/S0022-1031(02)00014-8</u>
- Gerencsér, L., Pérez Fraga, P., Lovas, M., Újváry, D., & Andics, A. (2019). Comparing interspecific socio-communicative skills of socialized juvenile dogs and miniature pigs. *Animal Cognition*, 22(6), 917-929. <u>https://doi.org/10.1007/s10071-019-01284-z</u>
- Godfray, H. C. J., Aveyard, P., Garnett, T., Hall, J. W., Key, T. J., Lorimer, J., Pierrehumbert, R.T., Scarborough, P., Springmann, M., & Jebb, S. A. (2018). Meat consumption,

health, and the environment. Science, 361(6399), eaam5324.

https://doi.org/10.1126/science.aam5324

González, N., Marquès, M., Nadal, M., & Domingo, J. L. (2020). Meat consumption: Which are the current global risks? A review of recent (2010–2020) evidences. *Food Research International*, 137, 109341. <u>https://doi.org/10.1016/j.foodres.2020.109341</u>

Gorsuch, R. L. (1983). Factor Analysis (2nd ed.). Erlbaum.

- Graça, J., Calheiros, M. M., & Oliveira, A. (2014). Moral disengagement in harmful but cherished food practices? An exploration into the case of meat. *Journal of Agricultural and Environmental Ethics*, 27(5), 749-765. https://doi.org/10.1007/s10806-014-9488-9
- Graça, J., Calheiros, M. M., & Oliveira, A. (2016). Situating moral disengagement:
 Motivated reasoning in meat consumption and substitution. *Personality and Individual Differences*, 90, 353-364. <u>https://doi.org/10.1016/j.paid.2015.11.042</u>
- Gradidge, S., & Zawisza, M. (2019). Why factual appeals about the abilities of sheep may fail. *Animal Sentience*, *25*(42).
- Gradidge, S., Zawisza, M., Harvey, A. J., & McDermott, D. T. (2021a). A structured literature review of the meat paradox. *Social Psychological Bulletin*, 16(3), 1-26. <u>https://doi.org/10.32872/spb.5953</u>
- Gradidge, S., Harvey, A. J., McDermott, D. T., & Zawisza, M. (2021b). Humankind's best friend vs. humankind's best food: Perceptions of identifiable dog vs. pig victims. *Human-Animal Interaction Bulletin*, 2021. https://doi.org/10.1079/hai.2021.0010
- Gradidge, S., Zawisza, M., Harvey, A. J., & McDermott, D. T. (2022). Farmyard animal or best friend? Exploring predictors of dog vs. pig pet speciesism. *People and Animals:*

The International Journal of Research and Practice, *5*(1), 11. <u>https://docs.lib.purdue.edu/paij/vol5/iss1/11</u>

Gray, P. B., & Young, S. M. (2011). Human–pet dynamics in cross-cultural perspective. *Anthrozoös, 24*(1), 17-30. <u>https://doi.org/10.2752/175303711X12923300467285</u>

Hare, B., Brown, M., Williamson, C., & Tomasello, M. (2002). The domestication of social cognition in dogs. *Science*, 298(5598), 1634-1636. https://doi.org/10.1126/science.1072702

- Harmon-Jones, E. (2000). Cognitive dissonance and experienced negative affect: Evidence that dissonance increases experienced negative affect even in the absence of aversive consequences. *Personality and Social Psychology Bulletin, 26*(12), 1490-1501.
 https://doi.org/10.1177/01461672002612004
- Harmon-Jones, E., & Allen, J. J. (2001). The role of affect in the mere exposure effect:
 Evidence from psychophysiological and individual differences approaches. *Personality and Social Psychology Bulletin, 27*(7), 889-898.
 https://doi.org/10.1177/0146167201277011
- Harmon-Jones, E., Brehm, J. W., Greenberg, J., Simon, L., & Nelson, D. E. (1996). Evidence that the production of aversive consequences is not necessary to create cognitive dissonance. *Journal of Personality and Social Psychology*, 70(1), 5.
 https://doi.org/10.1037/0022-3514.70.1.5
- Harrison, M. A., & Hall, A. E. (2010). Anthropomorphism, empathy, and perceived communicative ability vary with phylogenetic relatedness to humans. *Journal of Social, Evolutionary, and Cultural Psychology*, *4*(1), 34.
 https://doi.org/10.1037/h0099303

- Hayes, A. F. (2009). Beyond Baron and Kenny: Statistical mediation analysis in the new millennium. *Communication Monographs*, 76(4), 408-420. https://doi.org/10.1080/03637750903310360
- Hayes, A. F. (2018). Introduction to Mediation, Moderation, and Conditional Process Analysis: A Regression-Based Approach (2nd ed.). Guilford Publications, Inc.
- Hayes, A. F. (2022). The PROCESS macro for SPSS, SAS, and R. *PROCESS Macro*. Retrieved from: <u>http://processmacro.org/download.html</u>
- Hayes, A. F., & Preacher, K. J. (2014). Statistical mediation analysis with a multicategorical independent variable. *British Journal of Mathematical and Statistical Psychology*, 67(3), 451-470. <u>https://doi.org/10.1111/bmsp.12028</u>
- Hazel, S. J., Signal, T. D., & Taylor, N. (2011). Can teaching veterinary and animal-science students about animal welfare affect their attitude toward animals and human-related empathy? *Journal of Veterinary Medical Education*, 38(1), 74-83. https://doi.org/10.3138/jvme.38.1.74
- Herzog, H. A. (2007). Gender differences in human–animal interactions: A review. *Anthrozoös*, 20(1), 7-21. <u>https://doi.org/10.2752/089279307780216687</u>
- Herzog, H. (2014). Biology, culture, and the origins of pet-keeping. *Animal Behavior and Cognition, 1*(3), 296-308. <u>https://doi.org/10.12966/abc.08.06.2014</u>
- Herzog, H. A., Betchart, N. S., & Pittman, R. B. (1991). Gender, sex role orientation, and attitudes toward animals. *Anthrozoös*, 4(3), 184-191. https://doi.org/10.2752/089279391787057170
- Higgs, M. J., Bipin, S., & Cassaday, H. J. (2020). Man's best friends: Attitudes towards the use of different kinds of animal depend on belief in different species' mental

capacities and purpose of use. *Royal Society Open Science*, 7(2), 191162. https://doi.org/10.1098/rsos.191162

- Hills, A. M. (1995). Empathy and belief in the mental experience of animals. *Anthrozoös*, 8(3), 132-142. <u>https://doi.org/10.2752/089279395787156347</u>
- Horard-Herbin, M. P., Tresset, A., & Vigne, J. D. (2014). Domestication and uses of the dog in western Europe from the Paleolithic to the Iron Age. *Animal Frontiers*, 4(3), 23-31. <u>https://doi.org/10.2527/af.2014-0018</u>
- Horback, K. (2014). Nosing around: Play in pigs. *Animal Behavior and Cognition*, 2(2), 186-186. <u>https://doi.org/10.12966/abc.05.08.2014</u>
- Horne, Z., Rottman, J., & Lawrence, C. (2021). Can coherence-based interventions change dogged moral beliefs about meat-eating? *Journal of Experimental Social Psychology*, 96, 104160. <u>https://doi.org/10.1016/j.jesp.2021.104160</u>
- Hurley, S. (2016). Human–canine relationships in China. In: Pręgowski, M. (ed.), Companion Animals in Everyday Life. Palgrave Macmillan. <u>https://doi.org/10.1057/978-1-137-59572-0_9</u>
- Husnu, S., & Crisp, R. J. (2010). Elaboration enhances the imagined contact effect. *Journal of Experimental Social Psychology*, 46(6), 943-950. https://doi.org/10.1016/j.jesp.2010.05.014
- Hyers, L. L. (2006). Myths used to legitimize the exploitation of animals: An application of social dominance theory. *Anthrozoös*, 19(3), 194-210. <u>https://doi.org/10.2752/089279306785415538</u>
- IPCC (2021). IPCC sixth assessment report. *IPCC*. Retrieved from: https://www.ipcc.ch/report/ar6/wg1/

- Islam, M. R., & Khan, N. A. (2018). Threats, vulnerability, resilience and displacement among the climate change and natural disaster-affected people in South-East Asia: An overview. *Journal of the Asia Pacific Economy*, 23(2), 297-323. <u>https://doi.org/10.1080/13547860.2018.1442153</u>
- Izuma, K., & Murayama, K. (2019). Neural basis of cognitive dissonance. In Harmon-Jones,
 E. (Ed.), *Cognitive Dissonance: Reexamining a Pivotal Theory in Psychology* (pp. 227–245). American Psychological Association. <u>https://doi.org/10.1037/0000135-011</u>
- Izuma, K., Matsumoto, M., Murayama, K., Samejima, K., Sadato, N., & Matsumoto, K. (2010). Neural correlates of cognitive dissonance and choice-induced preference change. *Proceedings of the National Academy of Sciences*, 107(51), 22014-22019. <u>https://doi.org/10.1073/pnas.1011879108</u>
- Jain, K., & Srinivasan, N. (1990). An empirical assessment of multiple operationalizations of involvement. Advances in Consumer Research, 17, 594-602.
- Johnson, G. R. (2022). How many vegetarians and vegans are in the UK? *Finder*. Available at: <u>https://www.finder.com/uk/uk-diet-</u> <u>trends#:~:text=The%20UK%27s%20current%20diet%20vs,the%20vegan%20diet%2</u> <u>0(1%25)</u>.
- Johnston, A. M., Turrin, C., Watson, L., Arre, A. M., & Santos, L. R. (2017). Uncovering the origins of dog–human eye contact: Dingoes establish eye contact more than wolves, but less than dogs. *Animal Behaviour*, *133*, 123-129.

https://doi.org/10.1016/j.anbehav.2017.09.002

Joy, M. (2010). *Why We Love Dogs, Eat Pigs, and Wear Cows: An Introduction to Carnism.* Red Wheel Weiser. Julious, S. A. (2005). Sample size of 12 per group rule of thumb for a pilot study.
 Pharmaceutical Statistics: The Journal of Applied Statistics in the Pharmaceutical
 Industry, 4(4), 287-291. <u>https://doi.org/10.1002/pst.185</u>

- Kaminski, J., Waller, B. M., Diogo, R., Hartstone-Rose, A., & Burrows, A. M. (2019).
 Evolution of facial muscle anatomy in dogs. *Proceedings of the National Academy of Sciences*, *116*(29), 14677-14681. <u>https://doi.org/10.1073/pnas.1820653116</u>
- Keen, R., & Tiemeier, H. (2022). Covariate selection from data collection onwards: A methodology for neurosurgeons. *World Neurosurgery*, 161, 245-250. https://doi.org/10.1016/j.wneu.2021.11.057
- Keesing, F., & Ostfeld, R. S. (2021). Impacts of biodiversity and biodiversity loss on zoonotic diseases. *Proceedings of the National Academy of Sciences*, 118(17). <u>https://doi.org/10.1073/pnas.202354011</u>
- Kelley, K. (2022, Oct 12). Package 'MBESS'. *CRAN*. Retrieved from: <u>https://cran.r-</u> project.org/web/packages/MBESS/MBESS.pdf
- Kelley, K., & Pornprasertmanit, S. (2016). Confidence intervals for population reliability coefficients: Evaluation of methods, recommendations, and software for composite measures. *Psychological Methods*, 21(1), 69. <u>https://doi.org/10.1037/a0040086</u>
- Kendall, H. A., Lobao, L. M., & Sharp, J. S. (2006). Public concern with animal well-being:
 Place, social structural location, and individual experience. *Rural Sociology*, *71*(3), 399-428. <u>https://doi.org/10.1526/003601106778070617</u>
- Kim, H. M. (2006). The effect of salience on mental accounting: How integration versus segregation of payment influences purchase decisions. *Journal of Behavioral Decision Making*, 19(4), 381–391. <u>https://doi.org/10.1002/bdm.534</u>

- Kim, H. Y. (2013). Statistical notes for clinical researchers: Assessing normal distribution (2) using skewness and kurtosis. *Restorative Dentistry & Endodontics*, 38(1), 52. <u>https://doi.org/10.5395/rde.2013.38.1.52</u>
- Köhler, W. (1947). Gestalt Psychology (2nd ed.). New York: Liveright.
- Koplin, J., & Wilkinson, D. (2019). Moral uncertainty and the farming of human-pig chimeras. *Journal of Medical Ethics*, 45(7), 440-446. <u>https://doi.org/10.1136/medethics-2018-105227</u>
- Kozachenko, H. H., & Piazza, J. (2021). How children and adults value different animal lives. *Journal of Experimental Child Psychology*, 210, 105204. <u>https://doi.org/10.1016/j.jecp.2021.105204</u>
- Krings, V. C., Dhont, K., & Salmen, A. (2021). The moral divide between high-and lowstatus animals: The role of human supremacy beliefs. *Anthrozoös*, 34(6), 787-802. <u>https://doi.org/10.1080/08927936.2021.1926712</u>
- Leach, S., Sutton, R. M., Dhont, K., & Douglas, K. M. (2021). When is it wrong to eat animals? The relevance of different animal traits and behaviours. *European Journal of Social Psychology*, 51(1), 113-123. <u>https://doi.org/10.1002/ejsp.2718</u>
- Leyens, J. P., Rodriguez-Perez, A., Rodriguez-Torres, R., Gaunt, R., Paladino, M. P., Vaes,
 J., & Demoulin, S. (2001). Psychological essentialism and the differential attribution of uniquely human emotions to ingroups and outgroups. *European Journal of Social Psychology*, 31(4), 395-411. https://doi.org/10.1002/ejsp.50
- Li, P. J., Sun, J., & Yu, D. (2017). Dog "meat" consumption in China: A survey of the controversial eating habit in two cities. *Society & Animals*, 25(6), 513-532. <u>https://doi.org/10.1163/15685306-12341471</u>

- Li, J., He, D., Zhou, L., Zhao, X., Zhao, T., Zhang, W., & He, X. (2019). The effects of facial attractiveness and familiarity on facial expression recognition. *Frontiers in Psychology*, 10, 2496. <u>https://doi.org/10.3389/fpsyg.2019.02496</u>
- López-Cepero, J., Martos-Montes, R., & Ordóñez, D. (2021). Classification of animals as pet, pest, or profit: Consistency and associated variables among Spanish university students. *Anthrozoös*, 34(6), 877-888.

https://doi.org/10.1080/08927936.2021.1938408

- Losey, R. J. (2021). Domestication is not an ancient moment of selection for prosociality: Insights from dogs and modern humans. *Journal of Social Archaeology*, 14696053211055475. <u>https://doi.org/10.1177/1469605321105547</u>
- Loughnan, S., Haslam, N., & Bastian, B. (2010). The role of meat consumption in the denial of moral status and mind to meat animals. *Appetite*, *55*(1), 156-159. <u>https://doi.org/10.1016/j.appet.2010.05.043</u>
- Loughnan, S., Bastian, B., & Haslam, N. (2014). The psychology of eating animals. Current Directions in Psychological Science, 23(2), 104-108. https://doi.org/10.1177/096372141452578
- Luna, D., & Kim, H. M. C. (2009). How much was your shopping basket? Working memory processes in total basket price estimation. *Journal of Consumer Psychology*, 19(3), 346-355, <u>https://doi.org/10.1016/j.jcps.2009.03.003</u>
- Lund, T. B., McKeegan, D. E., Cribbin, C., & Sandøe, P. (2016). Animal ethics profiling of vegetarians, vegans and meat-eaters. *Anthrozoös, 29*(1), 89-106. <u>https://doi.org/10.1080/08927936.2015.1083192</u>

- Macdiarmid, J. I., Douglas, F., & Campbell, J. (2016). Eating like there's no tomorrow:
 Public awareness of the environmental impact of food and reluctance to eat less meat as part of a sustainable diet. *Appetite*, 96, 487-493.
 https://doi.org/10.1016/j.appet.2015.10.011
- Machovina, B., Feeley, K. J., & Ripple, W. J. (2015). Biodiversity conservation: The key is reducing meat consumption. *Science of the Total Environment*, 536, 419-431. <u>https://doi.org/10.1016/j.scitotenv.2015.07.022</u>
- Mantyka-Pringle, C. S., Martin, T. G., & Rhodes, J. R. (2012). Interactions between climate and habitat loss effects on biodiversity: A systematic review and meta-analysis. *Global Change Biology*, 18(4), 1239-1252. <u>https://doi.org/10.1111/j.1365-2486.2011.02593.x</u>
- Marino, L., & Colvin, C. M. (2015). Thinking pigs: A comparative review of cognition, emotion, and personality in Sus domesticus. *International Journal of Comparative Psychology, 28,* Article 23859.
- Maurer, D., Pathman, T., & Mondloch, C. J. (2006). The shape of boubas: Sound–shape correspondences in toddlers and adults. *Developmental Science*, 9(3), 316-322. https://doi.org/10.1111/j.1467-7687.2006.00495.x
- McConnell, A. R., Paige Lloyd, E., & Humphrey, B. T. (2019). We are family: Viewing pets as family members improves wellbeing. *Anthrozoös*, 32(4), 459-470. <u>https://doi.org/10.1080/08927936.2019.1621516</u>
- McCulloch, S. (2022). The UK must ban farrowing crates. *Veterinary Record*, *191*(6), 262-262. <u>https://doi.org/10.1002/vetr.2264</u>

McLeman, M. A., Mendl, M., Jones, R. B., White, R., & Wathes, C. M. (2005).
Discrimination of conspecifics by juvenile domestic pigs, Sus scrofa. *Animal Behaviour*, 70(2), 451-461. <u>https://doi.org/10.1016/j.anbehav.2004.11.013</u>

- McLeman, M. A., Mendl, M. T., Jones, R. B., & Wathes, C. M. (2008). Social discrimination of familiar conspecifics by juvenile pigs, Sus scrofa: Development of a non-invasive method to study the transmission of unimodal and bimodal cues between live stimuli. *Applied Animal Behaviour Science*, *115*(3-4), 123-137. https://doi.org/10.1016/j.applanim.2008.06.010
- McManus-Fry, E., Knecht, R., Dobney, K., Richards, M. P., & Britton, K. (2018). Doghuman dietary relationships in Yup'ik western Alaska: The stable isotope and zooarchaeological evidence from pre-contact Nunalleq. *Journal of Archaeological Science: Reports, 17*, 964-972. <u>https://doi.org/10.1016/j.jasrep.2016.04.007</u>
- Miller, & Stoica, M. (2004). Comparing the effects of a photograph versus artistic renditions of a beach scene in a direct-response print ad for a Caribbean resort island: A mental imagery perspective. *Journal of Vacation Marketing*, 10(1), 11–21, <u>https://doi.org/10.1177/135676670301000102</u>
- Miralles, A., Raymond, M., & Lecointre, G. (2019). Empathy and compassion toward other species decrease with evolutionary divergence time. *Scientific Reports*, 9(1), 1-8. <u>https://doi.org/10.1038/s41598-019-56006-9</u>
- Monteiro, C. A., Pfeiler, T. M., Patterson, M. D., & Milburn, M. A. (2017). The Carnism Inventory: Measuring the ideology of eating animals. *Appetite*, 113, 51-62. <u>https://doi.org/10.1016/j.appet.2017.02.011</u>

- Montoya, R. M., Horton, R. S., & Kirchner, J. (2008). Is actual similarity necessary for attraction? A meta-analysis of actual and perceived similarity. *Journal of Social and Personal Relationships*, 25(6), 889-922. <u>https://doi.org/10.1177/0265407508096700</u>
- Morris, P., Knight, S., & Lesley, S. (2012). Belief in animal mind: Does familiarity with animals influence beliefs about animal emotions? *Society & Animals*, *20*(3), 211-224.
- Mrkva, K., & Van Boven, L. (2020). Salience theory of mere exposure: Relative exposure increases liking, extremity, and emotional intensity. *Journal of Personality and Social Psychology*, *118*(6), 1118. <u>https://doi.org/10.1037/pspa0000184</u>
- Mundfrom, D. J., Shaw, D. G., & Ke, T. L. (2005). Minimum sample size recommendations for conducting factor analyses. *International Journal of Testing*, 5(2), 159-168. <u>https://doi.org/10.1207/s15327574ijt0502_4</u>
- Munro, L. (2005). Strategies, action repertoires and DIY activism in the animal rights movement. Social Movement Studies, 4(1), 75-94. <u>https://doi.org/10.1080/14742830500051994</u>
- Neves, J., Pestana, J., & Giger, J. C. (2022). Applying the Stereotype Content Model (SCM) and BIAS Map to understand attitudinal and behavioral tendencies toward the conservation of sharks. *Anthrozoös*, 35(3), 371-391. https://doi.org/10.1080/08927936.2021.1999608
- Newell, B. R., & Shanks, D. R. (2007). Recognising what you like: Examining the relation between the mere-exposure effect and recognition. *European Journal of Cognitive Psychology*, 19(1), 103-118. <u>https://doi.org/10.1080/09541440500487454</u>

- Norton, M. I., Frost, J. H., & Ariely, D. (2007). Less is more: The lure of ambiguity, or why familiarity breeds contempt. *Journal of Personality and Social Psychology*, *92*(1), 97. https://psycnet.apa.org/doi/10.1037/0022-3514.92.1.97
- O'Neill, D. G., Pegram, C., Crocker, P., Brodbelt, D. C., Church, D. B., & Packer, R. M. A. (2020). Unravelling the health status of brachycephalic dogs in the UK using multivariable analysis. *Scientific Reports*, 10(1), 1-13. <u>https://doi.org/10.1038/s41598-020-73088-y</u>
- Oh, M., & Jackson, J. (2011). Animal rights vs. cultural rights: Exploring the dog meat debate in South Korea from a world polity perspective. *Journal of Intercultural Studies*, 32(1), 31-56. <u>https://doi.org/10.1080/07256868.2010.491272</u>
- Packer, R. M., O'Neill, D. G., Fletcher, F., & Farnworth, M. J. (2019). Great expectations, inconvenient truths, and the paradoxes of the dog-owner relationship for owners of brachycephalic dogs. *PLoS One*, *14*(7), e0219918. <u>https://doi.org/10.1371/journal.pone.0219918</u>
- Pardo, A., & Román, M. (2013). Reflections on the Baron and Kenny model of statistical mediation. *Anales de Psicologia*, 29(2), 614-623. https://doi.org/10.6018/analesps.29.2.139241
- Pérez Fraga, P., Gerencsér, L., Lovas, M., Újváry, D., & Andics, A. (2021). Who turns to the human? Companion pigs' and dogs' behaviour in the unsolvable task paradigm. *Animal Cognition*, 24(1), 33-40. <u>https://doi.org/10.1007/s10071-020-01410-2</u>
- Pettigrew, T. F. (1998). Intergroup contact theory. *Annual Review of Psychology*, 49(1), 65-85. <u>https://doi.org/10.1146/annurev.psych.49.1.65</u>

Pettigrew, T. F., & Tropp, L. R. (2006). A meta-analytic test of intergroup contact theory. Journal of Personality and Social Psychology, 90(5), 751. https://doi.org/10.1037/0022-3514.90.5.751

- Pettigrew, T. F., & Tropp, L. R. (2008). How does intergroup contact reduce prejudice?
 Meta-analytic tests of three mediators. *European Journal of Social Psychology*, 38(6), 922-934. <u>https://doi.org/10.1002/ejsp.504</u>
- Philip, S.Y., Kew, S.F., van Oldenborgh, G.J., Yang, W., Vecchi, G.A., Anslow, F.S., Li, S., Seneviratne, S.I., Luu, L.N., Arrighi, J., Singh, R., van Aalst, M., Hauser, M., Schumacher, D.L., Marghidan, C.P., Ebi, K.L., Bonnet, R., Vautard, R., Tradowsky, J., Coumou, D., Lehner, F., Wehner, M., Rodell, C., Stull, R., Howard, R., Gillett, N., Otto, F.E.L. (2021). Rapid attribution analysis of the extraordinary heatwave on the Pacific Coast of the US and Canada June 2021. *World Weather Attribution*. Retrieved from: <u>https://www.worldweatherattribution.org/wp-content/uploads/NW-US-extremeheat-2021-scientific-report-WWA.pdf</u>
- Piazza, J., & Loughnan, S. (2016). When meat gets personal, animals' minds matter less: Motivated use of intelligence information in judgments of moral standing. *Social Psychological and Personality Science*, 7(8), 867-874. https://doi.org/10.1177/1948550616660159
- Piazza, J., Landy, J. F., & Goodwin, G. P. (2014). Cruel nature: Harmfulness as an important, overlooked dimension in judgments of moral standing. *Cognition*, 131(1), 108-124. <u>https://doi.org/10.1016/j.cognition.2013.12.013</u>
- Piazza, J., Ruby, M. B., Loughnan, S., Luong, M., Kulik, J., Watkins, H. M., & Seigerman, M. (2015). Rationalizing meat consumption. The 4Ns. *Appetite*, 91, 114-128. <u>https://doi.org/10.1016/j.appet.2015.04.011</u>

- Podberscek, A. L. (2009). Good to pet and eat: The keeping and consuming of dogs and cats in South Korea. *Journal of Social Issues*, 65(3), 615-632. <u>https://doi.org/10.1111/j.1540-4560.2009.01616.x</u>
- Podgórski, T., Lusseau, D., Scandura, M., Sönnichsen, L., & Jędrzejewska, B. (2014). Longlasting, kin-directed female interactions in a spatially structured wild boar social network. *PLoS One*, 9(6), e99875. <u>https://doi.org/10.1371/journal.pone.0099875</u>
- Poon, S. W. (2014). Dogs and British colonialism: The contested ban on eating dogs in colonial Hong Kong. *The Journal of Imperial and Commonwealth History*, 42(2), 308-328. <u>https://doi.org/10.1080/03086534.2013.851873</u>
- Possidónio, C., Graça, J., Piazza, J., & Prada, M. (2019). Animal images database: Validation of 120 images for human-animal studies. *Animals*, 9(8), 475. <u>https://doi.org/10.3390/ani9080475</u>
- Powell, G. M. (2010). The role of individual differences and involvement on attitudes toward animal welfare (Doctoral dissertation, Kansas State University).
- Prguda, E., & Neumann, D. L. (2014). Inter-human and animal-directed empathy: A test for evolutionary biases in empathetic responding. *Behavioural Processes*, 108, 80-86. <u>https://doi.org/10.1016/j.beproc.2014.09.012</u>
- Price, M., Meier, J., & Arbuckle, B. (2020). Canine economies of the ancient Near East and Eastern Mediterranean. *Journal of Field Archaeology*, 46(2), 81-92. <u>https://doi.org/10.1080/00934690.2020.1848322</u>
- Racca, A., Amadei, E., Ligout, S., Guo, K., Meints, K., & Mills, D. (2010). Discrimination of human and dog faces and inversion responses in domestic dogs (Canis familiaris).
 Animal Cognition, 13(3), 525-533. <u>https://doi.org/10.1007/s10071-009-0303-3</u>

- Ramachandran, V. S., & Hubbard, E. M. (2001). Synaesthesia A window into perception, thought and language. *Journal of Consciousness Studies*, 8(12), 3-34.
- Reimert, I., Bolhuis, J. E., Kemp, B., & Rodenburg, T. B. (2013). Indicators of positive and negative emotions and emotional contagion in pigs. *Physiology & Behavior*, 109, 42-50. <u>https://doi.org/10.1016/j.physbeh.2012.11.002</u>
- Reimert, I., Bolhuis, J. E., Kemp, B., & Rodenburg, T. B. (2015). Emotions on the loose:
 Emotional contagion and the role of oxytocin in pigs. *Animal Cognition*, 18(2), 517-532. <u>https://doi.org/10.1007/s10071-014-0820-6</u>
- Reis, H. T., Maniaci, M. R., Caprariello, P. A., Eastwick, P. W., & Finkel, E. J. (2011). Familiarity does indeed promote attraction in live interaction. *Journal of Personality* and Social Psychology, 101(3), 557. <u>https://doi.org/10.1037/a0022885</u>
- Richardson, J. T. (2011). Eta squared and partial eta squared as measures of effect size in educational research. *Educational Research Review*, 6(2), 135–147. <u>https://doi.org/10.1016/j.edurev.2010.12.001</u>
- Robbins, J. A., Danielson, J. A., Johnson, A. K., Parsons, R. L., Jorgensen, M. W., & Millman, S. T. (2021). Attitudes towards animals and belief in animal mind among first-year veterinary students before and after an introductory animal welfare course. *Animal Welfare*, 30(4), 409-418. <u>https://doi.org/10.7120/09627286.30.4.004</u>
- Roberts, S. E. (2017). The dog days of winter: Indigenous dogs, Indian hunters, and wintertime subsistence in the Northeast. *Northeastern Naturalist, 24*(sp7). https://doi.org/10.1656/045.024.s710
- Rodríguez-Pérez, A., Delgado-Rodríguez, N., Betancor-Rodríguez, V., Leyens, J. P., & Vaes, J. (2011). Infra-humanization of outgroups throughout the world. The role of
similarity, intergroup friendship, knowledge of the outgroup, and status. *Anales de Psicología*, *27*(3), 679-687.

- Rothgerber, H. (2014). Efforts to overcome vegetarian-induced dissonance among meat eaters. *Appetite*, 79, 32-41. <u>https://doi.org/10.1016/j.appet.2014.04.003</u>
- Rothgerber, H. (2015). Can you have your meat and eat it too? Conscientious omnivores, vegetarians, and adherence to diet. *Appetite*, *84*, 196-203. https://doi.org/10.1016/j.appet.2014.10.012
- Rothgerber, H. (2019). "But I don't eat that much meat": Situational underreporting of meat consumption by women. *Society & Animals*, *27*(2), 150-173.
- Rothgerber, H. (2020). Meat-related cognitive dissonance: A conceptual framework for understanding how meat eaters reduce negative arousal from eating animals. *Appetite*, 146, 104511. <u>https://doi.org/10.1016/j.appet.2019.104511</u>
- RSPCA (2022, August 2). New figures reveal increase in dog cruelty since start of the pandemic. *RSPCA*. Retrieved from: <u>https://www.rspca.org.uk/-/news-figures-revel-increase-in-dog-abuse-since-pandemic</u>
- Serpell, J. A. (2004). Factors influencing human attitudes to animals and their welfare. Animal Welfare.
- Sevillano, V., & Fiske, S. T. (2016a). Animals as social objects: Groups, stereotypes, and intergroup threats. *European Psychologist*, 21(3), 206. <u>https://doi.org/10.1027/1016-</u> 9040/a000268
- Sevillano, V., & Fiske, S. T. (2016b). Warmth and competence in animals. *Journal of Applied Social Psychology*, 46(5), 276-293. <u>https://doi.org/10.1111/jasp.12361</u>

Sevillano, V., & Fiske, S. T. (2019). Stereotypes, emotions, and behaviors associated with animals: A causal test of the stereotype content model and BIAS map. *Group Processes & Intergroup Relations*, 22(6), 879-900. <u>https://doi.org/10.1177/1368430219851560</u>

- Shah, A. M., & Han, J. J. (2022). First successful porcine to human heart transplantation performed in the United States. *Artificial Organs*, 46(4), 543-545. <u>https://doi.org/10.1111/aor.14203</u>
- Signal, T. D., & Taylor, N. (2007). Attitude to animals and empathy: Comparing animal protection and general community samples. *Anthrozoös*, 20(2), 125-130. <u>https://doi.org/10.2752/175303707X207918</u>
- Signal, T., Taylor, N., & Maclean, A. S. (2018). Pampered or pariah: Does animal type influence the interaction between animal attitude and empathy? *Psychology, Crime & Law, 24*(5), 527-537. <u>https://doi.org/10.1080/1068316X.2017.1399394</u>

Singer, P. (1990). Animal Liberation. Jonathan Cape Ltd.

Smith, M., Murrell, J. C., & Mendl, M. (2021). Two assays of working memory in companion dogs: The holeboard and disappearing object tasks. *Applied Animal Behaviour Science*, 234, 105179. <u>https://doi.org/10.1016/j.applanim.2020.105179</u>

Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B. L., Lassaletta, L., de Vries, W., Vermeulen, S. J., Herrero, M., Carlson, K. M., Jonell, M., Troell, M., DeClerck, F., Gordon, L. J., Zurayk, R., Scarborough, P., Rayner, M., Loken, B., Fanzo, J., Godfray, H.C.J., Tilmna, D., Rockström, J., & Willett, W. (2018). Options for keeping the food system within environmental limits. *Nature*, *562*(7728), 519–525. <u>https://doi.org/10.1038/s41586-018-0594-0</u>

- Stafford, T., & Grimes, A. (2012). Memory enhances the mere exposure effect. *Psychology* & *Marketing*, *29*(12), 995-1003. <u>https://doi.org/10.1002/mar.20581</u>
- Swami, V. (2021). Attraction Explained: The Science of How We Form Relationships. Routledge.
- Sytsma, J., & Machery, E. (2012). The two sources of moral standing. *Review of Philosophy* and Psychology, 3(3), 303-324. <u>https://doi.org/10.1007/s13164-012-0102-7</u>

Tabachnick, B. G., & Fidell, L. S. (2014). Using Multivariate Statistics (6th ed.). Pearson.

- Tabachnick, B. G., & Fidell, L. S. (2019). Using Multivariate Statistics (7th ed.) Allyn & Bacon.
- Tausch, N., & Hewstone, M. (2010). Intergroup contact and prejudice. In Dovidio, J. F., Hewstone, M., Glick, P., & Esses, V. M. (Eds.) *The Sage Handbook of Prejudice, Stereotyping, and Discrimination* (pp. 544-560). Sage Publications Ltd.
- Taylor, N., & Signal, T. D. (2005). Empathy and attitudes to animals. *Anthrozoös*, *18*(1), 18-27. <u>https://doi.org/10.2752/089279305785594342</u>
- Taylor, N., & Signal, T. D. (2009). Pet, pest, profit: Isolating differences in attitudes towards the treatment of animals. *Anthrozoös*, 22(2), 129-135. https://doi.org/10.2752/175303709X434158
- Tisdell, C., Nantha, H. S., & Wilson, C. (2007). Endangerment and likeability of wildlife species: How important are they for payments proposed for conservation? *Ecological Economics*, 60(3), 627-633. <u>https://doi.org/10.1016/j.ecolecon.2006.01.007</u>
- Topolski, R., Weaver, J. N., Martin, Z., & McCoy, J. (2013). Choosing between the emotional dog and the rational pal: A moral dilemma with a tail. *Anthrozoös, 26*(2), 253-263. <u>https://doi.org/10.2752/175303713X13636846944321</u>

- Vaes, J., Paladino, M. P., & Leyens, J. P. (2002). The lost e-mail: Prosocial reactions induced by uniquely human emotions. *British Journal of Social Psychology*, 41(4), 521-534. <u>https://doi.org/10.1348/014466602321149867</u>
- Valiente-Banuet, A., Aizen, M. A., Alcántara, J. M., Arroyo, J., Cocucci, A., Galetti, M., Garcia, M. B., Garcia, D., Gómez, J. M., Jordano, P., Medel, R., Navarro, L., Obeso, J. R., Oviedo, R., Ramirez, N., Rey, P. J., Traveset, A., Verdú, M., & Zamora, R. (2015). Beyond species loss: The extinction of ecological interactions in a changing world. *Functional Ecology*, *29*(3), 299-307. <u>https://doi.org/10.1111/1365-2435.12356</u>
- Vallejo, J. R., Santos-Fita, D., & González, J. A. (2017). The therapeutic use of the dog in Spain: A review from a historical and cross-cultural perspective of a change in the human-dog relationship. *Journal of Ethnobiology and Ethnomedicine*, 13(1), 1-17.
 https://doi.org/10.1186/s13002-017-0175-6
- Vaske, J. J., Beaman, J., & Sponarski, C. C. (2017). Rethinking internal consistency in Cronbach's alpha. *Leisure Sciences*, 39(2), 163-173. https://doi.org/10.1080/01490400.2015.1127189
- Vezzali, L., Capozza, D., Stathi, S., & Giovannini, D. (2012). Increasing outgroup trust, reducing infrahumanization, and enhancing future contact intentions via imagined intergroup contact. *Journal of Experimental Social Psychology*, 48(1), 437-440. <u>https://doi.org/10.1016/j.jesp.2011.09.008</u>

Walton, D. (2010). Appeal to Expert Opinion: Arguments from Authority. Penn State Press.

Ward, C., Bauer, E. B., & Smuts, B. B. (2008). Partner preferences and asymmetries in social play among domestic dog, Canis lupus familiaris, littermates. *Animal Behaviour*, 76(4), 1187-1199. <u>https://doi.org/10.1016/j.anbehav.2008.06.004</u>

- Webb, T. L., & Sheeran, P. (2006). Does changing behavioral intentions engender behavior change? A meta-analysis of the experimental evidence. *Psychological Bulletin*, *132*(2), 249–268. <u>https://doi.org/10.1037/0033-2909.132.2.249</u>
- West, S. G., Finch, J. F., & Curran, P. J. (1995). Structural equation models with nonnormal variables: Problems and remedies. In Hoyle, R. H. (Ed.), *Structural Equation Modeling: Concepts, Issues and Applications*. SAGE Publications.
- Westbury, H. R., & Neumann, D. L. (2008). Empathy-related responses to moving film stimuli depicting human and non-human animal targets in negative circumstances. *Biological Psychology*, 78(1), 66-74. <u>https://doi.org/10.1016/j.biopsycho.2007.12.009</u>
- Wilks, M., Caviola, L., Kahane, G., & Bloom, P. (2021). Children prioritize humans over animals less than adults do. *Psychological Science*, 32(1), 27-38. <u>https://doi.org/10.1177/0956797620960398</u>
- Wilkins, A. M., McCrae, L. S., & McBride, E. A. (2015). Factors affecting the human attribution of emotions toward animals. *Anthrozoös*, 28(3), 357-369. <u>https://doi.org/10.1080/08927936.2015.1052270</u>
- YouGov (2014). YouGov survey results. *YouGov*. Retrieved from: <u>http://cdn.yougov.com/cumulus_uploads/document/n9y67k53la/YG-Archive-140319-</u> <u>Fears.pdf</u>
- Zajonc, R. B. (1968). Attitudinal effects of mere exposure. *Journal of Personality and Social Psychology*, 9(2p2), 1.
- Zebrowitz, L. A., Bronstad, P. M., & Lee, H. K. (2007). The contribution of face familiarity to ingroup favoritism and stereotyping. *Social Cognition*, 25(2), 306. <u>https://doi.org/10.1521/soco.2007.25.2.306</u>

- Zebrowitz, L. A., White, B., & Wieneke, K. (2008). Mere exposure and racial prejudice:
 Exposure to other-race faces increases liking for strangers of that race. *Social Cognition, 26*(3), 259. <u>https://doi.org/10.1521/soco.2008.26.3.259</u>
- Zhao, X., Lynch, J. G., & Chen, Q. (2010). Reconsidering Baron and Kenny: Myths and truths about mediation analysis. *Journal of Consumer Research*, 37(2), 197-206. <u>https://doi.org/10.1086/651257</u>

Appendices

Appendix 1. Study 1 Full Measures

a. Empathy Towards Animals Scale (Powell, 2010, adapted from Davis, 1983)

Empathic concern subscale:

- 1. I often have tender, concerned feelings for animals who suffer misfortune
- 2. Sometimes I don't feel very sorry for animals in need*
- 3. When I see an animal being taken advantage of, I feel kind of protective toward them
- 4. The misfortunes of animals do not usually disturb me a great deal*
- When I see an animal being treated unfairly, I sometimes don't feel very much pity for them*
- 6. I am often quite touched by things I see happen to animals
- 7. I would describe myself as a pretty soft-hearted person

Perspective taking subscale:

- 8. Before scolding an animal, I try to imagine how I would feel if I were in their place
- 9. I sometimes try to understand by pets better by imagining how things look from their perspective
- 10. I sometimes find it difficult to see things from an animal's point of view*
- 11. When I'm upset at my pet, I usually try to "put myself in his shoes" for a while
- 12. When I see an animal in need, I imagine what my life would be like if I were an animal in the same situation

*This item is reverse-scored.

b. Animal Utility Scale (Kendall et al., 2006)

- As long as animals do not suffer pain, humans should be able to use them for any purpose
- 2. It is acceptable to use animals to test consumer products such as soaps, cosmetics, and household cleaners
- 3. Hunting animals for sport is an acceptable form of recreation
- c. Adapted Version of the Product Involvement Scale (Jain & Srinivasan, 1990; Kim, 2006; Luna & Kim, 2009)
- 1. I attach great importance to products made from pigs (e.g., pork, ham)
- 2. I am very interested in products made from pigs (e.g., pork, ham)
- 3. I am not indifferent to products made from pigs (e.g., pork, ham)*

*This third item was removed from the scale due to inadequate loading on the factor,

inadequate communality, and inadequate inter-item correlations with the first two items.

Appendix 2. Principal Axis Factor Analyses

Principal axis factoring does not require multivariate normality (Floyd & Widaman, 1995; Gorsuch, 1983), and so normality was not tested for any PAFAs. Inter-item correlations, factor loadings, and communalities are discussed in the relevant sections in the main text, as these results informed whether items were retained in a scale or removed. Oblique direct oblimin rotations were implemented for all PAFAs to allow factors to correlate (although all PAFAs revealed only one factor for each scale, as assessed by eigenvalues \geq one and through visual inspection of scree plots).

Assumptions specific to each PAFA are detailed below. General rules were applied for assumptions as follows: Firstly, recommended sample sizes for factor analysis are highly variable (see Field et al., 2012; Mundfrom et al., 2005), with some recommending absolute sample sizes (e.g., at least 200), and others recommending sample sizes dependent on the number of items in the scale (e.g., 20 participants per item; Mundfrom et al., 2005). The PAFAs in the current study revealed some communality values on the lower end of the scale, as well as scales having relatively small item to factor ratios (e.g., mostly 3:1 or 4:1, except for the active harm subscale at 5:1, as all scales have one factor only, and scales range from three to five items). Whilst an exact minimum sample size cannot be calculated, the sample sizes of the current studies (*Ns*=128 to 215) may be inadequate, as factor analyses with lower communalities and lower item to factor ratios require larger sample sizes (Mundfrom et al., 2005). I therefore proceeded with the PAFAs, but results may need to be interpreted with caution due to possibly inadequate sample sizes, especially for the PAFA on the premanipulation active help subscale in Study 6 (*N*=128).

Secondly, the Kaiser-Meyer-Olkin (KMO) measure of sampling adequacy for all PAFAs should ideally be > .8, although > .6 can still be considered acceptable (Backhaus et al., 2016), whilst Bartlett's test of sphericity should be significant for all PAFAs (Bartlett, 1951).

352

These assumptions ensure there is enough common variance among items for factor analysis to be appropriate.

a. Study 1 Animal Utility Scale

Both the KMO (.61) and Bartlett's test of sphericity, $\chi^2(3) = 74.57$, p < .001, were adequate, suggesting that factor analysis is appropriate for this scale. One factor was extracted, which explained 56.43% of total variance.

b. Study 1 Adapted Product Involvement Scale to Test Subjective Self-Relevance of Pigs

Whilst Bartlett's test of sphericity, $\chi^2(3) = 193.05$, p < .001, was adequate, KMO was inadequate (.56), suggesting that factor analysis is inappropriate for this scale. This lack of common variance is reflected in the extremely low communality value of the third item (alongside the third item's inadequate factor loading and correlations with the other two variables), hence the third item was removed from the scale.

c. Study 3 Adapted Warmth Subscale

Both the KMO (pre-manipulation: .72; post-manipulation .73) and Bartlett's test of sphericity, pre-manipulation $\chi^2(3) = 243.58$, post-manipulation $\chi^2(3) = 357.91$, *ps* < .001, were adequate at both timepoints, suggesting that factor analysis is appropriate for this scale. One factor was extracted at both timepoints, which explained 74.81% (pre-manipulation) and 80.8% (post-manipulation) of total variance.

d. Study 3 Adapted Competence Subscale

Both the KMO (pre-manipulation: .73; post-manipulation .76) and Bartlett's test of sphericity, pre-manipulation $\chi^2(3) = 355.37$, post-manipulation $\chi^2(3) = 528.03$, *ps* < .001, were adequate at both timepoints, suggesting that factor analysis is appropriate for this scale.

One factor was extracted at both timepoints, which explained 81.17% (pre-manipulation) and 88.02% (post-manipulation) of total variance.

e. Study 3 Adapted Active Help Subscale

Both the KMO (pre-manipulation: .81; post-manipulation .81) and Bartlett's test of sphericity, pre-manipulation $\chi^2(6) = 333.65$, post-manipulation $\chi^2(6) = 517.62$, ps < .001, were adequate at both timepoints, suggesting that factor analysis is appropriate for this scale. One factor was extracted at both timepoints, which explained 67.93% (pre-manipulation) and 75.6% (post-manipulation) of total variance.

f. Study 3 Adapted Active Harm Subscale

Both the KMO (pre-manipulation: .83; post-manipulation .82) and Bartlett's test of sphericity, pre-manipulation $\chi^2(10) = 338.66$, post-manipulation $\chi^2(10) = 404.69$, ps < .001, were adequate at both timepoints, suggesting that factor analysis is appropriate for this scale. One factor was extracted at both timepoints, which explained 57.01% (pre-manipulation) and 59.92% (post-manipulation) of total variance.

g. Study 4 Cognitive Capacities Subscale

Both the KMO (.64) and Bartlett's test of sphericity, $\chi^2(6) = 49.57$, p < .001, were adequate, suggesting that factor analysis is appropriate for this scale. One factor was extracted, which explained 47.79% of total variance.

h. Study 6 Pre-Manipulation Active Help Subscale

Both the KMO (.73) and Bartlett's test of sphericity, $\chi^2(6) = 74.83$, p < .001, were adequate, suggesting that factor analysis is appropriate for this scale. One factor was extracted, which explained 51.37% of total variance.

Appendix 3. Study 1 Participant Advertisements

a. Sona

Study Name: The effect of different characteristics on perceptions of animals

Description: I am conducting this 15 min study to examine possible factors explaining our preferences for different animals. You will be given a participant information sheet describing the study and then be asked to provide informed consent. You will then complete scales measuring perceptions of animals and pig products. A debrief and researchers' contact details will follow.

Eligibility: You must be 18 years of age or over due to ability to give informed consent.

Duration: 15 minutes

Credits: 0.25 credits

Researcher: Sarah Gradidge, sarah.gradidge@pgr.anglia.ac.uk

This study has received ethics approval by the Psychology School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

b. Social Media

Got 15 minutes and over 18? Want 0.25 SONA credits? Participate in my online survey on animal perception. More details under the link below. Thanks for participating and please share with others!

Qualtrics link here

This study has received ethics approval by the Psychology School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

c. Poster



d. Flyer

<u>Complete a survey</u> <u>investigating people's</u> <u>perceptions of animals!</u>

18 years of age or over?

Receive 0.25 SONA credits for a 15-min study!

Access the survey at https://tinyurl.com/yy3eo7yj or scan the QR code. Thank you for participating!



This study has received ethics approval by the Psychology School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

Appendix 4. Statistics Tables

Table A1

Inferential Statistics for Study 1 Multiple Regressions on Dog Warmth, Dog Competence, Pig Warmth, and Pig Competence

Regressor	B	SE	F	$\eta_{ m p}{}^2$	Adj. R^2
(OV1) Dog Warmth			12.45***		.45
Pigs' Behavioural Self-Relevance	.18	.09	4.08*	.02	
Pigs' Subjective Self-Relevance	03	.09	.11	.001	
Dog Familiarity	.56	.15	14.77***	.07	
Pig Familiarity	18	.13	1.76	.01	
Dog Similarity	.43	.13	10.17**	.05	
Pig Similarity	2	.15	1.85	.01	
Dog Pet Status	.72	.19	14***	.07	
Pig Pet Status	.47	.12	16.66***	.08	
Dog Profit Status	39	.12	11.59***	.06	
Pig Profit Status	.05	.09	.27	.001	
Empathic Concern Towards Animals	.12	.03	12.04***	.06	
Perspective Taking of Animals	.03	.03	1.12	.01	
Support for Animal Utility	05	.04	1.16	.01	
(Covariate) Diet	.34	.31	1.22	.01	
(Covariate) Gender	.24	.32	.57	.003	
(OV2) Dog Competence			6.38***		.28
Pigs' Behavioural Self-Relevance	07	11	20	0.02	
Pigs' Subjective Self-Relevance	.07 .04	.11 .11	.39 .11	.002 .001	
Dog Familiarity	05	.18	.06	<.001	
Pig Familiarity	04	.17	.05	<.001	
Dog Similarity	.38	.17	5.23*	.03	

Pig Similarity	16	.18	.79	.004	
Dog Pet Status	.66	.24	7.81**	.04	
Pig Pet Status	.52	.14	13.6***	.07	
Dog Profit Status	14	.14	.95	.01	
Pig Profit Status	17	.12	2.12	.01	
Empathic Concern Towards Animals	.13	.04	9.73**	.05	
Perspective Taking of Animals	.07	.04	3.96*	.02	
Support for Animal Utility	05	.05	.87	.004	
(Covariate) Diet	.6	.38	2.48	.01	
(Covariate) Gender	.08	.39	.04	<.001	
(OV3) Pig Warmth			8.3***		.34
Pigs' Behavioural Self-Relevance	.36	.13	7.15**	.04	
Pigs' Subjective Self-Relevance	31	.14	5.08*	.03	
Dog Familiarity	.25	.22	1.35	.01	
Pig Familiarity	.21	.2	1.07	.01	
Dog Similarity	22	.2	1.16	.01	
Pig Similarity	.5	.22	5.12*	.03	
Dog Pet Status	.84	.29	8.56**	.04	
Pig Pet Status	.64	.17	13.69***	.07	
Dog Profit Status	.11	.17	.39	.002	
Pig Profit Status	15	.14	1.18	.01	
Empathic Concern Towards Animals	.14	.05	7.69**	.04	
Perspective Taking of Animals	01	.05	.04	<.001	
Support for Animal Utility	1	.07	2.41	.01	
(Covariate) Diet	.16	.46	.12	.001	
(Covariate) Gender	.67	.47	1.99	.01	

(OV4) Pig Competence			8.41***		.35
Pigs' Behavioural Self-Relevance	.01	.14	.48	.002	
Pigs' Subjective Self-Relevance	27	.15	3.48	.02	
Dog Familiarity	16	.23	.46	.002	
Pig Familiarity	.56	.21	7.04**	.04	
Dog Similarity	16	.21	.6	.003	
Pig Similarity	.79	.23	11.95***	.06	
Dog Pet Status	.94	.3	9.74**	.05	
Pig Pet Status	.52	.18	8.28**	.04	
Dog Profit Status	004	.18	<.001	<.001	
Pig Profit Status	01	.15	.003	<.001	
Empathic Concern Towards Animals	.08	.05	2.52	.01	
Perspective Taking of Animals	.03	.05	.37	.002	
Support for Animal Utility	06	.07	.66	.003	
(Covariate) Diet	.6	.48	1.55	.01	
(Covariate) Gender	.29	.5	.33	.002	

Note. * $p \le .05$, ** $p \le .01$, *** $p \le .001$. OV refers to outcome variable.

Table A2Inferential State

Inferential Statistics for Study 2 2x2x2 MANCOVA on SCM/BIAS Map Variables		
DV	F	${\eta_{ m p}}^2$
<i>(IVs)</i> Behavioural Self-Relevance x Subjective Self-Relevance x Time Interaction	1.92	.05
(IV) Behavioural Self-Relevance x Time Interaction	.76	.02
(IV) Subjective Self-Relevance x Time Interaction	1.08	.03
(Covariate) Diet x Time Interaction	.42	.01
(Covariate) Gender x Time Interaction	1.25	.03

<i>(IVs)</i> Behavioural Self-Relevance x Subjective Self-Relevance Interaction	1.9	.05
(IV) Behavioural Self-Relevance Main Effect	1.33	.04
(IV) Subjective Self-Relevance Main Effect	.59	.02
(Covariate) Diet Main Effect	2.36*	.06
Warmth	1.31	.01
Competence	5.16*	.02
Active Help	7.75**	.03
Passive Help	9.99**	.04
Active Harm	.15	.001
Passive Harm	.29	.001
(Covariate) Gender Main Effect	3.31**	.08
Warmth	1.48	.01
Competence	6.55*	.03
Active Help	.06	<.001
Passive Help	.63	.003
Active Harm	8.91**	.04
Passive Harm	3.61	.02
Time Main Effect	2.71*	.07
Warmth	9.91**	.04
Competence	.56	.003
Active Help	5.43**	.02
Passive Help	2.66	.01
Active Harm	.78	.004
Passive Harm	4.07*	.02

Table A3

DV	F	${\eta_{ m p}}^2$
<i>(IVs) Behavioural Self-Relevance x Subjective Self-Relevance Interaction</i>	1.04	.01
(IV) Behavioural Self-Relevance Main Effect	4.47**	.06
Reported Weekly Meat Consumption	.3	.001
Post-Manipulation Dissonance	12.51***	.05
Post-Evaluation Dissonance	.78	.004
(IV) Subjective Self-Relevance Main Effect	5.29**	.07
Reported Weekly Meat Consumption	1.21	.01
Post-Manipulation Dissonance	9.01**	.04
Post-Evaluation Dissonance	.83	.004
(Covariate) Diet Main Effect	88.16***	.55
Reported Weekly Meat Consumption	257.58***	.54
Post-Manipulation Dissonance	10.73***	.05
Post-Evaluation Dissonance	.67	.003
(Covariate) Gender Main Effect	2.91*	.04
Reported Weekly Meat Consumption	3.59	.02
Post-Manipulation Dissonance	5.29*	.02
Post-Evaluation Dissonance	.39	.002

Inferential Statistics for Study 2 2x2 MANCOVA on Reported Weekly Meat Consumption, Post-Manipulation Dissonance and Post-Evaluation Dissonance

Table A4

Inferential Statistics for Study 3 3x2 ANCOVA for Subjective Self-Relevance Manipulation Check

Effect	F	${\eta_{ m p}}^2$
(IVs) Subjective Self-Relevance x Species Interaction	1.48	.01
(IV) Subjective Self-Relevance Main Effect	261.89***	.56
(IV) Species Main Effect	.14	.001
(Covariate) Neutrality	.62	.003
(Covariate) Gender	2.35	.01
<i>Note.</i> * $p \le .05$, ** $p \le .01$, *** $p \le .001$. IV = independent variable		

Table A5

Inferential Statistics for Study 3 3x2x2 MANCOVA on SCM/BIAS Map Variables, Reported Weekly Meat Consumption, Discomfort, and Behavioural Self-Relevance

Dv	F	$\eta_{ m p}$ 2
<i>(IVs)</i> Behavioural Self-Relevance/Time x Subjective Self- Relevance x Species Interaction	1.03	.04
(IVs) Behavioural Self-Relevance/Time x Subjective Self- Relevance Interaction	1.97*	.08
Warmth	1.95	.01
Competence	1.59	.01
Active Help	2.83	.01
Passive Help	.78	.004
Active Harm	.46	.002
Passive Harm	.01	<.001
Reported Weekly Meat Consumption	.004	<.001
Discomfort	.81	.004
Measured Behavioural Self-Relevance	12.17***	.06
(IVs) Behavioural Self-Relevance/Time x Species Interaction	4.76***	.18
Warmth	.13	.001

Competence	.22	.002
A otivo Holp	.22	.002
Active Help	2.1	.02
Passive Help	.95	.01
Active Harm	.26	.003
Passive Harm	1.44	.01
Reported Weekly Meat Consumption	.03	<.001
Discomfort	38.28***	.27
Measured Behavioural Self-Relevance	3.54*	.03
(IVs) Species x Subjective Self-Relevance Interaction	1.29	.06
(IV/Covariate) Behavioural Self-Relevance/Time x Neutrality	1.01	.04
(IV/Covariate) Behavioural Self-Relevance/Time x Gender Interaction	2.45*	.1
Warmth	1.08	.01
Competence	.03	<.001
Active Help	.002	<.001
Passive Help	.73	.003
Active Harm	3.9	.02
Passive Harm	3.97*	.02
Reported Weekly Meat Consumption	.22	.001
Discomfort	13.66***	.06
Measured Behavioural Self-Relevance	3.44	.02
(IV) Behavioural Self-Relevance/Time Main Effect	6.05***	.22
Warmth	.34	.002
Competence	.42	.002
Active Help	.06	<.001
Passive Help	.71	.003
Active Harm	.14	.001

Passive Harm	5.16*	.02
Reported Weekly Meat Consumption	2.17	.01
Discomfort	46.09***	.18
Measured Behavioural Self-Relevance	4.85*	.02
(IV) Subjective Self-Relevance Main Effect	1.11	.05
(IV) Species Main Effect	13.14***	.37
Warmth	34***	.25
Competence	27.54***	.21
Active Help	19.48***	.16
Passive Help	7.11***	.06
Active Harm	31.12***	.23
Passive Harm	25.46***	.2
Reported Weekly Meat Consumption	1.64	.02
Discomfort	16.74***	.14
Measured Behavioural Self-Relevance	134.66***	.57
(Covariate) Neutrality Main Effect	1.52	.06
(Covariate) Gender Main Effect	2.96**	.12
Warmth	2.58	.01
Competence	8.98**	.04
Active Help	3.71	.02
Passive Help	2.94	.01
Active Harm	.99	.01
Passive Harm	3.91*	.02
Reported Weekly Meat Consumption	.22	.001
Discomfort	19.6***	.09
Measured Behavioural Self-Relevance	1.26	.01

Table A6

Inferential Statistics for Study 4 One-Way ANCOVA for Fan	illiarity Manipula	tion Check
Effect	F	${\eta_{\mathrm{p}}}^2$
(IV) Familiarity Main Effect	1.14	.01
(Covariate) Neutrality	6.63*	.07
(Covariate) Gender	.83	.01
<i>Note.</i> * $p < .05$, ** $p \le .01$, *** $p \le .001$. IV = independent variable	le	

Law Cturde 10 117 ANCOUA 1 α

Table A7

Inferential Statistics for Study 4 One-Way MANCOVA for Similarity Manipulation Checks

DV	F	$\eta_{ m p}{}^2$
(IV) Similarity Main Effect	9.17***	.23
Perceived Similarity to Humans	19.46***	.17
Perceived Shared Emotional Capacities with Humans	11.2***	.11
Perceived Shared Cognitive Capacities with Humans	13.66***	.13
(Covariate) Neutrality	3.45*	.1
Perceived Similarity to Humans	7.95**	.08
Perceived Shared Emotional Capacities with Humans	.35	.004
Perceived Shared Cognitive Capacities with Humans	.24	.003
(Covariate) Gender	1.18	.04

Note. p < .05, p < .01, p < .01, p < .01. Multivariate effects from the MANCOVA are reported in *italics*, with univariate effects on individual DVs from the follow-up ANCOVAs being reported after significant multivariate effects only. IV = independent variable

DV	F	$\eta_{\rm p}^{-2}$
(IV) Condition x Time Interaction	3.16***	.09
Warmth	4.04**	.06
Competence	15.09***	.19
Active Help	5.15**	.07
Passive Help	4.38**	.06
Active Harm	.11	.002
Passive Harm	1.25	.02
(Covariate) Neutrality x Time Interaction	.61	.02
(Covariate) Gender x Time Interaction	.43	.01
(IV) Condition Main Effect	1.71*	.05
Warmth	2.57	.04
Competence	2.52	.04
Active Help	2.27	.03
Passive Help	.31	.01
Active Harm	1.54	.02
Passive Harm	1.26	.02
Time Main Effect	.86	.03
(Covariate) Neutrality	1.71	.05
(Covariate) Gender	1.52	.05

Table A8

Inferential Statistics for Study 4 4x2 MANCOVA on SCM/BIAS Map Variables

Table A9

DV	F	${\eta_{ m p}}^2$
(IVs) Similarity x Species x Time Interaction	1.66	.08
(IV) Similarity x Time Interaction	5.71***	.24
Warmth	8.02**	.04
Competence	23.41***	.12
Active Help	4.98*	.03
Passive Help	3.03	.02
Active Harm	7.81**	.04
Passive Harm	.02	<.001
Perceived Similarity to Humans	17.33***	.09
Perceived Shared Emotional Traits with Humans	29.75***	.15
Perceived Shared Cognitive Traits with Humans	40.27***	.19
(IV) Species x Time Interaction	1.03	.05
(IVs) Similarity x Species Interaction	.64	.03
(Covariate) Neutrality x Time Interaction	.53	.03
(Covariate) Gender x Time Interaction	.5	.03
(IV) Similarity Main Effect	2.84**	.13
Warmth	11.72***	.06
Competence	4.39*	.03
Active Help	7.64**	.04
Passive Help	.25	.001
Active Harm	3.73	.02
Passive Harm	2.23	.01
Perceived Similarity to Humans	2.01	.01

Inferential Statistics for Study 5 2x2x2 MANCOVA on SCM/BIAS Map Variables and Similarity Manipulation Checks

Perceived Shared Emotional Traits with Humans	13.26***	.07
Perceived Shared Cognitive Traits with Humans	1.48	.01
(IV) Species Main Effect	15.22***	.45
Warmth	72.84***	.3
Competence	90.12***	.34
Active Help	55.63***	.24
Passive Help	25.59***	.13
Active Harm	77.33***	31
Passive Harm	47.21***	.21
Perceived Similarity to Humans	6.49*	.04
Perceived Shared Emotional Traits with Humans	45.36***	.21
Perceived Shared Cognitive Traits with Humans	31.68***	.15
Time Main Effect	.54	.03
(Covariate) Neutrality	.5	.03
(Covariate) Gender	1.13	.06

Table A10

Inferential Statistics for Study 6 One-Way ANCOVA for Pet Status Manipulation Check		
Effect	F	${\eta_{ m p}}^2$
(IV) Pet Status Main Effect	5.06*	.04
(Covariate) Neutrality	6.28*	.05
(Covariate) Gender	1.68	.01
	1 . • 1 1	

Note. **p*<.05, ***p*≤.01, ****p*≤.001. IV = independent variable

DV	F	$\eta_{ m p}{}^2$
(IV) Pet Status x Time Interaction	2.46*	.11
Warmth	10.03**	.08
Competence	1	.01
Active Help	2.69	.02
Passive Help	1.39	.01
Active Harm	4.89*	.04
Passive Harm	4.43*	.03
(Covariate) Neutrality x Time Interaction	1.97	.09
(Covariate) Gender x Time Interaction	.64	.03
(IV) Pet Status Main Effect	1.18	.06
Time Main Effect	1.72	.08
(Covariate) Neutrality Main Effect	2.6*	.12
Warmth	7.09**	.05
Competence	5.07*	.04
Active Help	1.97	.02
Passive Help	7.08**	.05
Active Harm	.49	.004
Passive Harm	.17	.001
(Covariate) Gender Main Effect	.34	.02

 Table A11

 Inferential Statistics for Study 6 2x2 MANCOVA on SCM/BIAS Man Variables

Appendix 5. Study 2 Participant Advertisements

a. Sona

Study Name: Perceptions of animals

Description: I am conducting this 15-minute experiment to examine possible factors explaining our perceptions of, and behavioural intentions towards, different animals. You will be given a participant information sheet describing the experiment and then be asked to provide informed consent. You will then be introduced to an animal and asked to indicate your perceptions of and behavioural intentions towards it. A debrief and researchers' contact details will follow. Participants will have the opportunity to enter a prize draw at the end of survey to win one of two £50 Amazon gift vouchers.

Eligibility: You must be 18 years of age or over due to ability to give informed consent.

Duration: 15 minutes

Credits: 0.25 credits

Researcher: Sarah Gradidge, sarah.gradidge@pgr.anglia.ac.uk

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

b. Social Media

Got 15 minutes and over 18? Participate in my second online survey on perceptions of animals. Participants will have the opportunity to enter a prize draw at the end of survey to win one of two £50 Amazon gift vouchers. If you are an ARU undergraduate psychology student, you can also receive 0.25 SONA credits. More details under the link below. Please share with others!

Qualtrics link here

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

c. Poster



Complete a 15 minute online experiment investigating people's perceptions of animals and be in with a chance to win an Amazon gift voucher!





The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants

Appendix 6. Images for Self-Relevance Manipulation

The below images were used to accompany the text introducing the imagined restaurant scenario, as a way to enhance realism.





Image courtesy of peoplecreations on Freepik. Designed by Freepik. > Business photo created by peoplecreations





Image courtesy of Erik Mclean on Unsplash, Available at <u>https://unsplash.com/photos/8yWLSPvoeVI</u>



Appendix 7. Study 3 Prolific Participant Advertisement

Title of study: Perceptions of animals

Describe what participants will be doing in this study: This experiment aims to examine your perceptions of an animal. You will be asked how you feel about an animal, followed by reading an imagined scenario about being in a restaurant. After the imagined scenario, you will again indicate how you feel about the animal. You may also provide demographic information as follows: diet, gender, age, nationality, ethnicity, current country of residence, duration of living in current country of residence and religion. Note that partial data will be reimbursed depending on percentage of completion whilst adhering to Prolific's minimum hourly rate. If you provide partial data, you will be asked by the researcher to return your submission and you will receive payment via the Bonus Payments system. Also note that some participants may be asked if they recognise an animal: Those who say that they do recognise the animal will be excluded from the rest of the experiment, asked to return their submission and reimbursed for their time in completing the experiment.

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

Pre-screening:

- Participants age 18+
- Meat consumers only

378

Appendix 8. Study 4 Participant Advertisements

a. Sona

Study Name: Perceptions of animals 3

Description: This 15-minute experiment aims to examine your perceptions of an unrecognised animal. Firstly, you will read a participant information sheet and provide informed consent. You will then be asked if you recognise an animal and how you feel about the animal, followed by reading some texts about the animal. After the texts, you will again indicate how you feel about the animal, followed by a debrief and researchers' contact details. You will receive either 0.25 SONA credits or be able to enter a prize draw for one of two £50 Amazon gift vouchers instead.

Eligibility: You must be 18 years of age or over due to ability to give informed consent. You must not have completed any of the previous "Perceptions of animals" experiments

Duration: 15 minutes

Credits: 0.25 credits or entering the prize draw

Researcher: Sarah Gradidge, sarah.gradidge@pgr.aru.ac.uk

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of ARU's Policy and Code of Practice for the Conduct of Research with Human Participants.

b. Social Media

Got up to 15 minutes, over 18 and have not participated in any previous 'perceptions of animals' experiments? Participate in my fourth online survey on perceptions of animals. All participants can enter a prize draw to win one of two £50 Amazon gift vouchers. If you are an
ARU undergraduate psychology student, you can receive 0.25 SONA credits instead. More details under the link below. Please share with others!

Qualtrics link here

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of ARU's Policy and Code of Practice for the Conduct of Research with Human Participants.

Appendix 9. Study 5 Prolific Participant Advertisement

Title of study: Perceptions of animals

Describe what participants will be doing in this study: This experiment aims to examine your perceptions of an animal. You will be asked how you feel about an animal, followed by reading some text about the animal. After reading the text, you will again indicate how you feel about the animal. You may also provide demographic information as follows: diet, gender, age, nationality, ethnicity, current country of residence, duration of living in current country of residence and religion. Note that partial data will be reimbursed depending on percentage of completion whilst adhering to Prolific's minimum hourly rate. If you provide partial data, you will be asked by the researcher to return your submission and you will receive payment via the Bonus Payments system.

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

Pre-screening:

- Participants age 18+
- Meat consumers only

Appendix 10. Study 6 Participant Advertisements

a. Sona

Study Name: Perceptions of animals 4

Description: This 15-minute experiment aims to examine your perceptions of an unrecognised animal. Firstly, you will read a participant information sheet and provide informed consent. You will then be asked if you recognise an animal and how you feel about the animal, followed by reading some texts about the animal. After the texts, you will again indicate how you feel about the animal, followed by a debrief and researchers' contact details. You will receive either 0.25 SONA credits or be able to enter a prize draw for one of two £50 Amazon gift vouchers instead.

Eligibility: You must be 18 years of age or over due to ability to give informed consent. You must not have completed any of the previous "Perceptions of animals" experiments

Duration: 15 minutes

Credits: 0.25 credits or entering the prize draw

Researcher: Sarah Gradidge, sarah.gradidge@pgr.aru.ac.uk

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of ARU's Policy and Code of Practice for the Conduct of Research with Human Participants.

b. Social Media

Got up to 15 minutes, over 18 and have not participated in any previous 'perceptions of animals' experiments? Participate in my fifth online survey on perceptions of animals. All participants can enter a prize draw to win one of two £50 Amazon gift vouchers. If you are an ARU undergraduate psychology student, you can receive 0.25 SONA credits instead. More details under the link below. Please share with others!

Qualtrics link here

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of ARU's Policy and Code of Practice for the Conduct of Research with Human Participants.

c. Prolific

Title of study: Perceptions of animals

Describe what participants will be doing in this study: This experiment aims to examine your perceptions of an animal. You will be asked how you feel about an animal, followed by reading an imagined scenario about being in a restaurant. After the imagined scenario, you will again indicate how you feel about the animal. You may also provide demographic information as follows: diet, gender, age, nationality, ethnicity, current country of residence, duration of living in current country of residence and religion. Note that partial data will be reimbursed depending on percentage of completion whilst adhering to Prolific's minimum hourly rate. If you provide partial data, you will be asked by the researcher to return your submission and you will receive payment via the Bonus Payments system. Also note that some participants may be asked if they recognise an animal: Those who say that they do recognise the animal will be excluded from the rest of the experiment, asked to return their submission and reimbursed for their time in completing the experiment.

The study has received ethics approval by the School Research Ethics Panel (SREP) and ratified by the Faculty Research Ethics Panel under the terms of Anglia Ruskin University's Policy and Code of Practice for the Conduct of Research with Human Participants.

Pre-screening:

- Participants age 18+
- Meat consumers only